Advances in Experimental Medicine and Biology 1178 Proteomics, Metabolomics, Interactomics and Systems Biology

# Paul C. Guest Editor

# Reviews on Biomarker Studies in Aging and Anti-Aging Research



# Advances in Experimental Medicine and Biology

Proteomics, Metabolomics, Interactomics and Systems Biology

#### **Sub-Series Editor**

Daniel Martins-de-Souza University of Campinas (UNICAMP) Institute of Biology Laboratory of Neuroproteomics Campinas, Brazil Advances in Experimental Medicine and Biology presents multidisciplinary and dynamic findings in the broad fields of experimental medicine and biology. The wide variety in topics it presents offers readers multiple perspectives on a variety of disciplines including neuroscience, microbiology, immunology, biochemistry, biomedical engineering and cancer research. Advances in Experimental Medicine and Biology has been publishing exceptional works in the field for over 30 years and is indexed in Medline, Scopus, EMBASE, BIOSIS, Biological Abstracts, CSA, Biological Sciences and Living Resources (ASFA-1), and Biological Sciences. The series also provides scientists with up-to-date information on emerging topics and techniques. 2018 Impact Factor: 2.126

Content of this series is reviewed.

This series of volumes focuses on concepts, techniques and recent advances in the field of proteomics, interactomics, metabolomics and systems biology. Recent advances in various 'omics' technologies enable quantitative monitoring of myriad various biological molecules in a high-throughput manner, and allow determination of their variation between different biological states on a genomic scale. Now that the sequencing of various genomes, from prokaryotes to humans, has provided the list and linear sequence of proteins and RNA that build living organisms, defining the complete set of interactions that sustain life constitutes one of the key challenges of the postgenomic era. This series is intended to cover experimental approaches for defining protein-protein, protein-RNA, protein-DNA and protein-lipid interactions; as well as theoretical approaches dealing with data analysis, integration and modeling and ethical issues.

More information about this series at http://www.springer.com/series/15040

Paul C. Guest Editor

# Reviews on Biomarker Studies in Aging and Anti-Aging Research



*Editor* Paul C. Guest Laboratory of Neuroproteomics Department of Biochemistry and Tissue Biology Institute of Biology University of Campinas (UNICAMP) Campinas, Brazil

Proteomics, Metabolomics, Interactomics and Systems Biology ISSN 0065-2598 ISSN 2214-8019 (electronic) Advances in Experimental Medicine and Biology ISBN 978-3-030-25649-4 ISBN 978-3-030-25650-0 (eBook) https://doi.org/10.1007/978-3-030-25650-0

#### © Springer Nature Switzerland AG 2019

This work is subject to copyright. All rights are reserved by the Publisher, whether the whole or part of the material is concerned, specifically the rights of translation, reprinting, reuse of illustrations, recitation, broadcasting, reproduction on microfilms or in any other physical way, and transmission or information storage and retrieval, electronic adaptation, computer software, or by similar or dissimilar methodology now known or hereafter developed.

The use of general descriptive names, registered names, trademarks, service marks, etc. in this publication does not imply, even in the absence of a specific statement, that such names are exempt from the relevant protective laws and regulations and therefore free for general use.

The publisher, the authors, and the editors are safe to assume that the advice and information in this book are believed to be true and accurate at the date of publication. Neither the publisher nor the authors or the editors give a warranty, express or implied, with respect to the material contained herein or for any errors or omissions that may have been made. The publisher remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

This Springer imprint is published by the registered company Springer Nature Switzerland AG The registered company address is: Gewerbestrasse 11, 6330 Cham, Switzerland

### Preface

The lifespan of every organism is limited by the aging process, which involves physical decline, an increase in chronic diseases, and ultimately death. It has been an ongoing quest of mankind to understand the aging process and use this information to develop ways of extending both the health and lifespan of individuals. Through these efforts, researchers have gained new insights into the physiological and molecular aspects of aging using both epidemiological and model organism approaches, and this has led to significant advancements in potential antiaging strategies. This is important as most chronic diseases in the world are intertwined with the aging process and occur more frequently in the aged population. According to the World Health Organization, non-communicable diseases affect mainly adults and elderly individuals, and this imposes the greatest burden on global health with staggering costs to the healthcare services. This book presents a series of reviews in various aspects of aging and age-related disease research along with several methods which have shown progress as potential antiaging approaches.

Chapter 1 covers studies which have focused on long-lived mutant and naturally occurring animal species. Chapter 2 describes the association of glycolytic dysfunction with the accelerated aging of neuronal cells in schizophrenia patients. Chapter 3 covers studies on the screening of antiaging drugs and gives clues into further research of aging biomarkers and antiaging targets. Chapter 4 describes the effects of sex differences in the aging process and the role of sex hormones in this process. Chapter 5 covers the efficacy of the ketogenic diet in a variety of neurodegenerative, neurodevelopmental, and metabolic conditions throughout different stages of life. Chapter 6 reviews the potential role of CoO10 supplementation in the treatment of tissue fibrosis, implicated in the age-related loss of function of various organs including the heart. Chapter 7 reviews the evidence that associates dietary restriction, cardiovascular aging, and age-related cardiovascular diseases, and related strategies to prevent or retard age-related cardiovascular diseases in the elderly. Chapter 8 focuses on the possible use of fecal microbiota-related parameters and microbiota-derived metabolites as biomarkers of cognitive performance and dementia, with a spotlight on the most promising areas of future research. Chapter 9 looks at the potential impact of herbal products on the prevention, regeneration, and delayed

aging of skin. Chapter 10 explores the influence of epigenetics on aging and the potential of restoring age-related changes to a "younger" state. Chapter 11 describes the basics of adipose tissue biology, growth hormone secretion, and action, and how the interactions of these may play a critical role in determining lifespan and health-span. Chapter 12 provides a brief overview of cytoskeletal structure and function, and discusses the evidence which links cytoskeletal function and dynamics with aging and neurodegeneration. Finally, Chapter 13 details how the most promising avenues to halt the aging process have come from studies of the molecular pathways involved with caloric restriction, insulin/insulin-like growth factor signaling, and mitochondrial ROS production, in nematode, fly, and rodent models.

As a way of highlighting the growing interest in this topic throughout the world, the authors in this series come from all six of the world's habitable continents. This includes the countries Australia, Brazil, China, Indonesia, Iran, Italy, Greece, Russia, the United Kingdom, and the United States of America. As each review describes the cutting edge of research in this important field in a functional blend of scientific and layperson's language, this volume will be of interest to scientists, medical practitioners, members of health organizations, and pharmaceutical company employees, as well as those without professional or specialized knowledge.

Campinas, Brazil

Paul C. Guest

# Contents

1	Of Mice, Whales, Jellyfish and Men: In Pursuit of Increased Longevity Paul C. Guest	1
2	Effects on Glial Cell Glycolysis in Schizophrenia: An Advanced Aging Phenotype? Giuliana S. Zuccoli, Paul C. Guest, and Daniel Martins-de-Souza	25
3	Aging Biomarkers and Novel Targets for Anti-AgingInterventionsKang Xu, Yannan Guo, Zhongchi Li, and Zhao Wang	39
4	Sex Differences in Aging and Associated Biomarkers Natalie Thomas, Caroline Gurvich, and Jayashri Kulkarni	57
5	The Therapeutic Potential of Ketogenic Diet Throughout Life: Focus on Metabolic, Neurodevelopmental and Neurodegenerative Disorders Ann-Katrin Kraeuter, Paul C. Guest, and Zoltan Sarnyai	77
6	Coenzyme Q10 Supplementation in Fibrosis and Aging Iain P. Hargreaves and David Mantle	103
7	<b>Dietary Restriction, Cardiovascular Aging and Age-Related</b> <b>Cardiovascular Diseases: A Review of the Evidence</b> Behnaz Abiri and Mohammadreza Vafa	113
8	<b>Gut Microbiota and Microbiota-Related Metabolites</b> <b>as Possible Biomarkers of Cognitive Aging</b> Andrea Ticinesi, Antonio Nouvenne, Claudio Tana, Beatrice Prati, and Tiziana Meschi	129

9	The Impact of Herbal Products in the Prevention,Regeneration and Delay of Skin AgingMega Ferdina Warsito and Idha Kusumawati	155
10	Epigenetic Clock: Just a Convenient Marker or an ActiveDriver of Aging?Vasily V. Ashapkin, Lyudmila I. Kutueva, and Boris F. Vanyushin	175
11	<b>From White to Brown – Adipose Tissue Is Critical to the Extended</b> <b>Lifespan and Healthspan of Growth Hormone Mutant Mice</b> Justin Darcy and Andrzej Bartke	207
12	The Cytoskeleton as a Modulator of Agingand NeurodegenerationKonstantinos Kounakis and Nektarios Tavernarakis	227
13	Metabolic Biomarkers in Aging and Anti-Aging Research Paul C. Guest	247
Ind	ex	265

viii

## Chapter 1 Of Mice, Whales, Jellyfish and Men: In Pursuit of Increased Longevity



Paul C. Guest

#### 1 Introduction

On August 4, 1997, a French woman from Arles named Jeanne Louise Calment died at the age of 122 years and 164 days [1]. She was widely reported to have been the oldest human and there were documents to confirm this. Her incredible longevity attracted significant attention from the media worldwide and the secrets of how she achieved this long healthy life sparked considerable medical research and speculation. According to Dr. Jean Marie Robine, a public health researcher and biographer, Madame Calment ate pounds of chocolate, smoked cigarettes and drank red wine every day, and she rode a bicycle up to the age of 100 [2]. Dr. Robine stated, "I think she was someone who constitutionally and biologically speaking, was immune to stress." Madame Calment herself was fond of saying, "If you can't do anything about it, don't worry about it." The book speculates that this way of handling stress helped to strengthen Madame Calment's immune system, which normally declines with age [2].

It is a well known fact that women live longer than men, making sex one of the strongest predictors of human lifespan [3]. This is borne out by the current list of the oldest verified people (Tables 1.1 and 1.2). This shows that the top 10 females lived and average of 3% longer than the top 10 males. This is also borne out by World Health Organization (WHO) statistics for the year 2015 across the world, in terms of life expectancy at birth [4]. The WHO estimated life expectancies were 74.2 years for females and 69.8 years for males (a difference of 6% in favour of females). These statistics differ across the world and countries such as Japan having the

P. C. Guest (🖂)

Laboratory of Neuroproteomics, Department of Biochemistry and Tissue Biology, Institute of Biology, University of Campinas (UNICAMP), Campinas, Brazil

<sup>©</sup> Springer Nature Switzerland AG 2019

P. C. Guest (ed.), *Reviews on Biomarker Studies in Aging and Anti-Aging Research*, Advances in Experimental Medicine and Biology 1178, https://doi.org/10.1007/978-3-030-25650-0\_1

Name	Birth date	Death date	Age at death	Country
Jeanne Calment	Feb 21, 1875	Aug 4, 1997	122 years, 164 days	France
Sarah Knauss	Sep 24, 1880	Dec 30, 1999	119 years, 97 days	USA
Nabi Tajima	Aug 4, 1900	Apr 21, 2018	117 years, 260 days	Japan
Lucy Hannah	Jul 16, 1875	Mar 21,1993	117 years, 248 days	USA
Marie-Louise Meilleur	29 Aug, 1880	Apr 16, 1998	117 years, 230 days	Canada
Violet Brown	Mar 10, 1900	Sep 15, 2017	117 years, 189 days	Jamaica
Emma Morano	Nov 29, 1899	Apr 15, 2017	117 years, 137 days	Italy
Chiyo Miyako	May 2, 1901	Jul 22, 2018	117 years, 81 days	Japan
Misao Okawa	Mar 5, 1898	Apr 1, 2015	117 years, 27 days	Japan
María Capovilla	Sep 14, 1889	Aug 27, 2006	116 years, 347 days	Ecuador

 Table 1.1
 List of the top 10 oldest verified females as of May 8, 2019

Table 1.2 List of the top 10 oldest verified males as of May 8, 2019

Name	Birth date	Death date	Age at death	Country
Jiroemon Kimura	Apr 19, 1897	Jun 12, 2013	116 years, 54 days	Japan
Christian Mortensen	Aug 16, 1882	Apr 25, 1998	115 years, 252 days	USA
Emiliano Mercado del Toro	Aug 21, 1891	Jan 24, 2007	115 years, 156 days	Puerto Rico
Mathew Beard	Jul 91,870	Feb 16, 1985	114 years, 222 days	USA
Walter Breuning	Sep 21, 1896	Apr 14, 2011	114 years, 205 days	USA
Yukichi Chuganji	Mar 23, 1889	Sep 28, 2003	114 years, 189 days	Japan
Joan Riudavets	Dec 15, 1889	Mar 5, 2004	114 years, 81 days	Spain
Fred H. Hale, Sr.	Dec 1, 1890	Nov 19, 2004	113 years, 354 days	USA
Yisrael Kristal	Sep 15, 1903	Aug 11, 2017	113 years, 330 days	Israel
Johnson Parks	Oct 15, 1884	Jul 17, 1998	113 years, 275 days	USA

highest life expectancy with 86.8 years for females and 80.5 years for males, and Sierra Leone having the lowest at 50.8 years for females and 49.3 years for males (Fig. 1.1). One factor which might account for the lower life expectancy in Sierra Leone and other African countries may be related to healthcare expenditures. A study published in 2017 found a positive relationship between healthcare expenditures and life expectancy and a negative correlation with the number of neonatal, infant, and under-five deaths [5]. From these data, it is clear that environmental factors can also have an impact on life expectancy. A study of 28 European Union countries published in 2014 explored potential determinants of life expectancy at birth, namely gross domestic product (GDP) growth rate, population growth rate, level of education attained, education enrolment, GDP per capita and life expectancy in the period from 2001 to 2011 [6]. The results of this study found that GDP per capita and attained education level explained 72.6–82.6% of the differences, depending on the year of observation.

Non-communicable diseases account for 63% of all deaths worldwide and approximately three quarters of these were incurred by individuals over 60 years old [7]. In 2016, ischemic heart disease and stroke were among the highest causes of



**Fig. 1.1** Diagram of the world showing differing life expectancies of countries (according to the WHO, 2015). (Figure by JackintheBox - Own work, CC BY-SA 4.0, https://commons.wikimedia.org/w/index.php?curid=69178739)

death in the world with a combined 15.2 million deaths out of 56.9 million total deaths and chronic obstructive pulmonary disease, Alzheimer's disease and other dementias, lung and tracheal cancers and diabetes were responsible for a further 8.3 million deaths [8]. In addition, these age-related diseases are associated with prolonged disability and reduced productivity, and of course they also have a negative financial impact within families and heavy demands on healthcare systems [9].

Improvements in human longevity and healthy aging are among the most difficult challenges facing modern medicine. Life expectancy has more than doubled in the last hundred years and people aged over 85 (the "very old") constitute the fastest growing across all age groups due to increases in life expectancy and advances in medical support [10]. Several studies on the very old have found that the appearance or progression of non-communicable diseases is delayed in this group compared to other age groups, suggesting that healthspan and longevity are intimately linked [11–14]. These characteristics suggest that other factors beyond environmental conditions play an important role in regulation of lifespan.

The molecular mechanisms by which longevity has evolved are still mostly unexplained. However, the comparative analyses of genomes, transcriptomes, proteomes and metabolomes of long-lived and short-lived species, along with those of model organisms, has allowed us to increase our understanding of this process [15–23]. This review presents an overview of these studies with a focus on long-lived mutant mice and worms, and the longest living natural species from giant tortoises to whales, sharks, clams and the immortal jellyfish (Fig. 1.2). The main aim is to identify conserved genes and physiological networks involved in the aging process and longevity.



Fig. 1.2 Timeline showing the maximum lifespan recorded for some of the longest-lived species

#### 2 What Have We Learned from Studying the Very Old: The Human Perspective?

There has been an increase in research since the death of Jeanne Calment to identify the genetic, environmental and lifestyle components associated with extreme longevity. One approach has been to study population groups sometimes called the oldest-old, such as centenarians who lived long lives essentially free of disease. With this in mind, a number of genome-wide association studies (GWAS) of various human populations have been performed to identify genes associated with living a long time [12, 24–28]. Such studies have shown associations of *APOE* and *FOXO3A* variants with longevity and these findings have been confirmed in other population studies [29–31]. Not surprisingly, maintenance and repair of the genome have also been identified as important factors for supporting a long life. This has been shown by studies on centenarian populations which found that such individuals appear to have an enhanced DNA repair capacity with significantly lower genomic and cellular damage [32–35].

In an interesting approach, a genetic study of long-lived smokers identified an interaction network of 215 single nucleotide polymorphisms (SNPs) that was significantly associated with increased survival and low cancer rates [36]. The strategy was to potentially identify genes involved in enhanced maintenance and survival by

studying individuals who had lived long healthy lives despite the fact that they were smokers. The identified network was enriched in platelet-derived growth factor (*PDGF*), *RAS* and *RAP1* and pathways which have implications for cancer pathogenesis [37–39]. In addition, the results showed that pathways potentially linked with aging were enriched and of these, PI3K/AKT signalling showed the highest enrichment score. This pathway comprised genes related to stress resistance, DNA repair, cell death, protein turnover and antioxidant function [40]. The PI3K/AKT pathway is known to be activated by insulin and insulin-like growth factor (IGF) signalling, which have also been implicating in the aging process and longevity [41–43]. Another SNP in the forkhead transcription factor *FOXO3A* gene was identified consistent with a previous study which found that two SNPs in *FOXO3* were associated with extreme longevity in two different groups comprised of individuals with German [29] or Japanese [44] ancestry.

In addition to looking at gene polymorphisms, another study compared the circulating levels of specific proteins, lipids and glucose as potential biomarkers in long-lived and short-lived families as a means of identifying protective genes linked to healthy aging [45]. This showed that the oldest-old individuals had higher serum levels of apidonectin. This is interesting as adiponectin is produced by adipose tissue and has anti-atherogenic and insulin-sensitizing activities, and low levels of adiponectin have been used as a biomarker of metabolic syndrome [46, 47]. This is consistent with a recent study that used genetic analyses as a way of identifying circulating factors as potential biomarkers of aging, longevity and age-related diseases [48]. As part of this study, a panel of hormones was identified, consisting of adiponectin, IGF-1, resistin, ghrelin and growth hormone (Table 1.3).

An earlier database study identified 20 circulating biomarkers associated with mortality risk through examination of 11,555 different publications [49]. This most significant associations were found for brain natriuretic peptide, cholesterol fractions, C-reactive protein, erythrocyte sedimentation rate, fibrinogen, granulocytes, homocysteine, intercellular adhesion molecule-1, N terminal-pro brain natriuretic peptide, white blood cell count, granulocytes, homocysteine, intercellular adhesion molecule-1, neutrophils, osteoprotegerin, procollagen type III aminoterminal peptide, uric acid, soluble urokinase plasminogen activator receptor, tissue inhibitor of metalloproteinases 1 and tumour necrosis factor receptor II. The fact that only one biomarker, intracellular adhesion molecule 1, overlapped across both studies suggests that considerable further research is necessary.

In terms of microRNAs, miR-17-5p has been identified as potentially being involved in longevity and cancer through its role in cell cycle regulation, proliferation and apoptosis [50]. This microRNA is expressed highly in embryonic stem cells and a transgenic miR-17-5p mouse model showed an extended lifespan [51]. However, this microRNA may also be involved in cancer regulation and so further studies are again merited to determine its use as a potential biomarker or therapeutic target.

In the cases of hormones, most of the research has been conducted on insulin, IGF-1 and growth hormone. It is well known that serum levels of both of these hormones are altered with advancing age. One study of 49 centenarians found that

Table 1.3 Panels of potential circulating biomarkers associated with aging, longevity and age-related diseases identified through genetic analyses. The panels are divided into seven majorpathways [48]

Pathway	Potential biomarker papel component		
Calcium homoostasis	Colrationlin		
Calcium nonicostasis	Paquealcin		
	S100 soloine hinding protein D		
Cytoskalaton and hormonas	Alpha klotha		
Cytoskeleton and normones	Adiponectin		
	Eibroblect growth factor 21		
	Fibroblast growth factor 23		
	Ghrelin		
	Growth hormone		
	Insulin-like growth factor 1		
	Leptin		
	Resistin		
Fibrosis	Angiotensinogen		
	Plasminogen activator		
	Plasminogen activator inhibitor		
	Transforming growth factor beta		
Inflammation	Alpha-defensin		
	C-X3-C motif chemokine ligand 1		
	C-X-C motif chemokine ligand 10		
	Interleukin 6		
	Pentraxin		
	Intercellular adhesion molecule-1		
Mitochondria and apoptosis	Amyloid beta precursor protein		
	Fibronectin type III domain containing 5		
	Growth differentiation factor 15		
	Lactate dehydrogenase		
	Vimentin		
Neuromuscular junction	Agrin		
	Brain derived neurotrophic factor		
	Complement factor 3		
	High mobility group box 1		
	Interleukin 1 receptor like 1		
	Progranulin		
	Soluble receptor for advanced glycosylation end		
	products		
Other principles	Adenosylhomocysteinase		
	Glycoprotein nonmetastatic melanoma protein B		
	Keratin 18		
	Lactoferrin		
	Micro ribonucleic acid panel (to be defined)		

serum IGF-1 levels appeared to be correlated to short-term nutritional status and low levels of this hormone may be involved in progression of dementia [52]. A randomized control trial of overweight, sedentary individuals placed on a calorierestricted diet for 6 months found that fasting insulin levels and body temperature were both reduced [53] and survival studies of overweight and obese people have led to the suggestion that long-term calorie restriction could prevent weight gain and add 3–13 years to life expectancy [54]. In line with this, a recent study found that changes in the levels of insulin, IGF-1 and IGF binding proteins 1 and 2 could be used as biomarkers to monitor the effects of dietary restriction [55]. Another study which assessed various hormones and cytokines in centenarians found that low levels of leptin, IGF-1 and IGF binding protein 3, and high levels of tumor necrosis factor-alpha were associated with increased mortality [56]. An investigation of a healthy cohort of 476 individuals aged 16–104 years, living in the West Coast of Southern Italy, found that both blood levels of IGF-1 and leukocyte telomere length decreased significantly with age, consistent with the cellular senescence hypothesis of aging [57]. In addition, the role of eroded telomeres in cellular aging has been well established and has been reviewed elsewhere [58-60].

In possible contrast to these studies, the Baltimore Longitudinal Study of Aging, which compared the levels of various circulating biomarkers between participants who survived to at least 90 years of age and those who died between 72–76 years of age, found no differences in the circulating levels of ghrelin, insulin, leptin, interleukin 6, adiponectin and testosterone [61]. Taken together, these findings indicate that there is still a considerable amount of research warranted concerning the levels of circulating biomarkers of aging and longevity.

Another factor which should be taken into account in human aging and longevity studies is the effect of potential biomarkers related to oxidative damage. One study found a lipid profile characterized by a higher ratio of monounsaturated compared to polyunsaturated lipids, indicative of lower oxidative stress [62]. This was consistent with a combined nuclear magnetic resonance imaging metabonomic and shot-gun lipidomic analysis of human serum which found differences in the levels of 41 lipids associated with greater antioxidant capacity in centenarians compared to younger controls [63]. Another study tested the potential of administering a dietary antioxidant in the form of pomegranate juice in preventing memory loss in old subjects, using memory testing and functional magnetic brain imaging as the outcome measures [64]. This demonstrated an increase in visual memory task scores and functional brain activity in the group receiving the pomegranate juice. One inconsistent study of older community-dwelling adults found that urinary levels of the antioxidant resveratrol was not associated with inflammatory markers, cardiovascular disease or cancer, and was not predictive of all-cause mortality [65].

#### **3** The Lessons from Model Species

#### 3.1 Mice

The main parameters seen in association with increased longevity in mouse models are: (1) suppression of the growth hormone/IGF-1axis; (2) decreased metabolism; (3) increased resistance to oxidative stress; (4) reduced insulin secretion; (5) enhanced insulin sensitivity, (6) delayed maturation; and (7) reduced fecundity [66]. Studies of growth hormone-deficient mutant mice show that these animals live 50% longer and show differences in energy metabolism regulation, compared to wild-type mice [67]. In line with the above parameters, reduced growth hormone signalling is thought to increase longevity by leading to increased stress resistance, reduced growth, altered cytokine production by adipose tissue, increased insulin sensitivity and increased oxygen consumption [68]. Mice with an extended lifespan have also been produced by gene deletions or other manipulations leading to a deficiency in the levels of IGF-1 [66, 67, 69, 70], growth hormone [66–69], insulin [43, 70–72] and insulin receptor substrates (IRS) 1 [73] and 2 [74], and in the p70 ribosomal protein S6 kinase 1 (S6K1) [75]. S6K1 is involved in regulation of cell growth via the mammalian target of rapamycin (mTOR), a pathway known to be involved in aging through studies showing that the mTOR inhibitor rapamycin increases mouse longevity [76–78]. Furthermore, insulin is known to regulate nutrient and metabolic homeostasis via the IRS/PI3K/AKT signalling pathway that converges on regulation of FOXO1 and mTOR [79].

There is also an oxidative stress theory of aging which proposes that the rate of aging is regulated by accumulation of oxidative damage [80]. According to this hypothesis, damage occurs through accumulation of reactive oxygen species (ROS) which induces tissue degeneration. The oxidative stress theory of aging was later expanded to include the mitochondrial theory of aging, which proposes that ROS resulting from the mitochondrial respiratory chain damages macromolecules such as lipids, proteins and mitochondrial DNA [81]. This leads to a viscous cycle of defective mitochondrial function, in turn causing further ROS generation and oxidative damage. Both theories are supported by the fact that animals accumulate oxidative damage as they age [82]. Early on, this theory gained a lot of support although the results of recent studies suggest that oxidative stress or damage does not affect the aging process, measured as the mean or maximum lifespan [83]. However, a number of studies have demonstrated that reduced expression of antioxidants is associated with accelerated age-related diseases. For example, heterozygous knockout mice with reduced manganese superoxide dismutase (SOD2+/-) show increased incidence of neoplasia with age, compared to wildtype control mice  $(SOD2^{+/+})$  [84]. The SOD2<sup>+/-</sup> mice also have increased incidences of cardiac hypertrophy [85] and deficits in glucose metabolism [86]. Conversely, overexpression of SOD1 has been found to have beneficial effects in mouse models of atherosclerosis [87], metabolic dysfunction [88], Alzheimer's disease [89] and Parkinson's disease [90]. Similarly,

overexpression of *SOD2* has been found to prevent cardiac myopathy [91], age-related defects in glucose metabolism [88] and Alzheimer's disease [92].

Caloric restriction has been shown to decrease age-related chronic diseases and extend lifespan in many species, including mice and rats [93, 94]. In rodents and primates, caloric restriction has been found to lower circulating glucose, insulin, cholesterol, triglycerides and IGF-1 levels [95]. Many of the effects of calorie restriction are also seen in dwarf mice. The lifespan of the Ames and Snell dwarf mice is significantly longer than that of their phenotypically normal control siblings [96, 97]. The mean age at death of dwarf mice exceeds the corresponding value in the normal controls by approximately 40% and is thought to be due to delayed aging of the dwarf mice is also characterised by increased levels of the antioxidant enzymes catalase and SOD [98, 99], a reduced metabolic rate and core temperature [100], lower plasma glucose levels and increased insulin sensitivity [101], and reduced growth hormone and IGF-1 levels [102].

Finally, the *SIR2* gene is linked to longevity in yeast, Caenorhabditis elegans and Drosophila [103]. *SIR2* and the mammalian orthologue *SIRT1* are NAD-dependent deacetylases, and both are thought to link metabolism with lifespan [104]. This is consistent with the finding that some of the effects of moderate calorie restriction include changes in expression of *SIRT1*-related genes in mice and other species [105]. SIRT1 inhibits adipogenesis in white adipose cells via repression of peroxisome proliferator-activated receptor  $\gamma$  (PPAR $\gamma$ ), it regulates insulin secretion by pancreatic  $\beta$  cells and it deacetylates and activates expression of genes regulating gluconeogenesis in the liver [103].

To summarize this section, insulin signalling regulates nutrient and metabolic homeostasis via the IRS/PI3K/AKT signalling pathway that targets FOXO1 and mTOR. Insulin signalling also regulates the mitochondrial electron transport chain by suppression of FOXO1 to maintain the NAD+/NADH ratio, which in turn regulates the SIRT1 pathway required in mitochondrial function. Conversely, mitochondrial dysfunction can lead to high ROS levels that could further damage mitochondria and other cellular components leading to insulin resistance.

#### 3.2 Worms

FOXOs are transcription factors, featuring a conserved Forkhead box, involved in regulation of important cellular processes such as cell cycle arrest, apoptosis, metabolism, stress resistance, aging and longevity [106]. Invertebrates such as Caenorhabditis elegans (C. elegans) have only one FOXO gene, named *DAF-16*, named after the dauer phenotype of a developmentally arrested larval form of greater durability that is induced by crowding or starvation [107]. More recent investigations have found that *DAF-2*, *DAF-23* and *DAF-16* are all part of the same branch of the dauer regulatory pathway involved in transduction of a pheromone signal for dauer formation [108]. An elegant study carried out by Kenyon et al.,

published in 1993 showed that *DAF-2* C. elegans mutants have a lifespan that is approximately doubled and this effect is suppressed by a mutation in *DAF-16* [109]. These findings suggested that *DAF-2* mutations can extend lifespan through activation of *DAF-16*.

Several other gene mutations have been identified in C. elegans that lead to increased longevity and these can be assigned to pathways such as insulin/insulinlike signalling, the dietary restriction pathway and translation control [110–114]. A recent study showed that administration of the traditional Chinese herbal medicine Ganoderma lucidum promoted C. elegans to resist oxidative stresses induced by paraguat and the heavy metal  $Cr^{6+}$ , and it prolonged lifespan [115]. This protection occurred through the same pathways seen though dietary restriction and mTOR/ S6K signalling, respectively. Another traditional Chinese medicine was found to extend the lifespan of C. elegans though the insulin/IGF-1 signalling pathway [116]. Another study suggested that administration of the pro-antioxidant compound selenocysteine can mimic the effect of dietary restriction on lifespan in C. elegans [117]. This was consistent with previous studies of C. elegans showing that administration of selenium inhibited Mn-induced toxicity and led to increased expression of SOD-3 and nuclear localization of DAF-16 [118]. Furthermore, dietary supplementation with selenocysteine increased resistance to oxidative stress, extended lifespan without reduction of fertility, and delayed age-related decline of motility [119].

Another recent study showed that the levels of the miR-58 family of miRNAs, which are effectors of insulin signalling, are increased in long-lived C. elegans mutants, again providing a link to dietary restriction effects [120]. This was consistent another miRNA study which identified a number of factors involved in regulation of lifespan, as reviewed by Smith-Vikos and Slack in 2012 [121] and summarized in Table 1.4. The pathways involved were related to either insulin/IGF-1 signalling or cell cycle control. Another investigation into the role of miRNAs in regulation of C. elegans lifespan found that 6 miRNAs were either up- or down-regulated during aging and in *DAF-2* long-lived mutants and five of these showed an opposite regulation (Table 1.5) [122]. Specifically, miR-237, miR-62 and miR-252 were found to be decreased during aging and increased in the *DAF-2* mutants. Conversely, miR-239b and miR-246 were increased during aging and decreased in mutants. Thus, these miRNAs are potentially involved in regulation of genes that might represent good targets for regulation of lifespan.

Taken together, these studies of rodents and worms indicate the conservation of insulin/IGF signalling pathways involved in the aging process and longevity

**Table 1.4**miRNAs involved in the aging process identified through loss of function studies in C.elegans

miRNA	Loss of function mutant	Pathway
Lin-4	Shortened lifespan - aging phenotype	DAF-2 activation, DAF-16 repression
miR-71	Shortened lifespan - aging phenotype	Represses AGE-1/PDK-1/CDC-25.1/ CHK-1
miR-239	Longer lifespan	Activates AGE-1 and PDK-1

miRNA	Regulated during aging	Regulated in DAF-2 mutants
miR-237	Ļ	↑
miR-253	↑	↑
miR-62	$\downarrow$	↑
miR-252	Ļ	↑
miR-239b	↑	$\downarrow$
miR-246	↑	$\downarrow$

**Table 1.5** miRNAs up- or down-regulated >two-fold during normal aging (day 0–10) or in DAF-2 C. elegans mutants at day 0 or day 10 [122]



Fig. 1.3 Corresponding insulin/IGF signalling pathways form C. elegans and rodent studies indicating corresponding proteins or classes of proteins

(Fig. 1.3). Reduced insulin-like signalling through a cascade orthologous to the rodent and other mammalian insulin receptor/PI3K/AKT pathway, leads to an extended lifespan [123, 124]. Binding of the DAF-16 transcription factor to various genes leads to the lifespan extension phenotype.

#### 4 Giant Tortoise

Little has been published on the giant tortoise regarding lifespan research although this should be an excellent point of focus in this field. The Galapagos giant tortoise (*Chelonoidis nigra*) has one of the longest life spans among vertebrate species [125]. A female named "Harriet" was thought to have been collected from the Galapagos Islands by Charles Darwin and estimated to be 176 years old when she died in 2006 at the Famous Australia Zoo on the Sunshine Coast in Queensland, Australia [125–127]. I had the opportunity to visit the Galapagos Islands in 1987 and saw many of these marvellous creatures on various islands, some of which could have been alive at the time Charles Darwin visited these islands during the second voyage of the HMS Beagle in the latter half of 1835 [128].

The only molecular study carried out thus far on this species was just published in 2019 [129]. This was a global comparison analysis of the genomes of Lonesome George (the last known member of Chelonoidis abingdonii) and the Aldabra giant tortoise (Aldabrachelys gigantea) using both unsupervised and supervised analyses. This analysis resulted in identification of gene variants associated with DNA repair, mitochondrial function, inflammation and cancer development, which may be associated with some of the characteristics of these animals, including their gigantism and longevity.

#### 5 Bowhead Whales

The lifespan of some whale species is approximately double than that of humans [130]. The bowhead whale (Balaena mysticetus), a species of right whale (Balaenidaem) that lives in Arctic and sub-Arctic waters, and subsists on a lipidrich diet of krill and other small marine animals, has been estimated to live over 200 years [131]. This has been determined by chemical analyses of eye lenses of bowhead whales captured by native Alaskan hunters in the Arctic [132]. In addition, stone and ivory weapon fragments have been recovered from bowhead whales hunted near Alaska, USA, in the 1980s and 1990s although the dates of manufacture are not known. More recently bomb lance fragments that were likely to have been manufactured between 1879 and 1885 were recovered in 2007 from four bowhead whales [133]. From this information, the hunters estimated that the oldest of these whales was approximately 211 years old and the younger specimens were between 135 and 172 years old [134]. In addition to a longer lifespan, the bowhead whale has a lower incidence of disease in old age compared to humans [135]. Consistent with evolutionary theory, the longevity of whales has been associated with low fertility and extended developmental times, as seen for humans [136]. The longevity of whales is also consistent with theories that larger animal species tend to live longer [137, 138].

The molecular mechanisms underlying the increased longevity in the bowhead whale have not been fully elucidated, although cross species comparisons have allowed us to gain some insights. In genomic studies, the identification of loci most undergoing adaptation allows us to gain insights into evolutionary biology. This is achieved by comparing the codon substitution at presumably silent sites with those experiencing the most selection [139]. A higher rate of substitution occurs if natural selection favours changes in the protein sequence. Thus by comparing the genomes of long- versus short-lived species, we may be able to highlight those genes which are most associated with the aging process and longevity [24]. The publication of the genomes of the shorter-lived minke [140] and longer-lived bowhead [22] whales has made this possible. To determine selective pressure variation in bowhead whale, a protein-coding database was constructed for the bowhead, minke and orca whales, in a comparison with several terrestrial mammals of varying lifespans (chimpanzee, cow, dog, elephant, gibbon, gorilla, guinea pig, horse, human, macaque, marmoset, microbat, mouse, opossum, orangutan, platypus, rabbit and rat). In addition, the authors aligned each bowhead whale protein with orthologues from 9 other mammals (cow, dog, dolphin, elephant, horse, human, minke whale, mouse and rat) to identify the unique bowhead amino acid residues. Next gene expansion analysis was performed by construction of gene trees assembled from 21,069 (human), 22,275 (mouse), 19,292 (dog), 19,988 (cow), 15,769 (dolphin), 17,936 (platypus), 20,496 (minke whale) and 22,733 (bowhead whale) genes to identify duplications in the bowhead whale genome.

Their findings identified several genes under positive selection and mutations in genes specific in the bowhead whale associated with cancer and aging. They also indentified both gains and losses involving genes linked with DNA repair, cell-cycle regulation, cancer and aging. In addition, they found changes in genes related to thermoregulation, sensory perception, dietary adaptations and immune response. Several genes related to aging and cancer showed high selective pressure in whales, including suppressor of cytokine signaling 2 (SOCS2), aprataxin (APTX), noggin (NOG) and leptin (LEP). The highest selective pressure was seen for the genes FOXO3, ERCC3 (DNA repair), FGFR1 (fibroblast growth factor receptor) in the bowhead whale and amino acid changes unique to the bowhead whale were found in ERCC1 (DNA repair), HDAC1 and HDAC2 (histone deacetylases). In addition, gene duplications and unique amino acid changes were seen for PCNA (proliferating cell nuclear antigen; involved in DNA replication) and the amino acid sensor and mTORC1 activator LAMTOR1, which is associated with aging and cancer [141]. Finally, the UCP1 gene (uncoupling protein 1; involved in thermoregulation) has a premature stop codon in the bowhead whale, which is likely to change its function. This is interesting as bowhead whales need to maintain stable body temperatures in the cold water and this stop codon distinguishes them from smaller mammalian species.

In 2014, a genome-wide gene expression analysis of the bowhead whale was published based on *de novo* assembly of its mRNA transcriptome using the Illumina RNAseq technology [21]. To identify transcripts important in longevity, a comparison was carried out of the bowhead whale liver, kidney and heart transcriptomes

with those of the minke whale, cow, yak, Brandt's bat, Chinese tree shrew, naked mole rat, rhesus macaque, mouse and rat. A total of 45 transcripts were found to be differentially expressed in the bowhead whale liver compared to the other mammals and the greatest difference was a reduction in the levels of growth factor receptorbound protein 14 (GRB14). The encoded protein (GRB14) binds to the insulin and IGF-1 receptors, again implicating this signalling pathway in longevity and aging. Changes in proteins related to apoptosis and cell survival were also found, including Cbp/P300 interacting transactivator (CITED2) and TP53 apoptosis effector (PERP). In addition, altered levels of EPH receptor A2 (EPHA2) and replication protein A2 (RPA2) were detected, which promote vascular tone/proliferation and DNA repair, respectively. In the kidney, 53 transcripts were differentially expressed and many of these were associated with DNA repair and tumour suppression, and only 4 transcripts were altered in the heart transcriptome and one of these [argininosuccinate lyase (ASL)] is required in the production of nitrous oxide (NO), which is known to protect the heart during hypoxic events [142]. In the case of the whale, this could be related to its deep diving activities.

#### 6 Greenland Shark

The Greenland shark (Somniosus microcephalus) is a resident of the seas of North Atlantic and Arctic regions [143, 144]. This shark is known to mature slowly, reach sexual maturity after approximately 150 years and grow to a length of approximately 5 m [145]. A radiocarbon dating study of eye lens from 28 female Greenland sharks (0.81–5.02 m in total length) revealed a life span of greater than 272 years [145]. The age of the largest shark was approximately 392 years. Another investigation studied several metrics of oxidative status measured in skeletal muscle and red blood cells of the Greenland shark but found no correlation with the species lifespan [146]. Instead the authors proposed that these matrices were linked to ecological features, such as adaptation to the cold water environment and the extreme depth that sharks of this species dive to. Although further studies are required to understand the mechanisms of how these sharks reach this great age, the results indicated above place the Greenland shark as the longest-lived vertebrate on Earth [147].

#### 7 Ocean Quahog

The burrowing ocean quahog clam (Arctica islandica) is the longest lived of all noncolonial animal species on earth with a reported maximum species lifespan of more than 400 years [148, 149]. One study sampled the bivalve shell from a quahog clam specimen collected alive in 2006 from the north Icelandic shelf for radiocarbon analysis and annual band counting and found that this clam had lived for more than 405 years [150] and a more recent study found that one specimen had lived for 507 years [151]. Another study found that the extreme longevity of this species of clam might be due to a higher antioxidant capacity compared to shorter-lived clams [152]. These authors examined gill and mantle tissues of 4–192 year old quahogs and found that catalase and citrate synthase activities and glutathione concentrations declined rapidly within the first 25 years of life but thereafter remained stable for more than 150 years, and superoxide dismutase activity remained high over the entire life time. Another study tested this hypothesis by comparing reactive oxygen species production, resistance to oxidative stress and antioxidant defences in longlived ocean quahogs with the shorter-lived hard clam (Mercenaria mercenaria) [153]. These analyses revealed lower protein carbonyl levels in the ocean qualog compared with the hard clam, consistent with the possibility of reduced ROS production resulting in lower macromolecular damage. In addition, exposure to the oxidative stressor tert-butyl hydroperoxide (TBHP) revealed a greater resistance to oxidative stress-induced mortality in the quahog. However, the authors found that the quahog and hard clams had similar antioxidant capacities and homeostatic antioxidant responses following TBHP exposure. Therefore, variations in antioxidant capacities do not account for the difference in the longevities of these two clam species. Nevertheless, more recent studies found that the ocean quahog have a lower degree of lipid peroxidation [154], more resistance to genotoxic stressors [155] and greater maintenance of protein homeostasis [156], compared to other short-lived clams species. Taken together, these findings suggest that increased antioxidant capacity and resistance to stressors may be linked with the long lifespan of the ocean quahog, although other potential mechanisms should also be explored.

#### 8 Immortal Jellyfish

The immortal jellyfish (Turritopsis dohrnii) is thought to be the only animal capable of immortality [157, 158]. However, there is a trick to this. It's not because this jellyfish lives a long time. It's because it can effectively regenerate itself.

The adult form of the immortal jellyfish is approximately 0.5 cm wide. Once it has reproduced it can transform itself back into a juvenile polyp state. This involves retraction of the tentacles and shrinkage of the body using tissue from the circulatory canal and bell surface, in a process called transdifferentiation [159]. After or during this process, the jellyfish sinks to the seabed. The rejuvenated creature will then regain its sexual maturity in less than 30 days if the water temperature is at least 20 °C. This process has been observed to repeat itself indefinitely and so this jellyfish does not experience aging in the true sense [160]. Obviously, this incredible attribute of this creature has sparked considerable interest in the area of aging, stem cell and cancer research.

Although the transdifferentiation process has been documented, the underlying biological mechanisms of this process have not been fully elucidated. It is known that some jellyfish and other basal metazoans possess high levels of pluripotent stem cells capable of differentiating into all cell types [161]. However, during

evolution of more complex organisms, these capacities were thought to have been reduced or lost. One study carried out a phylogenetic analysis based on available 16 s rRNA gene and protein sequences of cytochrome oxidase subunit-I (COI or COX1) of the immortal jellyfish in comparison to similar species in order to identify its closest relatives [162]. This led to identification of 6 closely related species for further research. A more recent study sequenced the complete mitochondrial genome sequence of the immortal jellyfish, which contains genes encoding 13 proteins involved in oxidative phosphorylation and respiration, 2 transfer RNAs and 2 ribosomal RNAs [163]. Although this provided an initial characterization, thus far it has not allowed any insights into how this species can regenerate itself. This will require complete sequencing of the genome and appropriate cross comparisons with other species.

#### 9 Conclusions and Future Perspectives

Aging of differentiated adult cells occurs via several mechanisms. These include: (1) damage to DNA molecules; (2) perturbations in proteostasis; (3) mitochondrial dysfunction; (4) inappropriate insulin/IGF-1 signalling; and (5) damage resulting from oxidative stress. Most of the species discussed in this chapter characterized by long lifespans appear to have the capacity to reduce the effects of one or more of these pathways through evolution of specific counter measures. The study of model organisms such as long-lived rodent and worm mutants has allowed us to investigate these pathways in detail and identify potential modulatory agents. Furthermore, the investigation of differences in these pathways in short- and long-lived animal species has provided additional insights into how exceptional longevity can be achieved. Together, these approaches have improved our understanding of longevity regulation and provide a starting point for the development of interventions that can potentially maximise the human healthspan.

#### References

- Whitney CR (1997) Jeanne Calment, World's Elder, Dies at 122. The New York Times (5 August 1997). New York, NY, USA. Retrieved 16 January 2019. https://www.nytimes. com/1997/08/05/world/jeanne-calment-world-s-elder-dies-at-122.html
- Allard M, Lèbre V, Calment J, Robine J-M, Calment J (1999) Jeanne Calment: from Van Gogh's time to ours, 122 extraordinary years. Thorndike Press. ISBN: 0786217774
- Tower J (2017) Sex-specific gene expression and life span regulation. Trends Endocrinol Metab 28(10):735–747
- 4. https://www.who.int/gho/mortality\_burden\_disease/life\_tables/situation\_trends\_text/en/
- 5. Bein MA, Unlucan D, Olowu G, Kalifa W (2017) Healthcare spending and health outcomes: evidence from selected East African countries. Afr Health Sci 17(1):247–254
- 6. Bilas V, Franc S, Bosnjak M (2014) Determinant factors of life expectancy at birth in the European union countries. Coll Antropol 38(1):1–9

#### 1 Pursuing Increased Longevity

- 7. http://apps.who.int/medicinedocs/en/m/abstract/Js18806en/
- 8. https://www.who.int/news-room/fact-sheets/detail/the-top-10-causes-of-death
- 9. http://apps.who.int/medicinedocs/en/m/abstract/Js18804en/
- Hill TR, Mendonça N, Granic A, Siervo M, Jagger C, Seal CJ et al (2016) What do we know about the nutritional status of the very old? Insights from three cohorts of advanced age from the UK and New Zealand. Proc Nutr Soc 75(3):420–430
- Evert J, Lawler E, Bogan H, Perls T (2003) Morbidity profiles of centenarians: survivors, delayers, and escapers. J Gerontol A Biol Sci Med Sci 58(3):232–237
- Andersen SL, Sebastiani P, Dworkis DA, Feldman L, Perls TT (2012) Health span approximates life span among many supercentenarians: compression of morbidity at the approximate limit of life span. J Gerontol A Biol Sci Med Sci 67(4):395–405
- Sebastiani P, Sun FX, Andersen SL, Lee JH, Wojczynski MK, Sanders JL et al (2013) Families enriched for exceptional longevity also have increased health-span: findings from the long life family study. Front Public Health 1:38. https://doi.org/10.3389/fpubh.2013.00038
- Martin P, Kelly N, Kahana B, Kahana E, Willcox BJ, Willcox DC et al (2015) Defining successful aging: a tangible or elusive concept? Gerontologist 55(1):14–25
- 15. Ha MK, Soo Cho J, Baik OR, Lee KH, Koo HS, Chung KY (2006) Caenorhabditis elegans as a screening tool for the endothelial cell-derived putative aging-related proteins detected by proteomic analysis. Proteomics 6(11):3339–3351
- Bell R, Hubbard A, Chettier R, Chen D, Miller JP, Kapahi P et al (2009) A human protein interaction network shows conservation of aging processes between human and invertebrate species. PLoS Genet 5(3):e1000414. https://doi.org/10.1371/journal.pgen.1000414
- Austad SN (2010) Methusaleh's zoo: how nature provides us with clues for extending human health span. J Comp Pathol 142 Suppl 1:S10–S21. https://doi.org/10.1016/j.jcpa.2009.10.024
- Semeiks J, Grishin NV (2012) A method to find longevity-selected positions in the mammalian proteome. PLoS One 7(6):e38595. https://doi.org/10.1371/journal.pone.0038595
- Bodnar A (2013) Proteomic profiles reveal age-related changes in coelomic fluid of sea urchin species with different life spans. Exp Gerontol 48(5):525–530
- 20. De Waal EM, Liang H, Pierce A, Hamilton RT, Buffenstein R, Chaudhuri AR (2013) Elevated protein carbonylation and oxidative stress do not affect protein structure and function in the long-living naked-mole rat: a proteomic approach. Biochem Biophys Res Commun 434(4):815–819
- Seim I, Ma S, Zhou X, Gerashchenko MV, Lee SG, Suydam R et al (2014) The transcriptome of the bowhead whale Balaena mysticetus reveals adaptations of the longest-lived mammal. Aging (Albany NY) 6(10):879–899
- 22. Keane M, Semeiks J, Webb AE, Li YI, Quesada V, Craig T et al (2015) Insights into the evolution of longevity from the bowhead whale genome. Cell Rep 10(1):112–122
- Triplett JC, Swomley A, Kirk J, Lewis K, Orr M, Rodriguez K et al (2015) Metabolic clues to salubrious longevity in the brain of the longest-lived rodent: the naked mole-rat. J Neurochem 134(3):538–550
- Ma S, Gladyshev VN (2017) Molecular signatures of longevity: insights from cross-species comparative studies. Semin Cell Dev Biol 70:190–203
- 25. Willcox DC, Willcox BJ, Hsueh WC, Suzuki M (2006) Genetic determinants of exceptional human longevity: insights from the Okinawa Centenarian Study. Age 28(4):313–332
- Arnold J, Dai J, Nahapetyan L, Arte A, Johnson MA, Hausman D et al (2010) Predicting successful aging in a population-based sample of Georgia centenarians. Curr Gerontol Geriatr Res. pii:989315. https://doi.org/10.1155/2010/989315
- 27. Cho J, Martin P, Poon LW (2012) The older they are, the less successful they become? findings from the Georgia Centenarian Study. J Aging Res 2012:695854. https://doi. org/10.1155/2012/695854
- Brooks-Wilson AR (2013) Genetics of healthy aging and longevity. Hum Genet 132(12): 1323–1338
- Newman AB, Murabito JM (2013) The epidemiology of longevity and exceptional survival. Epidemiol Rev 35:181–197

- Flachsbart F, Caliebe A, Kleindorp R, Blanché H, von Eller-Eberstein H, Nikolaus S et al (2009) Association of FOXO3A variation with human longevity confirmed in German centenarians. Proc Natl Acad Sci U S A 106(8):2700–2705
- 31. Jacobsen R, Martinussen T, Christiansen L, Jeune B, Andersen-Ranberg K, Vaupel JW et al (2010) Increased effect of the ApoE gene on survival at advanced age in healthy and longlived Danes: two nationwide cohort studies. Aging Cell 9(6):1004–1009
- Beekman M, Blanché H, Perola M, Hervonen A, Bezrukov V, Sikora E et al (2013) Genomewide linkage analysis for human longevity: genetics of healthy aging study. Aging Cell 12(2):184–193
- 33. Humphreys V, Martin RM, Ratcliffe B, Duthie S, Wood S, Gunnell D et al (2007) Age-related increases in DNA repair and antioxidant protection: a comparison of the Boyd Orr cohort of elderly subjects with a younger population sample. Age Ageing 36(5):521–526
- 34. Chevanne M, Calia C, Zampieri M, Cecchinelli B, Caldini R, Monti D et al (2007) Oxidative DNA damage repair and parp 1 and parp 2 expression in Epstein-Barr virus-immortalized B lymphocyte cells from young subjects, old subjects, and centenarians. Rejuvenation Res 10(2):191–204
- Franzke B, Neubauer O, Wagner KH (2015) Super DNAging—new insights into DNA integrity, genome stability and telomeres in the oldest old. Mutat Res Rev Mutat Res 766:48–57
- 36. Kim YJ, Kim HS, Seo YR (2018) Genomic approach to understand the association of DNA repair with longevity and healthy aging using genomic databases of oldest-old population. Oxidative Med Cell Longev 2018:2984730–2984712. https://doi.org/10.1155/2018/2984730
- Levine ME, Crimmins EM (2016) A genetic network associated with stress resistance, longevity, and cancer in humans. J Gerontol A Biol Sci Med Sci 71(6):703–712
- Westermark B, Nister M, Heldin CH (1985) Growth factors and oncogenes in human malignant glioma. Neurol Clin 3(4):785–799
- 39. Haliotis T, Trimble W, Chow S, Mills G, Girard P, Kuo JF et al (1988) The cell biology of ras-induced transformation: insights from studies utilizing an inducible hybrid oncogene system. Anticancer Res 8(5A):935–945
- 40. Hattori M, Minato N (2003) Rap1 GTPase: functions, regulation, and malignancy. J Biochem 134(4):479–484
- 41. Nagano I, Murakami T, Manabe Y, Abe K (2002) Early decrease of survival factors and DNA repair enzyme in spinal motor neurons of presymptomatic transgenic mice that express a mutant SOD1 gene. Life Sci 72(4–5):541–548
- Gems D, Partridge L (2001) Insulin/IGF signalling and ageing: seeing the bigger picture. Curr Opin Genet Dev 11(3):287–292
- Richardson A, Liu F, Adamo ML, Van Remmen H, Nelson JF (2004) The role of insulin and insulin-like growth factor-I in mammalian ageing. Best Pract Res Clin Endocrinol Metab 18(3):393–406
- Mathew R, Pal Bhadra M, Bhadra U (2017) Insulin/insulin-like growth factor-1 signalling (IIS) based regulation of lifespan across species. Biogerontology 18(1):35–53
- 45. Morris BJ, Donlon TA, He Q, Grove JS, Masaki KH, Elliott A et al (2014) Association analyses of insulin signaling pathway gene polymorphisms with healthy aging and longevity in Americans of Japanese ancestry. J Gerontol A Biol Sci Med Sci 69(3):270–273
- 46. Kolovou V, Diakoumakou O, Papazafiropoulou AK, Katsiki N, Fragopoulou E, Vasiliadis I et al (2018) Biomarkers and gene polymorphisms in members of long- and short-lived families: a longevity study. Open Cardiovasc Med J 12:59–70
- 47. Ryo M, Nakamura T, Kihara S, Kumada M, Shibazaki S, Takahashi M et al (2004) Adiponectin as a biomarker of the metabolic syndrome. Circ J 68(11):975–981
- Bik W, Baranowska B (2009) Adiponectin a predictor of higher mortality in cardiovascular disease or a factor contributing to longer life? Neuro Endocrinol Lett 30(2):180–184
- 49. Cardoso AL, Fernandes A, Aguilar-Pimentel JA, de Angelis MH, Guedes JR, Brito MA et al (2018) Towards frailty biomarkers: candidates from genes and pathways regulated in aging and age-related diseases. Ageing Res Rev 47:214–277

- Barron E, Lara J, White M, Mathers JC (2015) Blood-borne biomarkers of mortality risk: systematic review of cohort studies. PLoS One 10(6):e0127550. https://doi.org/10.1371/journal. pone.0127550
- 51. Dellago H, Bobbili MR, Grillari J (2017) MicroRNA-17-5p: at the crossroads of cancer and aging a mini-review. Gerontology 63(1):20–28
- 52. Du WW, Yang W, Fang L, Xuan J, Li H, Khorshidi A et al (2014) miR-17 extends mouse lifespan by inhibiting senescence signaling mediated by MKP7. Cell Death Dis 5:e1355. https://doi.org/10.1038/cddis.2014.305
- 53. Arai Y, Hirose N, Yamamura K, Shimizu K, Takayama M, Ebihara Y et al (2001) Serum insulin-like growth factor-1 in centenarians: implications of IGF-1 as a rapid turnover protein. J Gerontol A Biol Sci Med Sci 56(2):M79–M82
- 54. Pennington CALERIE Team, Heilbronn LK, de Jonge L, Frisard MI, JP DL, Larson-Meyer DE et al (2006) Effect of 6-month calorie restriction on biomarkers of longevity, metabolic adaptation, and oxidative stress in overweight individuals: a randomized controlled trial. JAMA 295(13):1539–1548
- 55. Everitt AV, Le Couteur DG (2007) Life extension by calorie restriction in humans. Ann N Y Acad Sci 1114:428–433
- 56. Lettieri-Barbato D, Giovannetti E, Aquilano K (2016) Effects of dietary restriction on adipose mass and biomarkers of healthy aging in human. Aging (Albany NY) 8(12): 3341–3355
- 57. Arai Y, Takayama M, Gondo Y, Inagaki H, Yamamura K, Nakazawa S et al (2008) Adipose endocrine function, insulin-like growth factor-1 axis, and exceptional survival beyond 100 years of age. J Gerontol A Biol Sci Med Sci 63(11):1209–1218
- Barbieri M, Paolisso G, Kimura M, Gardner JP, Boccardi V, Papa M et al (2009) Higher circulating levels of IGF-1 are associated with longer leukocyte telomere length in healthy subjects. Mech Ageing Dev 130(11–12):771–776
- 59. Vaziri H, Benchimol S (1996) From telomere loss to p53 induction and activation of a DNAdamage pathway at senescence: the telomere loss/DNA damage model of cell aging. Exp Gerontol 31(1–2):295–301
- 60. Djojosubroto MW, Choi YS, Lee HW, Rudolph KL (2003) Telomeres and telomerase in aging, regeneration and cancer. Mol Cells 15(2):164–175
- 61. Lenart P, Krejci L (2016) DNA, the central molecule of aging. Mutat Res 786:1-7
- 62. Stenholm S, Metter EJ, Roth GS, Ingram DK, Mattison JA, Taub DD et al (2011) Relationship between plasma ghrelin, insulin, leptin, interleukin 6, adiponectin, testosterone and longevity in the Baltimore Longitudinal Study of Aging. Aging Clin Exp Res 23(2):153–158
- Gonzalez-Covarrubias V, Beekman M, Uh HW, Dane A, Troost J, Paliukhovich I et al (2013) Lipidomics of familial longevity. Aging Cell 12(3):426–434
- 64. Montoliu I, Scherer M, Beguelin F, DaSilva L, Mari D, Salvioli S et al (2014) Serum profiling of healthy aging identifies phospho- and sphingolipid species as markers of human longevity. Aging (Albany NY) 6(1):9–25
- 65. Bookheimer SY, Renner BA, Ekstrom A, Li Z, Henning SM, Brown JA et al (2013) Pomegranate juice augments memory and FMRI activity in middle-aged and older adults with mild memory complaints. Evid Based Complement Alternat Med 2013:946298. https:// doi.org/10.1155/2013/946298
- 66. Semba RD, Ferrucci L, Bartali B, Urpí-Sarda M, Zamora-Ros R, Sun K et al (2014) Resveratrol levels and all-cause mortality in older community-dwelling adults. JAMA Intern Med 174(7):1077–1084
- 67. Chen YF, Wu CY, Kao CH, Tsai TF (2010) Longevity and lifespan control in mammals: lessons from the mouse. Ageing Res Rev 9 Suppl 1:S28–S35
- Brown-Borg HM, Bartke A (2012) GH and IGF1: roles in energy metabolism of long-living GH mutant mice. J Gerontol A Biol Sci Med Sci 67(6):652–660
- Bartke A, Westbrook R, Sun L, Ratajczak M (2013) Links between growth hormone and aging. Endokrynol Pol 64(1):46–52

- Carter CS, Ramsey MM, Ingram RL, Cashion AB, Cefalu WT, Wang ZQ et al (2002) Models of growth hormone and IGF-1 deficiency: applications to studies of aging processes and lifespan determination. J Gerontol A Biol Sci Med Sci 57(5):B177–B188
- 71. Parr T (1999) Insulin exposure and unifying aging. Gerontology 45(3):121-135
- Katic M, Kahn CR (2005) The role of insulin and IGF-1 signaling in longevity. Cell Mol Life Sci 62(3):320–343
- Amrit FR, May RC (2010) Younger for longer: insulin signalling, immunity and ageing. Curr Aging Sci 3(3):166–176
- 74. Selman C, Lingard S, Choudhury AI, Batterham RL, Claret M, Clements M et al (2008) Evidence for lifespan extension and delayed age-related biomarkers in insulin receptor substrate 1 null mice. FASEB J 22(3):807–188
- Taguchi A, Wartschow LM, White MF (2007) Brain IRS2 signaling coordinates life span and nutrient homeostasis. Science 317:369–372
- 76. Selman C, Tullet JM, Wieser D, Irvine E, Lingard SJ, Choudhury AI et al (2009) Ribosomal protein S6 kinase 1 signaling regulates mammalian life span. Science 326:140–144
- 77. Kim DH, Sarbassov DD, Ali SM, King JE, Latek RR, Erdjument-Bromage H et al (2002) mTOR interacts with raptor to form a nutrient-sensitive complex that signals to the cell growth machinery. Cell 110(2):163–175
- Hayashi AA, Proud CG (2007) The rapid activation of protein synthesis by growth hormone requires signaling through mTOR. Am J Physiol Endocrinol Metab 292(6):E1647–E1655
- Cheng Z, Tseng Y, White MF (2010) Insulin signaling meets mitochondria in metabolism. Trends Endocrinol Metab 21(10):589–598
- Harman D (1956) Aging: a theory based on free radical and radiation chemistry. J Gerontol 11(3):298–300
- Jang YC, Van Remmen H (2009) The mitochondrial theory of aging: insight from transgenic and knockout mouse models. Exp Gerontol 44(4):256–260
- Yan LJ, Levine RL, Sohal RS (1997) Oxidative damage during aging targets mitochondrial aconitase. Proc Natl Acad Sci U S A 94(21):11168–11172
- Edrey YH, Salmon AB (2014) Revisiting an age-old question regarding oxidative stress. Free Radic Biol Med 71:368–378
- 84. Van Remmen H, Ikeno Y, Hamilton M, Pahlavani M, Wolf N, Thorpe SR et al (2003) Lifelong reduction in MnSOD activity results in increased DNA damage and higher incidence of cancer but does not accelerate aging. Physiol Genomics 16(1):29–37
- Richters L, Lange N, Renner R, Treiber N, Ghanem A, Tiemann K et al (2006) Exerciseinduced adaptations of cardiac redox homeostasis and remodeling in heterozygous SOD2knockout mice. J Appl Physiol (1985) 111(5):1431–1440
- 86. Hoehn KL, Salmon AB, Hohnen-Behrens C, Turner N, Hoy AJ, Maghzal GJ et al (2009) Insulin resistance is a cellular antioxidant defense mechanism. Proc Natl Acad Sci U S A 106:17787–17792
- Yang H, Roberts LJ, Shi MJ, Zhou LC, Ballard BR, Richardson A et al (2004) Retardation of atherosclerosis by overexpression of catalase or both cu/Zn-superoxide dismutase and catalase in mice lacking apolipoprotein E. Circ Res 95(11):1075–1081
- Liu Y, Qi W, Richardson A, Van Remmen H, Ikeno Y, Salmon AB (2013) Oxidative damage associated with obesity is prevented by overexpression of CuZn- or Mn-superoxide dismutase. Biochem Biophys Res Commun 438(1):78–83
- Borg J, Chereul E (2008) Differential MRI patterns of brain atrophy in double or single transgenic mice for APP and/or SOD. J Neurosci Res 86(15):3275–3284
- 90. Thiruchelvam M, Prokopenko O, Cory-Slechta DA, Richfield EK, Buckley B, Mirochnitchenko O (2005) Overexpression of superoxide dismutase or glutathione peroxidase protects against the paraquat + maneb-induced parkinson disease phenotype. J Biol Chem 280(23):22530–22539
- Shen X, Zheng S, Metreveli NS, Epstein PN (2006) Protection of cardiac mitochondria by overexpression of MnSOD reduces diabetic cardiomyopathy. Diabetes 55(3):798–805

- 1 Pursuing Increased Longevity
- 92. Dumont M, Wille E, Stack C, Calingasan NY, Beal MF, Lin MT (2009) Reduction of oxidative stress, amyloid deposition, and memory deficit by manganese superoxide dismutase overexpression in a transgenic mouse model of Alzheimer's disease. FASEB J 23(8):2459–2466
- Heilbronn LK, Ravussin E (2003) Calorie restriction and aging: review of the literature and implications for studies in humans. Am J Clin Nutr 78(3):361–369
- Smith JV, Heilbronn LK, Ravussin E (2004) Energy restriction and aging. Curr Opin Clin Nutr Metab Care 7(6):615–622
- Mahoney LB, Denny CA, Seyfried TN (2006) Caloric restriction in C57BL/6J mice mimics therapeutic fasting in humans. Lipids Health Dis 5:13. https://doi.org/10.1186/1476-511X-5-13
- 96. Silberberg R (1972) Articular aging and osteoarthrosis in dwarf mice. Pathol Microbiol (Basel) 38(6):417–430
- Brown-Borg HM, Borg KE, Meliska CJ, Bartke A (1996) Dwarf mice and the aging process. Nature 384(6604):33. https://doi.org/10.1038/384033a0
- Bartke A, Brown-Borg HM, Bode AM, Carlson J, Hunter WS, Bronson RT (1998) Does growth hormone prevent or accelerate aging? Exp Gerontol 33(7–8):675–687
- Hauck S, Bartke A (2000) Effects of growth hormone on hypothalamic Catalase and Cu/Zn superoxide dismutase. Free Rad Biol Med 28(6):970–978
- Hunter WS, Croson WB, Bartke A, Gentry MV, Meliska CJ (1999) Low body temperature in long-lived Ames dwarf mice at rest and during stress. Physiol Behav 67(3):433–437
- 101. Borg KE, Brown-Borg HM, Bartke A (1995) Assessment of the primary adrenal cortical and pancreatic hormone basal levels in relation to plasma glucose and age in the unstressed Ames dwarf mouse. Proc Soc Exp Biol Med 210(2):126–133
- 102. Holder AT, Wallis M, Biggs P, Preece MA (1980) Effects of growth hormone, prolactin and thyroxine on body weight, somatomedin-like activity and in-vivo sulphation of cartilage in hypopituitary dwarf mice. J Endocrinol 85(1):35–47
- 103. Bordone L, Cohen D, Robinson A, Motta MC, van Veen E, Czopik A et al (2007) SIRT1 transgenic mice show phenotypes resembling calorie restriction. Aging Cell 6(6):759–767
- Bordone L, Guarente L (2005) Calorie restriction, SIRT1 and metabolism: understanding longevity. Nat Rev Mol Cell Biol 6(4):298–305
- 105. Chen D, Steele AD, Lindquist S, Guarente L (2005) Increase in activity during calorie restriction requires Sirt1. Science 310(5754):1641
- Accili D, Arden KC (2004) FoxOs at the crossroads of cellular metabolism, differentiation, and transformation. Cell 117(4):421–426
- 107. Albert PS, Riddle DL (1988) Mutants of Caenorhabditis elegans that form dauer-like larvae. Dev Biol 126(2):270–293
- 108. Gottlieb S, Ruvkun G (1994) daf-2, daf-16 and daf-23: genetically interacting genes controlling Dauer formation in Caenorhabditis elegans. Genetics 137(1):107–120
- 109. Kenyon C, Chang J, Gensch E, Rudner A, Tabtiang R (1993) A C. elegans mutant that lives twice as long as wild type. Nature 366(6454):461–464
- 110. Lakowski B, Hekimi S (1998) The genetics of caloric restriction in Caenorhabditis elegans. Proc Natl Acad Sci U S A 95(22):13091–13096
- 111. Panowski SH, Wolff S, Aguilaniu H, Durieux J, Dillin A (2007) PHA-4/Foxa mediates dietrestriction-induced longevity of C. elegans. Nature 447(7144):550–555
- 112. Fuchs S, Bundy JG, Davies SK, Viney JM, Swire JS, Leroi AM (2010) A metabolic signature of long life in Caenorhabditis elegans. BMC Biol 8:14. https://doi. org/10.1186/1741-7007-8-14
- 113. Altintas O, Park S, Lee SJ (2016) The role of insulin/IGF-1 signaling in the longevity of model invertebrates, C. elegans and D. melanogaster. BMB Rep 49(2):81–92
- 114. Ewald CY, Castillo-Quan JI, Blackwell TK (2018) Untangling longevity, dauer, and healthspan in Caenorhabditis elegans insulin/IGF-1-signalling. Gerontology 64(1):96–104
- 115. Cuong VT, Chen W, Shi J, Zhang M, Yang H, Wang N et al (2019) The anti-oxidation and antiaging effects of Ganoderma lucidum in Caenorhabditis elegans. Exp Gerontol 117:99–105. https://doi.org/10.1016/j.exger.2018.11.016

- 116. Meng F, Li J, Rao Y, Wang W, Fu Y (2018) Gengnianchun extends the lifespan of Caenorhabditis elegans via the insulin/IGF-1 signalling pathway. Oxidative Med Cell Longev 2018:4740739. https://doi.org/10.1155/2018/4740739
- 117. Kim SH, Kim BK, Park SK (2018) Selenocysteine mimics the effect of dietary restriction on lifespan via SKN-1 and retards age-associated pathophysiological changes in Caenorhabditis elegans. Mol Med Rep 18(6):5389–5398
- 118. Wollenhaupt SG, Soares AT, Salgueiro WG, Noremberg S, Reis G, Viana C et al (2014) Seleno- and telluro-xylofuranosides attenuate Mn-induced toxicity in C. elegans via the DAF-16/FOXO pathway. Food Chem Toxicol 64:192–199
- 119. Kim JS, Kim SH, Park SK (2017) Selenocysteine modulates resistance to environmental stress and confers anti-aging effects in C. elegans. Clinics (Sao Paulo) 72:491–498
- 120. Zhang Y, Zhang W, Dong M (2018) The miR-58 microRNA family is regulated by insulin signaling and contributes to lifespan regulation in Caenorhabditis elegans. Sci China Life Sci 61(9):1060–1070
- 121. Smith-Vikos T, Slack FJ (2012) MicroRNAs and their roles in aging. Cell Sci 125(Pt 1): 7–17
- 122. de Lencastre A, Pincus Z, Zhou K, Kato M, Lee SS, Slack FJ (2010) MicroRNAs both promote and antagonize longevity in C. elegans. Curr Biol 20(24):2159–2168
- 123. Hubbard EJ (2011) Insulin and germline proliferation in Caenorhabditis elegans. Vitam Horm 87:61–77
- 124. Klotz LO, Sánchez-Ramos C, Prieto-Arroyo I, Urbánek P, Steinbrenner H, Monsalve M (2015) Redox regulation of FoxO transcription factors. Redox Biol 6:51–72
- 125. http://genomics.senescence.info/species/entry.php?species=Chelonoidis\_nigra
- 126. https://www.smh.com.au/national/harriet-finally-withdraws-after-176-years-20060624-gdntnq.html
- 127. Loire E, Chiari Y, Bernard A, Cahais V, Romiguier J, Nabholz B et al (2013) Population genomics of the endangered giant Galápagos tortoise. Genome Biol 14(12):R136. https://doi. org/10.1186/gb-2013-14-12-r136
- 128. http://darwin-online.org.uk/content/frameset?itemID=F1925&viewtype=text&pageseq=1
- 129. Quesada V, Freitas-Rodríguez S, Miller J, Pérez-Silva JG, Jiang ZF, Tapia W et al (2019) Giant tortoise genomes provide insights into longevity and age-related disease. Nat Ecol Evol 3(1):87–95
- 130. Frigerio NA, Sacher GA (1968) The determination of whale life-spans. ANL-7535. ANL Rep:116–118
- 131. Austad SN (2010) Methusaleh's Zoo: how nature provides us with clues for extending human health span. J Comp Pathol 142(Suppl 1):S10–S21
- 132. George JC, Bada J, Zeh J, Scott L, Brown SE, O'Hara T et al (1999) Age and growth estimates of bowhead whales (Balaena mysticetus) via aspartic acid racemization. Can J Zool 77(4):571–580
- 133. George JC, Bockstoce JR (2008) Two historical weapon fragments as an aid to estimating the longevity and movements of bowhead whales. Polar Biol 31:751–754
- 134. https://www.medicaldaily.com/can-marine-biology-help-us-live-forever-bowhead-whalecan-live-200-years-cancer-316424
- 135. Philo LM, Shotts EB, George JC (1993) Morbidity and mortality. In: Burns JJ, Montague JJ, Cowles CJ (eds) The bowhead whale. Allen Press, Lawrence, KS, pp 275–312. ISBN-10: 0935868623
- 136. Tacutu R, Craig T, Budovsky A, Wuttke D, Lehmann G, Taranukha D et al (2013) Human ageing genomic resources: integrated databases and tools for the biology and genetics of ageing. Nucleic Acids Res 41(Database issue):D1027–D1033. https://doi.org/10.1093/nar/ gks1155
- 137. Speakman JR (2005) Body size, energy metabolism and lifespan. J Exp Biol 208(Pt 9): 1717–1730
- Blagosklonny MV (2013) Big mice die young but large animals live longer. Aging (Albany NY) 5(4):227–233

#### 1 Pursuing Increased Longevity

- 139. Kryazhimskiy S, Plotkin JB (2008) The population genetics of dN/dS. PLoS Genet 4(12):e1000304. https://doi.org/10.1371/journal.pgen.1000304
- 140. Yim HS, Cho YS, Guang X, Kang SG, Jeong JY, Cha SS et al (2014) Minke whale genome and aquatic adaptation in cetaceans. Nat Genet 46(1):88–92
- 141. Cornu M, Albert V, Hall MN (2013) mTOR in aging, metabolism, and cancer. Curr Opin Genet Dev 23(1):53–62
- 142. Erez A, Nagamani SC, Shchelochkov OA, Premkumar MH, Campeau PM, Chen Y et al (2011) Requirement of argininosuccinate lyase for systemic nitric oxide production. Nat Med 17(12):1619–1626
- 143. Yano K, Stevens JD, Compagno LJV (2007) Distribution, reproduction and feeding of the Greenland shark Somniosus (Somniosus) microcephalus, with notes on two other sleeper sharks, Somniosus (Somniosus) pacificus and Somniosus (Somniosus) antarcticus. J Fish Biol 70:374–390. https://doi.org/10.1111/j.1095-8649.2007.01308.x
- 144. MacNeil MA, McMeans BC, Hussey NE, Vecsei P, Svavarsson J, Kovacs KM et al (2012) Biology of the Greenland shark Somniosus microcephalus. J Fish Biol 80(5):991–1018
- 145. Nielsen J, Hedeholm RB, Heinemeier J, Bushnell PG, Christiansen JS, Olsen J et al (2016) Eye lens radiocarbon reveals centuries of longevity in the Greenland shark (Somniosus microcephalus). Science 353(6300):702–704
- 146. Costantini D, Smith S, Killen SS, Nielsen J, Steffensen JF (2017) The Greenland shark: a new challenge for the oxidative stress theory of ageing? Comp Biochem Physiol A Mol Integr Physiol 203:227–232
- 147. https://www.sciencedaily.com/releases/2016/08/160811143218.htm
- 148. Strahl J, Philipp EE, Brey T, Broeg K, Abele D (2007) Physiological aging in the Icelandic population of the ocean quahog Arctica islandica. Aquat Biol 1:77–84
- Ridgway ID, Richardson CA (2011) Arctica islandica: the longest lived non colonial animal known to science. Rev Fish Biol Fisheries 21:297. https://doi.org/10.1007/s11160-010-9171-9
- 150. Wanamaker AD, Heinemeier J, Scourse JD, Richardson CA (2008) Very long-lived molluscs confirm 17th century AD tephra-based radiocarbon reservoir ages for north Icelandic shelf waters. Radiocarbon 50:1–14
- 151. Butler PG, Wanamaker ADJ, Scourse JD, Richardson CA, Reynolds DJ (2013) Variability of marine climate on the North Icelandic Shelf in a 1357-year proxy archive based on growth increments in the bivalve Arctica islandica. Palaeogeogr Palaeoclimatol Palaeoecol 373:141–151
- 152. Abele D, Strahl J, Brey T, Philipp EE (2008) Imperceptible senescence: ageing in the ocean quahog Arctica islandica. Free Radic Res 42:474–480
- 153. Ungvari Z, Ridgway I, Philipp EE, Campbell CM, McQuary P, Chow T et al (2011) Extreme longevity is associated with increased resistance to oxidative stress in Arctica islandica, the longest-living non-colonial animal. J Gerontol A Biol Sci Med Sci 66(7):741–750
- 154. Munro D, Blier PU (2012) The extreme longevity of Arctica islandica is associated with increased peroxidation resistance in mitochondrial membranes. Aging Cell 11(5):845–855
- 155. Ungvari Z, Sosnowska D, Mason JB, Gruber H, Lee SW, Schwartz TS et al (2013) Resistance to genotoxic stresses in Arctica islandica, the longest living noncolonial animal: is extreme longevity associated with a multistress resistance phenotype? J Gerontol A Biol Sci Med Sci 68(5):521–529
- 156. Sosnowska D, Richardson C, Sonntag WE, Csiszar A, Ungvari Z, Ridgway I (2014) A heart that beats for 500 years: age-related changes in cardiac proteasome activity, oxidative protein damage and expression of heat shock proteins, inflammatory factors, and mitochondrial complexes in Arctica islandica, the longest-living noncolonial animal. J Gerontol A Biol Sci Med Sci 69(12):1448–1461
- 157. https://immortal-jellyfish.com/
- 158. https://thebiologist.rsb.org.uk/biologist/158-biologist/features/1510-everlasting-life-theimmortal-jellyfish
- 159. Carla' EC, Pagliara P, Piraino S, Boero F, Dini L (2003) Morphological and ultrastructural analysis of Turritopsis nutricula during life cycle reversal. Tissue Cell 35(3):213–222

- 160. Petralia RS, Mattson MP, Yao PJ (2014) Aging and longevity in the simplest animals and the quest for immortality. Ageing Res Rev 16:66–82
- 161. Tanaka EM, Reddien PW (2011) The cellular basis for animal regeneration. Dev Cell 21(1):172–185
- 162. Devarapalli P, Kumavath RN, Barh D, Azevedo V (2014) The conserved mitochondrial gene distribution in relatives of Turritopsis nutricula, an immortal jellyfish. Bioinformation 10(9):586–591
- 163. Lisenkova AA, Grigorenko AP, Tyazhelova TV, Andreeva TV, Gusev FE, Manakhov AD et al (2017) Complete mitochondrial genome and evolutionary analysis of Turritopsis dohrnii, the "immortal" jellyfish with a reversible life-cycle. Mol Phylogenet Evol 107:232–238

## Chapter 2 Effects on Glial Cell Glycolysis in Schizophrenia: An Advanced Aging Phenotype?



Giuliana S. Zuccoli, Paul C. Guest, and Daniel Martins-de-Souza

#### 1 Introduction

Schizophrenia is a psychiatric disorder known to affect approximately 1% of the world population. It is manifested through a wide range of severe symptoms and onset occurs typically in late adolescence or early adulthood. However, the origins of schizophrenia are likely to be found early in development, long before the onset of symptoms [1, 2]. Over the years, many studies have been conducted in attempts to understand this disorder and it has been proposed that the disease occurs due to small flaws in several brain areas rather than more global damage in particular brain regions. Therefore, the disorder could arise from defective connections between components of the neural system [3].

Imaging, proteomic and genetic studies have found evidences of alterations in metabolic processes related to energy pathways in patients suffering from this disorder and, given the importance of proper functioning of energy handling in the brain (Fig. 2.1), this has been accepted as one important feature of schizophrenia [4–6]. In fact, the findings of proteomic profiling studies have suggested that, in the context of energy metabolism, glycolysis is the predominantly affected pathway in schizophrenia (Fig.2.2) [7].

Instituto Nacional de Biomarcadores em Neuropsiquiatria (INBION) Conselho Nacional de Desenvolvimento Científico e Tecnologico, São Paulo, Brazil

UNICAMP's Neurobiology Center, Campinas, Brazil e-mail: dmsouza@unicamp.br

G. S. Zuccoli · P. C. Guest

Laboratory of Neuroproteomics, Department of Biochemistry and Tissue Biology, Institute of Biology, University of Campinas (UNICAMP), Campinas, Brazil

D. Martins-de-Souza (🖂)

Laboratory of Neuroproteomics, Department of Biochemistry and Tissue Biology, Institute of Biology, University of Campinas (UNICAMP), Campinas, Brazil

<sup>©</sup> Springer Nature Switzerland AG 2019

P. C. Guest (ed.), *Reviews on Biomarker Studies in Aging and Anti-Aging Research*, Advances in Experimental Medicine and Biology 1178, https://doi.org/10.1007/978-3-030-25650-0\_2



**Fig. 2.1** Schematic representation of glucose handling during neurodevelopment and in the adult brain, showing the importance of cooperation between neurons, astrocytes and oligodendrocytes in the context of energy metabolism. ATP molecules marked in red represents the pathway that is the major source of energy: Glycolysis for neural progenitor cells, astrocytes and oligodendrocytes; Oxidative phosphorylation for neurons (cell body and axons)



Fig. 2.2 Evidence from proteomic studies showing compromised glycolysis in schizophrenia. Proteins in green represent those found to be altered in postmortem brain tissue from schizophrenia patients, whereas those in blue represent proteins found to be altered in cell-culture models of the disorder

Glucose is the obligatory energy substrate of the brain and this organ accounts for approximately one quarter of the whole-body glucose utilization despite the fact that it only accounts for 2% of the total body weight, [8]. In addition, glucose is an essential component of macromolecules such as glycolipids and glycoproteins in neural cells, and it enters the metabolic pathways that result in the synthesis of three key neurotransmitters of the brain, glutamate, GABA and acetylcholine [9].

Glucose is sequentially processed by glycolysis in the cell cytosol and processing through the tricarboxylic acid (TCA) cycle and oxidative phosphorylation takes place in the mitochondria [10]. The main energy-consuming process of the brain is the maintenance of ionic gradients across the plasma membrane, a condition that is crucial for excitability. Maintenance of these gradients is achieved predominantly through the activity of ionic pumps fueled by ATP [11]. Neurons represent approximately 80% the brain energy requirements whereas glia account for around 10% [12]. Axonal transport of molecules synthesized in the cell body along the axon is another process fueled by cellular energy metabolism [10].

Given the cellular heterogeneity of the brain, which is composed of neurons and glial cells, is important to note that different cell types have distinct metabolic profiles. In fact, neurons are predominantly oxidative whereas glycolysis predominates in astrocytes and oligodendrocytes [13, 14]. Hence, astrocytes and oligodendrocytes are responsible for providing lactate to the neuronal cell body and axon, respectively, where it can be further metabolized and provide ATP to insure the maintenance of neuronal function [15].

Given the evidences of compromised glycolysis in schizophrenia, is important to understand energy metabolism from a glial perspective, as opposed to a neuronalcentric view. In addition, metabolism in neuronal cells is altered in the aging process and various regions of the brain are thought to undergo accelerated aging in individuals with schizophrenia compred with healthy controls. Therefore, we present the role of glial cells in energy metabolism of the brain and consider the evidence for the advanced aging theory of schizophrenia in these cells. Focussed efforts in this area could lead to identification of novel biomarkers and potential drug targets for improved treatment of individuals with schizophrenia and other psychiatric or neurodegenerative disorders.

#### 2 The Role of Glia Cells in Brain Energy Metabolism

#### 2.1 Astrocytes

Astrocytes are the most abundant glial cells in the brain, exceeding the number of neurons by five fold. Astrocytes are involved in many brain functions, such as the formation of the blood-brain barrier (BBB), taking up neurotransmitters at glutamatergic synapses, immune function support, composing tripartite synapses with pre and post-synaptic neurons and providing energy support to neurons [16, 17].
Astrocytes are essentially the only cells containing glycogen in the adult central nervous system (CNS), and glycogen metabolism followed by glycolysis provides a source of lactate to other cells [18]. The most accepted hypothesis to explain this supply of lactate is the astrocyte-neuron lactate shuttle (ANLS) [19]. This hypothesis postulates that upon glutamate release from the presynaptic terminal of a neuron, the neurotransmitter activates the astrocyte membrane transporters and/or receptors leading to the transport of glucamate into astrocytes. This uptake of glutamate leads to the activation of glucose utilization in astrocytes and glucose uptake from the circulation through the glucose transporter GLUT1 [15]. Upon the synthesis of astrocyte-derived lactate via glycolysis, it diffuses out via the astrocyte monocarboxylate transporter types 1 and 4 (MCT-1 and has a high affinity for lactate [20]. Once inside the neuron cell body, lactate undergoes subsequent metabolic reactions in the tricarboxylic acid (TCA) cycle and oxidative phosphorylation to produce more ATP and other substances required for neuronal functioning (Fig. 2.1) [21].

In addition, studies of both astrocytes and oligodendrocytes have demonstrated that astrocytes also transfer energy metabolites directly to those cells. Connections between astrocytes and myelinating cells occur via gap junctions formed by connexins in the plasma membranes of both adjacent cells [22]. Those connexins form channels that promote the exchange of molecules, including lactate, between connected cells [23, 24]. Also, the formation of gap junctions between astrocytes and oligodendrocytes have an important role in K+ release during neuronal activity and in buffering extracellular glutamate [25].

Studies using a coculture model of astrocytes and neurons have shown that the enhancement of the astrocyte-neuron lactate shuttle process through over-expressing GLUT1 in astrocytes and MCT2 in neurons leads to neuroprotection against excito-toxic insults [26]. In contrast, the impairment of lactate delivery to neurons leads to neuronal damage. Therefore, lactate transfer from astrocytes to neurons seems to exert an important neuroprotective role and possibly indicates a therapeutic strategy under glucose-deprivation conditions and against neuronal insults associated with intracellular energy deficiency such as excitotoxicity [15].

# 2.2 Oligodendrocytes

Oligodendrocytes are the most abundant cell type in white matter and are responsible for providing protection and electrical insulation of axons, through myelination. Neurons may have axonal processes that exceed the length of the neuronal body and this represents a challenge when considering electrical impulse conduction, transport of molecules to the axon end and metabolic maintenance [27]. In this way, oligodendrocytes are crucial to circumvent those possible limiting factors, given that the myelination of axons allows a saltatory nerve conduction, in which neuronal action potentials are restricted to the nodes of Ranvier, increasing conduction speeds by up to 100-fold [28]. As mentioned above, there must be a metabolic support for axons of energy substrates in order to assure axonal propagation of action potentials and rapid transport along axons. This important task is fulfilled by oligodendrocytes since the myelin sheath can be considered metabolically active, allowing the movement of molecules to the subjacent axon [29–31]. The metabolic coupling between oligodendrocytes and axons begins with the transfer of lactate or pyruvate from oligodendrocytes to the peri axonal space via MCT-1. Following this, lactate or pyruvate can enter axons through MCT-2 and further metabolized to then be processed in the axonal mitochondria for ATP generation (Fig. 2.1) [32, 33]. This metabolic coupling assures axonal and neuronal health and survival, as studies showed that in both cell-culture and animal models, the disruption of MCT-1 expression in oligodendrocytes resulted in impairments in the delivery of lactate, leading to axonal damage and neuronal loss [22, 32, 33].

One important aspect to consider is that the constant over-production of lactate would result in lactic acidosis, which could be detrimental for white matter function [34]. Following the discovery that oligodendrocytes have functional NMDA receptors [35, 36], there was much interest in understanding the function of these receptors. NMDA receptor activation in oligodendrocytes promotes the increased incorporation of GLUT1 into the membranes of oligodendrocytes and myelin and enhances glucose import into those cells. Therefore, it has been hypothesized that a mechanism to prevent exacerbated production of lactate could be that oligodendrocytes adapt their energy metabolism based on activity-dependent glutamate release [37].

### **3** Astrocytes and Schizophrenia

Astrocytes have been associated with neuroinflammation in some brain areas [38] and gene sets from these cells have been associated with increased risk for schizophrenia [39]. In addition, gene expression analysis of postmortem brain tissue from schizophrenia patients has shown alterations in the expression levels of glial fibrillary acid protein (GFAP), which is an astrocyte marker [40, 41]. S100B is an astrocytic protein that has been implicated in the regulation of several intracellular activities and is used as a marker of astrocyte function [42]. Increased levels of this protein have been observed in serum [43] and cerebrospinal fluid [44] of schizophrenia patients and high amounts of S100B have been hypothesized to be predictive for negative symptoms [45].

The administration of MK-801 in animals has been proposed as one possible model of schizophrenia and reduced levels of glutamine and glutamate derived from acetate have been observed, indicating changes in astrocytes metabolism upon treatment with this drug [46]. One study that analyzed the effects of acute and long-term treatment of MK-801 on an astrocyte cell line revealed that the levels of aldolase C were affected under acute conditions and aldolase C and hexokinase 1 levels were altered following long-term treatment, both being enzymes involved in glycolysis (Fig. 2.2) [47].

The protein disrupted in schizophrenia-1 (DISC-1) is expressed in astrocytes and a study showed that a mutated DISC-1 protein was associated with decreased levels of d-serine, which is a co-ligand of NMDA receptors [48]. The co-culture of DISC-1 mutated astrocytes with neurons promoted deficits in synaptic and dendritic maturation, and this phenotype was reversed after d-serine treatment [49]. In addition, one study observed that treatment of a primary astrocyte culture with the antipsychotic drug clozapine increased d-serine levels released by these cells, which subsequently enhanced the glutamate release by neurons [50]. These results indicate that astrocytes could be a target cell in the treatment of schizophrenia. In a recent study, the knockdown of endogenous DISC-1 in primary astrocytes resulted in decreased levels of GLUT4, reduced glucose uptake and diminished production of lactate by astrocytes [51]. In addition, the same study showed that the expression of a truncated version of the DISC-1 protein in mice resulted in metabolic perturbations associated with altered behavior and memory deficits, and this phenotype was reversed upon systemic treatment with lactate. Hence, there is the indication that alterations in astrocytes could lead to abnormalities in energy supply and may be associated with some aspects of schizophrenia.

In the context of neurodevelopment, astrogenesis is a major feature during brain development and one recurrent finding using mouse models of early stress has been deficits in glial density [52]. Astrocytes are derived from neural stem cells and play an important role in regulating neural development by coordinating synapse formation and function, neuronal survival and axonal guidance [53]. Given the important role of these cells, the high energetic demand during neurodevelopment (Fig. 2.1) and evidences of compromised glycolysis in schizophrenia, studies using iPSC derived astrocytes could bring insights into the underlying mechanisms of astrogenesis during neurodevelopment in the context of schizophrenia.

### 4 Oligodendrocytes and Schizophrenia

Several studies have found abnormalities related to functional and structural dysconnectivity in schizophrenia [54–56] and it has been hypothesized that oligodendrocyte dysfunction with subsequent abnormalities in myelin maintenance and repair could lead to the disrupted connectivity related to the disorder [57, 58]. Imaging investigations have shown decreased white matter in schizophrenia patients [59, 60] and postmortem studies have found alterations related to morphology and density of oligodendrocytes from schizophrenia patients, which suggests disturbances in energy metabolism in these cells and possible effects on axonal integrity [61–64]. Microarray analysis of postmortem dorsolateral prefrontal cortex from schizophrenia patients has revealed that oligodendrocyte gene transcripts are specifically reduced in the disorder [65]. Some of these are involved in myelination, such as myelin-associated glycoprotein (MAG), transferrin and myelin and lymphocyte protein (MAL). In addition to the altered myelinating genes, CNP1 was also found to be dysregulated and this gene encodes an oligodendrocyte-specific protein located in the myelin sheath that is involved in oligodendrocyte differentiation and in maintaining axon function and survival [66].

Upon treatment of different cell lines with MK-801, which acts on the glutamatergic system and has been proposed as a pharmacological means of modeling schizophrenia, it was found that this glutamatergic manipulation resulted in significant alterations in proteins related to glycolysis in oligodendrocyte cell lines. Alterations in the levels of hexokinase 1, enolase 2, phosphoglycerate kinase, phosphoglycerate mutase 1 and triosephosphate isomerase have been observed following acute and long-term treatment (Fig. 2.2) [47, 67]. In addition, the subsequent treatment of these MK-801-treated oligodendrocytes with the antipsychotic drug clozapine reversed some of these glycolytic alterations, which leads to the possibility that certain antipsychotics may act as modulators of glycolysis in oligodendrocytes and thereby improve neuronal connectivity [68].

Given the neurodevelopmental origin of schizophrenia hypothesis, it has been hypothesized that early insults to the developing brain could promote an inflammatory environment, cause injuries to oligodendroglial progenitor cells and lead to the white matter abnormalities found in the disorder [69, 70]. Progenitor cells have a high division rate and, therefore, present an elevated energy demand, which is achieved by glycolysis (Fig. 2.1) [71, 72]. They present a complex program of differentiation until the formation of myelinating mature cells with several particularities related to their metabolism and physiology [73]. Therefore, alterations in energy metabolism in progenitor cells could have detrimental effects on oligodendrogenesis not only in the developing brain but in the adult brain as well, as they can differentiate into mature oligodendrocytes and promote re-myelination. In this way, the possibility to use human induced pluripotent stem cell (hiPSC) technology combined with the generation of oligodendrocyte progenitor cells to study the metabolic profile of progenitor cells in the context of schizophrenia could help to elucidate the role of those cells in the development of the disease [74, 75].

# 5 Effect of Aging on Oligodendrocyte and Astrocyte Metabolism

It is well known that the aging brain shows deficits in metabolic pathways such as those linked to glycolysis [76]. Aging appears to be associated with deteriorating systemic regulation of glucose metabolism which is linked in turn to declining brain glucose metabolism (Fig. 2.3). In addition, the progressive brain loss in schizophrenia appears to reflect an accelerated aging phenotype in various regions of the brain [77]. This has been seen through identification of a number of aging-related biomarkers in patients such as shorter telomere length in some brain areas and in various neuronal cell types [78]. Studies have also shown that telomere shortening occurs in peripheral blood cells in chronic stress [79], mood disorders [80] and schizophrenia [81], thus providing a potential peripheral biomarker of these diseases.



Fig. 2.3 The advanced aging hypothesis of schizophrenia: effects of disrupted glycolysis, oxidative damage and accelerated telomere shortening in glial cells of the brain

Telomeres are comprised of DNA and proteins at the end of each chromosome and these structures prevent the loss of genetic material during each cycle of cell replication. In differentiated cells, the telomeres shorten with each replicative division. This eventually induces DNA damage, leading to growth arrest and, ultimately, replicative senescence [82–84]. In undifferentiated stem cells, germ cells and progenitor cells, the loss of telomeres is protected by the enzyme telomerase, which adds TTAGGG repeats onto the ends of the chromosomes [85–89]. Experiments in telomerase-deficient mice provided the first proof that telomere erosion is a determinant of longevity and the occurrence of age-related pathologies [90, 91].

Studies have shown that schizophrenia patients exhibit an initial loss of gray matter followed by a deficit in white matter that progressively worsens with age [92]. Thus, the losses in white matter are likely to be linked more to aging and the effects on gay matter occur during the early stages of the disease in younger patients. Aging can cause a loss of normal astrocyte or oligodendrocyte function which reduces the ability of these cells to maintain a healthy environment in the brain [93, 94]. Culturing the oligodendrocyte cell line OLN-93 in the presence of glucose was

found to induce an aging phenotype [95]. This was seen as an elongation and thickening of cell processes, shortening of telomeres and increased expression of netrin. Consistent with our understanding of telomere shortening, this reflected a more mature state of oligodendrocyte development or differentiation. Furthermore, a laser capture micro-dissection study found an attenuated oxidative stress defence and deficient telomerase activity, contributing to telomere shortening in oligodendrocytes in major depressive disorder patients [96]. This suggests an aetiological link between telomere shortening and white matter abnormalities in schizophrenia, which may also occur in other psychiatric disorders. This is consistent with earlier studies which showed that differentiation of cultured oligodendrocyte precursor cells into mature oligodendrocytes is associated with lower activities of the telomerase enzyme [97]. These findings indicate that the population of neural stem cells that give rise to oligodendrocytes and astrocytes may be lower in patients with certain psychiatric or neurodegenerative disorders.

### 6 Conclusions

Many studies have suggested that energy metabolism dysfunction is an important feature of psychiatric disorders and the glycolysis pathway is believed to be the most affected in schizophrenia. In brain energy metabolism, glial cells are predominantly glycolytic and provide energy to the neuronal cell body and axon, respectively. Moreover, glial cells have important roles in neural development and in the adult nervous system and have surpassed their original definition as supportive cells. Thus, biomarkers associated with these cell types such as components of glycolysis and energy metabolism pathways may be useful in disease modelling and drug development studies. Finally, the hypothesis that schizophrenia may be seen as an advanced aging phenotype in some regions of the brain and neuronal cells such as glial cells is supported by the finding of accelerated telomere erosion in astrocytes and oligodendrocytes from schizophrenia patients. This may also be linked to advanced oxidative damage in these cells, consistent with the findings of studies on other diseases marked by accelerated telomere erosion [98]. Thus, studies of other biomarkers of the advanced aging phenotype, oxidative damage and disrupted glycolysis and energy production pathways may provide further useful information in studies of schizophrenia in both the acute and chronic stages (Fig. 2.3). In conclusion, evidence of glycolytic and aging-related alterations in astrocytes and oligodendrocytes in schizophrenia indicate the importance of studying this disorder from a glial-neuronal cell perspective.

Acknowledgements The authors are supported by FAPESP (São Paulo Research Foundation; grants 2017/25588-1, 2018/14666-4), Serrapilheira Institute (grant Serra-1709-16349), and CNPq (Brazilian National Council for Scientific and Technological Development; grant 302453/2017-2)

# References

- 1. Weinberger DR (1987) Implications of normal brain development for the pathogenesis of schizophrenia. Arch Gen Psychiatry 44(7):660–669
- Schmidt MJ, Mirnics K (2015) Neurodevelopment, GABA system dysfunction, and schizophrenia. Neuropsychopharmacology 40(1):190–206
- Fornito A, Harrison BJ (2012) Brain connectivity and mental illness. Front Psych 3:72. https:// doi.org/10.3389/fpsyt.2012.00072
- Cleghorn JM, Garnett ES, Nahmias C, Firnau G, Brown GM, Kaplan R et al (1989) Increased frontal and reduced parietal glucose metabolism in acute untreated schizophrenia. Psychiatry Res 28(2):119–133
- Hazlett EA, Buchsbaum MS, Kemether E, Bloom R, Platholi J, Brickman AM et al (2004) Abnormal glucose metabolism in the mediodorsal nucleus of the thalamus in schizophrenia. Am J Psychiatry 161(2):305–314
- Haznedar MM, Buchsbaum MS, Hazlett EA, Shihabuddin L, New A, Siever LJ (2004) Cingulate gyrus volume and metabolism in the schizophrenia spectrum. Schizophr Res 71(2–3):249–262
- Zuccoli GS, Saia-Cereda VM, Nascimento JM, Martins-de-Souza D (2017) The energy metabolism dysfunction in psychiatric disorders postmortem brains: focus on proteomic evidence. Front Neurosci 11:1–14
- Bélanger M, Allaman I, Magistretti PJ (2011) Differential effects of pro- and anti-inflammatory cytokines alone or in combinations on the metabolic profile of astrocytes. J Neurochem 116(4):564–576
- Deutch AY, Roth RH (2004) Pharmacology and biochemistry of synaptic transmission: classic transmitters. In: From molecules to networks (third edition): an introduction to cellular and molecular neuroscience. Academic Press Inc; Cambridge, MA, USA, pp 207–237. ISBN: 978-0-12-397179-1
- Magistretti PJ (2004) Brain energy metabolism. In: From molecules to networks (third edition): an introduction to cellular and molecular neuroscience. Academic Press Inc; Cambridge, MA, USA, pp 67–89. ISBN: 978-0-12-397179-1
- Magistretti PJ, Allaman I (2013) Brain energy metabolism. In: Pfaff DW, Volkow ND (eds) Neuroscience in the 21st century, 2nd edn. Springer; New York, NY, USA, pp 1591–1620. ISBN-10: 1493934732
- 12. Howarth C, Gleeson P, Attwell D (2012) Updated energy budgets for neural computation in the neocortex and cerebellum. J Cereb Blood Flow Metab 32(7):1222–1232
- Hyder F, Patel AB, Gjedde A, Rothman DL, Behar KL, Shulman RG (2006) Neuronalglial glucose oxidation and glutamatergic-GABAergic function. J Cereb Blood Flow Metab 26(7):865–877
- 14. Zhang Y, Chen K, Sloan SA, Bennett ML, Scholze AR, O'Keeffe S et al (2014) An RNA-sequencing transcriptome and splicing database of glia, neurons, and vascular cells of the cerebral cortex. J Neurosci 34(36):11929–11947
- Magistretti PJ, Allaman I (2015) A cellular perspective on brain energy metabolism and functional imaging. Neuron 86(4):883–901
- Perea G, Navarrete M, Araque A (2009) Tripartite synapses: astrocytes process and control synaptic information. Trends Neurosci 32(8):421–431
- 17. Sofroniew MV, Vinters HV (2010) Astrocytes: biology and pathology. Acta Neuropathol 119(1):7–35
- Brown AM, Ransom BR (2007) Astrocyte glycogen and brain energy metabolism. Glia 55(12):1263–1271
- Pellerin L, Magistretti PJ (1994) Glutamate uptake into astrocytes stimulates aerobic glycolysis: a mechanism coupling neuronal activity to glucose utilization. Proc Natl Acad Sci U S A 91(22):10625–10629

- Pierre K, Pellerin L (2005) Monocarboxylate transporters in the central nervous system: distribution, regulation and function. J Neurochem 94(1):1–14
- 21. Magistretti PJ (2006) Neuron-glia metabolic coupling and plasticity. J Exp Biol 209(12):2304-2311
- 22. Correale J, Ysrraelit MC, Benarroch EE (2018) Metabolic coupling of axons and glial cells: implications for multiple sclerosis progression. Neurology 90(16):737–744
- Orthmann-Murphy JL, Abrams CK, Scherer SS (2008) Gap junctions couple astrocytes and oligodendrocytes. J Mol Neurosci 35(1):101–116
- Suzuki A, Stern SA, Bozdagi O, Huntley GW, Walker RH, Magistretti PJ et al (2011) Astrocyteneuron lactate transport is required for long-term memory formation. Cell 144(5):810–823
- Ransom B, Behar T, Nedergaard M (2003) New roles for astrocytes (stars at last). Trends Neurosci 26(10):520–522
- 26. Bliss TM, Ip M, Cheng E, Minami M, Pellerin L, Magistretti P et al (2004) Dual-gene, dualcell type therapy against an excitotoxic insult by bolstering neuroenergetics. J Neurosci 24(27):6202–6208
- Trevisiol A, Nave KA (2015) Brain energy metabolism: conserved functions of glycolytic glial cells. Cell Metab 22(3):361–363
- Nave KA, Werner HB (2014) Myelination of the nervous system: mechanisms and functions. Annu Rev Cell Dev Biol 30:503–533
- Fünfschilling U, Supplie LM, Mahad D, Boretius S, Saab AS, Edgar J et al (2012) Glycolytic oligodendrocytes maintain myelin and long-term axonal integrity. Nature 485(7399):517–521
- Lee Y, Morrison BM, Li Y, Lengacher S, Farah MH, Hoffman PN et al (2012) Oligodendroglia metabolically support axons and contribute to neurodegeneration. Nature 487(7408):443–448
- Simons M, Nave K (2015) Oligodendrocytes: myelination and axonal support. Cold Spring Harb Perspect Biol 8(1):a020479. https://doi.org/10.1101/cshperspect.a020479
- Morrison BM, Lee Y, Rothstein JD (2013) Oligodendroglia: metabolic supporters of axons. Trends Cell Biol 23(12):644–651
- Saab AS, Tzvetanova ID, Nave KA (2013) The role of myelin and oligodendrocytes in axonal energy metabolism. Curr Opin Neurobiol 23(6):1065–1072
- Young PR, Snyder WR, Vacante DA, Waickus CM, Zygas AP, Grynspan F et al (1988) The acid instability of myelin. A model for myelin degeneration in multiple sclerosis. Med Hypotheses 26(1):31–37
- Káradóttir R, Cavelier P, Bergersen LH, Attwell D (2005) NMDA receptors are expressed in oligodendrocytes and activated in ischaemia. Nature 438(7071):1162–1166
- Salter MG, Fern R (2005) NMDA receptors are expressed in developing oligodendrocyte processes and mediate injury. Nature 438(7071):1167–1171
- 37. Saab AS, Tzvetavona ID, Trevisiol A, Baltan S, Dibaj P, Kusch K et al (2016) Oligodendroglial NMDA receptors regulate glucose import and axonal energy metabolism. Neuron 91(1):119–132
- Takahashi N, Sakurai T (2013) Roles of glial cells in schizophrenia: possible targets for therapeutic approaches. Neurobiol Dis 53:49–60
- Goudriaan A, de Leeuw C, Ripke S, Hultman CM, Sklar P, Sullivan PF et al (2014) Specific glial functions contribute to Schizophrenia susceptibility. Schizophr Bull 40(4):925–935
- 40. Sugai T, Kawamura M, Iritani S, Araki K, Makifuchi T, Imai C et al (2004) Prefrontal abnormality of schizophrenia revealed by DNA microarray: impact on glial and neurotrophic gene expression. Ann N Y Acad Sci 1025:84–91
- Webster MJ, O'Grady J, Kleinman JE, Weickert CS (2005) Glial fibrillary acidic protein mRNA levels in the cingulate cortex of individuals with depression, bipolar disorder and schizophrenia. Neuroscience 133(2):453–461
- 42. Brozzi F, Arcuri C, Giambanco I, Donato R (2009) S100B protein regulates astrocyte shape and migration via interaction with Src kinase: implications for astrocyte development, activation, and tumor growth. J Biol Chem 284(13):8797–8811

- 43. Schroeter ML, Abdul-Khaliq H, Frühauf S, Höhne R, Schick G, Diefenbacher A et al (2003) Serum S100B is increased during early treatment with antipsychotics and in deficit schizophrenia. Schizophr Res 62(3):231–236
- 44. Rothermundt M, Falkai P, Ponath G, Abel S, Bürkle H, Diedrich M et al (2004) Glial cell dysfunction in schizophrenia indicated by increased S100B in the CSF. Mol Psychiatry 9(10):897–899
- 45. Rothermundt M, Ponath G, Glaser T, Hetzel G, Arolt V (2004) S100B serum levels and long-term improvement of negative symptoms in patients with schizophrenia. Neuropsychopharmacology 29(5):1004–1011
- 46. Kondziella D, Brenner E, Eyjolfsson EM, Markinhuhta KR, Carlsson ML, Sonnewald U (2006) Glial-neuronal interactions are impaired in the schizophrenia model of repeated MK801 exposure. Neuropsychopharmacology 31(9):1880–1887
- 47. Guest PC, Iwata K, Kato TA, Steiner J, Schmitt A, Turck CW et al (2015) MK-801 treatment affects glycolysis in oligodendrocytes more than in astrocytes and neuronal cells: insights for schizophrenia. Front Cell Neurosci 9:180. https://doi.org/10.3389/fncel.2015.00180
- Ma TM, Abazyan S, Abazyan B, Nomura J, Yang C, Seshadri S et al (2013) Pathogenic disruption of DISC1-serine racemase binding elicits schizophrenia-like behavior via D-serine depletion. Mol Psychiatry 18(5):557–567
- 49. Xia M, Zhu S, Shevelkin A, Ross CA, Pletnikov M (2016) DISC1, astrocytes and neuronal maturation: a possible mechanistic link with implications for mental disorders. J Neurochem 138:518–524
- Tanahashi S, Yamamura S, Nakagawa M, Motomura E, Okada M (2012) Clozapine, but not haloperidol, enhances glial d -serine and L-glutamate release in rat frontal cortex and primary cultured astrocytes. Br J Pharmacol 165(5):1543–1555
- 51. Jouroukhin Y, Kageyama Y, Misheneva V, Shevelkin A, Andrabi S, Prandovszky E et al (2018) DISC1 regulates lactate metabolism in astrocytes: implications for psychiatric disorders. Transl Psychiatry 8(1):76. https://doi.org/10.1038/s41398-018-0123-9
- 52. Molofsky AV, Krencik R, Ullian EM, Tsai HH, Deneen B, Richardson WD et al (2012) Astrocytes and disease: a neurodevelopmental perspective. Genes Dev 26(9):891–907
- 53. Sloan SA, Barres BA (2014) Mechanisms of astrocyte development and their contributions to neurodevelopmental disorders. Curr Opin Neurobiol 27(6224):75–81
- 54. Burns J, Job D, Bastin ME, Whalley H, Macgillivray T, Johnstone EC et al (2003) Structural disconnectivity in schizophrenia: a diffusion tensor magnetic resonance imaging study. Br J Psychiatry 182:439–443
- 55. Schmitt A, Hasan A, Gruber O, Falkai P (2011) Schizophrenia as a disorder of disconnectivity. Eur Arch Psychiatry Clin Neurosci 261(S2):150–154
- Straube B, Green A, Sass K, Kircher T (2014) Superior temporal sulcus disconnectivity during processing of metaphoric gestures in Schizophrenia. Schizophr Bull 40(4):936–944
- 57. Davis KL, Stewart DG, Friedman JI, Buchsbaum M, Harvey PD, Hof PR et al (2003) White matter changes in schizophrenia: evidence for myelin-related dysfunction. Arch Gen Psychiatry 60(5):443–456
- Cassoli JS, Guest PC, Malchow B, Schmitt A, Falkai P, Martins-de-Souza D (2015) Disturbed macro-connectivity in schizophrenia linked to oligodendrocyte dysfunction: from structural findings to molecules. NPJ Schizophr 1:15034. https://doi.org/10.1038/npjschz.2015.34
- 59. Sigmundsson T, Suckling J, Maier M, Williams S, Bullmore E, Greenwood K et al (2001) Structural abnormalities in frontal, temporal, and limbic regions and interconnecting white matter tracts in schizophrenic patients with prominent negative symptoms. Am J Psychiatr 158(2):234–243
- Bora E, Fornito A, Radua J, Walterfang M, Seal M, Wood SJ et al (2011) Neuroanatomical abnormalities in schizophrenia: a multimodal voxelwise meta-analysis and meta-regression analysis. Schizophr Res 127(1–3):46–57
- 61. Uranova NA, Vikhreva OV, Rachmanova VI, Orlovskaya DD (2011) Ultrastructural alterations of myelinated fibers and oligodendrocytes in the prefrontal cortex in schizophrenia: a postmortem morphometric study. Schizophr Res Treatment 2011:325789. https://doi. org/10.1155/2011/325789

- 2 Glycolysis in Schizophrenia Glial Cells
- 62. Martins-de-Souza D, Harris LW, Guest PC, Bahn S (2011) The role of energy metabolism dysfunction and oxidative stress in schizophrenia revealed by proteomics. Antioxid Redox Signal 15(7):2067–2079
- Bernstein HG, Steiner J, Guest PC, Dobrowolny H, Bogerts B (2015) Glial cells as key players in schizophrenia pathology: recent insights and concepts of therapy. Schizophr Res 161(1):4–18
- Vikhreva OV, Rakhmanova VI, Orlovskaya DD, Uranova NA (2016) Ultrastructural alterations of oligodendrocytes in prefrontal white matter in schizophrenia: a post-mortem morphometric study. Schizophr Res 177(1–3):28–36
- 65. Hakak Y, Walker JR, Li C, Wong WH, Davis KL, Buxbaum JD et al (2001) Genome-wide expression analysis reveals dysregulation of myelination-related genes in chronic schizophrenia. Proc Natl Acad Sci U S A 98(8):4746–4751
- Nave KA, Ehrenreich H (2014) Myelination and oligodendrocyte functions in psychiatric diseases. JAMA Psychiat 71(5):582–584
- Cassoli JS, Iwata K, Steiner J, Guest PC, Turck CW, Nascimento JM et al (2016) Effect of MK-801 and clozapine on the proteome of cultured human oligodendrocytes. Front Cell Neurosci 10:52. https://doi.org/10.3389/fncel.2016.00052
- 68. Steiner J, Martins-de-Souza D, Schiltz K, Sarnyai Z, Westphal S, Isermann B et al (2014) Clozapine promotes glycolysis and myelin lipid synthesis in cultured oligodendrocytes. Front Cell Neurosci 8:384. https://doi.org/10.3389/fncel.2014.00384
- Leviton A, Gressens P (2007) Neuronal damage accompanies perinatal white-matter damage. Trends Neurosci 30(9):473–478
- Najjar S, Pearlman DM (2015) Neuroinflammation and white matter pathology in schizophrenia: Systematic review. Schizophr Res 161(1):102–112
- Karsten SL, Kudo LC, Jackson R, Sabatti C, Kornblum HI, Geschwind DH (2003) Global analysis of gene expression in neural progenitors reveals specific cell-cycle, signaling, and metabolic networks. Dev Biol 261(1):165–182
- Rafalski VA, Brunet A (2011) Energy metabolism in adult neural stem cell fate. Prog Neurobiol 93(2):182–203
- 73. Bradl M, Lassmann H (2010) Oligodendrocytes: biology and pathology. Acta Neuropathol 119(1):37–53
- 74. Minardi Nascimento J, Cereda V, Zuccoli G, Gouvêa-Junqueira D, Martins de-Souza D (2018) Modeling schizophrenia with human stem cells. In: Stem cell genetics for biomedical research: past, present, and future, 1st edn. Springer; New York, NY, USA, pp 13–26. ISBN-10: 3319906941
- Raabe FJ, Galinski S, Papiol S, Falkai PG, Schmitt A, Rossner MJ (2018) Studying and modulating schizophrenia-associated dysfunctions of oligodendrocytes with patient-specific cell systems. NPJ Schizophr 4(1):23. https://doi.org/10.1038/s41537-018-0066-4
- 76. Sebban C (1983) [Biochemical aspects of cerebral aging. Role of oxygen]. Presse Med 12(13):804–808
- 77. Schnack HG, van Haren NE, Nieuwenhuis M, Hulshoff Pol HE, Cahn W, Kahn RS (2016) Accelerated brain aging in schizophrenia: a longitudinal pattern recognition study. Am J Psychiatry 173(6):607–616
- 78. Shivakumar V, Kalmady SV, Rajasekaran A, Chhabra H, Anekal AC, Narayanaswamy JC et al (2018) Telomere length and its association with hippocampal gray matter volume in antipsychotic-naïve/free schizophrenia patients. Psychiatry Res Neuroimaging 282:11–17
- 79. Epel ES, Blackburn EH, Lin J, Dhabhar FS, Adler NE, Morrow JD et al (2004) Accelerated telomere shortening in response to life stress. Proc Natl Acad Sci U S A 101:17312–17315
- 80. Simon NM, Smoller JW, McNamara KL, Maser RS, Zalta AK, Pollack MH et al (2006) Telomere shortening and mood disorders: preliminary support for a chronic stress model of accelerated aging. Biol Psych 60:432–435
- Kao HT, Cawthon RM, Delisi LE, Bertisch HC, Ji F, Gordon D et al (2008) Rapid telomere erosion in schizophrenia. Mol Psychiatry 13(2):118–119

- Allsopp RC (1996) Models of initiation of replicative senescence by loss of telomeric DNA. Exp Gerontol 31(1–2):235–243
- Buchkovich KJ (1996) Telomeres, telomerase, and the cell cycle. Prog Cell Cycle Res 2:187–195
- 84. Preston RJ (1997) Telomeres, telomerase and chromosome stability. Radiat Res 147(5):529-534
- Lansdorp PM (1995) Telomere length and proliferation potential of hematopoietic stem cells. J Cell Sci 108(Pt 1):1–6
- 86. Hiyama K, Hirai Y, Kyoizumi S, Akiyama M, Hiyama E, Piatyszek MA et al (1995) Activation of telomerase in human lymphocytes and hematopoietic progenitor cells. J Immunol 155(8):3711–3715
- Moore MA (1997) Stem cell proliferation: ex vivo and in vivo observations. Stem Cells 15 Suppl 1:239–248
- Ravindranath N, Dalal R, Solomon B, Djakiew D, Dym M (1997) Loss of telomerase activity during male germ cell differentiation. Endocrinology 138(9):4026–4029
- Flores I, Benetti R, Blasco MA (2006) Telomerase regulation and stem cell behaviour. Curr Opin Cell Biol 18:254–260
- Blasco MA, Lee HW, Rizen M, Hanahan D, DePinho R, Greider CW (1997) Mouse models for the study of telomerase. Ciba Found Symp 211:160–170; discussion 170–176
- Rudolph KL, Chang S, Lee HW, Blasco M, Gottlieb GJ, Greider C et al (1999) Longevity, stress response, and cancer in aging telomerase-deficient mice. Cell 96(5):701–712
- 92. Cropley VL, Klauser P, Lenroot RK, Bruggemann J, Sundram S, Bousman C et al (2017) Accelerated gray and white matter deterioration with age in schizophrenia. Am J Psychiatry 174(3):286–295
- Peters A (2002) Structural changes in the normally aging cerebral cortex of primates. Prog Brain Res 136:455–465
- Palmer AL, Ousman SS (2018) Astrocytes and aging. Front Aging Neurosci 10:337. https:// doi.org/10.3389/fnagi.2018.00337
- 95. Dabouras V, Rothermel A, Reininger-Mack A, Wien SL, Layer PG, Robitzki AA (2014) Exogenous application of glucose induces aging in rat cerebral oligodendrocytes as revealed by alteration in telomere length. Neurosci Lett 368(1):68–72
- 96. Szebeni A, Szebeni K, DiPeri T, Chandley MJ, Crawford JD, Stockmeier CA et al (2014) Shortened telomere length in white matter oligodendrocytes in major depression: potential role of oxidative stress. Int J Neuropsychopharmacol 17(10):1579–1589
- 97. Caporaso GL, Chao MV (2001) Telomerase and oligodendrocyte differentiation. J Neurobiol 49(3):224–234
- Tse KH, Herrup K (2017) DNA damage in the oligodendrocyte lineage and its role in brain aging. Mech Ageing Dev 161(Pt A):37–50

# **Chapter 3 Aging Biomarkers and Novel Targets for Anti-Aging Interventions**



Kang Xu, Yannan Guo, Zhongchi Li, and Zhao Wang

# 1 Introduction

# 1.1 Aging of the Population

In 2017, the segment of the world population aged 60 and over reached 962 million, accounting for 13% of the world population, and the aged population is growing at a rate of 3% per year. At present, Europe has the highest proportion of the elderly, with a population 60 years and older accounting for 25% of the total population. By 2050, this proportion of the world population (except Africa) will reach 25% or even higher. At the same time, human life expectancy constantly rises, from 67 years in 2000 to 71 years in 2015, increasing by 4 years in a 5 year period [1]. Aging is unavoidable for mankind and the problem of population aging is a serious challenge faced by all countries. Aging has become a strategic issue affecting the long-term development of a country's economy. With the development of society and the gradual improvement of the public health system, people have begun to pay more and more attention to aging and age-related diseases. Aging is the strongest risk factor for diseases and dysfunction, ranging from neurodegeneration to cancer. This relationship likely arises from accumulation of genetic errors and effects of environmental factors [2]. In general, it is desirable to promote healthy aging.

Most species in nature obey the law of increasing mortality with age [3]. Aging and longevity are among the most complex topics in current biological and clinical research. Aging is the result of multifactorial effects, including genetic factors and genetic changes, external environmental stimuli and lifestyle factors. In aging research, we must be aware of the delicate balance between life extension and

K. Xu · Y. Guo · Z. Li · Z. Wang (🖂)

Protein Science Key Laboratory of the Ministry of Education, School of Pharmaceutical Sciences, Tsinghua University, Beijing, People's Republic of China e-mail: zwang@tsinghua.edu.cn

<sup>©</sup> Springer Nature Switzerland AG 2019

P. C. Guest (ed.), *Reviews on Biomarker Studies in Aging and Anti-Aging Research*, Advances in Experimental Medicine and Biology 1178, https://doi.org/10.1007/978-3-030-25650-0\_3

tumorigenesis. As reported in the past, the naked mole rat, the rodent species with the longest lifespan, has a maximum lifespan of over 30 years, surprisingly, without the extra incidence of cancer. Tian and colleagues have discovered that abundant high-molecular-mass hyaluronan (HA) in the skin and other organs/tissues plays a dominant role in longevity and cancer resistance of the naked mole rat [4]. We may obtain clues from such pioneering research to seek ways to slow down aging.

### 1.2 Biological and Chronological Age

Some scientists believe that chronological age is an essential risk factor for functional impairment and biological age may be a more accurate metric for prediction of aging [5]. Chronological age represents the actual age of a person, that is the time he or she has lived on this earth since birth. Chronological age insufficiently reflects an individual's general health status and susceptibility to diseases and disabilities associated with the process of aging. Biological age represents the physiological age of a person. Mostly, we have to take diet, exercise habits and other lifestyle factors and living environments into consideration. Patrick and colleagues have published a Health Aging and Body Composition Study and found that biological age is a better predictive factor for late-life depressive symptoms than chronological age [6]. Kim et al. has reported that biological age measurement can facilitate the assessment of colorectal adenoma risk associated with screening colonoscopy [7]. A cross-sectional study that has been conducted by Zhang and colleagues put forward a biological age-based formula to effectively monitor the degree and speed of aging in the Chinese population [8]. Biological age is better able to reflect a person's health state and indicates an individual's active-life expectancy, which is based on the studies above [9].

### 1.3 Aging Biomarkers and Breaking News

With the increasing worldwide life expectancy, the risk of aging-related diseases rises. Characterization of biomarkers associated with aging may open the door to aging predictions and antiaging-strategy discovery [10]. Over the years, numerous scientists have made important contributions to the search for accurate and adaptable aging biomarkers. Understanding the triggers for the aging process and the association between aging and diseases is important for the search and validation of aging biomarkers or a system that will boost basic gerontology and clinical translational research.

However, we have to deal with the complexity of the aging process, varying aging patterns and the diversity of aging causes. There are still no precise independent aging biomarkers that accurately reflect a person's aging status or predict aging speed and life expectancy.

#### 3 Biomarkers of Aging

Aging research has long been considered one of the most costly types of biological studies because it is time-consuming and expensive to use mammals as research subjects. Janssens et al. from the Karolinska Institute in Sweden has developed an innovative and age-stratified human-tissue transcriptomics-based cell culture system for geroprotector screening [11]. They found that monorden and tanespimycin, which are Hsp90 inhibitors, can induce a youthful transcriptional state in human cells, prolong the lifespan and help to maintain a healthy state in Caenorhabditis elegans. Nonetheless, the protective effect of these two compounds on humans remains to be investigated. From their article, maybe we can glean additional information related to aging and disease states of the cell from the transcriptome in specific tissues and at specific time points. Similarly, machine learning and artificial intelligence will greatly contribute to the efficiency of aging research.

This review collates current hot research topics and proposes promising aging biomarkers at the molecular, cellular and organismal levels, and we attempt to shed light on the relationship between various relevant factors. We provide a broader definition of aging biomarkers, including telomere length, protein expression levels and function, activation or inhibition of key signaling pathways, the gut microbiota, and metabolic patterns. We believe that combining different levels of aging biomarkers can better characterize aging and improve screening of aging interventions for targeted drugs.

### 2 Aging Biomarkers and Novel Targets for Interventions

# 2.1 The Molecular Level: The Telomere

Telomeres, the terminal nucleoprotein complexes of chromosomes in eukaryotes, play an important role in chromosomal DNA protection, gene expression, and regulation of stress-related signaling pathways by controlling cell senescence and organismal aging [12, 13]. Telomere length, reflecting a delicate balance between shortening and elongation of telomeres, is mainly controlled by telomerase and telomeric repeat-containing RNA (TERRA) [14]. A systemic knockout of a telomerase subunit in mice results in a decreased telomere length, accelerated organ dysfunction and a shortened lifespan [10, 15]. Reintroduction of telomerase into the G4<sup>TERT-ER</sup> mouse model can reverse the tissue degeneration and accelerated-aging phenotype, without an increased tumorigenesis risk. As a result, telomerase reintroduction has an excellent potential for organ or tissue regeneration and aging reversal [16]. There are also some chemical telomerase activators that are expected to be applied to clinical practice for aging research and treatment of degenerative disorders [10].

Telomere attrition may contribute to increased morbidity and mortality of agingrelated diseases [17]. As Wang and colleagues found, shorter telomeres correlate with a higher risk of all-cause mortality. Average telomere length varies and there are sex differences with women have a longer telomere length and life expectancy compared to men, perhaps owing to the hormonal differences, e.g., in estrogen levels, as well as the effect of the X chromosome [18, 19]. Decreasing immune surveillance and increasing inflammation are related to age, and both are linked to telomere shortening and lower telomerase activity [20, 21]. It has been proven that shorter telomere length is associated with oxidative-stress-induced diabetes, Alzheimer's disease, cardiovascular diseases and high levels of the proinflammatory cytokine tumor necrosis factor  $\alpha$  [22–24]. Accumulation of reactive oxygen species (ROS) may be a contributing factor of telomere-dependent senescence [25]. It is worth noting that mitochondria-derived oxidative stress plays a key role here because mitochondrial depolarization induced by carbonyl cyanide-4(trifluoromethoxy) phenylhydrazone, an uncoupler of oxidative phosphorylation in mitochondria, can bring about mitochondrial dysfunction, increased ROS production and subsequent telomere attrition and genomic instability [26]. Health-promoting daily activities and habits also contribute to the maintenance of telomere length, e.g., low alcohol consumption and cigarette smoking and following a healthy diet and a physicalactivity regimen [27].

Due to the ability to maintain telomere length, cancer cells can proliferate indefinitely [28]. However, the relevance of telomere length to cancer susceptibility is not clear because of the cancer type-specific characteristics [29].

Southern blotting, polymerase chain reaction (PCR)-based methods, single telomere length analysis (STELA) and fluorescence in situ hybridization (FISH) are relatively new technologies for measuring telomere length. Quantitative PCR and FISH are frequently applied in clinical and epidemiological studies [30].

The future research direction dealing with telomeres addresses various relevant questions including how to improve the accuracy of the methods measuring telomere length, telomerase activity, and telomeric repeat-containing RNA and whether or not telomere length and telomerase activity can be straightforward markers of aging and age-related diseases in clinical tests. Moreover, further investigations are still needed to look for specific pharmaceutical or nutraceutical approaches to maintain the telomere length and telomerase activity without increasing the risk of tumorigenesis and other adverse effects to achieve systemic healthy aging (Fig. 3.1) [30].

### 2.2 The Cellular Level

#### 2.2.1 Longevity Proteins

These proteins are needed to protect the cell/organism from future damage. Impairment of the function of longevity proteins raises the risk of diseases associated with aging (Fig. 3.2) [31].



Fig. 3.1 The relationship of telomeres with cell senescence and aging



**Fig. 3.2** Longevity proteins, the IGF-1 signaling pathway and mTORC1 signaling pathway affect the magnitude of a SASP, resulting in regulation of the aging process

**Sirtuins** Sir2 was discovered in yeast because of the significant correlation between the activity of Sirtuin family proteins and the rate of replicative aging [32, 33]. There are seven sirtuins in mammals which have different tissue expression patterns (and specificities for substrates) [34]. These are SIRT1, SIRT6 and SIRT7, which function in the nucleus, SIRT3, SIRT4 and SIRT5, which function in mitochondria, and SIRT2 which is active in the nucleus and cytoplasm. Sirtuins take part in various biological processes individually or cooperatively, including DNA damage repair, inflammation response, cell cycle regulation, and mitochondrial functions, and thus affect genomic stability, longevity, metabolic homeostasis, inflammation alleviation and health maintenance [35, 36].

SIRT1 and SIRT6 are two of the most promising regulators of longevity. Caloric restriction as a practical dietary regimen to protect from aging-related changes and to extend the lifespan for humans, can increase the expression and activity of SIRT1 and SIRT6 [37]. SIRT1 overexpression can attenuate many age-related disorders, including glucose resistance, neurodegenerative diseases and tumorigenesis but without the extension of the lifespan in a mouse model [38]. SIRT1 can also enhance DNA methylation, which is beneficial for genomic stability and longevity [36]. SIRT6 mainly functions through deacetylation and ADP ribosylation [39]. A systemic SIRT6 knockout in mice causes premature aging and death at 4 weeks of age [40]. SIRT6 overexpression extends the lifespan of male mice but does not affect female mice [41]. The latest studies have proved the new mechanism of SIRT6 action, including heterochromatin silencing, genome maintenance, epigenetic regulation of glucose and lipid metabolism, which are closely linked to aging and age-related diseases [39].

 $\alpha$ -Klotho (muscle regeneration) *Klotho*, a powerful aging suppressor gene, encodes a membrane-bound and circulating hormonal protein in mammals [42]. Mice with a systemic Klotho knockout present with a phenotype of accelerated aging, e.g., cognitive dysfunction and sarcopenia [43, 44]. After an acute muscle injury, there is increased expression of  $\alpha$ -Klotho in young skeletal muscle, resulting in impaired tissue regeneration. However, there are no significant changes in old muscles [45].

The insulin/insulin-like growth factor (IGF) pathway The IGF-1/insulin signaling pathway is one of the most conserved pathways involved in longevity and lifespan regulation [46]. The production of IGF-1 is induced by growth hormone mostly in the liver and partially in the heart, kidneys and cartilage. Serum IGF-1 concentrations could reflect the secretion and activity of growth hormone. The IGF-1 signaling pathway also participates in the control of tumorigenesis, diabetes, and cardiovascular diseases, which are common diseases associated with aging [47]. It has been proven that inhibition of IGF-1/insulin signaling can extend the lifespan of C. elegans [48]. In mice, growth hormone deficiency or resistance can extend the lifespan, with a decreased IGF-1 level [49]. Some epidemiological studies indicate

#### 3 Biomarkers of Aging

that there is a relationship between longevity, human healthy aging and mutation of the growth hormone and IGF-1 receptors [50, 51].

**Mammalian target of rapamycin (mTOR)** Kinase mTOR is a serine/threonine protein kinase, including mTORC1, a metabolic sensor for nutrients, growth factors, energy metabolism and stressful stimuli, and mTORC2 an essential regulator of glucose metabolism. mTOR could be regarded as a central regulatory element of glucose homeostasis, muscle mass and function, lipid homeostasis, immune function, brain function, cancer and a senescence-associated secretory phenotype (SASP), which are closely associated with longevity and aging [52, 53]. Rapamycin, an inhibitor of mTOR, can extend the lifespan of mice when fed late in life but with the adverse effect of increased glucose tolerance because of inhibition of mTORC2 [54]. These findings indicate that future work should focus on the discovery of a specific mTORC1 inhibitor.

#### 2.2.2 The SASP

Approximately 10 years ago, a SASP was first proposed by Coppé and colleagues [55]. This means that a senescent cell can produce and secrete some cytokines to affect surrounding cells and the microenvironment positively or negatively. Secreted factors include immunomodulators, inflammatory factors (and other interleukins and chemokines) and growth factors [56]. Senescence could occur in response to damage and danger signals, a metabolic abnormality, activated oncogenes or oncogenic mutations [57]. Due to differences in cell types and aging stimuli (including intensity and duration), the secretion associated with a SASP also varies. Some investigators have discovered that severe DNA damage causes persistent DNA damage response signals and initiation of a SASP [58]. In the initial stage, a SASP is implemented to enhance the senescence-related cell growth arrest, which is beneficial for clearance of senescent cells. However, with the rapid increase in the number and slowing clearance of senescent cells, chronic inflammation, as a hallmark of aging and a major contributor to age-related dysfunctions [58], and tumorigenesis are exacerbated [59, 60].

The classic SASP: cytokines IL-1α, as an important participant in a SASP, can also regulate the production and secretion processes of a SASP, especially those of IL-6, IL-8, vascular endothelial growth factors (VEGFs) and transforming growth factor β (TGF-β) [61, 62]. It is also a senescence marker of endothelial cells and of vascular aging [63]. The NF- $\kappa$ B signaling pathway, called the "master regulator of a SASP," plays a major role in the regulation of IL-6 and IL-8 expression [62]. GATA4 (which is degraded by p62-mediated autophagy [60]), upregulation of p38 MAPK [64], and activation of mTOR can increase NF- $\kappa$ B activity [65], resulting in a potent SASP. Many other SASP factors are influenced by the Janus kinase signal transducer and activator of transcription (JAK/STAT) pathway, which is NF- $\kappa$ B independent [66].

Statins, such as simvastatin, can inhibit a SASP, including expression of IL-6, IL-8, and MCP-1, and this effect slows down cancer cell proliferation [67]. Metformin, rapamycin, and JAK1/2 inhibitors have been reported to act as SASP inhibitors, which can also improve an aging state and alleviate age-related diseases in mammals at a different level [68]. The clinical application of SASP inhibitors requires more research and validation because of their potential adverse effects [69].

**Other factors** Extracellular vesicles (EVs) is a general name of a variety of small membranous vesicles that are released into the extracellular environment by most cell types. It has been reported that the EVs secreted by older cells are markedly different from those from younger cells. As a result, EVs can be regarded as one of the special participants of a SASP. Senescent cells secrete more EVs when considering the cell type and stimulation [70–72]. Cell senescence also alters the contents of EVs, including an increase in the levels of IL-6 and IL-12 mRNAs [73]. Naturally, the composition of EVs, such as microRNA content, also regulates cellular senescence [74].

Gan and colleague published some research results showing that the level of urinary 8-oxo-7,8-dihydroguanosine (8-oxoGsn) increases with age, in samples from 1228 healthy Chinese residents 2–90 years of age [75]. Because urine sampling and the detection methods of 8-oxoGsn are convenient and operational, urinary 8-oxoGsn could be regarded as a promising aging biomarker, especially in clinical situations.

**Detection of a SASP** For SASP discovery, mRNA profiling, proteomics, antibody arrays, and other multiplex assays are widely used. However, due to their high cost and low sensitivity, the highly sensitive and customizable sandwich enzyme-linked immunosorbent assay (ELISA), a variation of the ELISA, is recommended for laboratory research [76].

### 2.3 The Organismal Level

### 2.3.1 The Gut Microbiota and Aging

People that grow older will experience a series of changes associated with the gut microbiota. With the development of large-scale bacterial DNA sequencing, gut microbiology research has made a breakthrough in recent years [77]. The gut microbiota performs important functions in the processes of development, maturation and aging, and in the incidence of diseases. The gut microbiota could provide various vitamins, short-chain fatty acids, essential amino acids, peptides and other organic compounds that are essential for biological processes. At the same time, these microbes directly or indirectly participate in the regulation of digestion, nutrient absorption, and metabolism, which are closely related to inflammation and immunity [78, 79] (Fig. 3.3).



Fig. 3.3 Decreased microbiota diversity, increased tryptophan metabolism and elevated immunosenescence and inflamm-aging occur during the aging process, accelerate the aging rate and aggravate aging-related diseases

**Immuno-senescence and inflamm-aging** Long-term stimulation of the immune system can cause a decline of immune-system functioning, which is called immunosenescence. Low-grade chronic inflammation, along with many age-related diseases, metabolic syndrome and neurodegeneration, has been referred to as inflamm-aging [80]. Normal bacteria can act as antigens to stimulate the body's immune response. Immune aging is usually accompanied by up-regulation of the inflammatory response. During aging, a continuous gut microbiota imbalance leads to inflammatory responses to the intestinal mucosa. As a result, chronic inflammation develops throughout the body [81].

**Microbiota diversity** Changes in microbiota diversity with age are significant. Moreover, the diversity of the microbiota in elderly people is lower than that in younger people [82], with a reduction in relative abundance of *Bifidobacteria*, and increasing abundance of *Bacteroides* and *Enterobacteriaceae* [80]. The ratio of *Firmicutes* to *Bacteroides* can be a criterion of metabolic health and a decrease in this ratio is observed with aging [83].

*Bifidobacterium* species that are suggested to function in the maintenance of human health [84], as an important member of the gut microflora (especially probiotic species), show a decrease in relative abundance in elderly people as compared with younger adults [82]. It has been reported that long-term *Lactobacillus* and *Bifidobacterium* dietary supplementation can enhance memory and change metabolism of the brain in aged rat [85]. Compositional changes in *Bifidobacterium* take place with age. For example, there is a decrease in the abundance of *B. breve* in people who are over 50 years of age and a increase of *B. dentium* abundance in subjects over 60 years of age [84].

**Tryptophan metabolism** Tryptophan, as an intestinal microflora metabolite, performs a crucial function in the balance between gut immune tolerance and gut microbiota maintenance [86]. An enhancement of tryptophan metabolism is positively correlated with age, in agreement with the finding that elderly people have a lower serum tryptophan level [87]. There is another study which indicates that patients with senile dementia also have a decreased serum tryptophan level [88].

There are still many controversies regarding the notion that the gut microbiota is a specific and an unquestionable signature for evaluation of the aging status. However, there is reasonable evidence showing the importance of gut homeostasis for antiaging and microbiome-targeted interventions in antiaging medicine.

#### 2.3.2 The Metabolic Pattern and Aging

Elderly people have a different body composition, energy intake, energy expenditure, physical activity, fuel utilization and the ability to regulate the energy balance, in comparison with young people [89].

Nicotinamide adenine dinucleotide (NAD<sup>+</sup>) NAD<sup>+</sup>, as a pivotal cellular factor participating in different metabolic pathways, functions as both a co-substrate and co-enzyme in different cellular compartments. Under aging and high-fat diet, the NAD+ level falls. Simultaneously, a decreased NAD<sup>+</sup> concentration can accelerate the aging process. However, calorie restriction (CR), fasting, glucose deprivation and other lower-energy load conditions can efficiently raise the concentration of NAD<sup>+</sup>, resulting in an increasing lifespan and a better state of health in old age. It has been shown that a NAD<sup>+</sup> precursor, nicotinamide riboside, or nicotinamide mononucleotide dietary supplementation can contribute to health maintenance and increase longevity in mice [90, 91]. NAD<sup>+</sup> precursors could serve as therapeutic interventions for metabolic diseases, such as type 2 diabetes, fatty liver disease, glucose intolerance, and neurodegenerative disorders [90]. A reduced number and function of adult stem cells may induce injury and dysfunction of the tissue when it encounters external stimuli, including effects of the aging process. Nicotinamide riboside supplementation has been reported to induce the rejuvenation of muscle stem cells and intestinal stem cells in aged mice [92, 93].

Some researchers remain sceptical about the powers of NAD<sup>+</sup> because of some recent studies [94]. Nacarelli and colleagues have demonstrated that NAD<sup>+</sup> metabolism can promote an inflammatory environment and production and secretion of proinflammatory factors (i.e., SASP), which will facilitate tumorigenesis [94]. These results serve as a warning that the antiaging effect and potential tumorigenesis risk have to be balanced when oral NAD<sup>+</sup> is considered for clinical development as a nutraceutical (Fig. 3.4).

**CR and a ketogenic diet (KD)** The process of aging has been repeatedly proven to be closely related to metabolism and diet is the most direct factor affecting metabolism. How to regulate the aging process through dietary changes is a new



**Fig. 3.4** Lower energy loads and NAD<sup>+</sup> precursor supplementation could increase the NAD<sup>+</sup> level. Nevertheless, the effect of a high level of NAD<sup>+</sup> is controversial, including health maintenance, anabatic inflammatory responses and tumorigenesis. NMN nicotinamide mononucleotide, NR nicotinamide riboside

topic in aging intervention research. Dietary interventions, as the most common and direct method for combating aging and diseases, are gradually attracting more attention. CR and KDs have been studied and discussed for many decades. CR can convert glycolysis to  $\beta$ -oxidation, which means a shift from carbohydrate to fat metabolism in cells and in the organism. Balasubramanian et al. have concluded that the beneficial effect of CR is related to human aging and CR could protect the organism from arthritis, cardiovascular diseases, age-related diabetes, colon lesions and delay osteoporosis, sarcopenia, brain atrophy and cognitive decline, as well as decrease cancer incidence and progression [95]. Just as CR, the KD is associated with changes in metabolic patterns, from carbohydrates toward fatty acid oxidation and induces an increase in the levels of ketone bodies in mice [96]. Roberts et al. have shown that a KD could promote longevity and increase the health span in adult mice, with increased motor function and memory and a decreased tumor incidence. In their research, the KD was found to function by increasing protein acetylation and regulating mTOR signaling, and this was tissue specific [97]. Almost simultaneously, Newman and Verdin reported that a KD can reduce midlife mortality and improve memory in aging mice, in line with the conclusion of Roberts [98]. In our own research, we found that both a high-fat diet and a KD could reverse shortening of lifespan and attenuate the premature aging phenotype in a mouse model of progeria. We attribute this effect to a shift in metabolic patterns (unpublished observations). In 2014, a paper presented the idea that a high-fat diet is beneficial for



Fig. 3.5 CD and KD can convert carbohydrate metabolism to fat metabolism, which contributes to longevity

preventing premature aging in Cockayne syndrome, which has features that are similar to late-stage human aging [99]. All of these studies seemingly support this opinion that fat metabolism is a better choice for kerbing the aging process (Fig. 3.5).

It is also noteworthy that a KD shows promise for cancer therapy if tumor types and genetic changes are taken into account. In preclinical trials, there is evidence that some patients with glioblastoma, neuroblastoma, pancreatic cancer or lung cancer show better recovery when given an adjuvant KD therapy. The KD also has antitumor effects in stomach or liver cancer patients [100].

### **3** Conclusion and Perspectives

Focusing on the research on healthy aging and the rate of aging is more important than solely extending the lifespan. Various scientists have been contributing to the study of aging biomarkers and novel targets for anti-aging interventions and have made significant progress in these objectives. The biomarkers and targets mentioned in this review were hotspot issues in recent years and gained much recognition in the field of aging biology and geriatrics. What deserves our attention is that the effects of an individual's aging biomarkers vary at the cellular and systemic levels. For the exploration of aging biomarkers and anti-aging targets, we should pay more attention to the overall effect. Of course, large-scale screening at the cellular level has dramatically improved the efficiency of interventions and success rates.

There are some limitations in this review. For example, it does not cover some other phenomena that play an important role in the aging process, e.g., autophagy. Recently, the role of autophagy in the process of aging gradually attracted the attention of scientists. Autophagy is activated under stressful conditions, such as oxidative stress, starvation or hypoxia, to ensure cellular homeostasis. To a certain degree, activated autophagy, which is induced by overexpression of Atg5 and enhancement of lysosomal receptor of chaperone-mediated autophagy (CMA), may significantly attenuate age-related organismal damage and functional decline in aged mice [101, 102]. Notably, reduced autophagic activity is observed in aged muscle, thus resulting in lower myofiber function and muscle strength, and this effect is conserved between mice and humans [103]. In addition, the correlation between metabolic rate and aging is controversial. Thus, considerable more work and information is required to shed light on this topic.

The benefits of the advances in computer sciences, including meta-analyses and artificial intelligence, will improve the speed and efficiency of aging biomarker research [95]. We need more research and validation for creation and application of an aging biomarker assessment system in clinical practice.

Acknowledgements This work was financially supported by grants from the National Key R&D Program of China (2018YFC2000304 and 2018YFD0400204), National Natural Science Foundation of China (81871095), and the Key International S&T Cooperation Program of China (2016YFE113700).

### References

- United Nations DoEaS, Population Division (2017) World population prospects: the 2017 revision, key findings and advance tables. https://esa.un.org/unpd/wpp/Publications/Files/ WPP2017\_KeyFindings.pdf
- 2. Vijg J, Campisi J (2008) Puzzles, promises and a cure for ageing. Nature 454(7208):1065-1071
- 3. Jones OR, Scheuerlein A, Salguero-Gomez R, Camarda CG, Schaible R, Casper BB et al (2014) Diversity of ageing across the tree of life. Nature 505(7482):169–173
- Tian X, Azpurua J, Hine C, Vaidya A, Myakishev-Rempel M, Ablaeva J et al (2013) Highmolecular-mass hyaluronan mediates the cancer resistance of the naked mole rat. Nature 499(7458):346–349
- Lowsky DJ, Olshansky SJ, Bhattacharya J, Goldman DP (2014) Heterogeneity in healthy aging. J Gerontol A Biol Sci Med Sci 69(6):640–649
- Patrick JB, Melanie MW, Chen C, Morgan EL, Kristine Y, Steven PR, Bret RR (2018) Biological Age, Not Chronological Age, Is Associated with Late-Life Depression. The Journals of Gerontology: Series A 73(10):1370–1376
- Kim S-J, Kim BJ, Kang H (2017) Measurement of biological age may help to assess the risk of colorectal adenoma in screening colonoscopy. World J Gastroenterol 23(37):6877–6883
- Zhang W, Jia L, Cai G, Shao F, Lin H, Liu Z et al (2017) Model construction for biological age based on a cross-sectional study of a healthy Chinese han population. J Nutr Health Aging 21(10):1233–1239

- 9. Borkan GA, Norris AH (1980) Biological age in adulthood: comparison of active and inactive U.S. males. Hum Biol 52(4):787–802
- Bernardes de Jesus B, Blasco MA (2012) Potential of telomerase activation in extending health span and longevity. Curr Opin Cell Biol 24(6):739–743
- Janssens GE, Lin X-X, Millan-Arino L, Kavsek A, Sen I, Seinstra RI et al (2019) Transcriptomics-based screening identifies pharmacological inhibition of Hsp90 as a means to defer aging. Cell Rep 27(2):467–480.e6
- 12. Blackburn EH (1990) Telomeres: structure and synthesis. J Biol Chem 265(11):5919–5921
- Blackburn EH (2005) Telomeres and telomerase: their mechanisms of action and the effects of altering their functions. FEBS Lett 579(4):859–862
- 14. Honig LS, Kang MS, Cheng R, Eckfeldt JH, Thyagarajan B, Leiendecker-Foster C et al (2015) Heritability of telomere length in a study of long-lived families. Neurobiol Aging 36(10):2785–2790
- 15. Strong MA, Vidal-Cardenas SL, Karim B, Yu H, Guo N, Greider CW (2011) Phenotypes in mTERT(+/-) and mTERT(-/-) mice are due to short telomeres, not telomere-independent functions of telomerase reverse transcriptase. Mol Cell Biol 31(12):2369–2379
- Jaskelioff M, Muller FL, Paik J-H, Thomas E, Jiang S, Adams AC et al (2011) Telomerase reactivation reverses tissue degeneration in aged telomerase-deficient mice. Nature 469(7328):102–106
- Wang Q, Zhan Y, Pedersen NL, Fang F, Hagg S (2018) Telomere length and all-cause mortality: a meta-analysis. Ageing Res Rev 48:11–20
- Simoncini T, Hafezl-Moghadam A, Brazil DP, Ley K, Chin WW, Liao JK (2000) Interaction of oestrogen receptor with the regulatory subunit of phosphatidylinositol-3-OH kinase. Nature 407(6803):538–541
- 19. Austad SN (2006) Why women live longer than men: sex differences in longevity. Gend Med 3(2):79–92
- Sansoni P, Cossarizza A, Brianti V, Fagnoni F, Snelli G, Monti D et al (1993) Lymphocyte subsets and natural killer cell activity in healthy old people and centenarians. Blood 82(9):2767–2773
- Fulop T, Larbi A, Dupuis G, Le Page A, Frost EH, Cohen AA et al (2017) Immunosenescence and inflamm-aging as two sides of the same coin: friends or foes? Front Immunol 8:1960. https://doi.org/10.3389/fimmu.2017.01960
- 22. D'Mello MJ, Ross SA, Briel M, Anand SS, Gerstein H, Pare G (2015) Association between shortened leukocyte telomere length and cardiometabolic outcomes: systematic review and meta-analysis. Circ Cardiovasc Genet 8(1):82–90
- 23. Panossian LA, Porter VR, Valenzuela HF, Zhu X, Reback E, Masterman D et al (2003) Telomere shortening in T cells correlates with Alzheimer's disease status. Neurobiol Aging 24(1):77–84
- Dai D-F, Chiao YA, Marcinek DJ, Szeto HH, Rabinovitch PS (2014) Mitochondrial oxidative stress in aging and healthspan. Longev Healthspan 3:6. https://doi.org/10.1186/2046-2395-3-6
- Serra V, von Zglinicki T, Lorenz M, Saretzki G (2003) Extracellular superoxide dismutase is a major antioxidant in human fibroblasts and slows telomere shortening. J Biol Chem 278(9):6824–6830
- Liu L, Trimarchi JR, Smith PJS, Keefe DL (2002) Mitochondrial dysfunction leads to telomere attrition and genomic instability. Aging Cell 1(1):40–46
- Latifovic L, Peacock SD, Massey TE, King WD (2016) The influence of alcohol consumption, cigarette smoking, and physical activity on leukocyte telomere length. Cancer Epidemiol Biomark Prev 25(2):374–380
- 28. Shay JW (2016) Role of telomeres and telomerase in aging and cancer. Cancer Disgov 6(6):584–593
- Savage SA, Gadalla SM, Chanock SJ (2013) The long and short of telomeres and cancer association studies. J Natl Cancer Inst 105(7):448–449

- 3 Biomarkers of Aging
  - Mensà E, Latini S, Ramini D, Storci G, Bonafè M, Olivieri F (2019) The telomere world and aging: analytical challenges and future perspectives. Ageing Res Rev 50:27–42
  - Ames BN (2018) Prolonging healthy aging: longevity vitamins and proteins. Proc Natl Acad Sci U S A 115(43):10836–10844
  - 32. Sinclair DA, Guarente L (1997) Extrachromosomal rDNA circles—a cause of aging in yeast. Cell 91(7):1033–1042
  - 33. Guarente L (2000) Sir2 links chromatin silencing, metabolism, and aging. Genes Dev 14(9):1021–1026
  - Fang Y, Tang S, Li X (2019) Sirtuins in metabolic and epigenetic regulation of stem cells. Trends Endocrinol Metab 30(3):177–188
  - Watroba M, Szukiewicz D (2016) The role of sirtuins in aging and age-related diseases. Adv Med Sci 61(1):52–62
  - Wątroba M, Dudek I, Skoda M, Stangret A, Rzodkiewicz P, Szukiewicz D (2017) Sirtuins, epigenetics and longevity. Ageing Res Rev 40:11–19
  - Kanfi Y, Shalman R, Peshti V, Pilosof SN, Gozlan YM, Pearson KJ et al (2008) Regulation of SIRT6 protein levels by nutrient availability. FEBS Lett 582(5):543–548
  - Alcendor RR, Gao S, Zhai P, Zablocki D, Holle E, Yu X et al (2007) Sirt1 regulates aging and resistance to oxidative stress in the heart. Circ Res 100(10):1512–1521
  - Tasselli L, Zheng W, Chua KF (2017) SIRT6: novel mechanisms and links to aging and disease. Trends Endocrinol Metab 28(3):168–185
  - 40. Mostoslavsky R, Chua KF, Lombard DB, Pang WW, Fischer MR, Gellon L et al (2006) Genomic instability and aging-like phenotype in the absence of mammalian SIRT6. Cell 124(2):315–329
  - Kanfi Y, Naiman S, Amir G, Peshti V, Zinman G, Nahum L et al (2012) The sirtuin SIRT6 regulates lifespan in male mice. Nature 483(7388):218–221
  - 42. Xiao N-M, Zhang Y-M, Zheng Q, Gu J (2004) Klotho is a serum factor related to human aging. Chin Med J 117(5):742–747
  - Laszczyk AM, Fox-Quick S, Vo HT, Nettles D, Pugh PC, Overstreet-Wadiche L, King GD (2017) Klotho regulates postnatal neurogenesis and protects against age-related spatial memory loss. Neurobiol Aging 59:41–54
  - 44. Benoit B, Meugnier E, Castelli M, Chanon S, Vieille-Marchiset A, Durand C, Bendridi N, Pesenti S, Monternier PA, Durieux AC, Freyssenet D, Rieusset J, Lefai E, Vidal H, Ruzzin J (2017) Fibroblast growth factor 19 regulates skeletal muscle mass and ameliorates muscle wasting in mice. Nat Med 23:990–996
  - 45. Sahu A, Mamiya H, Shinde SN, Cheikhi A, Winter LL, Vo NV et al (2018) Age-related declines in alpha-Klotho drive progenitor cell mitochondrial dysfunction and impaired muscle regeneration. Nat Commun 9(1):4859. https://doi.org/10.1038/s41467-018-07253-3
  - 46. Kenyon C (2005) The plasticity of aging: insights from long-lived mutants. Cell 120(4):449-460
  - Lee W-S, Kim J (2018) Insulin-like growth factor-1 signaling in cardiac aging. Biochim Biophys Acta Mol Basis Dis 1864(5 Pt B):1931–1938
  - Tullet JM, Hertweck M, An JH, Baker J, Hwang JY, Liu S et al (2008) Direct inhibition of the longevity-promoting factor SKN-1 by insulin-like signaling in C. elegans. Cell 132(6):1025–1038
  - Bartke A (2008) Impact of reduced insulin-like growth factor-1/insulin signaling on aging in mammals: novel findings. Aging Cell 7(3):285–290
  - Ben-Avraham D, Govindaraju DR, Budagov T, Fradin D, Durda P, Liu B et al (2017) The GH receptor exon 3 deletion is a marker of male-specific exceptional longevity associated with increased GH sensitivity and taller stature. Sci Adv 3(6):e1602025. https://doi.org/10.1126/ sciadv.1602025
  - Suh Y, Atzmon G, Cho M-O, Hwang D, Liu B, Leahy DJ et al (2008) Functionally significant insulin-like growth factor I receptor mutations in centenarians. Proc Natl Acad Sci U S A 105(9):3438–3442

- Kennedy BK, Lamming DW (2016) The mechanistic target of rapamycin: the grand conduc-TOR of metabolism and aging. Cell Metab 23(6):990–1003
- Shimobayashi M, Hall MN (2014) Making new contacts: the mTOR network in metabolism and signalling crosstalk. Nat Rev Mol Cell Biol 15(3):155–162
- 54. Arriola Apelo SI, Neuman JC, Baar EL, Syed FA, Cummings NE, Brar HK, Pumper CP, Kimple ME, Lamming DW (2016) Alternative rapamycin treatment regimens mitigate the impact of rapamycin on glucose homeostasis and the immune system. Aging Cell 15:28–38
- 55. Coppé J-P, Patil CK, Rodier F, Sun Y, Muñoz DP, Goldstein J et al (2008) Senescenceassociated secretory phenotypes reveal cell-nonautonomous functions of oncogenic RAS and the p53 tumor suppressor. PLoS Biol 6(12):2853–2868. https://doi.org/10.1371/journal. pbio.0060301
- Campisi J (2011) Cellular senescence: putting the paradoxes in perspective. Curr Opin Genet Dev 21(1):107–112
- Schafer MJ, White TA, Iijima K, Haak AJ, Ligresti G, Atkinson EJ et al (2017) Cellular senescence mediates fibrotic pulmonary disease. Nat Commun 8:14532. https://doi.org/10.1038/ ncomms14532
- Rodier F, Coppé J-P, Patil CK, Hoeijmakers WA, Muñoz DP, Raza SR et al (2009) Persistent DNA damage signalling triggers senescence-associated inflammatory cytokine secretion. Nat Cell Biol 11(8):973–979
- Shelton DN, Chang E, Whittier PS, Choi D, Funk WD (1999) Microarray analysis of replicative senescence. Curr Biol 9(17):939–945
- 60. Kang C, Xu Q, Martin TD, Li MZ, Demaria M, Aron L et al (2015) The DNA damage response induces inflammation and senescence by inhibiting autophagy of GATA4. Science 349(6255):aaa5612. https://doi.org/10.1126/science.aaa5612
- Acosta JC, Banito A, Wuestefeld T, Georgilis A, Janich P, Morton JP et al (2013) A complex secretory program orchestrated by the inflammasome controls paracrine senescence. Nat Cell Biol 15(8):978–990
- 62. Orjalo AV, Bhaumik D, Gengler BK, Scott GK, Campisi J (2009) Cell surface-bound IL-1α is an upstream regulator of the senescence-associated IL-6/IL-8 cytokine network. Proc Natl Acad Sci U S A 106(40):17031–17036
- Mariotti M, Castiglioni S, Bernardini D, Maier JA (2006) Interleukin 1 alpha is a marker of endothelial cellular senescent. Immun Ageing 3:4. https://doi.org/10.1186/1742-4933-3-4
- 64. Freund A, Patil CK, Campisi J (2011) p38MAPK is a novel DNA damage responseindependent regulator of the senescence-associated secretory phenotype. EMBO J 30(8):1536–1548
- 65. Laberge R-M, Sun Y, Orjalo AV, Patil CK, Freund A, Zhou L et al (2015) MTOR regulates the pro-tumorigenic senescence-associated secretory phenotype by promoting IL1A translation. Nat Cell Biol 17(8):1049–1061
- Meyer SC, Levine RL (2014) Molecular pathways: molecular basis for sensitivity and resistance to JAK kinase inhibitors. Clin Cancer Res 20(8):2051–2059
- Liu S, Uppal H, Demaria M, Desprez P-Y, Campisi J, Kapahi P (2015) Simvastatin suppresses breast cancer cell proliferation induced by senescent cells. Sci Rep 5:17895. https:// doi.org/10.1038/srep17895
- Kirkland JL, Tchkonia T (2017) Cellular senescence: a translational perspective. EBioMedicine 21:21–28
- Lamming DW, Ye L, Katajisto P, Goncalves MD, Saitoh M, Stevens DM et al (2012) Rapamycin-induced insulin resistance is mediated by mTORC2 loss and uncoupled from longevity. Science 335(6076):1638–1643
- 70. Lehmann BD, Paine MS, Brooks AM, McCubrey JA, Renegar RH, Wang R et al (2008) Senescence-associated exosome release from human prostate cancer cells. Cancer Res 68(19):7864–7871
- Wang AL, Lukas TJ, Yuan M, Du N, Tso MO, Neufeld AH (2009) Autophagy and exosomes in the aged retinal pigment epithelium: possible relevance to drusen formation and age-related macular degeneration. PLoS One 4(1):e4160. https://doi.org/10.1371/journal.pone.0004160

- Takasugi M, Okada R, Takahashi A, Chen DV, Watanabe S, Hara E (2017) Small extracellular vesicles secreted from senescent cells promote cancer cell proliferation through EphA2. Nat Commun 8:15729. https://doi.org/10.1038/ncomms15728
- 73. Mitsuhashi M, Taub DD, Kapogiannis D, Eitan E, Zukley L, Mattson MP et al (2013) Aging enhances release of exosomal cytokine mRNAs by Aβ1-42-stimulated macrophages. FASEB J 27(12):5141–5150
- 74. van Balkom BW, De Jong OG, Smits M, Brummelman J, den Ouden K, de Bree PM et al (2013) Endothelial cells require miR-214 to secrete exosomes that suppress senescence and induce angiogenesis in human and mouse endothelial cells. Blood 121(19):3997–4006
- 75. Gan W, Liu XL, Yu T, Zou YG, Li TT, Wang S et al (2018) Urinary 8-oxo-7,8-dihydroguanosine as a potential biomarker of aging. Front Aging Neurosci 10:34. https://doi.org/10.3389/ fnagi.2018.00034
- Rodier F (2013) Detection of the senescence-associated secretory phenotype (SASP). Methods Mol Biol 965:165–173
- Thomas V, Clark J, Doré J (2015) Fecal microbiota analysis: an overview of sample collection methods and sequencing strategies. Future Microbiol 10(9):1485–1504
- Schneiderhan J, Master-Hunter T, Locke A (2016) Targeting gut flora to treat and prevent disease. J Fam Pract 65(1):34–38
- Rodriguez-Castaño GP, Caro-Quintero A, Reyes A, Lizcano F (2017) Advances in gut microbiome research, opening new strategies to cope with a western lifestyle. Front Genet 7:224. https://doi.org/10.3389/fgene.2016.00224
- Vaiserman AM, Koliada AK, Marotta F (2017) Gut microbiota: a player in aging and a target for anti-aging intervention. Ageing Res Rev 35:36–45
- Guigoz Y, Doré J, Schiffrin EJ (2008) The inflammatory status of old age can be nurtured from the intestinal environment. Curr Opin Clin Nutr Metab Care 11(1):13–20
- Hopkins M, Sharp R, Macfarlane G (2002) Variation in human intestinal microbiota with age. Dig Liver Dis 34 Suppl 2:S12–S18
- Rondanelli M, Giacosa A, Faliva MA, Perna S, Allieri F, Castellazzi AM (2015) Review on microbiota and effectiveness of probiotics use in older. World J Clin Cases 3(2):156–162
- Kato K, Odamaki T, Mitsuyama E, Sugahara H, J-z X, Osawa R (2017) Age-related changes in the composition of gut Bifidobacterium species. Cur Microbiol 74(8):987–995
- 85. O'Hagan C, Li JV, Marchesi JR, Plummer S, Garaiova I, Good MA (2017) Long-term multi-species Lactobacillus and Bifidobacterium dietary supplement enhances memory and changes regional brain metabolites in middle-aged rats. Neurobiol Learn Mem 144:36–47
- Gao J, Xu K, Liu H, Liu G, Bai M, Peng C et al (2018) Impact of the gut microbiota on intestinal immunity mediated by tryptophan metabolism. Front Cell Infect Microbiol 8:13. https:// doi.org/10.3389/fcimb.2018.00013
- Collino S, Montoliu I, Martin F-PJ, Scherer M, Mari D, Salvioli S et al (2013) Metabolic signatures of extreme longevity in northern Italian centenarians reveal a complex remodeling of lipids, amino acids, and gut microbiota metabolism. PLoS One 8(3):e56564. https://doi. org/10.1371/journal.pone.0056564
- Mace J, Porter R, Dalrymple-Alford J, Wesnes K, Anderson T (2010) Effects of acute tryptophan depletion on neuropsychological and motor function in Parkinson's disease. J Psychopharmacol 24(10):1465–1472
- Roberts SB, Rosenberg I (2006) Nutrition and aging: changes in the regulation of energy metabolism with aging. Physiol Rev 86(2):651–667
- 90. Verdin E (2015) NAD+ in aging, metabolism, and neurodegeneration. Science 350(6265):1208–1213
- Mendelsohn AR, Larrick JW (2017) The NAD+/PARP1/SIRT1 axis in aging. Rejuvenation Res 20(3):244–247
- 92. Zhang H, Ryu D, Wu Y, Gariani K, Wang X, Luan P et al (2016) NAD+ repletion improves mitochondrial and stem cell function and enhances life span in mice. Science 352(6292):1436–1443

- Igarashi M, Miura M, Williams E, Jaksch F, Kadowaki T, Yamauchi T et al (2019) NAD+ supplementation rejuvenates aged gut adult stem cells. Aging Cell 18:e12935. https://doi. org/10.1111/acel.12935
- 94. Nacarelli T, Lau L, Fukumoto T, Zundell J, Fatkhutdinov N, Wu S et al (2019) NAD+ metabolism governs the proinflammatory senescence-associated secretome. Nat Cell Biol 21(3):397–407
- 95. Balasubramanian P, Howell PR, Anderson RM (2017) Aging and caloric restriction research: a biological perspective with translational potential. EBioMedicine 21:37–44
- 96. Meidenbauer JJ, Ta N, Seyfried TN (2014) Influence of a ketogenic diet, fish-oil, and calorie restriction on plasma metabolites and lipids in C57BL/6J mice. Nutr Metab (Lond) 11:23. https://doi.org/10.1186/1743-7075-11-23
- Roberts MN, Wallace MA, Tomilov AA, Zhou Z, Marcotte GR, Tran D et al (2017) A ketogenic diet extends longevity and healthspan in adult mice. Cell Metab 26(3):539–546.e5
- 98. Newman JC, Verdin E (2017)  $\beta$ -Hydroxybutyrate: a signaling metabolite. Annu Rev Nutr 37:51–76
- 99. Scheibye-Knudsen M, Mitchell SJ, Fang EF, Iyama T, Ward T, Wang J et al (2014) A high-fat diet and NAD+ activate Sirt1 to rescue premature aging in cockayne syndrome. Cell Metab 20(5):840–855
- 100. Weber DD, Aminazdeh-Gohari S, Kofler B (2018) Ketogenic diet in cancer therapy. Aging (Albany NY) 10(2):164–165
- 101. Pyo J-O, Yoo S-M, Ahn H-H, Nah J, Hong S-H, Kam T-I et al (2013) Overexpression of Atg5 in mice activates autophagy and extends lifespan. Nat Commun 4:2300. https://doi. org/10.1038/ncomms3300
- 102. Zhang C, Cuervo AM (2008) Restoration of chaperone-mediated autophagy in aging liver improves cellular maintenance and hepatic function. Nat Med 14(9):959–965
- Jiao J, Demontis F (2017) Skeletal muscle autophagy and its role in sarcopenia and organismal aging. Curr Opin Pharmacol 34:1–6

# Chapter 4 Sex Differences in Aging and Associated Biomarkers



Natalie Thomas, Caroline Gurvich, and Jayashri Kulkarni

# 1 Introduction

Aging is a natural process defined by the gradual, time-dependent decline of physiological and behavioural function, and individuals of the same chronological age can show great variability in these parameters [1]. The capacity of biological systems to continuously adjust for optimal functioning despite ever changing environments is essential for healthy aging, and variability in these adaptive homeostatic mechanisms may reflect the heterogeneity observed [2].

Adaptive homeostasis refers to "The transient expansion or contraction of the homeostatic range in response to exposure to sub-toxic, non-damaging, signalling molecules or events, or the removal or cessation of such molecules or events" [3], and represents the framework in which the body dynamically maintains biological and behavioural processes required for life. The endocrine, metabolic and immune systems are well established in the literature as homeostatic mechanisms [3]. These systems are complex and integrated, involving feedback-dependent mechanisms and cross-talk across networks, thereby contributing to overall system regulation [4]. These integrated systems are ultimately governed by the complex interaction between genetic and environmental factors, including lifestyle factors, which may affect homeostatic mechanisms and ultimately change the progression of aging. Growing evidence suggests that the adaptability of these homeostatic systems decline with age, with many linked to age-related morbidities suggesting a shared pathophysiology of disrupted cellular homeostasis [2]. This change or decline in key homeostatic systems may therefore reflect the inability to activate or modulate several adaptive responses.

N. Thomas · C. Gurvich · J. Kulkarni (🖂)

Monash Alfred Psychiatry Research Centre, Monash University, Melbourne, VIC, Australia e-mail: jayashri.kulkarni@monash.edu

<sup>©</sup> Springer Nature Switzerland AG 2019

P. C. Guest (ed.), *Reviews on Biomarker Studies in Aging and Anti-Aging Research*, Advances in Experimental Medicine and Biology 1178, https://doi.org/10.1007/978-3-030-25650-0\_4

With an ever-increasing aging population, interest in biomarkers of healthy or accelerated aging is growing. Although there is no universally accepted definition of 'aging' biomarkers, attempts to identify objective biomarkers often point to mediators of homeostatic systems including physiological functions, cognitive processes, and endocrine and immune functions [5]. Importantly, sex differences are often reported in many of these adaptive systems, and this may reflect, to some extent, sex differences observed in aging and age-related disease states. Therefore, it is imperative to consider sex and gender differences in aging and age-related diseases.

Known sex differences in key homeostatic mediators may influence the differential susceptibility and resilience to aging outcomes between males and females. Therefore, within the context of biomarkers in aging, it is important to consider biological sex differences and the role of gender, which includes a person's psychosocial and cultural self-identification [6], and how these may influence analytical measurement. From a physiological standpoint, biological sex may play an important factor in the variability of biomarkers due to the capacity of sex chromosomes and sex hormones (i.e., estrogen, testosterone), to alter biomarkers present in blood, saliva, cerebral spinal fluid, or tissue samples [4]. From a behavioural standpoint, biological sex hormones, sex chromosomes, and gender may impart differential adaptive processes seen in the cognitive domain [7].

This chapter aims to outline sex differences in key homeostatic domains thought to be associated with the pathophysiology of aging, and which are often proposed as biomarkers of aging and age-related disease states. This includes summarising the available literature of sex-based differences and hormonal status with regards to the gonadal and adrenal endocrine systems and immune function.

### 2 Sex Differences in Age-Related Diseases

Sex differences in longevity, mortality and age-related diseases are well established in the literature. In developed nations, females tend to live longer than males and have notably lower death rates than men at all ages. However, it is also known that females suffer from higher levels of morbidity than men [8]. Salient gender-related differences have been associated with risk, clinical expression, treatment response and course of several age-related neurodegenerative disorders, such as Alzheimer's disease and Parkinson's disease [9].

Alzheimer's disease, the most common cause of dementia, disproportionately affects females, with a higher incidence compared to males [10]. Females have a faster rate of cognitive and functional decline following diagnosis and appear to suffer significantly greater cognitive impairments after accounting for sex differences in age, education and dementia severity [11]. Sexual dimorphisms in Alzheimer's disease pathologies have been reported in several mouse models, with females expressing greater pathology [12]. Sex has also been demonstrated to modify genetic risk factors for Alzheimer's disease. For example, the apolipoprotein E gene  $\varepsilon$ 4 allele confers a greater Alzheimer's disease risk in females, as compared to

males [13]. In contrast, males have at least a two-fold greater risk and prevalence of Parkinson's disease [14]. In comparison to females, males have a later onset of Parkinson's disease. Clinical and cognitive profiles also differ between males and females with Parkinson's disease. Males are more likely to present with rigidity, rapid eye movement behaviour disorder and sleep disturbances. The cognitive profile in males is more likely to include verbal fluency deficits and a lack of facial emotions. Females are more likely to present with a tremor dominant form of Parkinson's disease. A reduction in visuospatial cognition is more frequent in females [9, 15, 16]. Possible reasons for these sex differences in age-related, neuro-degenerative disorders and longevity are thought to be due to intrinsic differences based on genes, sex hormones, and reproductive physiology. These can confer differential risks of morbidity. In addition, extrinsic factors such as lifestyle, health habits, exercise, nutrition, and others may also have a connection with sex differences in biological vigor as potential moderators.

### 3 Endocrine Markers of Aging

The endocrine system represents a key facilitator of homeostatic function, maintaining critical functions including development, growth, metabolism, stress responses, reproduction, and sleep patterns. As such, it is one of the most important regulators of physiology over the lifespan. The hypothalamus can be considered as the link between the nervous system and endocrine system, continually adjusting according to internal and external environments using feedback mechanisms, and subsequently regulating all hormone-related physiology and behaviour. With advancing age, alterations in hormonal networks, with concomitant excesses or deficits in steroidal hormones occur [17]. This is further exacerbated by poor sensitivity of tissues to their action, and a decreased sensitivity toward feedback control [18]. These alterations are thought to be translated clinically to losses of function seen in aging of the reproduction system including menopause, developmental and maintenance growth axes, and the adrenal endocrine axis [18].

# 4 Aging of the Female and Male Reproductive System and the Aging Gonadotropic Axis

Sex steroid hormones, including estrogens and androgens (testosterones) are regulated by the hypothalamus pituitary gonadal (HPG) axis. While the endocrine feedback mechanisms are complex, a simplified account of the HPG axis suggests that the hypothalamus releases gonadotropin-releasing hormone (GnRH), stimulating the anterior pituitary to produce and secrete gonadotropins, luteinizing hormone (LH) and follicle-stimulating hormone (FSH). The ovary, comprised of two cellular components, is synergistically stimulated by FSH and LH, leading to the release of sex steroids [19]. FSH and LH then complete the endocrine feedback loop by posing both positive and negative feedback effects on the axes, depending on timing and context [20].

A clear change observed throughout aging is the decline in sex hormone production. As women age, the natural transition from reproductive years to menopause is associated with fluctuating and eventual decreasing levels of ovarian sex steroids. Suggested to be a predictor of reduced ovarian reserve, gonadotropin FSH increases pre-menopausally [21]. Subsequent to the menopausal transition, eventual decreases in FSH and LH occur [18]. Again, alteration at the level of the hypothalamuspituitary-gonadal axis is thought to principally contribute to this progressive decline. This appears to be moderated by increased levels of GnRH which increase with age with hypothalamic adaptations during aging, favouring its release [22, 23]. Moreover, the feedback effect of estrogen on FSH responsiveness is greater than on LH and is attenuated with aging [24].

This natural consequence of aging of the reproductive axis and sex-steroid privation leads to the loss of reproductive capability, in addition to a sequelae of clinical symptomatologies including loss of muscle and bone mass, visceral tissue accumulation, insulin resistance, and negative impacts on mood, cognition, and importantly, quality of life [25]. A recent meta-analysis of 310,329 women demonstrated that younger age at menopause (occurring prior to 45 years old) was associated with a higher risk of coronary heart disease, stroke, cardiovascular mortality, and of allcause mortality [26]. In addition, the occurrence of age-related morbidities including Alzheimer's disease and Parkinson's disease has led to the suggestion that the loss of sex steroid hormones, in both female and males, is a contributor to the aetiology [27, 28].

Although not as dramatic as seen during the female menopausal transition, a modest, progressive decline in anabolic androgen is observed in males throughout the aging process [29]. No specific age can be defined as to when this process begins, although several cross-sectional and longitudinal studies have confirmed that testosterone (free and bound) decline in aging, while LH increases with age [30-34]. Recently, weight and lifestyle factors have been shown to moderate these hormonal changes [30]. In addition, the levels of sex hormone binding globulin (SHBG) increase during aging which binds testosterone, resulting in relatively less free biologically-available testosterone [35]. Clinically, low testosterone levels have been associated with increased mortality in men, showing an approximate doubling in mortality risk compared to men with normal levels [36]. Associations between low testosterone and adverse outcomes have also been reported, including diabetes mellitus, osteoporosis, decreased libido, and cardiovascular disease [37-39]. In a recent biomarker analysis that included sex hormones in an aging 'panel', it was reported that low levels of testosterone are correlated with high-sensitivity cardiovascular risk factors, consistent with a potential increased risk of cardiovascular disease [40].

Fluctuating hormone levels throughout the menstrual cycle of reproductive aged women and in perimenopausal women should be also be considered, as female hormonal status has been shown to modulate many age-related biomarkers. As an example, stress-induced cortisol levels are associated with impaired emotional retrieval [41] and affect serum biomarkers in clinical tests only in some menstrual phases [42]. In addition, hormonal contraceptives used by many women of reproductive age have shown to vary clinical measurement of serum biomarkers [42], and this is rarely considered within the context of system dynamics and levels of molecular outputs. Although the contributions and mechanisms are yet to be fully elucidated, sex steroid hormones exhibit clear functional activity on age-related processes and may govern, in large part, the sexual dimorphisms observed in many age-related morbidities.

### 5 Dehydroepiandrosterone in Aging

Adrenal secretion of dehydroepiandrosterone (DHEA) also profoundly fluctuates across the lifespan and represents one of the largest endocrine changes observed in human aging. Although the physiological role of DHEA is not yet completely understood, its apparent influence in age and age-related disease states has yielded much interest.

DHEA is predominately synthesised within the adrenal zona reticularis of the adrenal cortex in response to adrenocorticotropic hormone (ACTH) and subsequently released into the bloodstream [43]. The majority of circulating DHEA is converted to its sulphated derivative DHEA-S via the enzyme steroid sulphotransferase, which is mostly bound to the carrier protein, albumin. As the precursor to the sex steroid hormones, it is also synthesised to a smaller extent in human gonads [44]. DHEA-S serves as a precursor to approximately 50% of androgens in adult men, 75% of active estrogens in premenopausal and nearly 100% in postmenopausal women [45]. More recently, reports pertaining to its de novo synthesis in human brain highlights the important neurosteroidal properties that DHEA may have, posited to have direct activity at estradiol receptors [46] and NMDA receptors [47], possess neuroprotective properties, and participate in neurogenesis, catechol-amine synthesis and secretion, in addition to anti-inflammatory effects [44]. Indeed, it has been reported that concentrations of DHEA/DHEA-S are considerably higher in the brain than in any other organ.

At birth, DHEA-S levels are markedly high until involution of the adrenal fetal zone, which leads to a cessation in DHEA production and it remains dormant until the childhood period, when production once again commences. Peak levels for both females and males are observed when they are in their thirties, after which point DHEA levels fall by approximately 2–3% each year [48], declining to levels that are approximately 20% of peak values in men and 30% of those in women [43, 49].

Several studies have reported the association between low levels of DHEA-S and DHEA and frailty in older populations [50–52]. In addition, several cross-sectional studies, including investigations of serum and cerebrospinal fluid (CSF), have implicated DHEA levels in a number of age-related pathologies including type 2 diabetes mellitus, psychiatric disease states, and neurocognitive disorders, including

Alzheimer's disease [29, 43]. Although many studies have provided support for a positive correlation between higher levels of DHEA and improved health outcomes, including muscle strength, higher bone density and increased longevity in males, other studies have demonstrated that DHEA is associated with negative health outcomes. For example, lower DHEA-S levels have been associated with greater risk of ischemic stroke, having been adjusted for appropriate potential confounders [53].

Due to such accumulating literature, DHEA has drawn much attention as a putative 'anti-aging' supplement, although little is known about the underlying mechanism with regards to its progressive decline and associated pathologies. Hypotheses pertaining to plausible biological mechanisms include the ability of DHEA/DHEA-S to modulate energy metabolism, systemic inflammation, and that it may counteract or modulate the stress endocrine hormone, cortisol [54–56]. Given the critical role of sex steroids and age-related associations, the function of DHEA as an alternative source of androgens and estrogens to local tissues may be a plausible way in which DHEA/DHEA-S protects against aging and age-related pathologies [57]. Although there is some evidence to suggest that DHEA supplementation may improve skin status and bone turnover in older females, there are few clinical studies powered enough to conclude the effects of DHEA on aging and age-related diseases. In addition, data on the safety profile of long-term DHEA supplementation are still lacking [58].

### 5.1 Sex Differences in DHEA

Differential trajectories of DHEA levels between the sexes begin following the onset of adrenarche, and differences in levels can again be observed during the perimenopausal period in women, and in older people [59]. Peak levels have also been reported to occur earlier for females [60]. The biological reason for this difference has yet to be elucidated but it has been suggested that higher levels observed in males may reflect chromosomal differences [61]. The gene for steroid sulfatase, which degrades androgens, is located on the X chromosome. Therefore, as males only have one copy of the gene, they consequently have less degradation and higher DHEA concentrations [57]. Although the literature reports somewhat inconsistent results, a recent review concluded that enough evidence supports the association between DHEA and aging in males, but in females this may not exist, or displays a non-linear, U-shaped association. Larger studies, with a prior stratification of sex, are required to confirm these findings [62].

#### 6 Analyses of Steroidal Hormones

The complexity of sex steroid measurement should not be underestimated, and there is current debate over the best technologies to use (see [63] for overview). These steroids exhibit only modest differences in structure, but which amount to profound

functional differences. Owing to such subtle differences in structure and small size, steroids are not easy to measure.

Although immunoassays have remained the mainstay of steroid analysis, the reliability of the specificity and sensitivity of these assays have come into questions, particularly commonly used commercially derived assays [64]. Unless preanalytical extraction methods or chromatographic steps are introduced to purify samples prior to the immune-assay, these assays are prone to inaccuracy due to their non-specificity [63]. Mass spectrometry has introduced far greater specificity of steroidal hormones and allows for specific analysis of different steroids (i.e., estrogen, and testosterone). Currently this technique provides the highest specificity due to mass spectrometry providing the exact structural information of the analyte, allowing for specificity between steroid analytes and their precursors. However, it is noted that costs and expertise required for such analyses may prevent many researchers from conducting such analyses. As technology advances and becomes cheaper, this gold standard technology will grow in use.

Relevant to this chapter, difficulty in the measurement of steroids has been proposed for the reason for the inconsistent findings seen in the DHEA aging literature. The low specificity and precision of immunoassays may have yielded incorrect measurements in subjects low in DHEA/ DHEA-S [62].

# 7 Aging of the Adrenal Endocrine System: The Hypothalamus-Pituitary-Adrenal Axis

Activation of the hypothalamus-pituitary- adrenal (HPA) endocrine axis and the subsequent stress response is a fundamental homeostatic mechanism that enables a wide array of physiological and behavioural responses to actual or perceived threat. Initiated at the level of the hypothalamus via corticotrophin releasing hormone (CRF), a signalling cascade then ensues with adrenocorticotropin hormone (ACTH) released from the anterior pituitary, followed by glucocorticoids being synthesised from cholesterol in the adrenal cortex, and released directly into the bloodstream. Glucocorticoids are lipophilic and largely transported bound to cortisol binding globulin (CBG) leaving only a small fraction (10%) biologically active [65]. As glucocorticoids can readily cross the blood-brain-barrier, the steroids consequently can exert action at both local central nervous system level and systemic peripheral levels, where it induces or represses the transcription of a plethora of target genes [66, 67]. In this way, glucocorticoids orchestrate a vast repertoire of adjustments to adapt to external or internal changes to the environment, ensuring optimal regulation of several interlinked regulatory systems including reproduction, growth, and immunity, and cognitive signalling processes [68]. In addition, fast acting, non-genomic action of glucocorticoids occur, engaging glutamatergic system alterations [69].
Glucocorticoids mediate its own activity and termination via two receptors; the high affinity type 1 mineralocorticoid receptor (MR) and lower affinity type 11 glucocorticoid receptors (GR) [70]. The high affinity of MR to glucocorticoids is thought to be responsible for regulating tonic activity and dictates the basal circadian and ultradian rhythms observed across the day [71]. GR are found widely distributed throughout the brain, with high density found in most neurons and glia. With lower affinity to physiological GC, these receptors only become functionally activated as the stress endocrine hormone cortisol rise following a stressor and play an important role in mediating GC effects on mobilising energy stores, inflammation and neural function [72]. Importantly, it is thought to be primarily responsible for blunting further activity of the HPA stress response. Both receptors act in synergy to mediate HPA dynamics important for several of these systems of homeostasis, eloquently fine tuning the HPA dynamics depending on the necessary action. An example of this is found within cognitive processes, such as learning and memory. GR and MR are both abundantly expressed in neurons of the hippocampus, prefrontal cortex, and amygdala [73] where they are thought to have opposing functions which regulate hippocampal synaptic neuroplasticity during the stress response, important in long term potentiation mechanism of learning and memory. Preclinical rat models have demonstrated that activation of the MRs may be a prerequisite for hippocampal plasticity, while GRs may exert an inhibitory effect of plasticity [74]. Although initial sensitivity to feedback of glucocorticoids due to the high expression of GRs and MRs, after repeated responses seen in acute and chronic stress, the hippocampus down-regulates receptor expression [75]. In line with this, although acute, moderated activation of the HPA axis is a necessary and beneficial response, such chronic activation may lead to tissue damage or receptor desensitisation, resulting in attenuated negative feedback, and ultimately adverse outcomes [76]. Individual differences in vulnerability and resistance to stress, governed by HPA dynamics [77] may thereby contribute to the heterogeneity of the aging, and agerelated diseases.

There is evidence that HPA axis dynamics [78] and GR expression levels in areas of the dentate gyrus [79] change over the natural process of aging and are postulated to be a major driver in accelerated aging, with lower diurnal cortisol levels having been associated with longevity [80]. In healthy humans, the HPA axis, and glucocorticoid output typically follows a diurnal pattern. The typical pattern displays a sharp increase upon awakening, known as the cortisol awakening response (CAR), followed by reduced levels throughout the day, until the sleep nadir [81]. Increased basal cortisol output [82], diurnal amplitude flattening [83], and an altered CAR response both in magnitude and in day to day variability [84] have all been associated in aging models [61]. In addition, attenuated negative feedback by glucocorticoid receptors at the level of the paraventricular nucleus, hypothalamus, hippocampus, and prefrontal cortex have been reported [61, 70]. Furthermore, the 11-β hydroxysteroid dehydrogenase enzymatic activity responsible for the interconversion of inactive cortisone to active cortisol has been reported to decrease with age, resulting in the inefficient inactivation of active cortisol to inactive cortisone [85]. The two isoforms of this enzyme have also been suggested as a potential key

mechanism in the acceleration of aging [86]. In addition, the systemic  $11-\beta$  hydroxysteroid dehydrogenase isoform 1 is predictive of progressive brain atrophy and cognitive decline [87].

Such age-related changes in HPA dynamics are often related with declined cognitive processes, cardiovascular alterations, and psychiatric disease states. For example, a higher 24 h urine output has been associated with Alzheimer's disease [88]. Mechanistically speaking, this may be a result of cortisol induced apoptosis and death of hippocampal neurons [89], as observed by decreased adult hippocampal neurogenesis in an Alzheimer's disease model [90].

#### 7.1 Sex Differences in HPA Axis Regulation

Research suggests there are substantial sex differences in HPA-axis dynamics and cortisol levels during regular diurnal maintenance and in response to stress, which may contribute to differential aging dynamics between the sexes [91]. In vitro models have shown that female rodents exhibit greater basal corticosterone production by the adrenal glands [92] and have a more robust HPA axis response to both physical and psychological stressors, thought to be a result of circulating estradiol levels [93]. Clinical investigations have also provided support for the sexual dimorphism of the HPA axis stress response [94]. The HPA axis stress response in females is characterised by a larger, more sustained secretion of ACTH and cortisol, suggesting enhanced activity and reduced negative feedback [95], while the pulsatile cortisol feedback on ACTH mediated by GC receptors appears to be influenced by sex [96]. In addition, there is evidence indicating that both GRs and MRs are less sensitive to cortisol modulation in females than males, suggesting reduced feedback by autoregulation of these receptors [70].

Although the exact mechanism of action remains to be elucidated, estradiol has been shown to enhance HPA activity, while testosterone appears to have an inhibitory effect by acting upon the hypothalamus [97]. In contrast, there is also evidence to indicate that estradiol, but not testosterone, heightens the cortisol-mediated negative feedback on pulsatile ACTH secretion in both aged men and women [96]. Further research must be conducted to understand this complex interplay between the HPA and HPG axis, across the life-course and considering sexual dimorphisms.

In line with these findings, a clear association between menstrual cycle phase and cortisol basal measurement has been shown. Women in the luteal phase of the menstrual cycle have a similar cortisol response to men, whereas in the follicular phase, attenuation of cortisol production has been shown [98]. Notably, increased CBG is known to occur in cases of estrogen excess (e.g., pregnancy and use of estrogen-containing oral contraceptives, resulting in higher concentrations of total cortisol [99], although free cortisol levels are usually unaltered in states of increased CBG [100]. It is also now appreciated that prenatal exposure to early life stressors and increased levels of glucocorticoids can influence later life behaviour, biological processes, and susceptibility for disease states. Support for sex differences in such early-life programming of the HPA in humans has been demonstrated in rodent models, with evidence for the female HPA axis being more vulnerable to stress and glucocorticoid programming, and female offspring demonstrating an increased HPA reactivity [101]. This may be one mechanism underlying sex differences in later life diseases and sex dimorphisms shown in aging.

## 7.2 Measurement of the HPA Axis

Glucocorticoids have traditionally been assayed using a range of substrates including blood, saliva and urine that provide a sampling procedure that is simple and relatively non-invasive [102]. More recently, cortisol has been measured using hair sampling, which gives a chronic reading of cortisol levels [103]. Salivary cortisol represents the bioavailable, active form of the molecule that escapes the CBGs in the salivary glands and saliva [65]. In addition, unbound cortisol is excreted in the urine and 24 h measurements correlate well with those in serum in cases of cortisol excess [65]. Hair analysis provides the opportunity to determine past cortisol exposure and represents a retrospective index of cumulative cortisol output which is not influenced by the circadian rhythm of the HPA-axis, or by acute stress. As hair grows at an average rate of 1 cm/month, the 1 cm segment closest to the scalp represents total cortisol levels during the most recent month, with the outer segments representing previous months [103–105]. However, several limitations should be noted, as different hair care routines (e.g., use of hair dye), seasonal variations, and storage length have all been documented to influence readings [106]. The time of sample collection is also a critical consideration as cortisol follows a circadian rhythm of secretion. The cortisol awakening response (CAR) is measured using a minimum of three separate sampling times using blood or saliva (upon awakening, 30 min after waking, and 45 min after waking) [107]. However, the majority of studies employ a 'minimal protocol' by analysing one time point only, which has potential implications for the reliability of measurement. For cross-sectional studies it is recommended that up to six consecutive days of samples should be collected to accurately assess single time-points, and that four measures at two consecutive weeks days are required to reliably measure the CAR as a trait measure [108]. HPA axis dynamics and feedback sensitivity can also be measured using a synthetic glucocorticoid, dexamethasone. In clinical testing of suspected hypercortisolemia, the dexamethasone suppression test (DST) can detect abnormal HPA axis activity by a failure to reduce total cortisol production via stimulating the negative feedback mechanism and ACTH production. Interestingly, this has been found to be moderated by genetic variation [109] and use of hormonal contraceptives. The false positive effect found in fertile women using hormonal contraceptives, due to the resultant increased levels of CBG and consequent increase in total cortisol concentrations,

was found to be eliminated after one-week cessation of contraceptive use [110]. Various stressful stimuli that can reliably induce the cortisol response in human research participants can also be used to evaluate HPA dynamics. These include standardised psychosocial protocols, such as the Trier Social Stress Test (TSST), and physical stressors (electrical stimulation).

## 8 Aging of the Immune System

A functional immune system is critical for protecting the host against infections and malignancies, regulation of wound healing, and ultimately for separation of the 'self' from surrounding organisms that compete for space and resources [111]. The immune system is coordinated into two sub-classifications that govern the immune response. The innate immune system mounts a fast, non-discriminatory response, initiated by macrophages and dendritic cells that produce inflammatory mediators, including histamine, bradykinin, serotonin, leukotrienes, prostaglandins, cytokines and several growth factors and enzymes. This promotes either tissue repair/regeneration with subsequent resolution of inflammation, or tissue remodelling/fibrosis [112]. The adaptive response is initiated if the innate immune response to a specific pathogen. This more sophisticated arm of the immune system requires recruitment of B and T cells, followed by antibody and cell-mediated responses, respectively, and consequently has a targeted, albeit slower action [113].

Like all successful adaptive homeostatic responses, subsequent to the activation of the immune response, an ensuing cessation or resolution of inflammation must occur, returning the tissue environment to the basal state allowing for tissue repair/regeneration. Recruitment of macrophages to an anti-inflammatory phenotype is critical to this process and involves the release of anti-inflammatory cytokines including interleukin (IL)-10 and TGF- $\beta$ , as well as lipid mediators [114]. An over-activated immune response leads to tissue remodelling (tissue fibrosis), autoimmune and neurodegenerative diseases, rather than tissue regeneration [114, 115]. Changes in inflammatory mediators have been identified in age-related syndromes including type 2 diabetes, Alzheimer's disease and cardiovascular disease [116].

During aging, the immune system loses the ability to protect against pathogens effectively, and fails to support appropriate wound healing [111]. Such a decline in effective immunological functioning results in these subjects not responding efficiently to novel or previously encountered antigens [117]. The adaptive immunological arm undergoes complex changes in the aging process that includes epigenetic and metabolic changes affecting naïve, memory and effector T cells, and B cells [118]. A decrease in the number of naïve T lymphocytes and increase in memory and effector T cells occurs with age, in addition to the reduced diversity in T-cell receptors with consequential diminished functionality of naïve and memory T-cells [116]. Clinically, this results in most vaccines being less immunogenic and efficient in the elderly [119].

Additionally, aging is also associated with progression toward a chronic, subclinical inflammatory state, characterised by elevation of peripheral proinflammatory chemical mediators in the absence of overt infection, a term coined 'inflamm-aging' [120]. Several human studies have demonstrated an association between elevated levels of cytokines, especially IL-6, and a decline in innate immune function in older individuals [121]. Such a peripheral inflammatory state has been associated with the susceptibility and progression of many agerelated disease states including cancer, diabetes [122], and cognitive and structural brain changes [123]. Moreover, it has been demonstrated as a key risk factor of mortality [124]. In addition, the innate arm of the immune system initiates cross talk with the adaptive arm through antigen presentation, co-stimulatory molecule expression and cytokine production, and consequently can contribute to adaptive immune responses [125]. More recently, it has been discovered that at sites of infection, adaptive immune memory cells can also regulate the innate inflammatory response, by the epigenetic reprogramming of innate immune cells by previous experiences [126]. Interestingly, it has been hypothesised that centenarians exert a counter action to this pro-inflammatory state, by producing increased amounts of anti-inflammatory cytokine mediators including IL-10 and TGF-β [127].

## 8.1 Sex Differences the Immunological Response

There is a growing body of evidence demonstrating inherent sex differences in the immune response, which includes both the innate and adaptive arms of the immune system. On average, females exhibit a stronger innate and adaptive immune response than males, resulting in improved pathogen clearance, vaccine efficiency, but greater predisposition to autoimmunity and chronic inflammatory disorders [128]. Sex chromosomes contribute to genetic differences due to the fact that many immunerelated genes (e.g., FoxP3 and toll like receptors 7 and 8) are present on the X chromosome. As the possibility of incomplete X-chromosome inactivation occurring in females exists, this can lead to over-expression in females of such immune related genes [129, 130]. Sex hormones can also interact with genetic and environmental factors that determine immunity and appear to influence multiple aspects of the immune system, including the contribution to cell differentiation, cytokine profiles, and epigenetic alterations [130, 131]. Estrogen receptors have been shown to regulate cells and pathways in both arms of the immune system in a predominately stimulatory fashion, in addition to immune cell development [132]. On the other hand, androgens have been shown to be primarily immunosuppressive, suppressing immune reactivity and inflammation [133]. Evidence for sex disparity does exists in the immunological response throughout aging, although the majority of studies do not stratify for sex. Females who are in the menopausal period of their reproductive lives show increased peripheral pro-inflammatory markers, higher numbers of natural killer immune cells with reduced cytotoxicity, and reduced numbers of B and T cells, relative to premenopausal females [134]. Research investigating differential sex influence in the innate immune system is still in its infancy, although data exist suggesting an elevated production of inflammatory cytokines and proteins in females compared to males persists among aged individuals [135, 136]. With regards to adaptive immunity, proliferation of T cells, central to the adaptive immune response, shows a decrease with aging, although this rate of decline was significantly lower in females, than in males [137]. Sex differences have also been noted to occur in response to vaccines protecting against influenza, tetanus, pertussis, shingles and pneumococcal infections, and the efficacy of vaccines in older individuals is consistently higher in females than in males [136].

Collectively, the major changes of the innate and adaptive immune system with aging result in vulnerability to certain infections, and decreased efficacy of many vaccines. Although sex differences in the aging immune system have not been studied extensively, addressing such sex differences may aid in greater effectiveness of vaccines and immunotherapies [130]. As several diseases associated with age are also sensitive to alterations of the immune system, it is important to evaluate potentially- associated sex differences in the underlying pathophysiological processes.

### 9 Conclusions

The capacity of homeostatic biological systems to continuously adjust for optimal functioning during healthy aging may reflect heterogeneity observed in the aging process. Importantly, such systems including endocrine, immune and cognitive systems appear to be moderated by sex, which may be a defining factor in determining sex-based differences in age-related processes, and diseases. Aging appears to be multi-factorial and characterised by the dysregulation of complex system dynamics, namely the interplay of these homeostatic mechanisms driving the aging process. It can be hypothesised that dysregulation in any of these systems may impose feedforward alterations on other systems, and dysregulation early in the accelerated aging process may be different for different people. This makes it particularly challenging to research this field, especially early in the aging process. Composite biomarker panels are required at multiple time points in order to target multiple systems and to better understand the chronicity of biomarkers during aging. Moreover, it is critical that future studies specifically address sex-effects when studying the aging process and age-related diseases, to enable sex differences, which are non-modifiable risk factors, to be differentiated from potentially modifiable lifestyle and environmental risk factors. This is crucial to advance therapeutic options that may include sex-based and hormone-based interventions. Policies put forward by the National Institute of Health regarding the requirement of sex and gender inclusion plans in both preclinical and clinical research, underscores this point. Biomarkers of aging cannot escape this necessity.

## References

- 1. Lowsky DJ, Olshansky SJ, Bhattacharya J, Goldman DP (2013) Heterogeneity in healthy aging. J Gerontol A Biol Sci Med Sci 69(6):640–649
- 2. Pomatto LCD, Davies KJA (2017) The role of declining adaptive homeostasis in aging. J Physiol 595(24):7275–7309
- 3. Davies KJ (2016) Adaptive homeostasis. Mol Asp Med 49:1-7
- Rios FJ, Moustaid-Moussa N, Martins JO (2018) Interplay between hormones, the immune system, and metabolic disorders. Mediat Inflamm 2018:8654212. https://doi. org/10.1155/2018/8654212
- Lara J, Cooper R, Nissan J, Ginty AT, Khaw K-T, Deary IJ et al (2015) A proposed panel of biomarkers of healthy aging. BMC Med 13:222. https://doi.org/10.1186/s12916-015-0470-9
- Podcasy JL, Epperson CN (2016) Considering sex and gender in Alzheimer disease and other dementias. Dialogues Clin Neurosci 18(4):437–446
- Gobinath A, Choleris E, Galea L (2017) Sex, hormones, and genotype interact to influence psychiatric disease, treatment, and behavioral research. J Neurosci Res 95(1–2):50–64
- 8. Austad SN, Fischer KE (2016) Sex differences in lifespan. Cell Metab 23(6):1022–1033
- Ullah MF, Ahmad A, Bhat SH, Abu-Duhier FM, Barreto GE, Ashraf GM (2019) Impact of sex differences and gender specificity on behavioral characteristics and pathophysiology of neurodegenerative disorders. Neurosci Biobehav Rev 102:95–105
- Laws KR, Irvine K, Gale TM (2018) Sex differences in Alzheimer's disease. Curr Opin Psychiatry 31(2):133–139
- 11. Mielke MM, Vemuri P, Rocca WA (2014) Clinical epidemiology of Alzheimer's disease: assessing sex and gender differences. Clin Epidemiol 6:37–48
- Jiao SS, Bu XL, Liu YH, Zhu C, Wang QH, Shen LL et al (2016) Sex dimorphism profile of Alzheimer's disease-type pathologies in an APP/PS1 Mouse Model. Neurotox Res 29(2):256–266
- Altmann A, Tian L, Henderson VW, Greicius MD, Alzheimer's Disease Neuroimaging Initiative Investigators (2014) Sex modifies the APOE-related risk of developing Alzheimer disease. Ann Neurol 75(4):563–573
- 14. Jurado-Coronel JC, Cabezas R, Avila Rodriguez MF, Echeverria V, Garcia-Segura LM, Barreto GE (2018) Sex differences in Parkinson's disease: features on clinical symptoms, treatment outcome, sexual hormones and genetics. Front Neuroendocrinol 50:18–30
- Georgiev D, Hamberg K, Hariz M, Forsgren L, Hariz GM (2017) Gender differences in Parkinson's disease: a clinical perspective. Acta Neurol Scand 136(6):570–584
- Liu R, Umbach DM, Peddada SD, Xu Z, Tröster AI, Huang X et al (2015) Potential sex differences in nonmotor symptoms in early drug-naive Parkinson disease. Neurology 84(21):2107–2115
- 17. Nugent BM, Tobet SA, Lara HE, Lucion AB, Wilson ME, Recabarren SE et al (2012) Hormonal programming across the lifespan. Horm Metab Res 44(8):577–586
- Diamanti-Kandarakis E, Dattilo M, Macut D, Duntas L, Gonos ES, Goulis DG et al (2017) MECHANISMS IN ENDOCRINOLOGY: aging and anti-aging: a combo-endocrinology overview. Eur J Endocrinol 176(6):R283–R308
- Raju GAR, Chavan R, Deenadayal M, Gunasheela D, Gutgutia R, Haripriya G et al (2013) Luteinizing hormone and follicle stimulating hormone synergy: a review of role in controlled ovarian hyper-stimulation. J Hum Reprod Sci 6(4):227–234
- Del Río JP, Alliende MI, Molina N, Serrano FG, Molina S, Vigil P (2018) Steroid hormones and their action in women's brains: the importance of hormonal balance. Front Public Health 6:141. https://doi.org/10.3389/fpubh.2018.00141
- Podfigurna A, Lukaszuk K, Czyzyk A, Kunicki M, Maciejewska-Jeske M, Jakiel G et al (2018) Testing ovarian reserve in pre-menopausal women: why, whom and how? Maturitas 109:112–117

- Gill S, Sharpless JL, Rado K, Hall JE (2002) Evidence that GnRH decreases with gonadal steroid feedback but increases with age in postmenopausal women. J Clin Endocrinol Metab 87(5):2290–2296
- Veldhuis JD (2013) Changes in pituitary function with ageing and implications for patient care. Nat Rev Endocrinol 9(4):205–215
- 24. Shaw ND, Histed SN, Srouji SS, Yang J, Lee H, Hall JE (2010) Estrogen negative feedback on gonadotropin secretion: evidence for a direct pituitary effect in women. J Clin Endocrinol Metab 95(4):1955–1961
- Bhatta S, Blair JA, Casadesus G (2018) Luteinizing hormone involvement in aging female cognition: not all is estrogen loss. Front Endocrinol (Lausanne) 9:544. https://doi.org/10.3389/ fendo.2018.00544
- 26. Muka T, Oliver-Williams C, Kunutsor S, Laven JSE, Fauser BCJM, Chowdhury R et al (2016) Association of age at onset of menopause and time since onset of menopause with cardiovascular outcomes, intermediate vascular traits, and all-cause mortality: a systematic review and meta-analysis. JAMA Cardiol 1(7):767–776
- 27. Hasanpour M, Nourazarian A, Geranmayeh MH, Nikanfar M, Khaki-Khatibi F, Rahbarghazi R (2018) The dynamics of neurosteroids and sex-related hormones in the pathogenesis of Alzheimer's disease. NeuroMolecular Med 20(2):215–224
- Gurvich C, Hoy K, Thomas N, Kulkarni J (2018) Sex differences and the influence of sex hormones on cognition through adulthood and the aging process. Brain Sci 8(9). pii: E163. https://doi.org/10.3390/brainsci8090163
- 29. Jones CM, Boelaert K (2015) The endocrinology of aging: a mini-review. Gerontology 61(4):291–300
- 30. Camacho EM, Huhtaniemi IT, O'Neill TW, Finn JD, Pye SR, Lee DM et al (2013) Ageassociated changes in hypothalamic-pituitary-testicular function in middle-aged and older men are modified by weight change and lifestyle factors: longitudinal results from the European Male Aging Study. Eur J Endocrinol 168(3):445–455
- 31. Lapauw B, Goemaere S, Zmierczak H, Van Pottelbergh I, Mahmoud A, Taes Y et al (2008) The decline of serum testosterone levels in community-dwelling men over 70 years of age: descriptive data and predictors of longitudinal changes. Eur J Endocrinol 159(4):459–468
- 32. Travison T, Araujo A, Kupelian V, O Donnell A, McKinlay J, Travison T et al (2007) The relative contributions of aging, health, and lifestyle factors to serum testosterone decline in men. J Clin Endocrinol Metab 92(2):549–555
- 33. Derby C, Zilber S, Brambilla D, Morales K, McKinlay J, Derby C et al (2006) Body mass index, waist circumference and waist to hip ratio and change in sex steroid hormones: the Massachusetts Male Aging Study. Clin Endocrinol 65(1):125–131
- 34. Feldman H, Longcope C, Derby C, Johannes C, Araujo A, Coviello A et al (2002) Age trends in the level of serum testosterone and other hormones in middle-aged men: longitudinal results from the Massachusetts Male Aging Study. J Clin Endocrinol Metab 87(2):589–598
- Wu FC, Tajar A, Beynon JM, Pye SR, Silman AJ, Finn JD et al (2010) Identification of lateonset hypogonadism in middle-aged and elderly men. New Eng J Med 363(2):123–315
- Shores MM, Matsumoto AM, Sloan KL, Kivlahan DR (2006) Low serum testosterone and mortality in male veterans. Arch Intern Med 166(15):1660–1665
- O'Donnell AB, Araujo AB, McKinlay JB, Kupelian V, Travison TG (2007) The relative contributions of aging, health, and lifestyle factors to serum testosterone decline in men. J Clin Endocrinol Metab 92(2):549–555
- Matsumoto AM (2002) Andropause: clinical implications of the decline in serum testosterone levels with aging in men. J Gerontol A Biol Sci Med Sci 57(2):M76–M99
- Forsberg CW, Smith NL, Shores MM, Matsumoto AM, Anawalt BD (2012) Testosterone treatment and mortality in men with low testosterone levels. J Clin Endocrinol Metab 97(6):2050–2058
- 40. Pastuszak AW, Kohn TP, Estis J, Lipshultz LI (2017) Low plasma testosterone is associated with elevated cardiovascular disease biomarkers. J Sex Med 14(9):1095–1103

- Maki PM, Mordecai KL, Rubin LH, Sundermann E, Savarese A, Eatough E et al (2015) Menstrual cycle effects on cortisol responsivity and emotional retrieval following a psychosocial stressor. Horm Behav 74:201–208
- Ramsey JM, Cooper JD, Penninx BW, Bahn S (2016) Variation in serum biomarkers with sex and female hormonal status: implications for clinical tests. Sci Rep 6:26947. https://doi. org/10.1038/srep26947
- 43. Yiallouris A, Tsioutis C, Agapidaki E, Zafeiri M, Agouridis AP, Ntourakis D et al (2019) Adrenal aging and its implications on stress responsiveness in humans. Front Endocrinol (Lausanne) 10:54. https://doi.org/10.3389/fendo.2019.00054
- 44. Powrie YSL, Smith C (2018) Central intracrine DHEA synthesis in ageing-related neuroinflammation and neurodegeneration: therapeutic potential? J Neuroinflammation 15(1):289. https://doi.org/10.1186/s12974-018-1324-0
- 45. Kroboth PD, Salek FS, Pittenger AL, Fabian TJ, Frye RF (1999) DHEA and DHEA-S: a review. J Clin Pharmacol 39(4):327–348
- 46. Miller KK, Al-Rayyan N, Ivanova MM, Mattingly KA, Ripp SL, Klinge CM et al (2013) DHEA metabolites activate estrogen receptors alpha and beta. Steroids 78(1):15–25
- 47. Webb SJ, Geoghegan TE, Prough RA, Michael Miller KK (2006) The biological actions of dehydroepiandrosterone involves multiple receptors. Drug Metab Rev 38(1–2):89–116
- Ferrari E, Cravello L, Muzzoni B, Casarotti D, Paltro M, Solerte SB et al (2001) Age-related changes of the hypothalamic-pituitary-adrenal axis: pathophysiological correlates. Eur J Endocrinol 144(4):319–329
- 49. van den Beld AW, Kaufman JM, Zillikens MC, Lamberts SWJ, Egan JM, van der Lely AJ (2018) The physiology of endocrine systems with ageing. Lancet Diabetes Endocrinol 6(8):647–658
- Voznesensky M, Walsh S, Dauser D, Brindisi J, Kenny AM (2009) The association between dehydroepiandosterone and frailty in older men and women. Age Ageing 38(4):401–406
- Rendina DN, Ryff CD, Coe CL (2017) Precipitous dehydroepiandrosterone declines reflect decreased physical vitality and function. J Gerontol A Biol Sci Med Sci 72(6):747–753
- Morrison MF, Katz IR, Parmelee P, Boyce AA, TenHave T (1998) Dehydroepiandrosterone sulfate (DHEA-S) and psychiatric and laboratory measures of frailty in a residential care population. Am J Geriatr Psychiatry 6(4):277–284
- 53. Jiménez MC, Sun Q, Schürks M, Chiuve S, Hu FB, Manson JE et al (2013) Low dehydroepiandrosterone sulfate is associated with increased risk of ischemic stroke among women. Stroke 44(7):1784–1789
- 54. Vieira-Marques C, Arbo BD, Cozer AG, Hoefel AL, Cecconello AL, Zanini P et al (2017) Sex-specific effects of dehydroepiandrosterone (DHEA) on glucose metabolism in the CNS. J Steroid Biochem Mol Biol 171:1–10
- Prall SP, Muehlenbein MP (2018) DHEA modulates immune function: a review of evidence. Vitam Horm 108:125–144
- Kamin HS, Kertes DA (2017) Cortisol and DHEA in development and psychopathology. Horm Behav 89:69–85
- Labrie F (2010) DHEA, important source of sex steroids in men and even more in women. Prog Brain Res 182:97–148
- 58. Samaras N, Samaras D, Frangos E, Forster A, Philippe J (2013) A review of age-related dehydroepiandrosterone decline and its association with well-known geriatric syndromes: is treatment beneficial? Rejuvenation Res 16(4):285–294
- Sulcová J, Hill M, Hampl R, Stárka L (1997) Age and sex related differences in serum levels of unconjugated dehydroepiandrosterone and its sulphate in normal subjects. J Endocrinol 154(1):57–62
- Orentreich N, Brind JL, Rizer RL, Vogelman JH (1984) Age changes and sex differences in serum Dehydroepiandrosterone sulfate concentrations throughout adulthood. J Clin Endocrinol Metab 59(3):551–555

- 61. Quinn TA, Robinson SR, Walker D (2018) Dehydroepiandrosterone (DHEA) and DHEA sulfate: roles in brain function and disease. In: Drevensek G (ed) Sex hormones in neurode-generative processes and diseases. IntechOpen. ISBN-10: 1789230144
- 62. Ohlsson C, Vandenput L, Tivesten A (2015) DHEA and mortality: what is the nature of the association? J Steroid Biochem Mol Biol 145:248–253
- Wudy SA, Schuler G, Sanchez-Guijo A, Hartmann MF (2018) The art of measuring steroids: principles and practice of current hormonal steroid analysis. J Steroid Biochem Mol Biol 179:88–103
- 64. Stanczyk FZ, Cho MM, Endres DB, Morrison JL, Patel S, Paulson RJ (2003) Limitations of direct estradiol and testosterone immunoassay kits. Steroids 68(14):1173–1178
- 65. El-Farhan N, Rees DA, Evans C (2017) Measuring cortisol in serum, urine and saliva are our assays good enough? Ann Clin Biochem 54(3):308–322
- 66. Galon J, Franchimont D, Hiroi N, Frey G, Boettner A, Ehrhart-Bornstein M et al (2002) Gene profiling reveals unknown enhancing and suppressive actions of glucocorticoids on immune cells. FASEB J 16(1):61–71
- Oakley RH, Cidlowski JA (2013) The biology of the glucocorticoid receptor: new signaling mechanisms in health and disease. J Allergy Clin Immunol 132(5):1033–1044
- Sapolsky RM, Romero LM, Munck AU (2000) How do glucocorticoids influence stress responses? Integrating permissive, suppressive, stimulatory, and preparative actions. Endocr Rev 21(1):55–89
- Finsterwald C, Alberini CM (2014) Stress and glucocorticoid receptor-dependent mechanisms in long-term memory: from adaptive responses to psychopathologies. Neurobiol Learn Mem 112:17–29
- Herman JP, McKlveen JM, Ghosal S, Kopp B, Wulsin A, Makinson R et al (2016) Regulation of the hypothalamic-pituitary-adrenocortical stress response. Compr Physiol 6(2):603–621
- den Boon FS, Sarabdjitsingh RA (2017) Circadian and ultradian patterns of HPA-axis activity in rodents: significance for brain functionality. Best Pract Res Clin Endocrinol Metab 31(5):445–457
- McEwen BS, De Kloet ER, Rostene W (1986) Adrenal steroid receptors and actions in the nervous system. Physiol Rev 66(4):1121–1188
- Lupien SJ, Maheu F, Tu M, Fiocco A, Schramek TE (2007) The effects of stress and stress hormones on human cognition: implications for the field of brain and cognition. Brain Cogn 65(3):209–237
- Avital A, Segal M, Richter-Levin G (2006) Contrasting roles of corticosteroid receptors in hippocampal plasticity. J Neurosci 26(36):9130–9134
- Vyas S, Rodrigues AJ, Silva JM, Tronche F, Almeida OFX, Sousa N et al (2016) Chronic stress and glucocorticoids: from neuronal plasticity to neurodegeneration. Neural Plast 2016:6391686. https://doi.org/10.1155/2016/6391686
- Herman JP (2013) Neural control of chronic stress adaptation. Front Behav Neurosci 7:61. https://doi.org/10.3389/fnbeh.2013.00061
- 77. Ancelin M-L, Scali J, Norton J, Ritchie K, Dupuy A-M, Chaudieu I et al (2017) Heterogeneity in HPA axis dysregulation and serotonergic vulnerability to depression. Psychoneuroendocrinology 77:90–94
- Gupta D, Morley JE (2014) Hypothalamic-pituitary-adrenal (HPA) axis and aging. Compr Physiol 4(4):1495–1510
- 79. Wang Q, Van Heerikhuize J, Aronica E, Kawata M, Seress L, Joels M et al (2013) Glucocorticoid receptor protein expression in human hippocampus; stability with age. Neurobiol Aging 34(6):1662–1673
- Noordam R, Jansen SWM, Akintola AA, Oei NYL, Maier AB, Pijl H et al (2012) Familial longevity is marked by lower diurnal salivary cortisol levels: the Leiden Longevity Study. PLoS One 7(2):e31166. https://doi.org/10.1371/journal.pone.0031166

- Weitzman ED, Fukushima D, Nogeire C, Roffwarg H, Gallagher TF, Hellman L (1971) Twenty-four hour pattern of the episodic secretion of cortisol in normal subjects. J Clin Endocrinol Metab 33(1):14–22
- Lupien S, Lecours AR, Lussier I, Schwartz G, Nair NP, Meaney MJ (1994) Basal cortisol levels and cognitive deficits in human aging. J Neurosci 14(5 Pt 1):2893–2903
- Deuschle M, Gotthardt U, Schweiger U, Weber B, Korner A, Schmider J et al (1997) With aging in humans the activity of the hypothalamus-pituitary-adrenal system increases and its diurnal amplitude flattens. Life Sci 61(22):2239–2246
- 84. Almeida DM, Piazza JR, Stawski RS (2009) Interindividual differences and intraindividual variability in the cortisol awakening response: an examination of age and gender. Psychol Aging 24(4):819–827
- 85. Fardella CE, Campino C, Carvajal CA, Baudrand R, Kalergis AM, Martinez-Aguayo A et al (2013) Age-related changes in 11β-hydroxysteroid dehydrogenase type 2 activity in normotensive subjects. Am J Hypertens 26(4):481–487
- Chapman K, Holmes M, Seckl J (2013) 11β-hydroxysteroid dehydrogenases: intracellular gate-keepers of tissue glucocorticoid action. Physiol Rev 93(3):1139–3206
- 87. MacLullich AM, Ferguson KJ, Reid LM, Deary IJ, Starr JM, Wardlaw JM et al (2012) 11beta-hydroxysteroid dehydrogenase type 1, brain atrophy and cognitive decline. Neurobiol Aging 33(1):207.e1-8
- Ennis GE, An Y, Resnick SM, Ferrucci L, O'Brien RJ, Moffat SD (2017) Long-term cortisol measures predict Alzheimer disease risk. Neurology 88(4):371–378
- Zhao H, Xu H, Xu X, Young D (2007) Predatory stress induces hippocampal cell death by apoptosis in rats. Neurosci Lett 421(2):115–120
- Donovan MH, Yazdani U, Norris RD, Games D, German DC, Eisch AJ (2006) Decreased adult hippocampal neurogenesis in the PDAPP mouse model of Alzheimer's disease. J Comp Neurol 495(1):70–83
- Rincón-Cortés M, Herman JP, Lupien S, Maguire J, Shansky RM (2019) Stress: influence of sex, reproductive status and gender. Neurobiol Stress 10:100155. https://doi.org/10.1016/j. ynstr.2019.100155
- Kitay JI (1961) Enhancement of steroidogenesis by rat adrenal slices in vitro with estradiol-17-beta. Nature 192:358–359
- Oyola MG, Handa RJ (2017) Hypothalamic-pituitary-adrenal and hypothalamic-pituitarygonadal axes: sex differences in regulation of stress responsivity. Stress 20(5):476–494
- 94. Toufexis D, Rivarola MA, Lara H, Viau V (2014) Stress and the reproductive axis. J Neuroendocrinol 26(9):573–586
- 95. Handa RJ, McGivern RF (2009) Stress response: sex differences. In: Squire LR (ed) Encyclopedia of neuroscience. Academic, Oxford, pp 511–517. ISBN: 9780080446172
- Roelfsema F, Aoun P, Veldhuis JD (2016) Pulsatile cortisol feedback on ACTH secretion is mediated by the glucocorticoid receptor and modulated by gender. J Clin Endocrinol Metab 101(11):4094–4102
- Viau V (2002) Functional cross-talk between the hypothalamic-pituitary-gonadal and -adrenal axes. J Neuroendocrinol 14(6):506–513
- Kirschbaum C, Kudielka BM, Gaab J, Schommer NC, Hellhammer DH (1999) Impact of gender, menstrual cycle phase, and oral contraceptives on the activity of the hypothalamuspituitary-adrenal axis. Psychosom Med 61(2):154–162
- 99. Simunkova K, Stárka L, Hill M, Kríz L, Hampl R, Vondra K (2008) Comparison of total and salivary cortisol in a low-dose ACTH (Synacthen) test: influence of three-month oral contraceptives administration to healthy women. Physiol Res 57(Suppl 1):S193–S199
- 100. Roche DJO, King AC, Cohoon AJ, Lovallo WR (2013) Hormonal contraceptive use diminishes salivary cortisol response to psychosocial stress and naltrexone in healthy women. Pharmacol Biochem Behav 109:84–90
- 101. Gifford RM, Reynolds RM (2017) Sex differences in early-life programming of the hypothalamic-pituitary-adrenal axis in humans. Early Hum Dev 114:7–10

- 102. Levine A, Zagoory-Sharon O, Feldman R, Lewis JG, Weller A (2007) Measuring cortisol in human psychobiological studies. Physiol Behav 90(1):43–53
- 103. Staufenbiel S, Penninx BW, Spijker A, Elzinga BM, van Rossum E (2012) Hair cortisol, stress exposure, and mental health in humans: a systematic review. Psychoneuroendocrinology 38(8):1220–1235
- Novak MA, Meyer JS (2012) Minireview: hair cortisol: a novel biomarker of hypothalamicpituitary-adrenocortical activity. Endocrinology 153(9):4120–4127
- 105. Noppe G, de Rijke YB, Dorst K, van den Akker ELT, van Rossum EFC (2015) LC-MS/ MS-based method for long-term steroid profiling in human scalp hair. Clin Endocrinol 83(2):162–166
- 106. Abell JG, Stalder T, Ferrie JE, Shipley MJ, Kirschbaum C, Kivimäki M et al (2016) Assessing cortisol from hair samples in a large observational cohort: the Whitehall II study. Psychoneuroendocrinology 73:148–156
- 107. Steptoe A, Serwinski B (2016) Cortisol awakening response. In: Fink G (ed) Stress: concepts, cognition, emotion, and behavior, 1st edn. Academic, pp 277–283. ISBN-10: 0128009519
- 108. Hellhammer J, Fries E, Schweisthal OW, Schlotz W, Stone A, Hagemann D (2007) Several daily measurements are necessary to reliably assess the cortisol rise after awakening: stateand trait components. Psychoneuroendocrinology 32(1):80–86
- 109. Fujii T, Hori H, Ota M, Hattori K, Teraishi T, Sasayama D et al (2014) Effect of the common functional FKBP5 variant (rs1360780) on the hypothalamic-pituitary-adrenal axis and peripheral blood gene expression. Psychoneuroendocrinology 42:89–97
- 110. Vastbinder M, Kuindersma M, Mulder AH, Schuijt MP, Mudde AH (2016) The influence of oral contraceptives on overnight 1 mg dexamethasone suppression test. Neth J Med 74(4):158–161
- 111. Weyand CM, Goronzy JJ (2016) Aging of the immune system. Mechanisms and therapeutic targets. Ann Am Thorac Soc 13(Suppl 5):S422–S428
- 112. Eming SA, Wynn T, Martin P (2017) Inflammation and metabolism in tissue repair and regeneration. Science 356(6342):1026–1030
- 113. Bonilla FA, Oettgen HC (2010) Adaptive immunity. J Allergy Clin Immunol 125(2 Suppl 2):S33–S40
- 114. Kulkarni OP, Lichtnekert J, Anders HJ, Mulay SR (2016) The immune system in tissue environments regaining homeostasis after injury: is "inflammation" always inflammation? Mediat Inflamm 2016:2856213. https://doi.org/10.1155/2016/2856213
- 115. Blach-Olszewska Z, Leszek J (2007) Mechanisms of over-activated innate immune system regulation in autoimmune and neurodegenerative disorders. Neuropsychiatr Dis Treat 3(3):365–372
- 116. Fulop T, Larbi A, Dupuis G, Le Page A, Frost EH, Cohen AA et al (2018) Immunosenescence and inflamm-aging as two sides of the same coin: friends or foes? Front Immunol 8:1960. https://doi.org/10.3389/fimmu.2017.01960
- 117. Ciabattini A, Nardini C, Santoro F, Garagnani P, Franceschi C, Medaglini D (2018) Vaccination in the elderly: the challenge of immune changes with aging. Semin Immunol 40:83–94
- Kondilis-Mangum HD, Wade PA (2013) Epigenetics and the adaptive immune response. Mol Asp Med 34(4):813–825
- 119. Weinberger B, Grubeck-Loebenstein B (2012) Vaccines for the elderly. Clin Microbiol Infect 18(Suppl 5):100–108
- 120. Franceschi C, Bonafe M, Valensin S, Olivieri F, De Luca M, Ottaviani E et al (2000) Inflammaging. An evolutionary perspective on immunosenescence. Ann N Y Acad Sci 908:244–254
- 121. Shaw AC, Goldstein DR, Montgomery RR (2013) Age-dependent dysregulation of innate immunity. Nat Rev Immunol 13(12):875–887
- 122. Rea IM, Gibson DS, McGilligan V, McNerlan SE, Alexander HD, Ross OA (2018) Age and age-related diseases: role of inflammation triggers and cytokines. Front Immunol 9:586. https://doi.org/10.3389/fimmu.2018.00586

- 123. Jiang J, Wen W, Sachdev PS (2016) Macrophage inhibitory cytokine-1/growth differentiation factor 15 as a marker of cognitive aging and dementia. Curr Opin Psychiatry 29(2):181–186
- 124. Sullivan DH, Roberson PK, Johnson LE, Mendiratta P, Bopp MM, Bishara O (2007) Association between inflammation-associated cytokines, serum albumins, and mortality in the elderly. J Am Med Dir Assoc 8(7):458–463
- 125. Leliefeld PHC, Koenderman L, Pillay J (2015) How neutrophils shape adaptive immune responses. Front Immunol 6:471. https://doi.org/10.3389/fimmu.2015.00471
- Cronkite DA, Strutt TM (2016) The regulation of inflammation by innate and adaptive lymphocytes. J Immunol Res 2018:1467538. https://doi.org/10.1155/2018/1467538
- 127. Minciullo PL, Catalano A, Mandraffino G, Casciaro M, Crucitti A, Maltese G et al (2016) Inflammaging and anti-inflammaging: the role of cytokines in extreme longevity. Arch Immunol Ther Exp (Warsz) 64(2):111–126
- 128. Klein SL, Flanagan K (2016) Sex differences in immune responses. Nat Rev Immunol 16(10):626–638
- 129. Gubbels Bupp MR (2015) Sex, the aging immune system, and chronic disease. Cell Immunol 294(2):102–110
- Edwards M, Dai R, Ahmed SA (2018) Our environment shapes us: the importance of environment and sex differences in regulation of autoantibody production. Front Immunol 9:478. https://doi.org/10.3389/fimmu.2018.00478
- 131. Jorgensen TN (2015) Sex disparities in the immune response. Cell Immunol 294(2):61-62
- Kovats S (2015) Estrogen receptors regulate innate immune cells and signaling pathways. Cell Immunol 294(2):63–69
- Gubbels Bupp MR, Jorgensen TN (2018) Androgen-induced immunosuppression. Front Immunol 9:794. https://doi.org/10.3389/fimmu.2018.00794
- 134. Flanagan KL, Fink AL, Plebanski M, Klein SL (2017) Sex and gender differences in the outcomes of vaccination over the life course. Annu Rev Cell Dev Biol 33:577–599
- 135. Furman D, Hejblum BP, Simon N, Jojic V, Dekker CL, Thiebaut R et al (2014) Systems analysis of sex differences reveals an immunosuppressive role for testosterone in the response to influenza vaccination. Proc Natl Acad Sci U S A 111(2):869–874
- Fink AL, Klein SL (2015) Sex and gender impact immune responses to vaccines among the elderly. Physiology (Bethesda) 30(6):408–416
- 137. Hirokawa K, Utsuyama M, Hayashi Y, Kitagawa M, Makinodan T, Fulop T (2013) Slower immune system aging in women versus men in the Japanese population. Immun Ageing 10(1):19. https://doi.org/10.1186/1742-4933-10-19

# Chapter 5 The Therapeutic Potential of Ketogenic Diet Throughout Life: Focus on Metabolic, Neurodevelopmental and Neurodegenerative Disorders



## Ann-Katrin Kraeuter, Paul C. Guest, and Zoltan Sarnyai

## 1 Introduction

Throughout the ages many great minds have prescribed fasting to improve physical and mental health. Nutritional interventions have been used to treat diseases throughout the ages. The Greek philosopher Hippocrates used food as medicine such as lentils for the treatment of ulcers and haemorrhoids. Galen, a Roman philosopher, believed that food could be the cure or the cause of diseases, stating that the excessive consumption of spicy foods (garlic leeks, onions) may cause a fever.

Everyone has a physician inside him or her; we just have to help it in its work. The natural healing force within each one of us is the greatest force in getting well. Our food should be our medicine. Our medicine should be our food. But to eat when you are sick is to feed your sickness.

Hippocrates

- Instead of using medicine, rather, fast a day. Plutarch
- I fast for greater physical and mental efficiency Plato

P. C. Guest

Laboratory of Neuroproteomics, Department of Biochemistry and Tissue Biology, Institute of Biology, University of Campinas (UNICAMP), Campinas, Brazil

© Springer Nature Switzerland AG 2019

A.-K. Kraeuter · Z. Sarnyai (🖂)

Laboratory of Psychiatric Neuroscience, Australian Institute of Tropical Health and Medicine, James Cook University, Townsville, QLD, Australia

Discipline of Biomedicine, College of Public Health, Medicine and Veterinary Sciences, James Cook University, Townsville, QLD, Australia e-mail: zoltan.sarnyai@jcu.edu.au

P. C. Guest (ed.), *Reviews on Biomarker Studies in Aging and Anti-Aging Research*, Advances in Experimental Medicine and Biology 1178, https://doi.org/10.1007/978-3-030-25650-0\_5

*Fasting is the greatest remedy, the physician within.* Phillippus Paracelsus

The best of all medicines is resting and fasting. To lengthen thy life, lessen thy meals. Benjamin Franklin

Although some of these ideas might be debatable with more recent research, these individuals hypothesised that food might be used as a therapeutic intervention. Fasting is one such nutritional treatment. The energy depots of mammals which enable fasting periods are the liver and adipose tissue [1]. Short periods of fasting or food restriction have been demonstrated to be beneficial in diabetes, cardiovascular disease, Alzheimer's disease, Parkinson's disease, Huntington's disease, stroke, cancer and obesity [1]. Periods of fasting result in an elevation of ketones [1]. Over a century ago, a study investigated the effects of starvation as a potential treatment for epilepsy [2]. The increase in ketone bodies, such as acetone and  $\beta$ -hydroxybutyrate, during starvation periods was first described by Woodyatt in 1921 [3]. Also in 1921, Wilder described that the benefits of fasting could be achieved in a state of ketosis [4].

The ketogenic diet is a form of fasting through altering energy metabolism and producing ketones [5]. The ketogenic diet is a high-fat, moderate protein and low carbohydrate diet, which uses fat as an alternative fuel to glucose. The diet is traditionally composed of three to four grams of long chain-saturated triglycerides for every gram of carbohydrate and proteins [6]. This modern form of fasting reduces circulating insulin and insulin signalling, and results in a metabolic shift to fatty acid utilization [7], which leads to increased production of ketone bodies as an alternative fuel source to glucose [6, 8]. During the ketogenic diet, blood levels of free fatty acids rise [6]. Free fatty acids are transported into the mitochondrial matrix of the liver with the help of carnitine. Acetyl-coenzyme A (acetyl-CoA) is produced through *B*-oxidation of the free fatty acids [6]. When glycolysis is predominant, acetyl-CoA is used to fuel the Krebs cycle. However, while in glucose-deprived states, such as ketosis, oxaloacetate is used for gluconeogenesis to maintain serum glucose levels. The decreased availability of oxaloacetate leads to a decreased amount of acetyl-CoA feeding into the Krebs cycle [8]. Consequently, excess acetyl-CoA will form the ketone acetoacetate, which spontaneously degrades into acetone or enzymatically metabolised into ß-hydroxybutyrate. Beta-hydroxybutyrate is formed in a reverse reaction with the help of the nicotinamide adenine dinucleotide hydride (NADH)-dependent mitochondrial enzyme ß-hydroxybutyrate dehydrogenase [8]. These ketone bodies then enter into the bloodstream.

Ketone bodies are not able to cross the blood-brain barrier without the aid of specific transporters. Beta-hydroxybutyrate and acetoacetate can be transported into the brain via the monocarboxylic acid transporter [9]. Interestingly, hippocampal mRNA levels of the monocarboxylic acid transporter are increased during the keto-genic diet [10]. Within the brain, beta-hydroxybutyrate is enzymatically converted by beta-hydroxybutyrate dehydrogenase into acetoacetate. Acetoacetate receives a coenzyme A from succinyl-CoA from the Krebs cycle to form acetoacetate-CoA.



Fig. 5.1 Number of articles published on the ketogenic diet over 1923–2018 (Scopus)

Acetoacetate-CoA will spontaneously degrade to acetyl-CoA, which is able to fuel the Krebs cycle without using glucose or its metabolite pyruvate [8] and excess acetone is rapidly excreted through the breath and urine.

Within the scientific community, the ketogenic diet has received increasing interest in recent years (Fig. 5.1). It has been used in the management of treatmentrefractory childhood epilepsy since the 1920s [6]. The potential of this diet as a nutritional therapeutic has been trialled in several disease areas, such as brain cancer [11], brain trauma [12], migraines [13, 14], type II diabetes, as well as neurodegenerative, neurodevelopmental and psychiatric disorders [15, 16]. The neurodegenerative disorders which have been tested in clinical trials using the ketogenic diet include Alzheimer's disease [16], Parkinson's disease [17, 18] and multiple sclerosis [19], and the neurodevelopmental and psychiatric diseases which have been trialled include autism spectrum disorder, anxiety and depression, [20–23], attention deficit hyperactivity disorder [23, 24], schizophrenia [25–28] and bipolar disorder [29, 30].

The pathophysiology of psychiatric conditions is complex and poorly understood [31]. Neurological disorders and substance abuse disorders have increased over recent years and have now closed the gap in terms of prevalence to cardiovascular diseases, which have a well-understood pathophysiology [31]. Current pharmacological treatments for most psychiatric conditions have limited efficacy and can results in considerable side effects. Although these treatments slow down disease progression they do not stop or cure the condition [32]. Therefore, alternative efficacious treatments with more tolerable side effects are needed. In this light, dietary interventions may have fewer side effects and result in better quality of life.

In this chapter, we have reviewed the efficacy of the ketogenic diet in a variety of neurodegenerative, neurodevelopmental and metabolic conditions throughout life. We will start with conditions affecting children, then we will consider metabolic disorders in adults and lastly we will cover conditions affecting the elderly. We have focused on application of the ketogenic diet in clinical studies as well as in preclinical models and we discuss the known benefits and drawbacks. We also highlight the need for further research in this area aimed at discovery of novel mechanistic targets of the ketogenic diet, which might result in novel treatment options and development of new drugs which maximise the potential benefits/risks ratio.

#### 2 Ketogenic Diet in Childhood

#### 2.1 Epilepsy

Over a century ago, one study investigated the effects of starvation as a potential treatment for epilepsy [2]. In addition, Wilder described for the first time that the benefits of fasting could be achieved in a state of ketosis [4]. It was proposed that a state of ketosis without actual starvation is more sustainable for the treatment of epilepsy in clinical trials [4]. In 1925, Peterman reported the composition of the ketogenic diet as similar to that which is used currently [33]. Acetone and acetoacetate have anticonvulsant effects, which can therefore prevent acutely provoked seizure [6]. This dietary intervention lost its popularity in the 1930s, after the introduction of the first effective anticonvulsant drug, diphenylhydantoin [34]. In 1972 at Johns Hopkins Hospital, Livingston reported the use of the ketogenic diet in 1000 children with epilepsy, resulting in 52% of the patients having complete seizure symptom control and 27% with improved symptoms [35]. In the 1990s, increased drug-resistance to commonly used anticonvulsant drugs resulted in a resurgence of the ketogenic diet as a treatment for epilepsy [36] (Fig. 5.2). Excessive neuronal excitability causes seizures, which may be ameliorated with the ketogenic diet through multiple pathways [37]. A number of recent meta-analyses have



Fig. 5.2 Number of articles published on the ketogenic diet in epilepsy from 1925–2018 (Scopus)

demonstrated the efficacy of the diet in epilepsy [38–40]. In addition, a recent study investigated the efficacy, tolerability and complications of an olive oil-based ketogenic diet treatment in children with epilepsy and found that previous adrenocorticotropic hormone use and the occurrence of constipation at baseline or during treatment were predictors of treatment efficacy [41]. Another study found that the ketogenic diet influenced both the taxonomic and functional composition of gut microbiota in children with severe epilepsy but could not determine whether these changes are involved in efficacy or the side effects of the diet [42]. Finally, a scoping review is underway to determine which outcomes should be measured and reported in studies of childhood epilepsy treated with the ketogenic diet therapy as a means of improving outcomes or helping to monitor side effects [43].

#### 2.2 Attention Deficit Hyperactivity Disorder (ADHD)

ADHD is the most common childhood neurodevelopmental disorder [44–46]. A meta-regression analysis estimated that around 5.29% of children and adolescents might have ADHD [47]. ADHD is characterised by impulsivity, motor hyperactivity, inappropriate behaviour and inattention [44–46]. Diseases associated with ADHD are autism spectrum disorder, communication, specific learning or motor disorders, intellectual disability and tic disorders [45]. In addition, ADHD is accompanied by psychiatric co-morbidities, such as conduct disorder, depression, bipolar disorder and substance abuse disorders [48].

The pathophysiology of ADHD is poorly understood but it has been suggested that abnormal dopaminergic, noradrenergic and serotonergic neurotransmission plays an important role [45]. Current treatment of ADHD includes the use of stimulants of neurotransmitters, dopamine and norepinephrine [45]. Such stimulants increase catecholamine availability and have been shown to be effective in children and adults [45, 49].

Preclinical research investigating the effects of the ketogenic diet in ADHD is limited due to the low availability of appropriate animal models of ADHD. Nevertheless, studies in rodents have found that the diet reduces activity levels compared to control animals [50, 51]. A study in dogs that showed certain behavioural features of ADHD found a decrease in chasing behaviour, but no change in excitability and trainability, after the dogs were fed a diet supplemented with medium chain triglycerides [52].

Thus far, one case report demonstrated that the ketogenic diet improved behaviour, hyperactivity and aggression in a child with ADHD and autism spectrum disorder [23]. Another study which investigated the effects of a ketogenic diet on seizure frequency found improved motor functioning, self-help skills, attention and social skills in children, who had abnormal emotional and behavioural responses at the study onset.

These limited results from human studies suggest that the ketogenic diet may have a potential to attenuate some of the features of ADHD in children.

#### 2.3 Autism Spectrum Disorders

Autism spectrum disorder is a life-long neurodevelopmental condition defined by behavioural abnormalities, such asocial behaviour, communication and repetitive behavioural patterns [53–55]. It is estimated that 1 in 88 children have autism spectrum disorder and is four time more common in males than females [53]. Furthermore, most patients with autism spectrum disorder had intellectual disability (55%) and below average intelligence (17%) in a conventional IQ test [53]. Comorbidities include seizures, gastrointestinal disturbances, motor impairments, auditory disorder, altered sleep, increased anxiety and heightened or reduced sensitivity to stimuli, such as sound or temperature [53, 56].

The aetiology of autism spectrum disorder is unknown but genetic and environmental factors leading to altered neurodevelopment has been suggested [53, 56]. Due to difficulties in identifying individual causative genes, research has focused on environmental factors, such as maternal immune activation, in utero exposure to pesticides, oxidative stress and environmental contaminants [53, 55, 56]. The medical management of autism spectrum disorder includes educational interventions and/or the use of antipsychotic medications [57]. These treatment options are not specific for autism spectrum disorder or its core symptoms and show poor symptom control [58].

Multiple preclinical studies have investigated the effects of the ketogenic diet in animal models of autism spectrum disorder. Improvements have been reported in the core symptoms of autism spectrum disorder with the diet including sociability [59–64], communication [59] and repetitive behaviours [59, 61–64]. Preclinical studies demonstrated that the ketogenic diet alters mitochondrial bioenergetics [65], neurotransmitter signalling [65], hormonal metabolism [65] and the gut microbiome [66], which may contribute to behavioural improvements.

The effects of the ketogenic diet on autism spectrum disorder behaviours have been study extensively in children. Studies have demonstrated that the diet was highly effective in improving mild to moderate symptoms of autism spectrum disorder [20, 22], including sociability, calmness, cognition, language function, stereotypical behaviour, IQ [21], while decreasing hyperactivity and aggressive behaviour [23]. A modified Atkins diet (60% fat, 30% protein, 10% carbohydrates) improved Childhood Autism Rating Scale and the Autism Treatment Evaluation Checklist scores in the areas of speech, sociability and cognition [67]. These promising early results support further research into the possible clinical application of the ketogenic diet in this neurodevelopmental condition.

#### **3** Ketogenic Diet in Adolescence/Early Adulthood

#### 3.1 Schizophrenia

Schizophrenia is a debilitating and life-long psychiatric disorder with an onset typically in early adolescence and is among the leading causes of disability worldwide [68–71]. Schizophrenia affects more than 21 million people (1%) worldwide [72].

Schizophrenia is characterized by three major symptom clusters classified as positive (psychotic), negative and cognitive symptoms [71]. Positive symptoms include visual and auditory hallucinations, disorganized thoughts and language, abnormal and disorganized behaviour and catatonia [73, 74]. Furthermore, individuals with schizophrenia have difficulties with normal emotional responses resulting in social withdrawal, anhedonia and a lack of motivation. These latter characteristics are termed the negative symptoms [71]. Cognitive symptoms are characterized by deficits in working memory [75]. The disease progresses in severity, usually beginning in adolescence with positive symptoms, which are followed later in life by appearance of negative and cognitive symptoms [73]. Schizophrenia has a high mortality rate because of the development of secondary diseases such as cardiovascular disorders and due to an increased proportion of suicides compared to the general population [76–78].

The pathophysiology of schizophrenia is complex. The symptoms are possibly underlined by hyperactivity of the mesolimbic dopamine pathway [79–84], disturbance in glutamatergic functioning [71], decreased activity in mesocortical dopamine projections to the prefrontal cortex [83], genetic factors [75, 85, 86], impairment in cerebral glucose metabolism [87–92] and environmental factors [75]. Current pharmaceutical treatments for schizophrenia include typical and atypical antipsychotic drugs, which act mostly on the D<sub>2</sub> dopamine receptors as antagonists. These are partially effective for positive symptoms but they can have significant side effects.

Several studies have demonstrated through in vivo imaging, post mortem brain analysis and animal studies that schizophrenia might be linked to abnormal cerebral glucose metabolism [91, 93-98]. On the basis of such findings it can be hypothesised that the ketogenic diet might be able to bypass glucose metabolism and use ketone bodies as an alternative fuel source. We demonstrated that the ketogenic diet effectively restored abnormal schizophrenia-like behaviours across the whole range of symptoms in an acute animal model of schizophrenia [99, 100]. Application of the ketogenic diet in an acute animal model of schizophrenia attenuated hyperactivity, stereotypical behaviour and increased sociability, working memory and prepulse inhibition of startle. Another group investigated a genetic mouse model of schizophrenia using the ketogenic diet [101]. The investigators used hippocampal auditory P20/P40 gating to assess the hippocampal inhibitory circuit by implanting depth electrodes into the hippocampus of DBA/2 mice, a translational endophenotype that mimics the inhibitory P50 sensory gating deficits in schizophrenia patients. This enabled determination of the ability to filter early neuronal responses to the second of a pair of identical and repeated auditory stimuli and the results showed that the ketogenic diet significantly improved auditory gating in the mice and reduced the second response.

According to Dohan in 1966 [102], schizophrenia-related hospitalizations decreased, because of limited bread consumption during World War II. Further studies demonstrate that a gluten free diet might be beneficial for patients with schizophrenia [103]. Two early case studies suggested that the ketogenic diet might have clinical efficacy in the management of schizophrenia [25, 26]. More importantly,

two recent case reports showed that the ketogenic diet improved auditory hallucinations, mood, energy, the ability to concentrate and sociability in patients with schizoaffective disorder [27]. This and another recent study showed improvement in the Positive and Negative Syndrome Scales symptom scores following the ketogenic diet [27, 28].

These studies demonstrate the potential positive effects of the ketogenic diet. However, additional larger scale studies are needed to fully evaluate the efficacy in a larger cohort of schizophrenia patients.

## 4 Ketogenic Diet in Adults

## 4.1 Obesity and Type II Diabetes

The global incidence of diabetes has increased due to an aging population, urbanization and unhealthy lifestyle choices [104]. It is predicted that by 2030, 439 million people globally will have diabetes, with 90% having type II diabetes. The main modifiable risk factors for type II diabetes are obesity and physical inactivity. Worryingly, type II diabetes has become increasingly prevalent in adolescence and even in children. It is predicted that obesity and overweight prevalence in adults will increase from 33% in 2005 to 57.8% in 2030 [104]. Obesity is the single most important predictor for type II diabetes [104] and a large proportion of patients with type II diabetes are overweight or obese [105].

Obesity is a major health risk associated with cardiovascular complications, such as high blood pressure, dyslipidaemia, abnormal function and size of the left ventricle, disturbed endothelial function and hyperinsulinaemia [106]. Other comorbidities of obesity include metabolic syndrome [107], irregular menstruation due to excess oestrogen levels [108], infertility of both genders [109, 110], non-alcoholic steatohepatitis, arthritis, obstructive sleep apnoea, stroke, arrhythmias and heart failure [108]. Obesity may result in a reduction of the life expectancy of a 40 yearold person by 7 years [111].

For decades, high fat intake has been linked to obesity, diabetes, and heart disease [112]. Interestingly, the debate in the media about which diet is most appropriate for effective weight-loss is still under debate. Furthermore, a lack of knowledge of health practitioners about current therapeutic interventions reduces the likelihood of obese patients receiving the best care of practice to succeed in losing weight [113]. A particular focus has been placed on the increase or decreased consumption of the macronutrients, fats and carbohydrates. However, the question remains to this date which diet is more effective for weight loss and overall health, a low-fat or low-carbohydrate diet.

After a report by the U.S. Senate Select Committee on Nutrition and Human Needs in 1977, which entailed reduced consumption of fats and increased carbohydrate consumption, obesity rose significantly within the U.S. population [112, 114].



**Fig. 5.3** Relationship between obesity and macronutrients, fat and carbohydrates. (Data by Austin et al. (2011))

Furthermore, a study demonstrated the effects of macronutrient change from 1971 to 2006 within the U.S. population [115]. Using the published data from the above study, we found that fat consumption had an inverse relationship with obesity, while carbohydrate intake increased in parallel with obesity (Fig. 5.3). To shed further light on this phenomenon, a prospective urban rural epidemiology study investigated the dietary intake of 13,335 individuals with a median follow-up of 7.4 years [116]. This investigation found that high carbohydrate intake was associated with a greater risk of mortality compared to fat intake. Fat intake was not associated with cardiovascular disease, myocardial infarction, or cardiovascular disease mortality.

Multiple studies have outlined the successful use of the ketogenic diet in obesity through an increase in satiety [117]. The ketogenic diet has gained popularity in obesity since the early 2000s from only two publications in 2002 to 60 publications in 2018 (Fig. 5.4). A recent review outlined the potential effect of the ketogenic diet as a weight loss treatment [118]. All of the studies reviewed showed a reduction in body weight and body fat. Another study compared the effects of a low-carbohydrate diet (70% kcal from fat) with those of a ketogenic (89% kcal from fat) and a control diet in C57Bl/6 mice (all diets maintained at 11.2 kcal/day) [119]. After 100 weeks, the animals fed the low-carbohydrate diet were significantly heavier compared to the ketogenic diet and control groups.

A number of clinical studies have also been conducted which addressed this subject. For example, a recent clinical study investigated 35 sedentary obese patients on the ketogenic diet for 12 weeks, which resulted in significantly reduced appetite and weight loss [120]. Another clinical study of 80 overweight or obese patients demonstrated that those placed on a low-calorie ketogenic diet lost significantly more weight than those on a low-calorie diet [121]. Similarly, another study found a significant weight loss with a ketogenic diet [122].



Fig. 5.4 Number of articles published on the ketogenic diet in obesity from 1965–2018 (Scopus)

The ketogenic diet has been shown to have positive effects on metabolic biomarkers. A critical review demonstrated that dietary carbohydrate restriction was beneficial for patients with type II diabetes by decreasing blood glucose levels and having a positive effect on lipid levels [123]. Furthermore, the patients required less insulin and therefore experienced fewer side effects compared to pharmacological treatments. Elevated blood glucose levels have been shown to cause kidney damage, damage of the retina, neurological problems and diabetes mellitus [124]. A low carbohydrate diet can reduce weight and improve glycaemic control [105]. However, long-term reduction of carbohydrates may lead to hypoglycaemia in people taking insulin [105]. A study demonstrated that 8 weeks of the ketogenic diet significantly reduced blood glucose concentrations and insulin levels [120], lowering the risk for metabolic syndrome and type II diabetes [106, 125, 126]. The effect of the ketogenic diet on glucose tolerance and insulin sensitivity has been controversial. One study demonstrated that mice treated for 1 month with the ketogenic diet at 11.2 kcal/ day showed impaired glucose tolerance compared to control animals [119]. However, other studies showed that mice fed ad libitum with a ketogenic diet had enhanced glucose tolerance [127]. Another study demonstrated that mice fed previously on a high fat diet for 12 weeks and switched to a ketogenic diet showed a significant improvement in glucose tolerance and insulin sensitivity [128]. Furthermore, Badman et al. showed improved glucose sensitivity in ob/ob mice fed the ketogenic diet [129]. In contrast, a number of studies have shown that the ketogenic diet led to increased hepatic insulin resistance [130] and decreased glucose tolerance [130, 131].

Consistently high levels of glycated haemoglobin (HbA1c) are associated with an increased risk of developing diabetes mellitus [132, 133]. In light of this, admin-

istration of a ketogenic diet for 16 weeks led to significantly decreased levels of HbA1c in humans [132]. In a similar manner, decreased levels of high-densitylipoprotein (HDL) increase the risk for coronary heart disease [134] and 16 weeks on the ketogenic diet led to an increase in circulating HDL levels [132]. On the other hand, lower levels of low-density-lipoprotein (LDL) are associated with decreased risk of cardiovascular disease [134] and 24 weeks on the ketogenic diet was found to result in significantly reduced LDL levels [135]. However, a more recent study found an increase in LDLs in obese patients after 12 weeks of the ketogenic diet [120]. Although cholesterol is important in physiological processes such as forming the scaffolding of cell membranes and production of steroid hormones, vitamin D and bile acid, increased levels of cholesterol may lead to cardiovascular disease [134]. In obese patients, total cholesterol was significantly reduced on the ketogenic diet within 8 weeks [135]. Likewise, triglycerides levels (also associated with cardiovascular disease) were significantly decreased in the obese patients after 24 weeks on the ketogenic diet, despite showing an initial increase [135]. Leptin is important in regulating food intake. Therefore, high levels of leptin may cause obesity [136]. In this light, it is interesting that the ketogenic diet significantly decreased leptin levels [120, 137].

These studies show that the ketogenic diet might be beneficial in the case of obesity. However it is still not clear if the ketogenic diet has benefits for type II diabetes due to the observed conflicting effects on insulin sensitivity and glucose tolerance.

## 5 Ketogenic Diet in Old Age

#### 5.1 Alzheimer's Disease

Alzheimer's disease is a progressive and ultimately fatal neurodegenerative disorder [138]. In 2015, 46.8 million people worldwide suffered from dementias including Alzheimer's disease [139, 140]. It is predicted that by 2050 the number of people suffering from dementia including Alzheimer's disease will nearly triple to approximately 131.5 million people [139, 140]. Alzheimer's disease is the most common form of dementia characterized by progressive cognitive impairments leading to deficits in speech, thoughts and problem solving, and causing disorientation, confusion, irritability and aggression [140, 141]. Brain imaging studies have shown that hallmark physical features of Alzheimer's disease include neurofibrillary tangles and β-amyloid plaques, causing cortical atrophy and decreased synaptic plasticity [140, 142]. Excessive neuronal loss will result in loss of bodily functions resulting in death [143].

Current treatment options for mild-moderate Alzheimer's disease include cholinesterase inhibitors, N-methyl d-aspartate (NMDA) receptor antagonists and nonpharmacological interventions [141, 144]. All of the above-mentioned interventions slow down cognitive decline and temporarily reduce functional deterioration but do not stop the neurodegeneration. Alzheimer's disease is occasionally referred to as Type-3 diabetes due to the association of cognitive impairments and diabetes mellitus [145]. Patients with diabetes have a 65% increased risk of developing Alzheimer's disease compared to patients without diabetes mellitus [145, 146]. Along similar lines, patients with Alzheimer's disease have been shown to be more likely to have insulin resistance [146]. However, the underlying molecular mechanisms for this linkage are not understood. Nevertheless, a number of studies have confirmed that insulin signalling is impaired in Alzheimer's disease and animal models of Alzheimer's disease [146]. Therefore, the ketogenic diet effecting insulin pathways might serve as a novel therapeutic approach for the treatment of Alzheimer's disease [145, 147].

Preclinical evidence has suggested that the ketogenic diet may have some efficacy in Alzheimer's disease. Studies in animal models of Alzheimer's disease have demonstrated behavioural improvements with the ketogenic diet and betahydroxybutyrate treatment, such as increased activity [148–151] and improved memory performance [150, 152]. These behavioural improvements were underlined by reduced levels of β-amyloid plaques [148–150, 152], reduced oxidative stress [148] and improved mitochondrial function [151, 152].

High fat, "Western" diets (HFD- high in carbohydrates and fat) have been implicated as a risk factor for Alzheimer's disease. A longitudinal community-based study from the Chicago Health and Aging Project investigated the effects of various dietary fats on Alzheimer's disease between 1993 and 1997 [153]. The subjects of the study showed a high correlation between consumption of high amounts of saturated and trans-unsaturated fats and development of Alzheimer's disease, whereas participants consuming higher amounts of omega-3 polyunsaturated fatty acids had a significantly reduced risk of developing the disorder. Total fat, animal fat and dietary cholesterol were not associated with an increased risk of developing Alzheimer's disease.

In a clinical study, 23 older adults with mild to moderate memory decline were assigned to either a high or low carbohydrate diet for 6 weeks [154]. Patients on the low carbohydrate diet improved in paired-associated learning after 6 weeks on the diet compared to baseline, whereas no difference was found for the high carbohydrate group. The subjects consuming the low carbohydrate diet showed significantly reduced weight, fasting glucose and fasting insulin levels. No changes in metabolic parameters were found for the high carbohydrate group after 6 weeks on their assigned diet compared to baseline. Finally, a significant positive correlation was identified between ketone levels and memory performance in the low carbohydrate group. This study demonstrated that decreased carbohydrate consumption results in ketosis and decreased insulin levels. However, the authors found no positive correlation between insulin levels and memory performance, which was suggested to be due to the limited sample size. Another clinical study demonstrated that calorie restriction in a larger cohort improved memory performance and they confirmed that this was correlated with decreased fasting insulin levels [155].

A case report demonstrated that supplementing a diet with ketogenic agents, such as ketone monoesters and medium chain fatty acids, resulted in cognitive improvement [156]. The patient rediscovered everyday tasks such as showering and shaving and their memory significantly improved after 6 weeks of treatment with ketone monoesters. The authors showed that the cognitive improvement was closely linked to increased beta-hydroxybutyrate levels. Another case report showed a significant improvement on Montreal Cognitive Assessment scores for a patient with mild Alzheimer's disease placed on the ketogenic diet [157].

Using a small, single-arm pilot clinical trial format, a study showed that a ketogenic diet supplemented with fatty acids resulted in a state of ketosis [16] and the patients showed significantly improvements in their Alzheimer's Disease Assessment Scale-cognitive subscale scores after 3 months. These studies indicate that a ketogenic diet might be beneficial for patients with Alzheimer's disease both from a metabolic and a cognitive perspective.

#### 5.2 Parkinson's Disease

Parkinson's disease is the second most common neurodegenerative disorder [158–160], affecting approximately 1% of the population aged 60 years or over [160–162]. Parkinson's disease is characterized by bradykinesia and akinesia (slowness or loss of movement), muscle rigidity, tremor and postural instability [160, 162, 163]. Non-motor symptoms include hyposmia, autonomic dysfunction, sleep dysfunction and a variety of psychiatric symptoms [159]. Genetic predisposition, environmental toxins and oxidative stress play a role in the pathophysiology of Parkinson's disease [163]. The core process of Parkinson's disease is the loss of dopaminergic neurons in the substantia niga region in the midbrain in combination with appearance of Lewy bodies and Lewy neuritis [159]. Patients with Parkinson's disease start showing symptoms when over 50% of the dopaminergic cells are lost [164]. Dopamine replacement therapy, such as L-dihydoxyphenylalanine (L-Dopa) and receptor agonists, is currently the most wildly used treatment [160]. Treatment can mitigate the symptoms but does not delay the neurodegeneration or regenerate the lost neurons [160].

Inflammation is a mayor hallmark of the disease, with an increase in inflammatory cytokines in the brain and the cerebrospinal fluid [158]. The ketogenic diet has been shown to have beneficial effects on inflammation [37] and may therefore might be a novel and safe treatment for patients with Parkinson's disease. Preclinical studies have investigated the effect of the ketogenic diet in a variety of different animal models of Parkinson's disease. Such studies have demonstrated increased motor performance following a ketogenic diet [165, 166] and administration of betahydroxybutyrate injections [167]. These studies further demonstrated that the ketogenic diet increased dopamine and its metabolites and normalised neuronal counts in the substantia nigra pars compacta [165, 168]. Likewsie, beta-hydroxybutyrate treatment increased dopamine metabolites, ATP production and mitochondrial oxygen consumption [167]. In addition, the ketogenic diet led to reduced levels of proinflammatory cytokines [165].

The ketogenic diet has been trialled in humans with Parkinson's disease and was able to improve resting tremors, freezing, balance, gait, mood and energy levels within 1 month [17]. Two studies demonstrated improvement on the Unified Parkinson's Disease Rating Scale following the ketogenic diet [17, 18]. Importantly, another study demonstrated that the diet did not interfere with the pharmacokinetics of L-Dopa [169].

## 6 Potential Side Effects of Ketogenic Diet

The ketogenic diet is a therapeutic nutritional intervention and may have some side effects. Some side effects are short or long lasting and can be temporary or lead to permanent changes. These side effects can led to discontinuation of the diet, although appropriate nutritional guidance and adjustments of the diet can reduce or eliminate side effects.

In a randomized controlled trial, 73 patients were recruited and 6 of these poorly tolerated the diet, three withdrew because of parental unhappiness and one child showed increased seizure activity [170]. Kang et al. investigated 129 children with epilepsy and assessed the safety of the ketogenic diet [171]. Minor side effects included extreme drowsiness, dehydration (46.5%), abdominal pain (9%), hunger (22%), lack of energy (24%) and gastrointestinal disturbance such as constipation (33%), vomiting (24%, 27.9%), diarrhoea (13%, 32.6%), constipation (2.3%) and infectious diseases like pneumonia, cystitis and non-specific febrile illnesses (9.3%) and hypercholesterolemia (14.7%), hyperuricemia (26.4%), symptomatic hypoglycaemia (7%), hypoproteinemia (5.4%), hypomagnesaemia (4.7%), repetitive hypernatremia (4.7%) and hepatitis (2.3%). Kang et al. reported that all of the above-mentioned side effects were transient and required only short-term medication. Constipation is caused by the high-fat and low-fibre content of the diet [172]. In line with this, a randomized control trial showed that 33% of children on the ketogenic diet were constipated [170]. On the other hand, the ketogenic diet may cause diarrhoea due to ineffective absorption and the high fat content of the diet [171]. Constipation and diarrhoea can be prevented through administration of vitamins and supplements. Supplement treatment will also guarantee delivery of trace minerals that are reduced within a standard ketogenic diet. The lack of essential minerals may cause reduced energy, which was seen in 24% of patients on the diet [170]. The associated dehydration caused dried mucous membranes [173]. The vomiting, constipation and diarrhoea might be due to the rapid change in diet. Protein malnutrition, especially in children, has to be prevented due to potential stunted growth [174, 175]. With this in mind, a retrospective review of 32 infants showed that 96.4% of patients showed appropriate growth [176].

Kidney stones may occur in 3.1% of patients on the ketogenic diet [171], which can be mediated through increased fluid intake and reduced intake of medication

[177, 178]. A review of 58 children at the Johns Hopkins Hospital between 1980 and 1985 revealed that three patients on the diet developed kidney stones, two had hyperuricemia, one had acidosis and one had hypocalcaemia [179].

At Johns Hopkins Hospital in 1972, 1000 patients with epilepsy on the ketogenic diet showed only a few serious complications [35]. A prospective study of 150 children showed that two children died when on the ketogenic diet, but these children were suffering prior to the study from cardiac dysrhythmia and died of cardiac arrest 1 month after starting the diet. In a further study, two children developed hypoproteinemia, one child developed Fanconi's renal tubular acidosis and two children showed an increased liver function test. Notably, these children were on valproate [180], which has been shown to induce Fanconi's renal tubular acidosis and an increased liver function test [171]. Therefore, it is still not clear whether the ketogenic diet or the anti-seizure medication caused these side effects. Another rare side effect may be acute haemorrhagic pancreatitis, caused by hyperlipidaemia and hypertriglyceridemia [173, 181].

In conclusion, the ketogenic diet is generally safe and unwanted side effects can be eliminated by slightly adjusting the dietary composition. It is rarely required to medicate the side effects or discontinue the diet. However, the level of ketosis must be regulated tightly to ensure efficacy with minimum side effects. A recent study introduced a compact and inexpensive breath sensor to monitor ketosis online [182]. They tested the device on 11 subjects undergoing a 36 h ketogenic diet treatment and the sensor recognized the onset and progression of ketosis, and the readings were consistent with those from capillary blood  $\beta$ -hydroxybutyrate (BOHB) measurements. Interestingly the sensor was capable of detecting those individuals with lower tolerance to the diet. It should be noted that when deficiencies are identified such as depletion of key vitamins and minerals these can be included in the diet as supplements [183].

#### 7 Conclusion

This accumulation of preclinical and clinical evidence indicates that the ketogenic diet could potentially be an effective therapeutic avenue for a variety of disorders with metabolic and inflammatory components. However, current evidence is lacking in randomised, properly controlled, larger scale clinical trials to fully evaluate the efficacy of the ketogenic diet and the potential side effects in the aforementioned diseases and disorders (Table 5.1). The exact mechanism of action of the ketogenic diet has yet to be established in the individual diseases and disorders. The ketogenic diet appears to have fewer side-effects than most current treatment options and might therefore be a potential novel and safer treatment avenue for diseases. Although the ketogenic diet was effective in treating a variety of diseases, it cannot be neglected that the main complication with this diet is non-compliance. Therefore, substances acting like the ketogenic diet and therefore increasing ketosis might have similar beneficial effects as highlighted in this chapter. One recent study showed

		Case	Uncontrolled	Randomised, controlled
Disorder	Preclinical	study	trial	trial
Childhood				
Epilepsy	$\checkmark$		$\checkmark$	$\checkmark$
ADHD	$\checkmark$		$\checkmark$	
ASD	$\checkmark$	$\checkmark$	$\checkmark$	
Adolescence/Early adulthood				
Schizophrenia	$\checkmark$	$\checkmark$		
Adult				
Obesity/Type-2	$\checkmark$		$\checkmark$	
diabetes				
Elderly				
Alzheimer's	$\checkmark$		$\checkmark$	
Parkinson's	$\checkmark$		$\checkmark$	

Table 5.1 Available evidence of diseases through-out life

ADHD attention-deficit hyperactivity disorder, ASD autism spectrum disorder,  $\sqrt{\text{published}}$  evidence available

that the ketogenic diet may even have uses as an effective alternative therapeutic strategy for super-refractory status epilepticus patients in intensive care units for reducing seizure frequency and weaning from prolonged mechanical ventilation [184]. However, close monitoring and preventive management of potential adverse effects are critical elements for success. It may even be possible that certain populations respond differently to the ketogenic diet depending on the condition itself as well as certain biomarker readings. Thus, it may be beneficial to carry out further research aimed at identification of such biomarkers. Future research should also be aimed at discovery of novel mechanistic targets of the ketogenic diet, which might result in novel treatment options.

## References

- Mattson MP, Longo VD, Harvie M (2017) Impact of intermittent fasting on health and disease processes. Ageing Res Rev 39:46–58
- Guelpa G, Marie A (1911) La lutte contre l'epilepsie par ladesintoxication et par la re-education alimentaire. Revuede Therapie Medico-Chirurgide 78:8–13
- 3. Woodyatt R (1921) Objects and method of diet adjustment in diabetics. Arch Intern Med 28:125–141
- 4. Wilder R (1921) The effect on ketonemia on the course of epilepsy. Mayo Clin Bull 2:307
- Seyfried TN (2014) Ketone strong: emerging evidence for a therapeutic role of ketone bodies in neurological and neurodegenerative diseases. J Lipid Res 55(9):1815–1817
- 6. Bough KJ, Rho JM (2007) Anticonvulsant mechanisms of the ketogenic diet. Epilepsia 48(1):43–58
- Paoli A, Rubini A, Volek JS, Grimaldi KA (2013) Beyond weight loss: a review of the therapeutic uses of very-low-carbohydrate (ketogenic) diets. Eur J Clin Nutr 67(8):789–796

#### 5 Ketogenic Diet Throughout Life

- Hartman AL, Gasior M, Vining EP, Rogawski MA (2007) The neuropharmacology of the ketogenic diet. Pediatr Neurol 36(5):281–292
- Leino RL, Gerhart DZ, Duelli R, Enerson BE, Drewes LR (2001) Diet-induced ketosis increases monocarboxylate transporter (MCT1) levels in rat brain. Neurochem Int 38(6):519–527
- Noh HS, Lee HP, Kim DW, Kang SS, Cho GJ, Rho JM et al (2004) A cDNA microarray analysis of gene expression profiles in rat hippocampus following a ketogenic diet. Brain Res Mol Brain Res 129(1–2):80–87
- 11. Seyfried TN, Mukherjee P (2005) Targeting energy metabolism in brain cancer: review and hypothesis. Nutr Metab (Lond) 2:30. https://doi.org/10.1186/1743-7075-2-30
- Stafstrom CE, Rho JM (2012) The ketogenic diet as a treatment paradigm for diverse neurological disorders. Front Pharmacol 3:59. https://doi.org/10.3389/fphar.2012.00059
- Kossoff EH, Huffman J, Turner Z, Gladstein J (2010) Use of the modified Atkins diet for adolescents with chronic daily headache. Cephalalgia 30(8):1014–1016
- Maggioni F, Margoni M, Zanchin G (2011) Ketogenic diet in migraine treatment: a brief but ancient history. Cephalalgia 31(10):1150–1151
- Barañano KW, Hartman AL (2008) The ketogenic diet: uses in epilepsy and other neurologic illnesses. Curr Treat Options Neurol 10(6):410–419
- Taylor MK, Sullivan DK, Mahnken JD, Burns JM, Swerdlow RH (2018) Feasibility and efficacy data from a ketogenic diet intervention in Alzheimer's disease. Alzheimers Dement (N Y) 4:28–36
- Vanitallie TB, Nonas C, Di Rocco A, Boyar K, Hyams K, Heymsfield SB (2005) Treatment of Parkinson disease with diet-induced hyperketonemia: a feasibility study. Neurology 64(4):728–730
- Phillips MCL, Murtagh DKJ, Gilbertson LJ, Asztely FJS, Lynch CDP (2018) Low-fat versus ketogenic diet in Parkinson's disease: a pilot randomized controlled trial. Mov Disord 33(8):1306–1314
- Choi IY, Piccio L, Childress P, Bollman B, Ghosh A, Brandhorst S (2016) A diet mimicking fasting promotes regeneration and reduces autoimmunity and multiple sclerosis symptoms. Cell Rep 15(10):2136–2146
- 20. Evangeliou A, Vlachonikolis I, Mihailidou H, Spilioti M, Skarpalezou A, Makaronas N et al (2003) Application of a ketogenic diet in children with autistic behavior: pilot study. J Child Neurol 18(2):113–118
- Herbert MR, Buckley JA (2013) Autism and dietary therapy: case report and review of the literature. J Child Neurol 28(8):975–982
- 22. Lee RWY, Corley MJ, Pang A, Arakaki G, Abbott L, Nishimoto M et al (2018) A modified ketogenic gluten-free diet with MCT improves behavior in children with autism spectrum disorder. Physiol Behav 188:205–211
- 23. Żarnowska I, Chrapko B, Gwizda G, Nocuń A, Mitosek-Szewczyk K, Gasior M (2018) Therapeutic use of carbohydrate-restricted diets in an autistic child; a case report of clinical and 18FDG PET findings. Metab Brain Dis 33(4):1187–1192
- Pulsifer MB, Gordon JM, Brandt J, Vining EP, Freeman JM (2001) Effects of ketogenic diet on development and behavior: preliminary report of a prospective study. Dev Med Child Neurol 43(5):301–306
- Pacheco A, Easterling WS, Pryer MW (1965) A pilot study of the ketogenic diet in Schizophrenia. Am J Psychiatry 121:1110–1111
- 26. Kraft BD, Westmen EC (2009) Schizophrenia, gluten, and low-carbohydrate, ketogenic diets: a case report and review of the literature. Nutr Metab (Lond) 6:10. https://doi.org/10.1186/1743-7075-6-10
- Palmer CM (2017) Ketogenic diet in the treatment of schizoaffective disorder: two case studies. Schizophr Res 189:208–209
- Gilbert-Jaramillo J, Vargas-Pico D, Espinosa-Mendoza T, Falk S, Llanos-Fernàndez K, Guerrero-Haro J et al (2018) The effects of the ketogenic diet on psychiatric symptomatol-

ogy, weight and metabolic dysfunction in schizophrenia patients. Clin Nutr Metab 1(2). https://doi.org/10.15761/CNM.1000105

- Yaroslavsky Y, Stahl Z, Belmaker RH (2002) Ketogenic diet in bipolar illness. Bipolar Disord 4(1):75
- Phelps JR, Siemers SV, El-Mallakh RS (2013) The ketogenic diet for type II bipolar disorder. Neurocase 19(5):423–426
- Murray CJ, Vos T, Lozano R, Naghavi M, Flaxman AD, Michaud C et al (2012) Disabilityadjusted life years (DALYs) for 291 diseases and injuries in 21 regions, 1990-2010: a systematic analysis for the Global Burden of Disease Study 2010. Lancet 380(9859):2197–2223
- Cummings JL, Mega MS (2003) Neuropsychiatry and behavioral neuroscience. Oxford University Press, Cary, NC, USA. ISBN: 9780195138580
- 33. Peterman M (1925) The ketogenic diet in epilepsy. JAMA 84:1979-1983
- 34. Lutas A, Yellen G (2013) The ketogenic diet: metabolic influences on brain excitability and epilepsy. Trends Neurosci 36(1):32–40
- 35. Livingston S (1972) Dietary treatment of epilepsy. In: Livingston S (ed) Comprehensive management of epilepsy in infancy, childhood and adolescence. Charles C. Thomas, Springfield, pp 378–405. ASIN: B0006C2W3U
- 36. Kwan P, Brodie MJ (2000) Early identification of refractory epilepsy. N Engl J Med 342(5):314–319
- 37. Rho JM (2015) How does the ketogenic diet induce anti-seizure effects? Neurosci Lett 637:4–10
- Youngson NA, Morris MJ, Ballard JWO (2017) The mechanisms mediating the antiepileptic effects of the ketogenic diet, and potential opportunities for improvement with metabolismaltering drugs. Seizure 52:15–19
- McDonald TJW, Cervenka MC (2018) The expanding role of ketogenic diets in adult neurological disorders. Brain Sci 8(8). pii: E148. https://doi.org/10.3390/brainsci8080148
- 40. van Berkel AA, IJff DM, Verkuyl JM (2018) Cognitive benefits of the ketogenic diet in patients with epilepsy: a systematic overview. Epilepsy Behav 87:69–77
- Guzel O, Uysal U, Arslan N (2019) Efficacy and tolerability of olive oil-based ketogenic diet in children with drug-resistant epilepsy: a single center experience from Turkey. Eur J Paediatr Neurol 23(1):143–151
- 42. Lindefeldt M, Eng A, Darban H, Bjerkner A, Zetterström CK, Allander T et al (2019) The ketogenic diet influences taxonomic and functional composition of the gut microbiota in children with severe epilepsy. NPJ Biofilms Microbiomes 5:5. https://doi.org/10.1038/ s41522-018-0073-2
- Carroll J, Martin-McGill K, Cross H, Hickson M, Collinson A (2019) Outcome measurement and reporting in childhood epilepsy treated with ketogenic diet therapy: a scoping review protocol. JBI Database System Rev Implement Rep. https://doi.org/10.11124/ JBISRIR-2017-003924. [Epub ahead of print]
- 44. Howard AL, Robinson M, Smith GJ, Ambrosini GL, Piek JP, Oddy WH (2011) ADHD is associated with a "Western" dietary pattern in adolescents. J Atten Disord 15(5):403–411
- 45. Thapar A, Cooper M (2016) Attention deficit hyperactivity disorder. Lancet 387(10024):1240–1250
- 46. Dutra TG, Baltar A, Monte-Silva KK (2016) Motor cortex excitability in attention-deficit hyperactivity disorder (ADHD): a systematic review and meta-analysis. Res Dev Disabil 56:1–9
- 47. Polanczyk G, de Lima MS, Horta BL, Biederman J, Rohde LA (2007) The worldwide prevalence of ADHD: a systematic review and metaregression analysis. Am J Psychiatry 164(6):942–948
- Wilens TE, Biederman J, Spencer TJ (2002) Attention deficit/hyperactivity disorder across the lifespan. Annu Rev Med 53:113–131
- Moriyama TS, Polanczyk GV, Terzi FS, Faria KM, Rohde LA (2013) Psychopharmacology and psychotherapy for the treatment of adults with ADHD-a systematic review of available meta-analyses. CNS Spectr 18(6):296–306

- 5 Ketogenic Diet Throughout Life
  - Murphy P, Likhodii SS, Hatamian M, McIntyre Burnham W (2005) Effect of the ketogenic diet on the activity level of Wistar rats. Pediatr Res 57(3):353–357
  - Murphy P, Burnham WM (2006) The ketogenic diet causes a reversible decrease in activity level in long-Evans rats. Exp Neurol 201(1):84–89
  - 52. Packer RM, Law TH, Davies E, Zanghi B, Pan Y, Volk HA (2016) Effects of a ketogenic diet on ADHD-like behavior in dogs with idiopathic epilepsy. Epilepsy Behav 55:62–68
  - Hampson DR, Blatt GJ (2015) Autism spectrum disorders and neuropathology of the cerebellum. Front Neurosci 9:420. https://doi.org/10.3389/fnins.2015.00420
  - Fuccillo MV (2016) Striatal circuits as a common node for autism pathophysiology. Front Neurosci 10:27. https://doi.org/10.3389/fnins.2016.00027
  - 55. Li Q, Zhou JM (2016) The microbiota-gut-brain axis and its potential therapeutic role in autism spectrum disorder. Neuroscience 324:131–139
  - Vogel Ciernia A, LaSalle J (2016) The landscape of DNA methylation amid a perfect storm of autism aetiologies. Nat Rev Neurosci 17(7):411–423
  - Kern JK, Geier DA, Sykes LK, Geier MR (2015) Relevance of neuroinflammation and encephalitis in autism. Front Cell Neurosci 9:519. https://doi.org/10.3389/fncel.2015.00519
  - 58. Anagnostou E, Soorya L, Brian J, Dupuis A, Mankad D, Smile S et al (2014) Intranasal oxytocin in the treatment of autism spectrum disorders: a review of literature and early safety and efficacy data in youth. Brain Res 1580:188–198
  - Ruskin DN, Svedova J, Cote JL, Sandau U, Rho JM, Kawamura M Jr et al (2013) Ketogenic diet improves core symptoms of autism in BTBR mice. PLoS One 8:e65021. https://doi. org/10.1371/journal.pone.0065021
  - 60. Ahn Y, Narous M, Tobias R, Rho JM, Mychasiuk R (2014) The ketogenic diet modifies social and metabolic alterations identified in the prenatal valproic acid model of autism spectrum disorder. Dev Neurosci 36(5):371–380
  - Verpeut JL, DiCicco-Bloom E, Bello NT (2016) Ketogenic diet exposure during the juvenile period increases social behaviors and forebrain neural activation in adult Engrailed 2 null mice. Physiol Behav 161:90–98
  - 62. Castro K, Baronio D, Perry IS, Riesgo RDS, Gottfried C (2017) The effect of ketogenic diet in an animal model of autism induced by prenatal exposure to valproic acid. Nutr Neurosci 20(6):343–350
  - Ruskin DN, Fortin JA, Bisnauth SN, Masino SA (2017) Ketogenic diets improve behaviors associated with autism spectrum disorder in a sex-specific manner in the EL mouse. Physiol Behav 168:138–145
  - 64. Ruskin DN, Murphy MI, Slade SL, Masino SA (2017) Ketogenic diet improves behaviors in a maternal immune activation model of autism spectrum disorder. PLoS One 12:e0171643. https://doi.org/10.1371/journal.pone.0171643
  - Mychasiuk R, Rho JM (2017) Genetic modifications associated with ketogenic diet treatment in the BTBR(T+Tf/J) mouse model of autism spectrum disorder. Autism Res 10(3):456–471
  - 66. Newell C, Bomhof MR, Reimer RA, Hittel DS, Rho JM, Shearer J (2016) Ketogenic diet modifies the gut microbiota in a murine model of autism spectrum disorder. Mol Autism 7(1):37. https://doi.org/10.1186/s13229-016-0099-3
  - El-Rashidy O, El-Baz F, El-Gendy Y, Khalaf R, Reda D, Saad K (2017) Ketogenic diet versus gluten free casein free diet in autistic children: a case-control study. Metab Brain Dis 32(6):1935–1941
  - Bowden NA, Weidenhofer J, Scott RJ, Schall U, Todd J, Michie PT et al (2006) Preliminary investigation of gene expression profiles in peripheral blood lymphocytes in schizophrenia. Schizophr Res 82(2–3):175–183
  - Zipursky RB, Reilly TJ, Murray RM (2013) The myth of schizophrenia as a progressive brain disease. Schizophr Bull 39(6):1363–1372
  - 70. Evensen S, Wisløff T, Lystad JU, Bull H, Ueland T, Falkum E (2016) Prevalence, employment rate, and cost of Schizophrenia in a High-Income Welfare Society: a population-based study using comprehensive health and welfare registers. Schizophr Bull 42(2):476–483

- 71. Owen MJ, Sawa A, Mortensen PB (2016) Schizophrenia. Lancet 388(10039):86-97
- 72. https://www.who.int/mental\_health/management/schizophrenia/en/
- Andreasen N (1995) Symptoms, signs, and diagnosis of schizophrenia. Lancet 346(8973): 477–481
- 74. Tandon R, Keshavan MS, Nasrallah HA (2008) Schizophrenia, "Just the Facts": what we know in 2008 part 1: overview. Schizophr Res 100(1–3):4–19
- 75. van Os J, Kapur S (2009) Schizophrenia. Lancet 374(9690):635–645
- Palmer B, Pankratz V, Bostwick J (2005) The lifetime risk of suicide in schizophrenia: a reexamination. Arch Gen Psychiatry 62(3):247–253
- 77. Keltner N, Grant J (2006) Smoke, smoke, smoke that cigarette. Perspect Psychiatr Care 42(4):256–261
- 78. Lozano R, Naghavi M, Foreman K, Lim S, Shibuya K, Aboyans V et al (2012) Global and regional mortality from 235 causes of death for 20 age groups in 1990 and 2010: a systematic analysis for the Global Burden of Disease Study 2010. Lancet 380(9859):2095–2128
- 79. van Rossum JM (1966) The significance of dopamine-receptor blockade for the mechanism of action of neuroleptic drugs. Arch Int Pharmacodyn Ther 160(2):492–494
- Seeman P, Niznik HB (1990) Dopmine receptors and transporters in Parkinson's disease and schizophrenia. FASEB J 4(10):2737–2744
- 81. Laruelle M, Abi-Dargham A, van Dyck CH, Gil R, D'Souza CD, Erdos J et al (1996) Single photon emission computerized tomography imaging of amphetamine-induced dopamine release in drug-free schizophrenic subjects. Proc Natl Acad Sci U S A 93(17):9235–9240
- Seeman P, Kapur S (2000) Schizophrenia: more dopamine, more D2 receptors. Proc Natl Acad Sci U S A 97(14):7673–7675
- Toda M, Abi-Dargham A (2007) Dopamine hypothesis of schizophrenia: making sense of it all. Curr Psychiatry Rep 9(4):329–336
- Frohlich J, Horn JD (2014) Reviewing the ketamine model for schizophrenia. J Psychopharmacol 28(4):287–302
- Craddock N, O'Donovan MC, Owen MJ (2005) The genetics of schizophrenia and bipolar disorder: dissecting psychosis. J Med Genet 42(3):193–204
- Harrison PJ, Weinberger DR (2005) Schizophrenia genes, gene expression, and neuropathology: on the matter of their convergence. Mol Psychiatry 10(1):40–68; image 5
- Magistretti PJ, Allaman I (2015) A cellular perspective on brain energy metabolism and functional imaging. Neuron 86(4):883–901
- Kann O (2016) The interneuron energy hypothesis: implications for brain disease. Neurobiol Dis 90:75–85
- Pillinger T, Beck K, Gobjila C, Donocik JG, Jauhar S, Howes OD (2017) Impaired glucose homeostasis in first-episode Schizophrenia: a systematic review and meta-analysis. JAMA Psychiat 74(3):261–269
- 90. Steiner J, Berger M, Guest PC, Dobrowolny H, Westphal S, Schiltz K (2017) Assessment of insulin resistance among drug-naive patients with first-episode Schizophrenia in the context of hormonal stress axis activation. JAMA Psychiat 74(9):968–970
- Sullivan CR, O'Donovan SM, McCullumsmith RE, Ramsey A (2018) Defects in bioenergetic coupling in Schizophrenia. Biol Psychiatry 83(9):739–750
- 92. Chouinard VA, Henderson DC, Dalla Man C, Valeri L, Gray BE, Ryan KP et al (2018) Impaired insulin signaling in unaffected siblings and patients with first-episode psychosis. Mol Psychiatry. https://doi.org/10.1038/s41380-018-0045-1. [Epub ahead of print]
- Dwyer DS, Bradley RJ, Kablinger AS, Freeman AM 3rd (2001) Glucose metabolism in relation to schizophrenia and antipsychotic drug treatment. Ann Clin Psychiatry 13(2):103–113
- 94. Fujimoto T, Takeuch K, Matsumoto T, Kamimura K, Hamada R, Nakamura K et al (2007) Abnormal glucose metabolism in the anterior cingulate cortex in patients with schizophrenia. Psychiatry Res 154(1):49–58
- Krivoy A, Fischel T, Weizman A (2008) The possible involvement of metabotropic glutamate receptors in schizophrenia. Eur Neuropsychopharmacol 18(6):395–405

- 96. Beasley CL, Dwork AJ, Rosoklija G, Mann JJ, Mancevski B, Jakovski Z et al (2009) Metabolic abnormalities in fronto-striatal-thalamic white matter tracts in schizophrenia. Schizophr Res 109(1–3):159–166
- Harris LW, Guest PC, Wayland MT, Umrania Y, Krishnamurthy D, Rahmoune H et al (2013) Schizophrenia: metabolic aspects of aetiology, diagnosis and future treatment strategies. Psychoneuroendocrinology 38(6):752–766
- Sullivan CR, Koene RH, Hasselfeld K, O'Donovan SM, Ramsey A, McCullumsmith RE (2018) Neuron-specific deficits of bioenergetic processes in the dorsolateral prefrontal cortex in schizophrenia. Mol Psychiatry. https://doi.org/10.1038/s41380-018-0035-3. [Epub ahead of print]
- Kraeuter AK, Loxton H, Lima BC, Rudd D, Sarnyai Z (2015) Ketogenic diet reverses behavioral abnormalities in an acute NMDA receptor hypofunction model of schizophrenia. Schizophr Res 169(1–3):491–493
- 100. Kraeuter AK, van den Buuse M, Sarnyai Z 2018. Ketogenic diet prevents impaired prepulse inhibition of startle in an acute NMDA receptor hypofunction model of schizophrenia. Schizophr Res. pii: S0920–9964(18)30662–5. https://doi.org/10.1016/j.schres.2018.11.011. [Epub ahead of print]
- Tregellas JR, Smucny J, Legget KT, Stevens KE (2015) Effects of a ketogenic diet on auditory gating in DBA/2 mice: a proof-of-concept study. Schizophr Res 169(1–3):351–354
- 102. Dohan FC (1966) Cereals and schizophrenia data and hypothesis. Acta Psychiatr Scand 42(2):125–152
- 103. Kalaydjian AE, Eaton W, Cascella N, Fasano A (2006) The gluten connection: the association between schizophrenia and celiac disease. Acta Psychiatr Scand 113(2):82–90
- 104. Chen L, Magliano DJ, Zimmet PZ (2012) The worldwide epidemiology of type 2 diabetes mellitus—present and future perspectives. Nat Rev Endocrinol 8(4):228–236
- 105. Dyson P (2014) Popular diets: are they effective for people with type 2 diabetes? Pract Diabetes 31:187–192
- 106. Reilly JJ, Methven E, McDowell ZC, Hacking B, Alexander D, Stewart L et al (2003) Health consequences of obesity. Arch Dis Child 88(9):748–752
- 107. https://www.idf.org/
- 108. Haslam DW, James WP (2005) Obesity. Lancet 366(9492):1197-1209
- 109. Esposito K, Giugliano F, Di Palo C, Giugliano G, Marfella R, D'Andrea F et al (2004) Effect of lifestyle changes on erectile dysfunction in obese men: a randomized controlled trial. JAMA 291(24):2978–2984
- 110. Green BB, Weiss NS, Daling JR (1988) Risk of ovulatory infertility in relation to body weight. Fertil Steril 50(5):721–726
- 111. Peeters A, Barendregt JJ, Willekens F, Mackenbach JP, Al Mamun A, Bonneux L et al (2003) Obesity in adulthood and its consequences for life expectancy: a life-table analysis. Ann Intern Med 138(1):24–32
- 112. Ludwig DS, Willett WC, Volek JS, Neuhouser ML (2018) Dietary fat: from foe to friend? Science 362(6416):764–770
- 113. Yancy WS Jr, Olsen MK, Guyton JR, Bakst RP, Westman EC (2004) A low-carbohydrate, ketogenic diet versus a low-fat diet to treat obesity and hyperlipidemia: a randomized, controlled trial. Ann Intern Med 140(10):768–777
- 114. Ludwig DS (2016) Lifespan weighed down by diet. JAMA 315(21):2269-2270
- 115. Austin GL, Ogden LG, Hill JO (2011) Trends in carbohydrate, fat, and protein intakes and association with energy intake in normal-weight, overweight, and obese individuals: 1971-2006. Am J Clin Nutr 93(4):836–843
- 116. Dehghan M, Mente A, Zhang X, Swaminathan S, Li W, Mohan V et al (2017) Associations of fats and carbohydrate intake with cardiovascular disease and mortality in 18 countries from five continents (PURE): a prospective cohort study. Lancet 390(10107):2050–2062
- 117. Gibson AA, Seimon RV, Lee CM, Ayre J, Franklin J, Markovic TP et al (2015) Do ketogenic diets really suppress appetite? A systematic review and meta-analysis. Obes Rev 16(1):64–76

- 118. Tinsley GM, Willoughby DS (2016) Fat-free mass changes during ketogenic diets and the potential role of resistance training. Int J Sport Nutr Exerc Metab 26(1):78–92
- 119. Roberts MN, Wallace MA, Tomilov AA, Zhou Z, Marcotte GR, Tran D et al (2017) A ketogenic diet extends longevity and healthspan in adult mice. Cell Metab 26(3):539–546.e5
- 120. Mohorko N, Černelič-Bizjak M, Poklar-Vatovec T, Grom G, Kenig S, Petelin A et al (2019) Weight loss, improved physical performance, cognitive function, eating behavior, and metabolic profile in a 12-week ketogenic diet in obese adults. Nutr Res 62:64–77
- 121. Genco A, Ienca R, Ernesti I, Maselli R, Casella G, Bresciani S (2018) Improving weight loss by combination of two temporary antiobesity treatments. Obes Surg 28(12):3733–3737
- 122. Castro AI, Gomez-Arbelaez D, Crujeiras AB, Granero R, Aguera Z, Jimenez-Murcia S et al (2018) Effect of a very low-calorie ketogenic diet on food and alcohol cravings, physical and sexual activity, sleep disturbances, and quality of life in obese patients. Nutrients 10(10). pii: E1348. https://doi.org/10.3390/nu10101348
- 123. Feinman RD, Pogozelski WK, Astrup A, Bernstein RK, Fine EJ, Westman EC (2015) Dietary carbohydrate restriction as the first approach in diabetes management: critical review and evidence base. Nutrition 31(1):1–13
- 124. Lindenmayer JP, Czobor P, Volavka J, Citrome L, Sheitman B, McEvoy JP et al (2003) Changes in glucose and cholesterol levels in patients with Schizophrenia treated with typical or atypical antipsychotics. Am J Psychiatry 160(2):290–296
- 125. Modan M, Halkin H, Almog S, Lusky A, Eshkol A, Shefi M et al (1985) Hyperinsulinemia: a link between hypertension obesity and glucose intolerance. J Clin Invest 75(3):809–817
- 126. Thio LL, Erbayat-Altay E, Rensing N, Yamada KA (2006) Leptin contributes to slower weight gain in Juvenile Rodents on a ketogenic diet. Pediatr Res 60(4):413–417
- 127. Douris N, Melman T, Pecherer JM, Pissios P, Flier JS, Cantley LC et al (2015) Adaptive changes in amino acid metabolism permit normal longevity in mice consuming a lowcarbohydrate ketogenic diet. Biochim Biophys Acta 1852(10 Pt A):2056–2065
- 128. Kennedy AR, Pissios P, Otu H, Roberson R, Xue B, Asakura K et al (2007) A high-fat, ketogenic diet induces a unique metabolic state in mice. Am J Physiol Endocrinol Metab 292(6):E1724–E1739
- 129. Badman MK, Kennedy AR, Adams AC, Pissios P, Maratos-Flier E (2009) A very low carbohydrate ketogenic diet improves glucose tolerance in ob/ob mice independently of weight loss. Am J Physiol Endocrinol Metab 2215(5):1197–1204
- 130. Jornayvaz FR, Jurczak MJ, Lee HY, Birkenfeld AL, Frederick DW, Zhang D et al (2010) A high-fat, ketogenic diet causes hepatic insulin resistance in mice, despite increasing energy expenditure and preventing weight gain. Am J Physiol Endocrinol Metab 299(5):E808–E815
- 131. Garbow JR, Doherty JM, Schugar RC, Travers S, Weber ML, Wentz AE et al (2011) Hepatic steatosis, inflammation, and ER stress in mice maintained long term on a very lowcarbohydrate ketogenic diet. Am J Physiol Gastrointest Liver Physiol 300(6):956–967
- 132. Yancy WS Jr, Foy M, Chalecki AM, Vernon MC, Westman EC (2005) A lowcarbohydrate, ketogenic diet to treat type 2 diabetes. Nutr Metab (Lond) 2:34. https://doi. org/10.1186/1743-7075-2-34
- Rustad JK, Musselman DL, Nemeroff CB (2011) The relationship of depression and diabetes: pathophysiological and treatment implications. Psychoneuroendocrinology 36(9):1276–1286
- 134. Gordon DJ, Probstfield JL, Garrison RJ, Neaton JD, Castelli WP, Knoke JD et al (1989) High-density lipoprotein cholesterol and cardiovascular disease. Four prospective American studies. Circulation 79(1):8–15
- 135. Dashti HM, Mathew TC, Hussein T, Asfar SK, Behbahani A, Khoursheed MA et al (2004) Long-term effects of a ketogenic diet in obese patients. Exp Clin Cardiol 9(3):200–205
- 136. Zirlik S, Hauck T, Fuchs FS, Neurath MF, Konturek PC, Harsch IA (2011) Leptin, Obestatin and Apelin levels in patients with obstructive sleep apnoea syndrome. Med Sci Monit 17(3):CR159–164

- 137. Sumithran P, Prendergast LA, Delbridge E, Purcell K, Shulkes A, Kriketos A et al (2013) Ketosis and appetite-mediating nutrients and hormones after weight loss. Eur J Clin Nutr 67(7):759–764
- 139. https://www.alz.co.uk/research/WorldAlzheimerReport2015.pdf
- 140. Cuyvers E, Sleegers K (2016) Genetic variations underlying Alzheimer's disease: evidence from genome-wide association studies and beyond. Lancet Neurol 15(8):857–868
- 141. Cummings JL (2004) Alzheimer's disease. N Engl J Med 351(1):56-67
- 142. Van der Auwera I, Wera S, Van Leuven F, Henderson ST (2005) A ketogenic diet reduces amyloid beta 40 and 42 in a mouse model of Alzheimer's disease. Nutr Metab (Lond) 2:28. https://doi.org/10.1186/1743-7075-2-28
- Coyle JT, Price DL, DeLong MR (1983) Alzheimer's disease: a disorder of cortical cholinergic innervation. Science 219(4589):1184–1190
- 144. Birks J (2006) Cholinesterase inhibitors for Alzheimer's disease. Cochrane Database Syst Rev:CD005593. https://doi.org/10.1002/14651858.CD005593
- 145. Arvanitakis Z, Wilson RS, Bienias JL, Evans DA, Bennett DA (2004) Diabetes mellitus and risk of Alzheimer disease and decline in cognitive function. Arch Neurol 61(5):661–666
- 146. Ferreira LSS, Fernandes CS, Vieira MNN, De Felice FG (2018) Insulin resistance in Alzheimer's disease. Front Neurosci 12:830. https://doi.org/10.3389/fnins.2018.00830
- 147. Chesneau V, Vekrellis K, Rosner MR, Selkoe DJ (2000) Purified recombinant insulindegrading enzyme degrades amyloid β-protein but does not promote its oligomerization. Biochem J 351(Pt 2):509–516
- 148. Beckett TL, Studzinski CM, Keller JN, Paul Murphy M, Niedowicz DM (2013) A ketogenic diet improves motor performance but does not affect beta-amyloid levels in a mouse model of Alzheimer's disease. Brain Res 1505:61–67
- 149. Brownlow ML, Benner L, D'Agostino D, Gordon MN, Morgan D (2013) Ketogenic diet improves motor performance but not cognition in two mouse models of Alzheimer's pathology. PLoS One 8:e75713. https://doi.org/10.1371/journal.pone.0075713
- 150. Kashiwaya Y, Bergman C, Lee JH, Wan R, King MT, Mughal MR et al (2013) A ketone ester diet exhibits anxiolytic and cognition-sparing properties, and lessens amyloid and tau pathologies in a mouse model of Alzheimer's disease. Neurobiol Aging 34(6):1530–1539
- 151. Pawlosky RJ, Kemper MF, Kashiwaya Y, King MT, Mattson MP, Veech RL (2017) Effects of a dietary ketone ester on hippocampal glycolytic and tricarboxylic acid cycle intermediates and amino acids in a 3xTgAD mouse model of Alzheimer's disease. J Neurochem 141(2):195–207
- 152. Yin JX, Maalouf M, Han P, Zhao M, Gao M, Dharshaun T et al (2016) Ketones block amyloid entry and improve cognition in an Alzheimer's model. Neurobiol Aging 39:25–37
- 153. Morris MC, Evans DA, Bienias JL, Tangney CC, Bennett DA, Aggarwal N et al (2003) Dietary fats and the risk of incident Alzheimer disease. Arch Neurol 60(2):94–200
- 154. Krikorian R, Shidler MD, Dangelo K, Couch SC, Benoit SC, Clegg DJ (2012) Dietary ketosis enhances memory in mild cognitive impairment. Neurobiol Aging 33(2):425.e19-27. https:// doi.org/10.1016/j.neurobiolaging.2010.10.006
- 155. Witte AV, Fobker M, Gellner R, Knecht S, Flöel A (2009) Caloric restriction improves memory in elderly humans. Proc Natl Acad Sci U S A 106(4):1255–1260
- 156. Newport MT, VanItallie TB, Kashiwaya Y, King MT, Veech RL (2015) A new way to produce hyperketonemia: use of ketone ester in a case of Alzheimer's disease. Alzheimers Dement 11(1):99–103
- 157. Morrill SJ, Gibas KJ (2019) Ketogenic diet rescues cognition in ApoE4+ patient with mild Alzheimer's disease: a case study. Diabetes Metab Syndr Clin Res Rev 13:1187–1191
- Dzamko N, Geczy CL, Halliday GM (2015) Inflammation is genetically implicated in Parkinson's disease. Neuroscience 302:89–102
- Kalia LV, Lang AE (2016) Parkinson disease in 2015: evolving basic, pathological and clinical concepts in PD. Nat Rev Neurol 12(2):65–66
- Zhu B, Caldwell M, Song B (2016) Development of stem cell-based therapies for Parkinson's disease. Int J Neurosci 126(11):955–962
- 161. de Lau LM, Bornebroek M, Witteman JC, Hofman A, Koudstaal PJ, Breteler MM (2005) Dietary fatty acids and the risk of Parkinson disease: the Rotterdam study. Neurology 64(12):2040–2045
- 162. http://www.parkinsonsnsw.org.au/wp-content/uploads/2015/03/AE-Report\_2011.pdf
- 163. Murakami K, Miyake Y, Sasaki S, Tanaka K, Fukushima W, Kiyohara C et al (2010) Dietary glycemic index is inversely associated with the risk of Parkinson's disease: a case-control study in Japan. Nutrition 26(5):515–521
- 164. McKeown MJ, Peavy GM (2015) Biomarkers in Parkinson disease: It's time to combine. Neurology 84(24):2392–2393
- 165. Yang X, Cheng B (2010) Neuroprotective and anti-inflammatory activities of ketogenic diet on MPTP-induced neurotoxicity. J Mol Neurosci 42(2):145–153
- 166. Shaafi S, Najmi S, Aliasgharpour H, Mahmoudi J, Sadigh-Etemad S, Farhoudi M et al (2016) The efficacy of the ketogenic diet on motor functions in Parkinson's disease: a rat model. Iran J Neurol 15(2):63–69
- 167. Tieu K, Perier C, Caspersen C, Teismann P, Wu DC, Yan SD et al (2003) D-β-Hydroxybutyrate rescues mitochondrial respiration and mitigates features of Parkinson disease. J Clin Invest 112(6):892–901
- 168. Cheng B, Yang X, An L, Gao B, Liu X, Liu S (2009) Ketogenic diet protects dopaminergic neurons against 6-OHDA neurotoxicity via up-regulating glutathione in a rat model of Parkinson's disease. Brain Res 1286:25–31
- 169. Elbarbry F, Nguyen V, Mirka A, Zwickey H, Rosenbaum R (2019) A new validated HPLC method for the determination of levodopa: application to study the impact of ketogenic diet on the pharmacokinetics of levodopa in Parkinson's participants. Biomed Chromatogr 33(1):e4382. https://doi.org/10.1002/bmc.4382
- 170. Neal EG, Chaffe H, Schwartz RH, Lawson MS, Edwards N, Fitzsimmons G et al (2008) The ketogenic diet for the treatment of childhood epilepsy: a randomised controlled trial. Lancet Neurol 7(6):500–506
- 171. Kang HC, Chung DE, Kim DW, Kim HD (2004) Early- and late-onset complications of the ketogenic diet for intractable epilepsy. Epilepsia 45(9):1116–1123
- 172. Wheless JW (2001) The ketogenic diet: an effective medical therapy with side effects. J Child Neurol 6(9):633–635
- 173. Pietzak MM, Thomas DW (2000) Pancreatitis in childhood. Pediatr Rev 21(12):406-412
- 174. Wheless JW (2009) The Ketogenic diet: an effective medical therapy with side effects. J Child Neurol 16(9):633–635
- 175. Rogovik AL, Goldman RD (2010) Ketogenic diet for treatment of epilepsy. Can Fam Physician 56(6):540–542
- 176. Nordli DR Jr, Kuroda MM, Carroll J, Koenigsberger DY, Hirsch LJ, Bruner HJ et al (2001) Experience with the ketogenic diet in infants. Pediatrics 108(1):129–133
- 177. Herzberg GZ, Fivush BA, Kinsman SL, Gearhart JP (1990) Urolithiasis associated with the ketogenic diet. J Pediatr 117(5):743–745
- 178. Furth SL, Casey JC, Pyzik PL, Neu AM, Docimo SG, Vining EP et al (2000) Risk factors for urolithiasis in children on the ketogenic diet. Pediatr Nephrol 15(1–2):125–128
- 179. Kinsman SL, Vining EP, Quaskey SA, Mellits D, Freeman JM (1992) Efficacy of the ketogenic diet for intractable seizure disorders: review of 58 cases. Epilepsia 33(6):1132–1136
- Freeman JM, Vining EP, Pillas DJ, Pyzik PL, Casey JC, Kelly LM (1998) The efficacy of the ketogenic diet-1998: a prospective evaluation of intervention in 150 children. Pediatrics 102(6):1358–1363
- 181. Stewart WA, Gordon K, Camfield P (2001) Acute pancreatitis causing death in a child on the ketogenic diet. J Child Neurol 16(9):682. https://doi.org/10.1177/088307380101600910

- 182. Güntner AT, Kompalla JF, Landis H, Theodore SJ, Geidl B, Sievi NA et al (2018) Guiding ketogenic diet with breath acetone sensors. Sensors (Basel) 18(11). pii: E3655. https://doi. org/10.3390/s18113655
- McDonald TJW, Cervenka MC (2018) Ketogenic diets for adult neurological disorders. Neurotherapeutics 15(4):1018–1031
- 184. Park EG, Lee J, Lee J (2019) The ketogenic diet for super-refractory status epilepticus patients in intensive care units. Brain Dev. pii: S0387–7604(18)30173–6. https://doi.org/10.1016/j. braindev.2018.12.007. [Epub ahead of print]

# Chapter 6 Coenzyme Q10 Supplementation in Fibrosis and Aging



Iain P. Hargreaves and David Mantle

# 1 Introduction

Coenzyme Q10 (CoQ10) is a naturally occurring vitamin-like substance, first characterised in 1957 by Professor Fred Crane at the University of Wisconsin, USA [1]. CoQ10 is a lipophilic molecule which is a member of a group of compounds known as ubiquinones because of their ubiquitous distribution in nature, being found in animal, plants and microorganisms [1]. Ubiquinones are composed of a benzoquinone nucleus and an isoprenoid side chain, which varies in length among the different ubiquinone species with CoQ10 having a side chain composed of ten isoprenoid subunits (Fig. 6.1) [1]. CoQ10 is the predominant ubiquinone species found in human tissues [1]. CoQ10 plays an essential role in cellular energy generation within the mitochondrial respiratory chain (MRC). The role of CoQ10 is of particular importance in tissues with a high energy requirement, such as cardiac muscle. In addition to its role in cellular energy generation, CoQ10 also serves as an important lipid soluble antioxidant and antiinflammatory agent within the body [1-3]. The objective of this article is to review the potential role of CoQ10 as a biomarker of aging, specifically with regards to the prevention of tissue fibrosis in the heart, which has been implicated in age-related dysfunction of this organ.

I. P. Hargreaves  $(\boxtimes)$ 

School of Pharmacy and Biomolecular Sciences, Liverpool John Moores, Liverpool, UK e-mail: i.hargreaves@ucl.ac.uk

D. Mantle Pharma Nord (UK) Ltd, Morpeth, Newcastle, UK

© Springer Nature Switzerland AG 2019

P. C. Guest (ed.), *Reviews on Biomarker Studies in Aging and Anti-Aging Research*, Advances in Experimental Medicine and Biology 1178, https://doi.org/10.1007/978-3-030-25650-0\_6



**Fig. 6.2** Diagram of the mitochondrial respiratory chain (MRC) and complex V illustrating proton (H+) movement during oxidative phosphorylation. Q: Coenzyme Q<sub>10</sub>. Cyt C: Cytochrome c

## 2 Functions of CoQ10

CoQ10 serves as an electron carrier within the MRC (Fig. 6.2), where it is involved in the transfer of electrons derived from complex I (NADH:ubiquinone reductase) and complex II (succinate dehydrogenase) to complex III (ubiquinol cytochrome c reductase), allowing a continuous passage of electrons within the MRC, which is required in the process of oxidative phosphorylation and the concomitant production of ATP, the energy currency of the cell [1]. Tissues with a high energy requirement, especially the heart and skeletal muscles, contain higher numbers of mitochondria within their cells and are particularly reliant on maintaining adequate tissue CoQ10 levels for normal functioning [2]. Thus, the heart and skeletal muscles typically contain about 1000 mg of CoQ10, out of a total body pool of 1500– 2000 mg. CoQ10 occurs in cells in two closely related forms, oxidised (ubiquinone) and fully reduced (ubiquinol) [1]. The interconversion between these two forms is essential for the normal functioning of CoQ10, and this interconversion is principally mediated by the selenium-containing enzyme, thioredoxin reductase [4].

CoQ10 is also important within the body serving as a major fat-soluble antioxidant, protecting cell membranes (particularly those of the mitochondria) and circulatory lipoproteins from the damaging effects of free radical species [e.g., reactive oxygen species (ROS)] [1]. CoQ10 is the only lipid soluble antioxidant produced within the body [5]. The antioxidant function of CoQ10 is attributed to its ubiquinol form, which must be constantly regenerated from CoQ10 to maintain its antioxidant capacity [5]. In addition to directly preventing lipid peroxidation, ubiquinol is able to enhance the cellular antioxidant capacity by maintaining the antioxidants, vitamin C and E, in their active fully reduced forms [6]. Most recently, a gene expression profiling study showed that CoQ10 influences the expression and regulation of hundreds of genes in the human body [7]. In cell culture, CoQ10 has been shown to reduce the activity of inflammatory markers, suggesting CoQ10 may have anti-inflammatory action via gene expression modification, most notably in elderly individuals [8, 9].

#### 3 Synthesis and Deficiency of CoQ10

Although some CoO10 is obtained from the diet, most of the daily requirement is synthesized within the body, particularly by the liver, although all organs of the body have the capacity to synthesize CoQ10 [1]. Based on a total body pool of 2 g and an average tissue turnover time of 4 days, the human requirement for CoQ10 has been estimated at approximately, 500 mg/day [10]. A relatively small proportion of this daily requirement is obtained from the normal diet, typically up to 5 mg/day [10]. The synthesis of CoQ10 is a complex, multistage process requiring a number of amino acids, vitamins and trace element precursors and cofactors, and a deficiency in any of these can adversely affect the normal production of CoO10 [1]. It is noteworthy that CoQ10 shares a common biosynthetic pathway with cholesterol [1]. As people age, it has been reported that the capacity of the body to synthesize CoQ10 decreases [5]. Optimal production of CoQ10 occurs around the mid-twenties, with a continual decrease thereafter to approximately 50% at age 70. CoQ10 levels can also be depleted by intense exercise, certain types of prescription medicines, and illnesses [1, 5]. Dietary supplementation with coenzyme CoQ10 therefore provides a mechanism to maintain adequate levels of CoO10 within the body.

In humans, at least 13 genes are thought to be involved in the biosynthesis of CoQ10, and mutations in 10 of these genes have been associated with primary CoQ10 deficiency, a condition which results from a genetic defect in the CoQ10 biosynthetic pathway [11–13]. The first cases of primary CoQ10 deficiency were reported in 1989 by Ogasahara and colleagues [14]. The two patients were sisters born to unrelated parents and they presented with recurrent rhabdomyolysis, which was associated with developmental delay and seizures [14]. Subsequently, a number of other patients have been described with primary CoQ10 deficiency, which appears to have a heterogeneous clinical presentation. This can be divided into five distinct clinical phenotypes: (1) encephalomyopathic (as in the case of the two siblings described by Ogasahara et al. [14]; (2) cerebral ataxic; (3) infantile multisystem disease; (4) steroid resistant nephrotic syndrome; and (5) isolated myopathy [15].

In most cases of suspected primary CoQ10 deficiency a family history suggests an autosomal mode of inheritance, however, genetic diagnosis is complicated by the fact that the CoQ10 biosynthetic pathway has not been fully elucidated in humans, and diagnosis generally relies on the determination of the endogenous CoQ10 status of the patient [16]. Interestingly, a CoQ10 deficiency can also result from a disorder not associated with a genetic defect in the CoQ10 biosynthetic pathway [17]. This type of CoQ10 deficiency is known as a secondary CoQ10 deficiency which is thought to be more prevalent than a primary deficiency, and has been associated with diseases such as MRC disorders, cardiovascular disease, Parkinson's disease and sepsis [1, 17]. The cause of secondary CoQ10 deficiency in disease has not yet been fully elucidated and a number of theories have been suggested to account for this loss of CoQ10 including increased oxidative stress-induced catabolism and/or ROS induced inhibition of the CoQ10 biosynthetic pathway enzymes [17].

# 4 Laboratory Measurement of CoQ10 Status

Quantification of CoQ10 is not usually included in routine biochemical analysis of blood by hospital pathology laboratories. The most common laboratory procedures used to assess CoO10 status are based on high-pressure liquid chromatography (HPLC) with either ultraviolet (HPLC-UV) or electrochemical detection (HPLC-ED) [16]. CoQ10 levels are usually determined in plasma isolated from blood, with normal plasma levels typically in the range of  $0.5-1.7 \mu M$  [16]. In view of the fact that CoQ10 levels are dependent upon the circulatory lipoprotein status (lipoproteins are the major carriers of CoO10 in the circulation), it has been suggested that plasma CoQ10 levels should be expressed as a ratio to total plasma cholesterol status to take into account the lipoprotein status of the blood [1]. Furthermore, dietary intake has also been reported to influence plasma CoQ10 status, contributing up to 25% in some cases of the total amount of this isoprenoid in the circulation [1, 10]. Accordingly, it has been suggested that plasma may not be an appropriate surrogate for the assessment of endogenous CoQ10 status, and skeletal muscle is generally considered as the tissue of choice for this determination [16]. In addition to direct assessment in skeletal muscle, decreased activity of the linked MRC enzymes, NADH:cytochrome c reductase (complex I + III) and/or succinate:cytochrome c reductase (complex II + III) may also indicate evidence of a CoQ10 deficiency since the activity of these linked enzymes is dependent upon the endogenous CoQ10 status of the tissue [16]. However, in view of the invasive nature of a muscle biopsy, blood mononuclear cells or fibroblast skin cells have also been suggested as alternative surrogates to assess endogenous CoQ10 status [16]. With regards to the neurological dysfunction associated with CoQ10 deficiency, the potential to assess cerebral CoQ10 status would be of diagnostic value. However, since it is not possible to directly determine the CoQ10 status of brain tissue, cerebral spinal fluid (CSF) has been used as an alternative surrogate for this evaluation [16]. Tentative references ranges for CSF CoQ10 status have now been established at 1.18-4.91 nM, although highly sensitive mass spectrometry techniques are required for this determination, in view of the low levels of CoQ10 detected in this matrix [18]. Furthermore, considering the increasing number of reported cases of steroid resistant nephrotic syndrome associated with CoQ10 deficiency, Yubero et al. [19] have

developed a reliable method for the determination of the CoQ10 status of kidney epithelial cells isolated from urine and it is hoped that such an approach will prevent the need for an invasive needle biopsy to assess the CoQ10 status of this organ.

#### 5 Aging and Fibrosis

Fibrosis is the formation of fibrous connective tissue, particularly collagen, in response to an injury. Fibrosis is an adaptive response to tissue injury, and is an essential part of the normal processes of wound healing and tissue repair [20]. In younger individuals, such fibrous material is replaced over time by new functional tissue. However, in older people, tissue scarring tends to persist and may continue to form and accumulate. Uncontrolled continuation of fibrosis can result in scarring and the permanent remodelling of the organ which can result in an eventual loss of function [21]. Progressive fibrosis is a hallmark of the aging process and has been implicated in the pathogenesis of diseases of the heart, lungs, liver, kidneys and bone marrow [22]. It has been estimated that organ fibrosis may be the root cause of over 800,000 deaths per year, approximately 50% of the total number of human deaths [20]. However, at present there is no effective treatment for fibrosis or information available for designing appropriate therapeutic strategies [21, 22].

## 5.1 Fibrosis Mechanism

Fibrous connective tissue such as collagen is produced by specific types of cells called fibroblasts, which are present in most organs. Under normal circumstances, fibroblasts produce collagen in a controlled manner, to provide a "scaffold" for the structural support of the various types of functional cells in the different organs. Following tissue injury, fibroblasts are activated via inflammatory cytokines to produce collagen as part of the scar formation/healing process [23]. It is well established that fibrosis is linked to inflammation, and there is evidence that even low-grade but persistent inflammation is sufficient to promote, for example, cardio-vascular fibrosis where an increased amount of collagen can contribute to the increased stiffness of both arteries and the heart wall [21].

# 5.2 Preclinical Studies on CoQ10 Supplementation and Fibrosis

A number of studies have demonstrated the beneficial effects of CoQ10 supplementation on fibrosis related parameters (including oxidative stress/inflammation) in various animal models of fibrosis. These include dimethylnitrosamine-induced liver fibrosis in mice [24]. In this model, CoQ10 treatment was found to block the activation of the stella cells (which is a central event in liver fibrosis) via activation of the nuclear factor, erythroid 2–like (Nrf2) which also resulted in an increased expression of the enzymes involved in the synthesis of the antioxidant, glutathione (GSH) in the hepatic cells. In sub-optimal nutrition-induced liver fibrosis in rats, CoQ10 supplementation was found to reverse fibrosis via suppression of the expression of cytokine transforming growth factor beta 1 ( $Tgf\beta 1$ ) via a mechanism involving the upregulation of Nrf2-antioxidant response element (ARE)-associated genes [25]. In a study by Chen et al. which investigated the effect of CoQ10 supplementation on doxorubicin induced cardiac fibrosis in rats, it was found that CoQ10 was able to ameliorate the fibrosis by decreasing the expression of both  $Tgf\beta 1$  and connective tissue growth factor (CTGF) [26].

In a study which assessed isoprenaline induced cardiac fibrosis in rats, CoQ10 was found to inhibit fibrosis in the heart as well as the kidney via its ability to reduce oxidative stress and prevent inflammatory cell infiltration in the tissues. However, the details of the mechanism of action of CoQ10 were not elucidated in the study [27]. In a recent study, Xue et al. [28] reported that CoQ10 supplementation was able to suppress the activation of mouse pancreatic stellate cells (PCSs), which has been associated in the development of pancreatic fibrosis, by its ability to decrease intracellular ROS levels and induce the mammalian target of rapamycin (mTOR) cell signalling pathway. CoO10 supplementation was also found to ameliorate methotrexate induced lung and liver fibrosis in rats by the attenuation of hepatic oxidative stress which, contrary to the previous study [28], was found to result in a down-regulation of mTOR expression and increased evidence of autophagy [29]. The reasons for the disparity between the two studies are uncertain, but may represent a species or organ specific mechanism of fibrosis induction. Interestingly, in the aforementioned studies, CoQ10 appears to attenuate fibrosis by its ability either directly or via activation of cell signalling pathways to ameliorate cellular oxidative stress.

#### 5.3 Cardiovascular Fibrosis

In the normal heart, contractile cells (myocytes) occupy approximately 75% of the tissue volume, with the remainder comprising other cell types, which are predominatly fibroblasts [21]. Collagen secreted by fibroblasts forms a supportive scaffold for the myocytes, as well as providing a means of transmission for myocyte generated force within the heart [21]. With regard to the normal ageing process, postmortem analysis of cardiac tissue from human subjects without cardiovascular disease found that the collagen content increased by approximately 50% between the third and seventh decades of life [30].

Unlike other organs, the heart has a limited regenerative capacity following injury, with the repair process involving the removal of necrotic cells, followed by fibrotic scar tissue replacement [21]. Two common aspects of cardiovascular disease in which fibrosis plays a major role are myocardial infarction and heart

failure [20]. Myocardial infarction is caused by the blockage of the coronary arteries, which results in the death of the cardiomyocyte cells responsible for contraction. This in turn causes fibroblasts to produce excess collagen, which helps to preserve the structural integrity of the tissue, but which stiffens cardiac muscle, impairing heart contractility and relaxation, ultimately leading to impaired cardiac function and eventual heart failure [31]. Excessive production of collagen by cardiac fibroblasts can also cause fibrotic thickening of the heart valves, again ultimately resulting in heart failure [30, 31].

### 6 CoQ10 and the KISEL-10 Study

In the KISEL-10 randomised controlled clinical trial, normal elderly individuals were supplemented with coenzyme Q10 (200 mg/day) and selenium (200 µg/day) for 4 years. The supplemented individuals showed a 53% reduction in cardiovascular related mortality risk compared to placebo [32]. Supplemental selenium was included in this study as this trace metal functions as a co-factor for the enzyme thioredoxin reductase, which is responsible for the interconversion of the ubiquinone and ubiquinol forms of CoQ10. In addition, there is evidence suggesting that a selenium deficiency may be prevalent within the general populations of a number of European countries, including Sweden [37]. Furthermore, selenium is also required for the biological activity of the antioxidant enzyme, glutathione peroxidase which together with ubiquinol contributes to the cellular antioxidant defense [38]. A more recent study by Alehagen et al. [33] using data derived from the KISEL-10 study has now identified significant reductions in the blood levels for a range of biochemical markers of fibrosis, including cathepsin S, endostatin, galectin-3, growth differentiation factor 15 (GDF-15), matrix metalloproteinase-1 (MMP1), MMP9 and tissue inhibitor of metalloproteinase 1 (TIMP1). It is therefore concluded that the improvement in heart function and reduced risk of cardiovascular mortality following supplementation with coenzyme Q10 and selenium results from a reduction in cardiovascular fibrosis. It is of note that the reduction in systemic fibrosis markers may also help to prevent fibrotic degeneration of other tissues such as the lungs and liver [33].

#### 7 Safety of Coenzyme Q10 Supplementation

The safety of CoQ10 has been investigated by Hidaka et al. [34] and Hosoe et al. (2007) [35]. These authors reported that CoQ10 is generally well tolerated, with no serious adverse effects being detected in long term use. Rarely, some individuals may experience mild gastrointestinal disturbances, although this is not dose related. There are no known toxic side effects, and CoQ10 cannot be overdosed. CoQ10 is well tolerated in healthy adults at an intake of 900 mg/day, and in rats at a dose of

up to 1200 mg/kg/day. In addition, Yamaguchi et al. [36] reported that CoQ10 had no genotoxic activity. The safety of CoQ10 has been confirmed in more than 200 randomised controlled trials, over a wide range of disorders including cardiovascular disease, Parkinson's diseases and mitochondrial disease [1].

#### 8 Requirements for Coenzyme Q10 Supplementation

CoQ10 is a lipid soluble substance absorbed from the digestive tract in the same manner as other dietary fats. Because of its hydrophobicity and large molecular weight, absorption of CoQ10 is in general slow and somewhat limited. Oil based formulations show the highest bioavailability. Absorption of CoQ10 is non-linear, with increasing doses absorbed to a decreasing degree. CoQ10 is therefore best administered in split doses (typically 100 mg two or three times daily).

When first manufactured, CoQ10 is produced in a crystalline form which cannot be absorbed from the digestive tract. In CoQ10 supplements, this crystalline form must be further treated to break it down into individual molecules to enable absorption and, most importantly, crystals should not re-form within the capsule. Supplement manufacturers vary in their ability to fulfil these requirements, and previous clinical trial studies reporting lack of benefit in a variety of disorders may have failed because of insufficient dosage and/or lack of bioavailability of the particular supplement used.

#### 9 Conclusions

Oral supplementation with CoQ10 and selenium provides a means of correcting dietary deficiencies to which older individuals may be subject. In addition, supplementation significantly reduces the levels of fibrotic markers in elderly individuals, thereby reducing the extent of cardiovascular fibrosis to which older individuals are subjected, and improving cardiovascular function and reducing risk of cardiovascular associated mortality. Although the factors responsible for the anti-fibrotic action of CoQ10 have yet to be fully elucidated, its antioxidant and anti-inflammatory functions are thought to be major contributors to its clinical efficacy. Further studies may lead to its use as an intervention to promote healthy aging. This is particularly pertinent to the treatment of idiopathic pulmonary fibrosis (IPF), a progressive and ultimately fatal lung disorder disproportionately affecting the elderly, with two-thirds of patients who present with IPF being older than 60 [39]. The management of IPF via prescription-type drugs has proved to be particularly refractory, and alternative therapeutic strategies for the treatment of this disorder are clearly warranted and CoQ10 may be an appropriate consideration.

# References

- 1. Hargreaves IP (2003) Ubiquinone: cholesterol's reclusive cousin. Ann Clin Biochem 40(Pt3):207–218
- 2. Crane FL (2001) Biochemical functions of coenzyme Q10. J Am Coll Nutr 20(6):591-598
- Fan L, Feng Y, Chen GC, Qin LQ, Fu CL, Chen LH (2003) Effects of coenzyme Q10 supplementation on inflammatory markers: a systematic review and meta-analysis of randomized controlled trials. Pharmacol Res 119:128–136
- 4. Xia L, Nordman T, Olsson JM, Damdimopoulos A, Björkhem-Bergman L, Nalvarte I et al (2003) The mammalian cytosolic selenoenzyme thioredoxin reductase reduces ubiquinone. A novel mechanism for defense against oxidative stress. J Biol Chem 278(4):2141–2146
- 5. Mantle D (2015) Coenzyme Q10 and cardiovascular disease: an overview. Br J Cardiol 22(4):1–7
- 6. Navas P, Villalba JM, de Cabo R (2007) The importance of plasma membrane coenzyme Q in aging and stress responses. Mitochondrion 7(Suppl):S34–S40
- Gutierrez-Marisca FM, Yubero-Serrano EM, Villalba JM, Lopez-Miranda J (2018) Coenzyme Q10: from bench to clinic in aging diseases, a translational review. Crit Rev Food Sci Nutr 16(1):1–18
- Schmelzer C, Lindner I, Rimbach G, Niklowitz P, Menke T, Döring F (2008) Functions of coenzyme Q10 in inflammation and gene expression. Biofactors 32(1–4):179–183
- Yubero-Serrano EM, Gonzalez-Guardia L, Rangel-Zuñiga O, Delgado-Lista J, Gutierrez-Mariscal FM, Perez-Martinez P et al (2012) Mediterranean diet supplemented with coenzyme Q10 modifies the expression of proinflammatory and endoplasmic reticulum stress-related genes in elderly men and women. J Gerontol A Biol Sci Med Sci 67(1):3–10
- Weber C, Bysted A, Hilmer G (1997) The coenzyme Q10 content of the average Danish diet. Int J Vitam Nutr Res 67(2):123–129
- Doimo M, Desbats MA, Cerqua C, Cassina M, Trevisson E, Salviati L (2014) Genetics of coenzyme q10 deficiency. Mol Syndromol 5(3–4):156–162
- 12. Awad AM, Bradley MC, Fernández-Del-Río L, Nag A, Tsui HS, Clarke CF (2018) Coenzyme Q<sub>10</sub> deficiencies: pathways in yeast and humans. Essays Biochem 62(3):361–376
- Yubero D, Montero R, Santos-Ocaña C, Salviati L, Navas P, Artuch R et al (2018) Molecular diagnosis of coenzyme Q<sub>10</sub> deficiency: an update. Expert Rev Mol Diagn 18(6):491–498
- Ogasahara S, Engel AG, Frens D, Mack D (1989) Muscle coenzyme Q deficiency in familial mitochondrial encephalomyopathy. Proc Natl Acad Sci U S A 86(7):2379–2382
- Emmanuele V, López LC, Berardo A, Naini A, Tadesse S, Wen B et al (2012) Heterogeneity of coenzyme Q10 deficiency: patient study and literature review. Arch Neurol 69(8):978–983
- Yubero D, Montero R, Artuch R, Land JM, Heales SJ, Hargreaves IP (2014) Biochemical diagnosis of coenzyme Q10 deficiency. Mol Syndromol 5(3–4):147–155
- Neergheen V, Hargreaves IP (2018) Secondary coenzyme Q10 deficiency: causes and consequence. In: Grigoryeva S (ed) Coenzyme Q10- uses, health effects and role in disease. Nova Science Publishers, New York, pp 89–111. ISBN-10: 1536132845
- 18. Duberley KE, Hargreaves IP, Chaiwatanasirikul KA, Heales SJ, Land JM, Rahman S et al (2013) Coenzyme Q10 quantification in muscle, fibroblasts and cerebrospinal fluid by liquid chromatography/tandem mass spectrometry using a novel deuterated internal standard. Rapid Commun Mass Spectrom 27(9):924–930
- Yubero D, Montero R, Ramos M, Neergheen V, Navas P, Artuch R et al (2015) Determination of urinary coenzyme Q10 by HPLC with electrochemical detection: reference values for a paediatric population. Biofactors 41(6):424–430
- Murtha LA, Schuliga MJ, Mabotuwana NS, Hardy SA, Waters DW, Burgess JK et al (2017) The processes and mechanisms of cardiac and pulmonary fibrosis. Front Physiol 8:777. https:// doi.org/10.3389/fphys.2017.00777
- 21. Biernacka A, Frangogiannis NG (2011) Aging and cardiac fibrosis. Aging Dis 2(2):158-173

- 22. Jiang S, Li T, Yang Z, Yi W, Di S, Sun Y et al (2017) AMPK orchestrates an elaborate cascade protecting tissue from fibrosis and aging. Ageing Res Rev 38:18–27
- Nanthakumar CB, Hatley RJ, Lemma S, Gauldie J, Marshall RP, Macdonald SJ (2015) Dissecting fibrosis: therapeutic insights from the small molecule toolbox. Nat Rev Drug Discov 14(10):693–720
- 24. Choi HK, Pokharel YR, Lim SC, Han HK, Ryu CS, Kim SK et al (2009) Inhibition of liver fibrosis by solubilized coenzyme Q10: role of Nrf2 activation in inhibiting transforming growth factor-beta1 expression. Toxicol Appl Pharmacol 240(3):377–384
- 25. Tarry-Adkins JL, Fernandez-Twinn DS, Hargreaves IP, Neergheen V, Aiken CE, Martin-Gronert MS et al (2016) Coenzyme Q10 prevents hepatic fibrosis, inflammation, and oxidative stress in a male rat model of poor maternal nutrition and accelerated postnatal growth. Am J Clin Nutr 103(2):579–588
- 26. Chen PY, Hou CW, Shibu MA, Day CH, Pai P, Liu ZR et al (2017) Protective effect of Co-enzyme Q10 On doxorubicin-induced cardiomyopathy of rat hearts. Environ Toxicol 32(2):679–689
- 27. Ulla A, Mohamed MK, Sikder B, Rahman AT, Sumi FA, Hossain M et al (2017) Coenzyme Q10 prevents oxidative stress and fibrosis in isoprenaline induced cardiac remodeling in aged rats. BMC Pharmacol Toxicol 18(1):1–12
- Xue R, Wang J, Yang L, Liu X, Gao Y, Pang Y et al (2019) Coenzyme Q10 ameliorates pancreatic fibrosis via the ROS triggered mTOR signalling pathway. Oxidative Med Cell Longev 2019:8039694. https://doi.org/10.1155/2019/8039694
- Mohamed D, Khairy E, Tawfek SS, Habib EK, Fetouh MA (2019) CoQ10 attenuates lung and liver fibrosis via modulation of autophagy in methotrexate treated rat. Biomed Pharmacother 109:892–901
- Gazoti Debessa CR, Mesiano Maifrino LB, Rodrigues de Souza R (2001) Age related changes in collagen network of the human heart. Mech Ageing Dev 122(10):1049–1058
- Konstam MA, Kramer DG, Patel AR, Maron MS, Udelson JE (2011) Left ventricular remodelling in heart failure: current concepts in clinical significance and assessment. JACC Cardivasc Imaging 4(1):98–108
- 32. Alehagen U, Johansson P, Björnstedt M, Rosén A, Dahlström U (2013) Cardiovascular mortality and N-terminal-proBNP reduced after combined selenium and coenzyme Q10 supplementation: a 5-year prospective randomized double-blind placebo-controlled trial among elderly Swedish citizens. Int J Cardiol 167(5):1860–1866
- 33. Alehagen U, Aaseth J, Alexander J, Svensson E, Johansson P, Larsson A (2018) Less fibrosis in elderly subjects supplemented with selenium and CoQ10- a mechanism behind reduced cardiovascular mortality. Biofactors 44(2):137–147
- Hidaka T, Fujii K, Funahashi I, Fukutomi N, Hosoe K (2008) Safety assessment of coenzyme Q10 (CoQ10). Biofactors 32(1–4):199–208
- 35. Hosoe K, Kitano M, Kishida H, Kubo H, Fujii K, Kitahara M (2007) Study on safety and bioavailability of ubiquinol (Kaneka QH) after single and 4-week multiple oral administration to healthy volunteers. Regul Toxicol Pharmacol 47(1):19–28
- 36. Yamaguchi N, Nakamura K, Oguma Y, Fujiwara S, Takabe M, Sono A et al (2009) Genotoxicity studies of ubidecarenone (coenzyme Q10) manufactured by bacteria fermentation. J Toxicol Sci 34(4):389–397
- 37. Mantle D, Hargreaves I (2019) Coenzyme Q10 and degenerative disorders affecting longevity: an overview. Antioxidants (Basel) 8(2). pii: E44. https://doi.org/10.3390/antiox8020044
- Stepien K, Heaton R, Rankin S, Murphy A, Bentley J, Sexton D (2017) Evidence of oxidative stress and secondary mitochondrial dysfunction in metabolic and non-metabolic disorders. J Clin Med 6(7). pii: E71. https://doi.org/10.3390/jcm6070071
- Hecker L (2018) Mechanisms and consequences of oxidative stress in lung disease: therapeutic implications for an aging population. Am J Phys Lung Cell Mol Phys 314(4):L642–L653

# Chapter 7 Dietary Restriction, Cardiovascular Aging and Age-Related Cardiovascular Diseases: A Review of the Evidence



Behnaz Abiri and Mohammadreza Vafa

#### 1 Introduction

Calorie restriction (CR) and alternate-day fasting (ADF) (or intermittent fasting) are two of the various forms of dietary restriction (DR) [1]. CR, explained as a decrease in calorie intake without malnutrition, has been demonstrated to increase life span, ameliorate many functional indexes, and decrease metabolic risk factors for chronic metabolic disorders in several mammalian species [2, 3]. Calorie restricted diets have included decreasing food intake to 60-85% of daily energy requirements [1]. In addition to noticeable life span prolongation, CR has significant impacts on agerelated physiological and pathophysiological changes [4–9]. CR is also expected to decrease cardiovascular morbidity and mortality by creating pleiotropic cardiovascular preservation [6–9]. Several investigations have demonstrated that CR significantly reduces oxidative damage in the aging heart [7]. The concentrations of 8-oxo-2'-deoxyguanosine were found to be lower in cardiac mitochondria acquired from rats receiving CR than in those from *ad libitum* (AL)-fed groups [7]. Recent studies demonstrated that CR makes an active defense response that enhances survival during stressful states [8, 10]. At the core of this response are the so-called longevity regulatory pathways, which consist of insulin/insulin-like growth factor 1, the mammalian target of rapamycin (mTOR), AMP-activated kinase (AMPK), and

B. Abiri

M. Vafa (🖂)

Department of Nutrition, Faculty of Paramedicine, Ahvaz Jundishapur University of Medical Sciences, Ahvaz, Iran

Department of Nutrition, School of Public Health, Iran University of Medical Sciences, Tehran, Iran

Pediatric Growth and Development Research Center, Institute of Endocrinology and Metabolism, Iran University of Medical Sciences, Tehran, Iran e-mail: vafa.m@iums.ac.ir

<sup>©</sup> Springer Nature Switzerland AG 2019

P. C. Guest (ed.), *Reviews on Biomarker Studies in Aging and Anti-Aging Research*, Advances in Experimental Medicine and Biology 1178, https://doi.org/10.1007/978-3-030-25650-0\_7

NAD-dependent deacetylases, which are also called sirtuins. These signaling pathways make a network with positive and negative feedback control [11, 12]. They also play a major role in the evolution of CR-induced cardiovascular health.

ADF is another dietary regimen that has also been investigated. ADF regimens commonly imply a "feast day" on which food is eaten ad libitum and this is changed with a "fast day" on which food is not consumed or is decreased. The feast and fast phases are typically 24 h each, but they may differ. An important point about the ADF approach is that total calorie intake is not to be restricted. Instead, the frequency of food intake is changed [1]. Findings from both animal and human studies demonstrate that ADF may reduce the risk of cardiovascular diseases (CVDs), while results from animal experiments propose a protective impact on cancer risk. In regards to diabetes prevention, animal models show a beneficial impact. Indirect evidence proposes that the two diets may share similar mechanisms of action. These findings propose that ADF may function by promotion of resistance to oxidative attack, which is a key characteristic of the stress resistance theory.

Because of the elevation cardiovascular risk factors with age and also due to the anti-aging influences of restricted diets, this paper reviews the evidence that associates DR, cardiovascular aging and age-related cardiovascular diseases, along with implications for strategies to prevent or retard age-related cardiovascular diseases in the elderly population.

#### 2 Methods

The data were obtained according to the results of original articles associated with CR, ADF, cardiovascular aging, age-related CVD, cardiovascular protection, agerelated cardiomyopathy and atherosclerosis. For this purpose, we mostly used the online PubMed database and Google scholar search engine, with the use of following keywords: "calorie restriction", "alternate-day fasting", "cardiovascular aging", "age-related cardiovascular diseases", "cardiac aging" and "aging cardiomyopathy". Then, we chose the pertinent free- access full texts and reviewed the appropriate articles. We contemplated suitable published articles in the English language with no restrictions regarding the dates of articles. Our review also incorporated animal and human studies. In addition, we searched some of the references of the selected articles to make the associated topics more clear as appropriate. We summarized the information about the some of the most important papers in Tables 7.1 and 7.2.

#### 3 Probable Indicators of DR in Cardiovascular Diseases

The prevalence of left ventricular hypertrophy (LVH), atrial fibrillation (AF), and congestive heart failure (CHF) increases significantly with aging [13]. The poor consequences of CVD in old age can be described, at least in part, by cardiac

First Author (reference no)	Taffet [29]	Shinmura [30]	Ahmet [33]	Yan [34]
Species	B6D2-F1 hybrid mouse	Ficsher-344 rat	Ficsher-344 rat	129/Sv mouse
Age at the time of investigation	30–35 month old	30 month old	24 month old	24 month old
Length of time of CR	28-33 months	22 months	22 months	2 months
LV diastolic function	Improved	Improved	Improved	Not mentioned
LV systolic function	Unchanged	Unchanged	Improved	Improved
Cardiac fibrosis	Not investigated	Unchanged	Reduced	Reduced
Cardiomyocyte apoptosis	Not investigated	Reduced	Not investigated	Reduced
Cardiomyocyte size	Not investigated	Reduced	Reduced	Reduced
Total cardiomyocyte number	Not investigated	Not investigated	Unchanged	Unchanged

 Table 7.1
 Results from studies of the impact of calorie restriction (CR) on left ventricular (LV) function and cardiac aging

senescence both at the cellular and organ levels. To investigate the probable indicators for DR in age-associated CVDs, a discussion of the impacts of DR on atherosclerosis and vascular aging is required. Atherosclerosis and vascular aging are dependent on each other and are specific constituents to what is currently mentioned as clinical vascular disease [14]. Thus, aged blood vessels produce the milieu in which atherosclerosis can burgeon and DR can hinder both atherosclerosis and vascular aging [6, 9].

Moreover, LVH and CHF are diseases related to cardiac aging [13]. So, the prevention of CHF related to LVH, which is described as heart failure with preserved ejection fraction (HFpEF), is considered as a good indicator for DR. Next to population aging, obesity and overeating are common health problems in developed countries. The aggregation of visceral adipose tissue related to overeating and obesity results in metabolic syndrome and type 2 diabetes mellitus and, consequently, an elevation in morbidity and mortality from atherosclerotic diseases [15, 16]. Evidence demonstrates that overeating and obesity speed up the aging process, while DR can delay this. Thus, CVDs related to metabolic disorders might be the best indicator for DR.

There is also evidence that CR can decrease the incidence of neurodegenerative disorders in animal models of Alzheimer's disease, Huntigton's disease, amyotrophic lateral sclerosis, Parkinson's disease and spinal muscular atrophy [11]. These results lead to the proposal that DR can prevent the progression of genetic and/or degenerative disorders regardless of aging or obesity status. Hence, it has been speculated that DR may delay the progress of genetic cardiomyopathies such as mitochondrial myopathy. While, no experimental investigations have been done regarding this issue, the subject remains unresolved.

First Author			Length of	
(reference no)	Species	Age	time of CR	Impacts of CR
Shimura [6]	C57BL6 mouse	6 month old	3 months	Decreased myocardial I/R injury
Abete [40]	Wister rat	6 month old 24 month old	5 months 12 months	Restored the impact of ischemic preconditioning in aged hearts
Broderick [41]	Wister rat	10 month old	8 months	Decreased myocardial I/R injury
Chandrasekar [42]	Ficsher-344 rat	12 month old	11.5 months	Decreased myocardial inflammation after I/R
Long [44]	Sprague Dawley rat	20 month old	8 months	Restored the impact of ischemic preconditioning, but decreased cardiac performance after I/R injury
Shimura [46]	Ficsher-344 rat	4 month old 24 month old	5 weeks 5 weeks	Decreased myocardial I/R injury in both investigated ages of rats
Shimura [48]	C57BL6 mouse	4 month old	5 weeks	Decreased myocardial I/R injury
Shimura [47]	Ficsher-344 rat	12 month old	6 months	Decreased myocardial I/R injury and restored the impact of ischemic preconditioning
Edwards [43]	B6D2F1 mouse	30 month old	6 months	Decreased myocardial I/R injury
Sung [49]	C57BL6 mouse	4 month old	5 weeks	Decreased myocardial I/R injury
Peart [45]	C57BL6 mouse	12 month old	3 months	Decreased myocardial I/R injury

**Table 7.2** Results from studies of the impact of calorie restriction (CR) on myocardial ischemiareperfusion (I/R) injury and ischemic precondition

# 4 Impacts of DR on Vascular Aging and Atherosclerosis: Evidence from Experimental Animal Studies

Evidence reveals that increased production of reactive oxygen species (ROS) results in endothelial dysfunction with aging in animal models [6, 9, 12]. One of the results of elevated oxidative stress with aging is functional inactivation of nitric oxide (NO) due to high levels of superoxide generation. Damaged NO bioavailability due to age-associated oxidative stress in the coronary circulation and other vascular beds leads to severe dysfunction of vasodilation [6, 9, 12]. In addition to keeping normal organ blood flow, NO generated by endothelial cells exerts vasoprotective and cardioprotective impacts, consisting of inhibition of platelet accumulation, inhibited adhesion of inflammatory factors to endothelial cells, disturbance of proinflammatory cytokine-induced signaling pathways, apoptosis inhibition, conservation of endothelial progenitor cell function and control of tissue energy metabolism [6, 9, 12]. The disruption of NO bioavailability becomes worse due to an age-associated decrease in endothelial NO synthase (eNOS) expression, decreased availability of tetrahydrobiopterin, reduced availability of intracellular L-arginine and disturbed balance between eNOS and inducible NO synthase (iNOS). Age-associated impairment of NO bioavailability is likely to lead to elevated vascular inflammation and atherogenesis and result in a cellular energetic imbalance [6, 9]. In addition to inactivation of NO and the resulting macromolecular harm, ROS play major roles in the pathophysiological signaling pathways of endothelial and smooth muscle cells [6, 9].

CR regulates and activates eNOS, promotes NO bioavailability and ameliorates endothelial dysfunction in aged animals and humans [6, 9, 12, 17, 18]. Moreover, decreases in the age-related enhancement in oxidative stress are also believed to contribute to the anti-aging action of CR [6, 7, 9, 12]. Indeed, recent investigations have shown that CR effectively reduces the vascular production of ROS in aged animals [6, 7, 9, 12].

Considerable experimental and clinical evidence demonstrates that aging is related to chronic low grade inflammation that makes the vasculature susceptible to atherosclerosis. Recent studies have revealed important crosstalk between inflammation, ROS generation, and endothelial dysfunction in the pathogenesis of vascular senescence [6, 9]. Among the agents involved in this crosstalk, activation of nuclear factor- kB (NF-kB) has an important role in endothelial activation and proinflammatory gene expression with advancing age [6, 9, 19]. In this regard, CR decreases vascular NF-kB production and endothelial activation in aged rats [19]. CR can also disturb other pro-inflammatory signaling pathways, consisting of c-Jun N-terminal kinase (JNK), p38 kinase, and activator protein-1 DNA-binding activity [6, 9].

The sirtuin family has been demonstrated to play a notable role in different aspects of CR [6–10]. Sirt1 is expressed in high abundance in the cardiovascular system and its production is promoted by CR [6, 7]. Pharmacological over-expression or activation of Sirt1 by resveratrol exerts significant anti-oxidative and anti-inflammatory influences, which appear to result in decreased cellular senescence [6, 7, 9, 10, 20]. Furthermore, endothelial-specific over-expression of Sirt1 greatly decreases the atherosclerosis progression in mice with apolipoprotein (Apo)-E deficiency [21]. CR may control both eNOS expression and activity by Sirt1 activation. Recent investigations have shown that Sirt1 and eNOS localize in endothelial cells and Sirt1 deacetylates eNOS, promoting eNOS activity and enhancing NO in endothelial cells [20].

There have been few studies that assess the influence of ADF on vascular aging and atherosclerosis in rodents. Rodents and monkeys kept on ADF show increased insulin sensitivity [22]. Rodents maintained on ADF demonstrate reductions in resting blood pressure (BP) and heart rate (HR) [22, 23]. These results suggest that ADF can also prevent vascular aging and atherosclerosis. However, more evaluations are required to compare the influence of ADF with that of CR on vascular aging and atherosclerosis.

# 5 Impacts of CR on Vascular Aging and Atherosclerosis: Evidence from Human Studies

Elevated thickening and stiffness of large arteries and endothelial dysfunction in apparent healthy old individuals, along with enhance in systolic blood pressure and pulse pressure, forecast a higher risk for progression of clinical atherosclerosis, acute coronary syndrome (ACS) and stroke [14]. Such age-associated vascular alterations can be attributed to vascular aging. In this context, vascular aging is a risk factor for possible clinical disease manifestation.

Atherosclerosis and the following cardiovascular difficulties are considerable causes of death worldwide. The risk factors for atherosclerosis consist of hypertension, type 2 diabetes, higher concentrations of serum total and LDL cholesterol and smoking. Aging is also a major risk factor for atherosclerosis. Early or accelerated vascular aging can be increased by cardiovascular risk factors and cellular senescence is also seen in patients with atherosclerosis [14]. Thus, atherosclerosis is a disorder resulting from both organismal aging and cellular senescence.

CR has been demonstrated to delay vascular aging in both rodents and humans [6, 9, 12, 17, 24–27]. The underlying mechanisms of the beneficial cardiovascular impacts of CR are multifaceted, consisting of ameliorations in systemic risk factors for atherosclerosis such as dyslipidemia and insulin resistance and BP lowering, along with direct anti-aging impacts on the vasculature [6, 9].

# 6 Impacts of ADF on Vascular Aging and Atherosclerosis: Evidence from Human Studies

ADF includes an AL feed day alternated with a 25% calorie intake fast day. While the intervention duration has been 12 weeks at most, ADF was found to be beneficial for weight loss (-6.5%) and for ameliorating the atherosclerosis risk factors in normal weight and overweight adults [28]. The ADF protocol appears to be hopeful for treating overweight persons and preventing the progress of type 2 diabetes in patients with CVDs.

# 7 Impacts of DR on Cardiac Aging and Age-Related Left Ventricular Diastolic Dysfunction: Animal Calorie Restriction Experiments

For the first time, Taffet et al. [29] indicated that long term CR ameliorates agerelated alterations in late diastolic function in mice. In another study, Shinmura et al. [30] used Fischer-344 rats in an evaluation of the influence of long term CR on cardiac senescence, with the diet initiated at the age of 8 months and continued until the age of 30 months. The results of this study showed that long term CR ameliorated diastolic function in the aged myocardium by improving the age-related failure in myocyte relaxation. Moreover, the results demonstrated that a reduced size of cardiomyocytes may help by improving left ventricular (LV) diastolic dysfunction in rats on CR diets. In addition, the results suggested that CR impacts on age-related changes in cytoskeletal proteins, resulting in the amelioration of LV diastolic function.

Dhahbi et al. [31] indicated that long term CR decreases the myocardial collagen and extracellular matrix content and reduces cardiac fibrosis related to aging. Therefore, CR-induced alterations in cardiac connective tissue may help in the improvement of diastolic function, particularly late diastolic function, as reported by Taffet et al. [29]. Oppositely, a study carried out by Shinmura et al. [30] did not find a significant reduction in cardiac fibrosis in the hearts of rats receiving CR. These results proposed that long term CR may improve the age-related failure of early diastolic function by keeping the function of the sarcoplasmic reticulum (SR). These findings are different from those of Taffet et al. [29] as this latter group of researchers found no amelioration in early diastolic cardiac function in mice. However, the results reported by Shinmura et al. [30] are consistent with those of Seymour et al. [32], who found that CR ameliorated cardiac remodeling and diastolic dysfunction in Dahl-ss rats.

Recently, another research group reported the impact of long term CR on cardiac aging and assessed LV function in Fischer-344 rats at an earlier age. In comparison to AL-fed rats, 24-month old rats receiving CR had decreased levels of cardiac and aortic fibrosis, elevated density of cardiomyocytes that had smaller size, decreased diastolic dysfunction, and normal systolic performance and arterioventricular coupling [33]. Taken together, these results demonstrated that CR prevents age-related LV dysfunction by retarding cardiac aging.

On the other hand, Yan et al. [34] found that short term CR may reverse agerelated LV dysfunction. When short term (2 months) CR was initiated after agerelated LV dysfunction was generated in 20-month-old mice, leading to reduced cardiac function, and increased LV weight, the myocardial fibrosis and apoptosis were reversed and LV performance was found to be similar to that of young mice or mice that had been on CR diets from younger ages. The researchers also reported that short term CR prevented changes in cytoskeletal proteins which led to aging cardiomyopathy, but not the reduction in cardiomyocyte number related to advancing age [34].

#### 8 Mechanisms by Which CR Delays Cardiac Aging

The mechanisms by which long term CR delays cellular aging and diminishes the physiological function of organs have not been completely clarified. Aging happens, in part, due to the aggregation of oxidative harm caused by oxidative free

radicals that are produced continually during the metabolic processes [7]. Oppositely, CR reduces the age-related aggregation of oxidative harm to lipids, proteins and DNA. Also, expression of protein carbonyls is lower in hearts of rats receiving CR diets compared with hearts of AL-fed rats [7, 30]. It is probable that long term CR delays cellular aging and decreases age-associated functional losses by diminishing oxidative harm in the aged heart. However, there is still no concrete evidence indicating that the diminishing of oxidative harm is the first means by which CR prevents cardiac aging.

Another probable mechanism by which long term CR may delay cardiac aging is through the elevation of autophagy [7, 30]. Autophagy under basal states has a housekeeping role in the turnover of cytoplasmic components. Increased autophagy during CR has been suggested to be protective because this process leads to degradation and removal of harmed organelles. When autophagy is not functioning perfectly in the aged heart, the aggregation of damaged SR and mitochondria may be lethal by leading to diastolic dysfunction. Damaged autophagy in the aged heart may lead to the aggregation of lipofuscin, as well as in the inhibition of autophagy. It was reported that long term CR decreases the aggregation of lipofuscin [30] and the researchers proposed that long term CR disturbs this cycle in the aged heart. Furthermore, it was demonstrated that increased autophagy is related to suppression of the mTOR pathway in the heart [30]. The activation of mTOR leads to a negative regulatory impact on the induction of autophagy [10]. Recently, a proteome analysis showed that either 10 weeks of CR or rapamycin treatment ameliorated protein turnover in aged hearts [35]. It is possible that the inhibition of mTOR signaling during CR conditions has a key role in the making of CR-induced cardioprotection.

# 9 Impacts of DR on Cardiac Aging and Age-Related Left Ventricular Diastolic Dysfunction: Animal ADF Experiments

While the protective impact of CR against age-related cardiac dysfunction has been accepted as proven in rodents, the impact of ADF on cardiac senescence and cardiac function remains unclear. Castello et al. [36] demonstrated the impacts of long term ADF, from 2 to 24 months of age, on cardiac fibrosis and oxidative stress related to aging in rats. The results showed that ADF significantly decreased cardiac oxidative harm, the expression of pro-inflammatory cytokines, NF-kB DNA binding capacitance, and cardiac fibrosis, compared to the same parameters in AL-fed 24 months old rats. The researchers also reported that ADF decreased cardiomyocyte hypertrophy in aged rats, possibly by repressing extracellular signal-regulated kinase 1 and 2 and restoring repressor of cytokine signaling 3 and signal transducer and activator of transcription 3 activities, alterations which have been associated with advancing age [37].

In contrast, a study by Ahmet et al. [38] concluded that ADF has an adverse impact on cardiac function. In this study, ADF was initiated at 4 months and kept until 10 months of age. ADF led to 9% lowering in cardiomyocyte diameter and a 3-fold elevation in myocardial fibrosis. Also, the results showed a significant lowering in LV diastolic compliance and a defective elevation in systolic pump function. The mechanism by which ADF may lead to the diastolic dysfunction with a decreased cardiac reserve remains unclear, although the researchers proposed that the defect in important nutrients or the activation of the renin-angiotensin-aldosterone system may be involved. Due to these discrepancies across studies, further studies are warranted.

# 10 Impacts of Dietary Restriction on Cardiac Aging and Age-Related LV Diastolic Dysfunction: Human Calorie Restriction Trials

In one study, Meyer et al. [39] showed that CR is useful for LV diastolic function in humans. The E/A ratio (the ratio of peak velocity of blood flow due to gravity in early diastole to the peak velocity flow caused by atrial contraction in late diastole) was greater in the CR group than in the group receiving a Western diet, although the differences were not significant. The researchers assumed that the useful impact of CR on LV diastolic function resulted from reductions in systolic BP and systemic inflammation caused by CR.

Although ADF effectively decreases body weight even in humans, more studies investigating the impact of this diet on cardiovascular functions are required before it can be applied in clinical studies in humans.

# 11 Impacts of Calorie Restriction on Myocardial Ischemia-Reperfusion Injury

Lifelong CR significantly reduces myocardial oxidative stress and the inflammatory response related to advanced age [4, 7–9]. In this regard, some researchers have investigated the impact of long term CR on the grade of myocardial ischemia-reperfusion injury and the progress of ischemic preconditioning (PC) [6, 40–49]. Broderick et al. showed that CR for 8 months ameliorated recovery of cardiac function after 25 min of ischemia in rat hearts [41]. The researchers also demonstrated that the cardioprotective impact of CR was related to amelioration in mitochondrial respiration. Abete et al. [40] showed that CR can restore the cardioprotective impact of ischemia-reperfusion but considerably restored the PC impact in aged hearts. Long et al. [44] showed that CR for 6 months restored the PC impact in

middle-aged rat hearts. However, in their study, recovery of cardiac output and aortic flow was decreased in CR fed rats compared to AL-fed rats.

In another study [47], 6 month old rats were exposed to CR for 6 months to determine whether or not this diet can ameliorate myocardial ischemic tolerance and restore the progress of ischemic PC. The results showed that 6 months of CR ameliorated the recovery of LV function after ischemia-reperfusion and decreased infarct volume in the hearts of the 12 month old rats. Moreover, CR restored the ischemic PC impact in middle-aged rats. Altogether, the results of recent studies supported the potential of CR intervention aimed at lowering myocardial harm to decrease ischemic stress in humans.

Although, several studies have shown that fasting decreases myocardial ischemiareperfusion injury in rodents, it remains unclear whether or not ADF decreases myocardial ischemia-reperfusion injury.

# 12 Impacts of DR on Myocardial Infarction and Post-Infarct LV Remodeling

The influences of DR on infarct volume after permanent coronary ligation and post infract LV remodeling were investigated in vivo. After 3 months of receiving an AL or ADF diet, myocardial infarction (MI) was made by permanent coronary ligation in 5 month old rats. Infarct volume was investigated by use of triphenyltetrazolium chloride spotting 24 h after MI and found to be significantly smaller in rats receiving ADF [50]. In addition, post-infarct LV remodeling, myocardial infarction extension and LV function assessed by echocardiography 10 weeks after MI were significantly ameliorated in ADF fed rats. Thus, ADF may protect the myocardium against ischemic damage and decrease post infarct LV remodeling, potentially via anti-apoptotic and anti-inflammatory mechanisms. Oppositely, this research group indicated that 3 months of CR prior to coronary ligation could not protect the hearts against ischemic damage or post infarct LV remodeling in 5 month old rats [33].

When ADF was initiated 2 weeks after permanent coronary ligation, the survival of rats after MI was considerably enhanced [51]. Moreover, ADF decreased post infarct remodeling and ameliorated LV function after myocardial infarction. Molecular investigations showed an up-regulation of angiogenic agents consisting of hypoxia-inducible factor  $1\alpha$ , brain-derived neurotrophic factor, and vascular endothelial growth factor in the hearts during fasting states. These results suggested that the protective influence of ADF against post infarct LV remodeling was greater than that of CR.

# 13 Dietary Restriction to Protect Against Metabolic Cardiomyopathy Related to Obesity and Type 2 Diabetes Mellitus

Overeating and obesity are serious health problems in societies throughout the world. Obesity-related metabolic disorders, including metabolic syndrome and type 2 diabetes, quicken atherosclerosis and vascular aging, leading to the progression of ischemic heart disease (IHD), stroke, peripheral arterial disease, and other CVDs [15, 16]. In addition, metabolic disorders lead to cardiac hypertrophy, HFpEF, and pulmonary artery hypertension. Hence, the impact of DR on cardiac dysfunction related to metabolic disorders is of considerable clinical interest.

Ob/ob mice demonstrate noticeable obesity, insulin resistance, cardiac hypertrophy, mitochondrial impairment and ectopic lipid aggregation, resulting from leptin deficiency. In a study, Sloan et al. [52] compared the impact of CR on myocardial metabolism and function in 4 week old ob/ob mice to that of leptin supplementation. While CR led to normal body weight and glucose tolerance, fat mass content and circulating lipid concentrations remained elevated in CR fed ob/ob mice. Palmitate oxidation in the heart remained increased in ob/ob mice receiving CR and became normal following intraperitoneal or intracerebroventricular leptin application. These findings suggest that damaged hypothalamic leptin signaling can elevate myocardial fatty acid oxidation in spite of CR.

Moreover, the useful impacts of CR were impaired in ob/ob mice when CR was initiated from middle age. AlGhatrif et al. [53] showed that CR can restore LV diastolic function, reverse myocardial steatosis, and ameliorate insulin sensitivity when this diet was initiated at 2 months of age in ob/ob mice, but none of these alterations were reported when CR was initiated at 6–7 months of age. However, CR resulted in reduced cardiac oxidative stress and normal NOS activity.

Recently, some clinical studies have indicated that CR combined with significant body weight loss has cardiac-specific impacts and improves LV diastolic function in healthy individuals [39], and also in patients with type 2 diabetes [15]. Therefore, the clinical use of CR and the progress of CR mimetics have potential as novel therapeutic approaches in the treatment of subjects with diastolic dysfunction. However, the mechanisms by which CR ameliorates cardiac diastolic dysfunction in humans remain unclear. Recent evaluations have led to the proposal that myocardial triglyceride content is an independent predictor of diastolic function in old age [54] and in subjects with type 2 diabetes [15]. The reduction in myocardial triglyceride content due to CR is related to amelioration in LV diastolic function [15, 54]. It is probable that elevated autophagy aids the degradation of fatty acid intermediates that are potentially toxic.

# 14 Impacts of Dietary Restriction on Autonomic Nerve Function

An age-related reduction in autonomic nervous system function is partly responsible for the elevation in arrhythmias and syncope with aging [55]. Some researchers have investigated the impact of DR on autonomic nerve function in rodents or humans [23, 55]. To investigate the impact of DR on beat-to-beat heart rate variability (HRV) and diastolic BP variability (DBPV), Sprague Dawley rats inserted with telemetric transmitters were exposed to AL, ADF or CR feeding at 3 months of age [23]. Both ADF and CR led to elevations in the low frequency (LF) and the high frequency (HF) constituents of the HRV spectrum within 1 month. ADF, but not CR, reduced the LF and LF/HF ratio of the DPV spectrum. These results suggested that DR leads to reduced sympathetic nervous system activity and increased parasympathetic tone. Furthermore, the impact on parasympathetic tone appears to be more powerful from ADF than that from CR.

In one study, Stein et al. [55] assessed HRV in 22 subjects aged 35–82 years following CR compared to 20 age-matched controls eating a Western diet (WD). All HRV values were significantly greater in the CR group compared to the same values in the WD group. These results propose that CR reduces sympathetic tone, elevates parasympathetic tone and restores circadian rhythm variance. In conclusion, DR may exert direct systemic impacts that oppose the expected age-related alterations in autonomic nervous system function in humans.

#### 15 Conclusions

In summary, dietary restriction consists of calorie restriction and and alternate day fasting as approved nutritional interventions that manifest anti-aging impacts and delay the aging process. Dietary restriction also shows protection in all cell tissues and organs including the heart and vasculature bed. It also impacts on energy and oxygen radical metabolism, inflammatory agents and cellular stress response systems. Through these mechanisms, dietary restriction can protect the cells and organs against the damaging impact of aging. Because aging is associated with serious chronic disorders such as cardiovascular diseases and type 2 diabetes, and the proportionate population of elderly people has had an elevating trend over the last century, achieving the most effective nutritional intervention that achieves healthier aging with fewer disabling problems can ameliorate the public health state and enhance quality of life.

# References

- 1. Varady KA, Hellerstein MK (2007) Alternate-day fasting and chronic disease prevention: a review of human and animal trials. Am J Clin Nutr 86:7–13
- 2. Weindruch R, Sohal RS (1997) Caloric intake and aging. N Engl J Med 337:986-994
- Mattison JA, Lane MA, Roth GS, Ingram DK (2003) Calorie restriction in rhesus monkeys. Exp Gerontol 38:35–46
- 4. Masoro EJ (2005) Overview of caloric restriction and ageing. Mech Ageing Dev 126:913-922
- McCay CM, Crowell MF, Maynard LA (1989) The effect of retarded growth upon the length of life span and upon the ultimate body size. 1935. Nutrition 5:155–171
- Shinmura K (2011) Cardiovascular protection afforded by caloric restriction: essential role of nitric oxide synthase. Geriatr Gerontol Int 11:143–156
- 7. Shinmura K (2013) Effects of caloric restriction on cardiac oxidative stress and mitochondrial bioenergetics: potential role of cardiac sirtuins. Oxidative Med Cell Longev 2013:528935
- 8. Speakman JR, Mitchell SE (2011) Caloric restriction. Mol Asp Med 32:159-221
- 9. Ungvari Z, Parrado-Fernandez C, Csiszar A, De Cabo R (2008) Mechanisms underlying caloric restriction and lifespan regulation. Circ Res 102:519–528
- North BJ, Sinclair DA (2012) The intersection between aging and cardiovascular disease. Circ Res 110:1097–1108
- Schroeder JE, Richardson JC, Virley DJ (2010) Dietary manipulation and caloric restriction in the development of mouse models relevant to neurological diseases. Biochim Biophys Acta 1802:840–846
- Zanetti M, Cappellari GG, Burekovic I, Barazzoni R, Stebel M, Guarnieri G (2010) Caloric restriction improves endothelial dysfunction during vascular aging: effects on nitric oxide synthase isoforms and oxidative stress in rat aorta. Exp Gerontol 45:848–855
- Lakatta EG (2003) Arterial and cardiac aging: major shareholders in cardiovascular disease enterprises. Circulation 107:490–497
- 14. Lakatta EG (2003) Arterial and cardiac aging: major shareholders in cardiovascular disease enterprises: part I: aging arteries: a "set up" for vascular disease. Circulation 107:490–497
- 15. Hammer S, Snel M, Lamb HJ, Jazet IM, van der Meer RW, Pijl H et al (2008) Prolonged caloric restriction in obese patients with type 2 diabetes mellitus decreases myocardial triglyceride content and improves myocardial function. J Am Coll Cardiol 52:1006–1012
- Riordan MM, Weiss EP, Meyer TE, Ehsani AA, Racette SB, Villareal DT et al (2008) The effects of caloric restriction-and exercise-induced weight loss on left ventricular diastolic function. Am J Physiol Heart Circ Physiol 294:H1174–H1182
- 17. Holloszy JO, Fontana L (2007) Caloric restriction in humans. Exp Gerontol 42:709-712
- Nisoli E, Tonello C, Cardile A, Cozzi V, Bracale R, Tedesco L et al (2005) Calorie restriction promotes mitochondrial biogenesis by inducing the expression of eNOS. Science 310:314–317
- Donato AJ, Eskurza I, Silver AE, Levy AS, Pierce GL, Gates PE et al (2007) Direct evidence of endothelial oxidative stress with aging in humans: relation to impaired endothelium-dependent dilation and upregulation of nuclear factor-kappaB. Circ Res 100:1659–1666
- Mattagajasingh I, Kim CS, Naqvi A, Yamamori T, Hoffman TA, Jung SB et al (2007) SIRT1 promotes endothelium-dependent vascular relaxation by activating endothelial nitric oxide synthase. Proc Natl Acad Sci U S A 104:14855–14860
- Zhang QJ, Wang Z, Chen HZ, Zhou S, Zheng W, Liu G et al (2008) Endothelium-specific overexpression of class III deacetylase SIRT1 decreases atherosclerosis in apolipoprotein E-deficient mice. Cardiovasc Res 80:191–199
- 22. Mattson MP, Wan R (2005) Beneficial effects of intermittent fasting and caloric restriction on the cardiovascular and cerebrovascular systems. J Nutr Biochem 16:129–137
- Mager DE, Wan R, Brown M, Cheng A, Wareski P, Abernethy DR et al (2006) Caloric restriction and intermittent fasting alter spectral measures of heart rate and blood pressure variability in rats. FASEB J 20:631–637

- 24. Heilbronn LK, de Jonge L, Frisard MI, DeLany JP, Larson-Meyer DE, Rood J et al (2006) Effect of 6-month calorie restriction on biomarkers of longevity, metabolic adaptation, and oxidative stress in overweight individuals: a randomized controlled trial. JAMA 295:1539–1548
- 25. Csiszar A, Labinskyy N, Jimenez R, Pinto JT, Ballabh P, Losonczy G et al (2009) Antioxidative and anti-inflammatory vasoprotective effects of caloric restriction in aging: role of circulating factors and SIRT1. Mech Ageing Dev 130:518–527
- 26. Fontana L, Meyer TE, Klein S, Holloszy JO (2004) Long-term calorie restriction is highly effective in reducing the risk for atherosclerosis in humans. Proc Natl Acad Sci U S A 101:6659–6663
- Pearson KJ, Lewis KN, Price NL, Chang JW, Perez E, Cascajo MV et al (2008) Nrf2 mediates cancer protection but not prolongevity induced by caloric restriction. Proc Natl Acad Sci U S A 105:2325–2330
- Varady KA, Bhutani S, Klempel MC, Kroeger CM, Trepanowski JF, Haus JM et al (2013) Alternate day fasting for weight loss in normal weight and overweight subjects: a randomized controlled trial. Nutr J 12:146
- Taffet GE, Pham TT, Hartley CJ (1997) The age-associated alterations in late diastolic function in mice are improved by caloric restriction. J Gerontol A Biol Sci Med Sci 52:B285–B290
- 30. Shinmura K, Tamaki K, Sano M, Murata M, Yamakawa H, Ishida H et al (2011) Impact of long-term caloric restriction on cardiac senescence: caloric restriction ameliorates cardiac diastolic dysfunction associated with aging. J Mol Cell Cardiol 50:117–127
- Dhahbi JM, Tsuchiya T, Kim HJ, Mote PL, Spindler SR (2006) Gene expression and physiologic responses of the heart to the initiation and withdrawal of caloric restriction. J Gerontol A Biol Sci Med Sci 61:218–231
- 32. Seymour EM, Parikh RV, Singer AA, Bolling SF (2006) Moderate calorie restriction improves cardiac remodeling and diastolic dysfunction in the Dahl-SS rat. J Mol Cell Cardiol 41:661–668
- Ahmet I, Tae HJ, de Cabo R, Lakatta EG, Talan MI (2011) Effects of calorie restriction on cardioprotection and cardiovascular health. J Mol Cell Cardiol 51:263–271
- 34. Yan L, Gao S, Ho D, Park M, Ge H, Wang C et al (2013) Calorie restriction can reverse, as well as prevent, aging cardiomyopathy. Age (Dordr) 35:2177–2182
- 35. Dai DF, Karunadharma PP, Chiao YA, Basisty N, Crispin D, Hsieh EJ et al (2014) Altered proteome turnover and remodeling by short-term caloric restriction or rapamycin rejuvenate the aging heart. Aging Cell 13:529–539
- 36. Castello L, Froio T, Maina M, Cavallini G, Biasi F, Leonarduzzi G et al (2010) Alternate-day fasting protects the rat heart against age-induced inflammation and fibrosis by inhibiting oxidative damage and NF-kB activation. Free Radic Biol Med 48:47–54
- 37. Castello L, Maina M, Testa G, Cavallini G, Biasi F, Donati A et al (2011) Alternate-day fasting reverses the age-associated hypertrophy phenotype in rat heart by influencing the ERK and PI3K signaling pathways. Mech Ageing Dev 132:305–314
- Ahmet I, Wan R, Mattson MP, Lakatta EG, Talan MI (2010) Chronic alternate-day fasting results in reduced diastolic compliance and diminished systolic reserve in rats. J Card Fail 16:843–853
- Meyer TE, Kovács SJ, Ehsani AA, Klein S, Holloszy JO, Fontana L (2006) Long-term caloric restriction ameliorates the decline in diastolic function in humans. J Am Coll Cardiol 47:398–402
- 40. Abete P, Testa G, Ferrara N, De Santis D, Capaccio P, Viati L et al (2002) Cardioprotective effect of ischemic preconditioning is preserved in food-restricted senescent rats. Am J Physiol Heart Circ Physiol 282:H1978–H1987
- Broderick TL, Driedzic WR, Gillis M, Jacob J, Belke T (2001) Effects of chronic food restriction and exercise training on the recovery of cardiac function following ischemia. J Gerontol A Biol Sci Med Sci 56:B33–B37
- Chandrasekar B, Nelson JF, Colston JT, Freeman GL (2001) Calorie restriction attenuates inflammatory responses to myocardial ischemia-reperfusion injury. Am J Physiol Heart Circ Physiol 280:H2094–H2102

- Edwards AG, Donato AJ, Lesniewski LA, Gioscia RA, Seals DR, Moore RL (2010) Lifelong caloric restriction elicits pronounced protection of the aged myocardium: a role for AMPK. Mech Ageing Dev 131:739–742
- 44. Long P, Nguyen Q, Thurow C, Broderick TL (2002) Caloric restriction restores the cardioprotective effect of preconditioning in the rat heart. Mech Ageing Dev 123:1411–1413
- 45. Peart JN, See Hoe L, Pepe S, Johnson P, Headrick JP (2012) Opposing effects of age and calorie restriction on molecular determinants of myocardial ischemic tolerance. Rejuvenation Res 15:59–70
- 46. Shinmura K, Tamaki K, Bolli R (2005) Short-term caloric restriction improves ischemic tolerance independent of opening of ATP-sensitive K+ channels in both young and aged hearts. J Mol Cell Cardiol 39:285–296
- 47. Shinmura K, Tamaki K, Bolli R (2008) Impact of 6-mo caloric restriction on myocardial ischemic tolerance: possible involvement of nitric oxide-dependent increase in nuclear Sirt1. Am J Physiol Heart Circ Physiol 295:H2348–H2355
- Shinmura K, Tamaki K, Saito K, Nakano Y, Tobe T, Bolli R (2007) Cardioprotective effects of short-term caloric restriction are mediated by adiponectin via activation of AMP-activated protein kinase. Circulation 116:2809–2817
- Sung MM, Soltys CL, Masson G, Boisvenue JJ, Dyck JR (2011) Improved cardiac metabolism and activation of the RISK pathway contributes to improved post-ischemic recovery in calorie restricted mice. J Mol Med (Berl) 89:291–302
- 50. Ahmet I, Wan R, Mattson MP, Lakatta EG, Talan M (2005) Cardioprotection by intermittent fasting in rats. Circulation 112:3115–3121
- Katare RG, Kakinuma Y, Arikawa M, Yamasaki F, Sato T (2009) Chronic intermittent fasting improves the survival following large myocardial ischemia by activation of BDNF/VEGF/ PI3K signaling pathway. J Mol Cell Cardiol 46:405–412
- 52. Sloan C, Tuinei J, Nemetz K, Frandsen J, Soto J, Wride N et al (2011) Central leptin signaling is required to normalize myocardial fatty acid oxidation rates in caloric-restricted ob/ob mice. Diabetes 60:1424–1434
- 53. AlGhatrif M, Watts VL, Niu X, Halushka M, Miller KL, Vandegaer K et al (2013) Beneficial cardiac effects of caloric restriction are lost with age in a murine model of obesity. J Cardiovasc Transl Res 6:436–445
- 54. van der Meer RW, Rijzewijk LJ, Diamant M, Hammer S, Schär M, Bax JJ et al (2008) The ageing male heart: myocardial triglyceride content as independent predictor of diastolic function. Eur Heart J 29:1516–1522
- 55. Stein PK, Soare A, Meyer TE, Cangemi R, Holloszy JO, Fontana L (2012) Caloric restriction may reverse age-related autonomic decline in humans. Aging Cell 11:644–650

# Chapter 8 Gut Microbiota and Microbiota-Related Metabolites as Possible Biomarkers of Cognitive Aging



Andrea Ticinesi, Antonio Nouvenne, Claudio Tana, Beatrice Prati, and Tiziana Meschi

## 1 Gut Microbiota and Cognitive Aging

#### 1.1 The "Gut-Brain Axis" in Dementia

In the last decade, several studies have highlighted that the intestinal microbiome, i.e. the ensemble of bacteria symbiotically living with the host in the gut lumen, may influence the physiopathology of a large number of human diseases, not involving only the gastrointestinal system [1-3]. For example, alterations of the composition of gut microbiota may promote the progression of chronic liver disease [4], chronic renal failure [5] and even the formation of kidney stones [6]. By modulating insulin sensitivity, anabolism and systemic inflammation, the gut microbiota could also have relevance in the onset of age-related bone diseases and sarcopenia [7, 8].

In this context, the existence of a possible "gut-brain axis" influencing cognition in aging has been hypothesized [9, 10]. A large number of animal studies, recently reviewed elsewhere [10], have demonstrated that alterations of the gut microbiota composition, the so-called dysbiosis, can be associated with reduction of performances at cognitive tasks and, conversely, the administration of probiotics or

A. Ticinesi (🖂) · A. Nouvenne

Geriatric Rehabilitation Department, University-Hospital of Parma, Parma, Italy

Microbiome Research Hub, University of Parma, Parma, Italy e-mail: aticinesi@ao.pr.it

C. Tana · B. Prati Geriatric Rehabilitation Department, University-Hospital of Parma, Parma, Italy

T. Meschi

Geriatric Rehabilitation Department, University-Hospital of Parma, Parma, Italy

Microbiome Research Hub, University of Parma, Parma, Italy

Department of Medicine and Surgery, University of Parma, Parma, Italy

<sup>©</sup> Springer Nature Switzerland AG 2019

P. C. Guest (ed.), *Reviews on Biomarker Studies in Aging and Anti-Aging Research*, Advances in Experimental Medicine and Biology 1178, https://doi.org/10.1007/978-3-030-25650-0\_8

functional foods can prevent cognitive decline in transgenic models of Alzheimer's disease. The hypothesized mechanisms involved in these phenomena include bacterial production of amyloid proteins able to promote  $\beta$ -amyloid deposition in the brain tissue, modulation of vagal nerve activity, production of bacterial metabolites that may act as endocrine modulators or neurotransmitters, modulation of neuroin-flammation and microbiota-mediated transformation of nutrients into substances exerting toxic or protective activity on brain cells [10, 11].

Therefore, the current scientific literature suggests that the intestinal microbiome can influence the onset of dementia and its progression in multiple ways, making the "gut-brain axis" a promising area of research for the future.

# 1.2 Clinical Relevance of the "Gut-Brain Axis" in Humans with Dementia

Despite the results of pre-clinical studies, the translation of the "gut-brain axis" concept into human health is still uncertain. In fact, only a few investigators have comprehensively compared the intestinal microbiota composition of older patients with cognitive complaints (ranging from mild cognitive impairment to overt dementia) with that of healthy active older subjects (Table 8.1). The results of these studies suggest the presence of reduced gut microbiota biodiversity (i.e., lower numbers of species) and specific compositional signatures, such as over-representation of Bacteroides and Enterobacteriaceae and under-representation of Dialister, in subjects with cognitive symptoms [12–15]. In the largest of these studies, comparing the fecal microbiota composition of 34 older patients with dementia and 94 agematched controls, dementia was associated with increased biodiversity, reduced representation of *Bacteroides*, and an increased Firmicutes/Bacteroidetes ratio [16]. These associations were independent of other known biomarkers of dementia, including ApoE ɛ4 polymorphism, presence of lacunar infarcts at brain computed tomography (CT) or amyloid deposition detected by positron-emission tomography (PET) [16].

These findings, although promising, should be interpreted with caution. Most of the existing studies were observational, with a cross-sectional design preventing to draw solid conclusions on causal inference. Low sample sizes of these studies also represent an issue (Table 8.1) and generalizability to the general population with cognitive impairment or dementia is not automatic.

These limitations could be overcome by rigorous and well-designed intervention studies, implying the manipulation of the intestinal microbiome by administration of probiotics or functional foods. However, despite the large number of human trials with pre- or probiotics, studies specifically focused on dementia are still lacking [17]. The few studies published to date included only subjects without cognitive complaints [17] or with advanced dementia [18–20], and thus had a low chance of detecting significant variations in cognitive performance. Moreover, the investigators

er subjects er subjects Setting and country 18 memory	main	Number and characteristics of participants 73 patients with dementia (40 amvloid-	he fecal microbiot Mean age $71 \pm 7$ (cases)	a composition between subjects suffering Gut microbiota-related parameters differing between cases and controls Abundance of <i>Eubacterium</i>	rom dementia or cognitive symptoms Implications The fecal microbiota of patients with dementia has a higher mo-
Northern Italy positive and amyloid-neg anyloid-neg 10 controls	dementia (4( positive and amyloid-neg 10 controls	) amyloid- 33 (ative)	68 ± 8 (Controls)	rectale(anti-inflammatory) Abundance of <i>Escherichia/Shigella</i> (pro-inflammatory)	dementia has a higher pro- inflammatory activity than co
Memory Clinic 25 patients w and Registry for Alzheimer's of Prevention of 25 controls Alzheimer's disease in the United States	25 patients w Alzheimer's ( 25 controls	ith disease	$71 \pm 7$ (cases) 69 \pm 8 (controls)	Overall biodiversity ( $\alpha$ -diversity) Inter-individual variability ( $\beta$ -diversity) Abundance of the Firmicutes and Bacteroidetes phyla Abundance of 13 taxa, including <i>Bifidobacterium</i> , <i>Alistipes</i> , <i>Bilophila</i> and <i>Clostridium</i>	Subjects with Alzheimer's disease have a distinct composition of fecal microbiome, with reduced biodiversity and abundance of putative anti-inflammatory taxa
Community 43 healthy com recreation and dwellers (25 wi wellness center cognitive functi in the United 18 with previou States cognitive impai	<ul><li>43 healthy com dwellers (25 wi cognitive functi cognitive functi 18 with previou unknown mild cognitive impai</li></ul>	munity th intact ons and sly rment)	64 ± 7	Abundance of 4 bacterial phyla (Bacteroidetes, Firmicutes, Proteobacteria, Verrucomorbia)	The gross composition of fecal microbiota in terms of phylum abundance is correlated with cognitive performance; the strongest correlation is observed for Verrucomorbia abundance
<ul><li>4 hospital</li><li>4 pospital</li><li>4 patients with dementia</li><li>4 controls</li><li>4 controls</li></ul>	43 patients with dementia 43 controls		70 ± 9 (cases and controls)	Inter-individual variability (β-diversity) Abundance of several species including Bacteroides, Actinobacteria, Ruminococcus, Lachnospiraceae, Selenomonadales	Subjects with dementia have a distinct fecal microbiota composition than controls at both community and species level
Memory clinic 34 patients with in Japan dementia 94 controls	34 patients with dementia 94 controls		76 (interquartile range 69–81)	Overall biodiversity (α-diversity) Firmicutes/Bacteroidetes ratio Abundance of <i>Bacteroides</i> , <i>Bifidobacterium</i> and Lactobacillales	The gut microbiota composition was associated with the status of suffering from dementia independently of other known and well-established biomarkers of dementia

8 Microbiome and Cognition in Aging

in these studies concentrated on microbiological and inflammation-related, rather than on clinical outcomes [18–20]. In the only study which assessed cognition as the primary outcome, the intervention consisted in a 12-week resistance training program associated with the administration of a *Bifidobacterium* spp blend, so that the improvement of executive functions detected in the intervention arm could not be attributed with complete certainty to the benefits of gut microbiome manipulation [21].

In this scenario, the possibility of obtaining a clinically relevant modulation of cognitive symptoms with microbiome-targeted interventions in older patients still remains speculative.

# 1.3 Gut Microbiota at the Cross-Road Between Environment and Human Pathophysiology

An increasing body of evidence suggests that the human microbiome composition is strongly influenced by environmental factors [1]. These factors include dietary patterns and habits [2], exercise [22], drugs [23, 24], the presence of diseases [23, 24], climate and pollution [25]. Moreover, the gut microbiota composition is shaped in each human being according to his or her genetic background, and is greatly influenced by immune system functionality [1, 20]. Aging itself represents a process able to influence the gut microbiota composition in an independent way, implying a progressive reduction of biodiversity and resilience to stressors [26]. Finally, growing evidence indicates that the output of metagenomics analyses of fecal microbiota is dependent on the bacterial load, which is ultimately associated with stool moisture [27, 28]. Intestinal motility, number of bowel movements and hydration status are thus important variables influencing the microbiome composition.

In the largest population-based study to date, investigating the human gut microbiota composition and its determinants, the authors were able to identify a long list of environmental factors associated with a distinct microbiological signature [29]. These results put into question the role of the intestinal microbiome as an independent mediator of human health and disease. They also suggest the existence of complex relationships between the environment and the host physiopathology, with gut microbiota possibly positioned at the cross-road [8, 30]. In fact, it is very difficult to consider all of the possible environmental factors influencing the microbiome composition, acting as potential confounders in pathophysiological associations [1, 30].

From a clinical perspective, this means that the disease-associated microbiome alterations most likely represent only one component of a large network of pathophysiological factors [30]. It also helps to understand why microbiome-targeted interventions, including administration of probiotics and functional foods, rarely provide clinically significant results outside the field of gastroenterology. If the microbiome is a cross-road mediator between the environment and host pathophysiology, then environmental interventions have a greater potential of influencing dis-

ease onset and course than microbiome manipulations targeted to a single or a limited number of bacterial species.

Despite these possible limitations, the intestinal microbiome remains a promising target for anti-aging interventions [31], particularly in the field of cognitive disorders, in which studies conducted in mouse models have provided encouraging results [10]. However, much research must still be done before we can recommend therapeutic strategies centered on the microbiome for patients with cognitive impairment, mainly because of objective difficulties in identification and preparation of probiotic blends or functional foods able to induce significant modification in the overall structure of gut communities [32].

Possibly, the nearest goal in this field of medicine is the use of gut microbiota composition and gut microbiota-related metabolites as biomarkers of disease. The existing studies exploring the relationship between microbiome and dementia highlight that a certain number of abnormalities in microbiome composition and functionality may be of potential interest as biomarkers. To be useful as biomarker of cognitive function, a physiological parameter should be measurable, correlated with cognitive performance or other indicators of cognitive impairment, able to predict the onset and course of cognitive impairment as well as the therapeutic response to treatments [33, 34]. Correlation with other measures of disease and prediction of outcomes are thus necessary before a biomarker can enter into clinical practice [33, 34]. Unfortunately, most of the research on the gut-brain axis has been focused on correlations rather than predictions [1]. However, there are many microbiome-related parameters deserving careful evaluation as possible biomarkers of cognitive aging in future studies.

# 2 Gut Microbiota Composition as Biomarker of Cognitive Function

# 2.1 Microbiome Biodiversity

The latest next-generation sequencing techniques (16S rRNA microbial profiling, shotgun metagenomics) allow detection of virtually all different taxa harbored in biological samples [35, 36]. After bioinformatics processing, the output of these analyses retrieves several indexes of biodiversity, such as the Shannon Index, the Simpson Index or the Chao1 Index [37]. In microbial ecology, biodiversity (also called alpha diversity) is defined as species richness and relative species abundance in space and time [38]. Thus, the more elevated are biodiversity indices, the higher is the number of species harbored in the fecal sample [37].

A high fecal microbiome biodiversity is generally considered to be a marker of good health status, implying a virtuous symbiotic equilibrium between microbial communities and host [3]. Conversely, reduced fecal microbiome biodiversity is generally associated with poor health status, and represents the main feature of gut

microbiota dysbiosis [3]. For example, studies performed in patients with critical illnesses admitted to intensive care units globally show a high level of fecal microbiota dysbiosis, with reduced species richness [39]. Moreover, older patients with prolonged hospital stay who develop *Clostridium difficile* infection, one of the most frequent healthcare-associated infections, exhibit a reduced level of fecal microbiota biodiversity, compared with those who have milder health problems and shorter hospital stays [40].

However, an elevated fecal microbiota biodiversity is not necessarily associated with a healthy microbiota. When patients are treated with systemic antibiotic therapy, a transient increase in gut microbiota biodiversity may be observed due to overexpansion of microbial populations, mainly pathobionts that are not sensitive to the administered treatment [41, 42]. Stressful events such as exercise to exhaustion, may also promote increased gut microbiota biodiversity due to increased colonization and expansion of opportunistic pathogens at the expense of symbionts [22]. Thus, a finding of increased biodiversity in fecal microbiota should be carefully interpreted, especially in the light of which taxa are present and their relative abundance.

In mouse models of senescence or dementia, the association between gut microbiota biodiversity and cognitive performance has provided inconsistent results. For example, in a group of senescence-accelerated mouse prone 8 (SAMP8) mice exhibiting symptoms of severe cognitive dysfunction, the gut microbiota biodiversity measured by Chao1 and Shannon indices was significantly reduced in comparison with controls [43]. However, another study showed that older mice with mild symptoms of cognitive dysfunction exhibited an increased, rather than decreased, gut microbiota biodiversity in comparison with younger mice [44]. Antibioticinduced reduction of gut microbiota biodiversity resulted in worsened cognitive performance of mice in one study [45] but was associated with reduced A $\beta$  amyloid deposition in transgenic mouse models of Alzheimer's disease [46, 47]. In germfree mice, harboring no intestinal microbiota, a massive reduction of A $\beta$  amyloid brain deposition was also documented in comparison with controls [48].

In humans, the association between gut microbiota biodiversity and cognitive functions has been assessed in different settings. In patients with cirrhosis undergoing liver transplantation, a high gut microbiota diversity was associated with reduced risk of encephalopathy and thus cognitive symptoms [49]. However, species richness in gut microbiota was unable to predict long-term cognitive outcomes in a group of unselected patients with cirrhosis [50]. In obese subjects, the Shannon index of fecal microbiota was correlated with anatomical and functional parameters of brain magnetic resonance imaging (R2\* and fractional anisotropy of the hypothalamus, caudate nucleus and hippocampus) and cognitive performance related to speed, attention and flexibility [51].

In a group of 85 HIV-infected subjects naïve to antiretroviral therapy, fecal microbiota biodiversity was significantly lower in the presence of HIV-associated neurological disease with cognitive symptoms [52]. However, this difference was not independent of educational level and HIV-infection-related clinical data, such as CD4 T-cell count, suggesting that gut microbiota dysbiosis is not pathophysiologically involved in the cognitive dysfunction [52].

In the largest human study to date, Verdi et al. [53] analyzed the fecal microbiota of 1551 individuals over the age of 40 and correlated microbial variables with the cognitive performance assessed by several tests (verbal fluency, Mini-Mental State Examination, Deary-Liewald Reaction Time test, Paired Associated Learning from the Cambridge Neuropsychological Test Automated Battery). Among the results of these tests, only the Deary-Liewald Reaction Time test score was inversely associated with alpha diversity independently of covariates, suggesting that the possible link between gut microbiota biodiversity and cognition may be mediated by several other environmental and host-related factors [53].

Fecal microbiota biodiversity was analyzed in patients with dementia in only three studies [13, 16, 54]. Araos et al. analyzed the fecal microbiota of 85 long-term-care facility residents with advanced dementia, demonstrating a very high level of dysbiosis with reduced species richness in all subjects [54]. Their results suggest that advanced dementia may be associated with low alpha diversity indices but the absence of a control group in the study design prevents solid conclusions. In another study performed in 25 patients with dementia and 25 controls, the Shannon index was significantly reduced alpha diversity and cognitive symptoms. However, in the most recent and largest study comparing alpha diversity between 34 demented patients and 94 controls, Saji and colleagues documented that the Shannon index was significantly lower in controls, rather than in demented patients (Table 8.1) [16].

Overall, the current literature state-of-art does not support the use of fecal microbiota biodiversity indices as biomarkers of cognitive aging. The studies performed on this topic are too scarce, with reduced sample sizes, and have therefore provided inconclusive results with interpretations that may be challenging. However, they do suggest that an association between alpha diversity and cognitive symptoms may exist, although this is not independent of covariates. Future studies should better investigate this association, considering larger sample sizes, thorough selection of covariates and the use of longitudinal designs to assess whether variations in gut microbiome composition can predict variations in cognitive function.

#### 2.2 Firmicutes to Bacteroidetes Ratio

Bioinformatics processing of next-generation sequencing metagenomics output allows identification of the number of bacterial taxa harbored in fecal samples as well as assignment of each of these to the corresponding taxonomic level and calculation of their relative abundance [35, 36]. These data are generally used to determine the relative abundance of the two most represented phyla in fecal microbiota, i.e., Bacteroidetes and Firmicutes, and their ratio. The Firmicutes/Bacteroidetes ratio is an index representing the overall qualitative composition of the fecal microbiota. The intestinal microbiota can in fact physiologically assume different states, called enterotypes, with a predominance of either Bacteroidetes or Firmicutes, according to individual factors and dietary habits [55]. The Firmicutes/Bacteroidetes ratio increases from infancy to adult life and then declines again with senescence [56]. The predominance of Firmicutes in fecal microbiota composition has also been associated with the presence of metabolic imbalances, especially obesity, insulin resistance and metabolic syndrome [57, 58]. In fact, it generally increases with increasing body mass indices [59]. The Firmicutes/Bacteroidetes ratio has thus become an important parameter in the evaluation of the relationship between gut microbiota, obesity and obesity-related disorders [57].

An altered Firmicutes/Bacteroidetes ratio can however also represent an index of dysbiosis in other diseases, including dementia. In mice genetically prone to dementia (3xtg breed), gut microbiota analyses revealed a marked elevation of the Firmicutes/Bacteroidetes ratio compared to controls [60]. In mice with diet-induced obesity, over-expression of Firmicutes and reduced expression of Bacteroidetes was associated with impaired cognitive performance, especially in recognition memory and spatial memory tasks [61]. Finally, in older mice, an increase in the Firmicutes/Bacteroidetes ratio was associated with compromised cognition and increased anxiety behaviors, and this was not observed in younger mice [44].

The Firmicutes/Bacteroidetes ratio was calculated in human beings with dementia in only two studies, where it appeared significantly higher than in healthy controls [15, 16]. However, in a population of cognitively healthy older adults, Manderino et al. showed that greater proportions of Firmicutes and smaller proportions of Bacteroidetes in fecal microbiota were associated with better performances at cognitive tests [14].

Thus, the current literature state-of-art seems to support the concept that the presence of cognitive impairment may be associated with variations in the Firmicutes/Bacteroidetes ratio in fecal microbiota, making this parameter a promising potential biomarker of cognitive aging, deserving investigation in future studies. The capacity of this ratio to predict the onset and worsening of cognitive symptoms in human beings, and its association with obesity and metabolic imbalance should be particularly assessed.

#### 2.3 Abundance of Specific Taxa in Fecal Microbiota

Apart from the overall composition of the fecal microbiota, variations in the representation of specific taxa may show a significant association with the presence of cognitive symptoms. The outputs of metagenomics analyses and their bioinformatics elaborations generally retrieve, for each detected bacterial taxon, a ratio of relative abundance, representing the proportion of bacteria belonging to that taxon on the total bacterial load of the sample [28]. Thus, these relative abundances do not represent the absolute quantities of bacteria harbored in each sample, and this concept should be considered for interpreting any result [28]. However, relative abundances could represent promising biomarkers of health status, especially if their association with diseases is reproducible across different studies. The main bacterial taxa found over-represented or under-represented in fecal samples of either patients or animal models with cognitive symptoms, cognitive impairment or dementia, according to the current literature state-of-art, are summarized in Table 8.2. The findings of studies on mouse models [43, 60, 62–64] showed some degree of inconsistency. For example, in mice with cognitive impairment, Christensenellaceae and Ruminococcaceae were found to be depleted in one study [43] and over-represented in another [60]. However, depletion of Bifidobacteria and over-representation of Anaeroplasmatales were both associated with dementia in two studies [60, 63]. The largest study which compared senescence-accelerated mouse prone 8 with senescence-accelerated mouse resistant 1 breeds, showed that the presence of cognitive symptoms was associated with depletion of 26 taxa and over-representation of only one taxon [43].

Human studies which investigated the relative abundances of specific microbial taxa in fecal microbiota composition in relation to cognitive performance were performed in healthy subjects [19, 53, 65, 66], patients with cirrhosis [67–69], Parkinsonism [70, 71], or dementia/cognitive impairment [12–16]. An overview of taxa that resulted in significant differences between subjects with and without signs of cognitive dysfunction is presented in Table 8.2. In a large population of twins aged 40 or older, Verdi and colleagues showed that under-representation of taxa belonging to the Burkholderiales order was associated with worse performance at cognitive tests independently of covariates such as frailty and treatment with proton pump inhibitors [53]. These findings suggest that microbiota depletion of Burkholderiales may represent a promising biomarker of cognitive aging, although the cross-sectional design of the study prevents drawing any conclusions on prediction of cognitive outcomes. In their studies on patients with cirrhosis, Bajaj and colleagues found other putative microbial biomarkers of cognitive impairment [67– 69], namely Alcaligenaceae, Porphyromonadaceae and Enterobacteriaceae with positive associations, and Lactobacillales and Lachnospiraceae having negative associations with the condition (Table 8.2). These studies provide interesting data on the possible usefulness of gut microbiota-related parameters for predicting cognitive outcomes. However, they were conducted using a population of patients exhibiting a high burden of gut microbiota dysbiosis, due to cirrhosis. Therefore, their results may not be immediately transferred to the elderly population with cognitive impairment or dementia.

Decreased relative abundance of Lachnospiraceae, *Butyricicoccus* and *Clostridium* XIVb, and increased relative abundance of Lactobacillaceae and Christensenellaceae were associated with the presence of cognitive impairment in two distinct studies performed on patients with Parkinsonism [70, 71]. Interestingly, a decreased relative abundance of Lachnospiraceae associated with cognitive symptoms was also observed in patients with cirrhosis and, most importantly, in one of the few studies specifically conducted in the field of human dementia [15, 69]. Similarly, a decreased relative abundance of *Clostridium* XIVb was also detected in the study by Vogt and colleagues [13], in which the fecal microbiota of 25 patients with Alzheimer's disease was compared with that of 25 healthy controls (Table 8.1).
symptoms, cognitive impairment or de	smentia		
Mouse studies		Human studies	
Over-representation in cognitive	Under-representation in cognitive	Over-representation in cognitive	Under-representation in cognitive
ппрантпепичетна	ппрантлепичетна	ипрантлепиченна	ппрантпепцает
Anaeroplasmatales [60, 64]	Bifidobacterium [60, 64]	Alcaligenaceae [67]	Adlerkreutzia [13]
Anaerostipes [64]	Christensenellaceae [43]	Alistipes [13]	Bacteroidaceae [15, 16]
Christensenellaceae [60]	Desulfovibrio [43]	Bacteroides [13]	Bifidobacterium [13]
Cyanobacteria [64]	Lachnospiraceae [43]	Bilophila [13]	Burkolderiales [53]
Erysipelotrichiaceae [63]	Prevotella [43]	Blautia [13]	Butyricicoccus[70]
Helicobacter [65]	Ruminococcaceae [43]	Caldiserica [66]	Clostridium XIVb [13, 69, 70]
Odoribacter [65]	Subdoligranulum [43]	Christensenellaceae [71]	Dialister [13]
Proteobacteria [63]		Enterobacteriaceae [68]	Eubacterium [12]
Ruminococcaceae [60]		Enterococcaceae [15]	Lachnospiraceae [15, 69, 71]
Sutterella [63]		Escherichia/Shigella [12]	Lactobacillales [68]
Tenericutes [64]		Gemella [13]	Parabacteroides [19]
Turicibacter [60]		Helicobacter [72]	Phascolarctobacterium [19]
		Lactobacillaceae [71]	Ruminococcaceae [19, 69]
		Phascolarctobacterium [13]	Turicibacter [13]
		Porphyromonadaceae [67, 68]	Veillonellaceae [15]
		Prevotellaceae [65]	Verrucomorbia [14]
		Proteobacteria [14]	
		Ruminococcaceae [15, 65]	
		Tenericutes [66]	
		Thermodesulfobacterium [66]	
			;

Table 8.2 Overview of the main bacterial taxa found as over- or under-represented in the fecal microbiota of animal models or human beings with cognitive

138

Some taxa are listed both as over- or under-represented because of some inconsistency across the results of different studies

Thus, the abundance of Lachnospiraceae and Clostridium XIV b may deserve greater attention in future studies as possible biomarkers of cognitive aging.

The study by Vogt and colleagues [13] was also the most accurate one to date in the identification of microbial taxa with altered relative abundance in Alzheimer's disease, and with significant correlation of these with other biomarkers, including p-tau and  $\beta$ -amyloid deposition. Specifically, the relative abundance of *Blautia*, *Bacteroides*, *Phascolarctobacterium*, *Alistipes*, *Bilophila* and *Gemella* was increased in Alzheimer's disease, while the relative abundance of *Clostridum*, *Bifidobacterium*, *Dialister*, *Adlerkreutzia* and *Turicibacter* was decreased [13]. Future, larger studies should better clarify whether these bacteria may assume the role of microbial biomarkers of dementia or cognitive impairment.

It is also noteworthy that a significant association between the representation of some taxa in the fecal microbiome and imaging biomarkers of cognitive dysfunction [66] or genetic risk factors for Alzheimer's disease (APOE polymorphisms) [65] has been detected in two distinct studies. Other investigations recently reviewed by Franceschi et al. [72] also suggest that the presence of Helicobacter pylori in the intestinal microbiome could be involved in the development of Alzheimer's disease, by promoting systemic chronic inflammation or molecular mimicry mechanisms. These aspects will need further confirmation in future studies but may represent the bases for the discovery of other microbial biomarkers of dementia and cognitive impairment.

In summary, the current literature state-of-art suggests a list of possible gut microbiota-related biomarkers of cognitive aging, shown in Table 8.2. However, the evidence is too scarce to recommend the use of fecal microbiota analyses as a method to identify the risk of cognitive impairment or dementia. In addition, these studies were conducted in heterogeneous settings, included small numbers of participants and had a cross-sectional design, preventing study of the association of microbiome composition with clinical outcomes. Finally, some inconsistencies are present among the results of different studies, reinforcing the need to develop larger and sounder investigations in this field.

## **3** Microbiota-Related Metabolites as Biomarkers of Cognitive Function

## 3.1 The Role of Microbial Metabolites as Biomarkers in the Gut-Brain Axis

Many substances synthesized by components of the gut microbiota, either as metabolic byproducts or constituents of bacterial structures, may have a role in influencing the gut-brain axis functionality in cognitive function [10]. These substances, their characteristics and possible pathophysiological relevance have been recently reviewed by many authors [73–77]. These include neurotransmitters [73, 74], metabolites derived by amino acid catabolism [75], short-chain fatty acids (SCFAs) and other lipids [76], components of the outer membrane of bacteria [76] and even bacterial amyloid proteins [77]. Despite their possible involvement in the development and course of dementia, only a limited number of them could have potential usefulness as biomarkers of cognitive aging. In fact, neurotransmitters, such as acetylcholine, norepinephrine, histamine and  $\gamma$ -aminobutyric acid, are also synthetized by the host, and their microbial origin cannot be easily detected with standard laboratory methods [73]. Bacterial amyloid proteins may be involved in the pathophysiology of Alzheimer's disease by promoting the formation of  $\beta$ -amyloid protein aggregates in the brain by molecular mimicry but they cannot easily be detected in biological samples [77]. Metabolites derived by amino acid catabolism and SCFAs have recently received considerable attention by researchers and show, in some cases, correlations with cognitive function [75, 76]. However, some technical issues associated with laboratory methods of detection are currently limiting their study and implementation in clinical practice.

Along with substances synthesized by the intestinal microbiota, there are also a number of substances derived from diet and transformed by certain metabotypes of gut microbiota that may have a relevant pathophysiological involvement in the gutbrain axis [8, 10]. However, since the intake of the precursors of these substances is variable and not easy to standardize in the diet, their use as biomarkers is in doubt. The most known of these substances is trimethylamine N-oxide (TMAO), which is produced by the metaorganismal metabolism of dietary choline and has been implicated in the pathogenesis of several human diseases, including cardiovascular, cerebrovascular and metabolic diseases [78]. Other microbial products of nutrients introduced with diet include polyphenol metabolites and urolithins [8]. Finally, recent studies link cognitive performance to vitamin K status, which is partly produced by the intestinal microbiota [79], so that vitamin K metabolism may represent another promising area of research in the field of cognitive aging.

The putative biomarkers of cognitive aging derived by microbiota metabolism can be detected in several biological samples, ranging from blood to cerebrospinal fluid to feces. In fact, most of the substances produced by gut microbiota are absorbed into the systemic circulation and can be detected in blood. This phenomenon is particularly enhanced in the presence of a "leaky gut". This means that increased intestinal mucosa permeability iscaused by inflammatory conditions of the gastrointestinal tract or by the presence of an extreme gut microbiota dysbiosis [80].

From the systemic circulation, some microbiota-derived substances may reach the central nervous system and may be best detected in cerebrospinal fluid samples. Some investigators have also studied metabolic byproducts of intestinal bacteria in stool samples, since their concentration may be directly related to the microbiota composition and functionality. SCFAs are among the most studied microbiotaderived biomarkers of disease in stool samples [81]. However, these molecules show low stability in fecal samples leading to huge intra-individual variability, depending on the timing of analyses after sample collection [81].

# 3.2 Structural and Functional Components of Bacteria as Biomarkers of Cognitive Aging

The structural and functional components of gut microbiota that have been detected in biological samples of the host and related with the onset of cognitive dysfunction or dementia, include lipopolysaccharide (LPS) and rhamnolipids (Table 8.3). LPS can be detected in human blood samples at low concentrations even in the absence of bacteremia or sepsis. This phenomenon is an expression of gut microbiota

Possible biomarker	Category/Origin	Biological sample of detection	Comment
Lipopolysaccharide	Constituent of bacterial wall, exotoxin	Blood Cerebrospinal fluid Brain biopsies	Increased in case of altered gut mucosa permeability or gut microbiota dysbiosis; Enhances neuroinflammation
Rhamnolipids	Constituent of bacterial wall, exotoxin	Blood Cerebrospinal fluid	Found in cerebrospinal fluid and blood in only one study, promote inflammation
Kynurenine Kynurenic acid 3-hydroxy- kynurenine	Products of amino acid metabolism (tryptophan)	Blood	Only indirectly influenced by gut microbiota metabolism of tryptophan
Indole-3 acetic acid Indoxyl sulfate	Products of amino acid metabolism (tryptophan)	Blood	Products of microbiota-driven degradation of tryptophan; Uremic toxins found in advanced chronic kidney failure
NADH:Ubiquinone reductase	Bacterial enzyme	Feces	Enzyme related to bacterial synthesis of tryptophan through the shikimate pathway
Acetate, butyrate, propionate	Short-chain fatty acids	Blood Feces	Products of bacterial metabolism of carbohydrates; Show neuroprotective, anti-inflammatory and pro-anabolic effects
Deoxycholic acid	Bile acid	Blood Feces	Concentrations depend on the gut microbiota metabolism of bile acids
Trimethylamine-N oxide	Product of microbial metabolism of dietary choline	Blood Cerebrospinal fluid	Marker of increased cardiovascular and cerebrovascular risk
Polyphenol metabolites	Product of microbial metabolism of nutrients	Blood	Effect depends on the individual metabotype (composition and functionality of gut microbiota)
Vitamin K	Bacterial cofactor	Blood	Possibly influencing cognitive function; role not yet assessed in relation to microbiota

**Table 8.3** Overview of the main bacterial products or metabolites that may be potentially useful as biomarkers of cognitive aging and dementia, according to the current scientific literature

dysbiosis or altered intestinal permeability, and causes chronic subclinical activation of inflammation that is involved in the pathophysiology of several non-communicable diseases, including atherosclerosis, diabetes and obesity [82]. Since inflammation is also an important mechanism triggering age-related cognitive impairment and the onset of dementia, some researchers have investigated the possible association between the presence of LPS in biological samples of the host and dementia [10].

In mice with diet-induced obesity, an association between reduced performance at recognition and spatial memory tasks and increased serum levels of LPS was detected [61]. These findings were confirmed by an intervention study where genistein was administered in association with a high-fat diet to mice [83], suggesting that LPS levels may be inversely correlated with cognitive performance. However, the intracerebral injection of LPS in mouse models of Alzheimer's disease did not induce significant alterations in cognitive function [84].

Human serum levels of LPS have not been associated with cognitive performance by any study to date. However, investigations on brain biopsies have demonstrated that lysates from the hippocampus and superior temporal lobe neocortex of patients with Alzheimer's disease had significantly higher concentrations of LPS than brain lysates from non-demented controls [85–87]. Therefore, the relationship between LPS and cognitive performance in humans deserves more investigations and LPS could represent a promising biomarker of cognitive health.

Andreadou et al. have also demonstrated that rhamnolipids, other bacterial virulence factors possibly derived from the gut microbiota, exhibit higher serum and cerebrospinal fluid levels in patients suffering from Alzheimer's disease than in non-demented controls [88]. These findings have not been replicated in other studies but the role of rhamnolipids as a possible biomarker of cognitive aging and pathophysiology of dementia should be investigated.

### 3.3 Microbiota, Amino Acid Metabolism and Cognitive Aging

Studies performed in mice have demonstrated that the decline of cognitive performance is associated with a distinct metabolic profile in serum and in the brain [44, 60]. In mouse genetically prone to dementia, an overabundance of ketone bodies, lactate and amino acids and depletion of unsaturated fatty acids and choline, was demonstrated in serum and feces, suggesting a role of the intestinal microbiota metabolism in driving these changes [60, 65]. Brain metabolomics of aged mice showed significantly lower concentrations of amino acids (including methionine, phenylalanine, cysteine and creatine) and lipidic cofactors (including hydroxycholesterol, prostagrandins, phosphocholine and docosapentaenoate) than control young mice [44]. Part of these brain metabolic signatures may be directly influenced by the gut microbiota. In fact, older mice with signs of cognitive impairment had significantly higher brain levels of two metabolites related to microbial metabolism of the amino acid tryptophan, i.e., 3-indoxyl-sulfate and phenol sulfate, compared to young mice [44]. Interestingly, these metabolites have been associated with neurological toxicity and inflammation [89]. In patients who underwent bone marrow transplantation, the levels these molecules were predictive of clinical outcomes, reflecting the presence of gut microbiota dysbiosis [90].

The metabolism of tryptophan has also been associated with brain function in humans. Aging is associated with reduced serum levels of tryptophan and increased activation of the main human metabolic pathway of tryptophan degradation, i.e. the kynurenine pathway [91]. The serum levels of some kynurenine pathway intermediates, such as kynurenic acid and 3-hydroxykynurenine, have also shown a positive association with the level of cognitive impairment [91]. In a recent large population-based study, the kynurenine-tryptophan ratio was inversely associated with the performance at the Kendrick Object Learning Test and Controlled Oral Word Association Test [92]. These findings can be explained by a possible neurotoxic effect of kynurenine pathway intermediates.

Gut microbiota can contribute to tryptophan degradation with its own metabolic pathways. The most studied intermediate is indole-3 acetic acid (IAA), which is known as one of the main uremic toxins in patients suffering from chronic kidney failure [93]. Serum IAA levels were found to be negatively associated with Mini-Mental State Examination test score in patients undergoing chronic hemodialysis [94] and with depressive symptoms in subjects with severe chronic kidney disease not undergoing hemodialysis [95]. Another intermediate of microbial tryptophan degradation is indoxyl sulfate, which showed serum concentrations that were associated with poor executive functions in a group of 199 patients with early-stage chronic kidney disease [96]. Interestingly, indoxyl sulfate is able to induce apoptosis in human astrocytes, suggesting a possible involvement of this microbiota-derived metabolite in the pathophysiology of dementia [97].

Moreover, tryptophan is autonomously produced by bacteria through the shikimate pathway, a seven-step metabolic route allowing the synthesis of aromatic amino acids from two intermediates in carbohydrate metabolism, i.e., phosphoenolpyruvic acid and erythrose 4-phosphate. In a study performed on 20 patients with Alzheimer's disease and 5 healthy controls, Paley et al. were able to identify gut bacterial gene sequences unique to patients with Alzheimer's disease, belonging to the enzyme NADH:Ubiquinone reductase that is related to the optimal functioning of the shikimate pathway [98]. These findings suggest that the gut microbiota of patients with Alzheimer's disease may have an enhanced production of tryptophan through the shikimate pathway and that this phenomenon may have relevance in the physiopathology of dementia.

In summary, the current scientific literature does not provide sufficient data to recommend the adoption of bacterial products of amino acid metabolism as biomarkers of cognitive aging. However, the complex interplay between the gut microbiota and the host in tryptophan metabolism may have relevance in the field of cognitive impairment and should be further investigated in the future. Kynurenine, kynurenic acid, 3-hydroxykynurenine, IAA and indoxyl sulfate could represent good candidates as biomarkers of cognitive aging.

## 3.4 Microbiota-Derived SCFAs and Cognitive Aging

SCFAs, namely acetate, butyrate and propionate, are among the most studied metabolic products of the gut microbiota. They are derived from dietary carbohydrates and absorbed into the systemic circulation, where they can exert a wide range of physiologic functions, including control of body weight, regulation of insulin sensitivity, regulation of inflammation and immune system activation [99, 100]. These mechanisms may have particular relevance for central nervous system diseases, in which inflammation is also involved in the pathogenesis of Alzheimer's disease [101]. Butyrate has been implicated as a central mediator of the possible gut-brain axis, and the effects of the administration of butyrate or butyrate-producing bacteria on brain functions have been investigated through several animal studies, recently reviewed by Stilling and colleagues [102]. However, few of these studies were focused on dementia or cognitive impairment and, therefore, their clinical relevance remains uncertain.

In a study performed in mice, Fröhlich and colleagues demonstrated that the administration of antibiotics to induce a deep gut microbiota dysbiosis, was associated with cognitive dysfunction and lower colonic concentrations of SCFAs compared to administration of placebo [45]. Conversely, the administration of a symbiotic (*Enterococcus faecium* plus inulin) in a mouse model of cognitive aging was associated with an increase in the fecal concentration of butyrate and better cognitive performances than the placebo [103]. The administration of a butyrate-producing species, Clostridium butyricum, as a probiotic to mouse models of vascular dementia and cerebral ischemia/reperfusion injury resulted in improved cognitive function or reduced neurologic deficits [104, 105]. In another study, the administration of *Bifidobacterium breve* strain A1, a metabolically active strain able to produce acetate, to an Alzheimer's disease mouse model resulted in improvements in cognitive function or in a less pronounced cognitive decline [106].

These data globally indicate that SCFAs, and particularly butyrate, could play an important physiopathological role in cognitive impairment, and thus may represent good biomarkers of cognitive aging. Fecal samples of patients with Alzheimer's disease are generally abundant in butyrate-producing bacteria [107] but no comparison with healthy controls is currently available on this specific point. The role of SCFAs as biomarkers of cognition in older subjects should thus be more investigated in future studies. Unfortunately, the volatile nature of these compounds makes them difficult to detect in human samples, and analyses generally show a low degree of reproducibility [81].

## 3.5 Other Microbiota-Derived Compounds and Cognitive Aging

#### 3.5.1 Bile Acids

Bile acids are not a primary product of intestinal microbiota but their metabolism is influenced by the microbiome composition and functionality. In fact, dysbiosis can be associated with important alterations of bile acid metabolism, contributing to the pathogenesis of several diseases [108]. Tauroursodeoxycholic acid, a bile acid found in bears and subject to gut microbiota metabolism, showed neuroprotective properties especially in animal models of Huntington's disease [109]. These properties have also been recently demonstrated in mouse models of dementia, in which the administration of tauroursodeoxycolic acid was able to slow down amyloid precursor protein processing and amyloid- $\beta$  deposition [110, 111].

These data have raised speculations that other bile acids produced by humans and subject to gut microbiota metabolism, may have a role in the pathophysiology of dementia and could serve as biomarkers of cognitive aging. Olzarán and colleagues were able to identify a metabolic fingerprint of Alzheimer's disease by analyzing plasma samples of a large group of patients with mild cognitive impairment or dementia [112]. Interestingly, they found that the concentrations of one bile acid (deoxycolic acid) were independently associated with the presence of cognitive symptoms [112]. Another recent study showed that specific bile acid profiles in serum were associated with other well-established biomarkers of Alzheimer's disease normally found in the cerebrospinal fluid (amyloid  $\beta_{1-42}$  and p-tau<sub>181</sub>) and with imaging markers of brain atrophy [113], reinforcing a possible role of bile acids as microbiota-derived biomarkers of cognitive aging.

#### 3.5.2 TMAO

TMAO is a metabolic product derived from microbial metabolism of dietary choline and serum levels of this molecule have been associated with an increased risk of atherosclerosis in animal models and human beings [78]. Thus, it has been considered as a marker of risk of vascular dementia [78]. However, recent studies have highlighted that it may also have relevance in the pathogenesis of Alzheimer's disease. First, the administration of exogenous TMAO to mice genetically prone to dementia resulted in an increased number of senescent cells in hippocampus, reduced synaptic plasticity-related protein expression and reduced cognitive performance [114]. Moreover, in a large sample of older patients, either with Alzheimertype dementia, mild cognitive impairment or no cognitive symptoms, the levels of TMAO detected in cerebrospinal fluid were significantly different according to cognitive status and related with other biomarkers of Alzheimer's disease (i.e., phosphorylated tau, total tau and neurofilament light chain protein) [115]. Thus, the elevation of TMAO in cerebrospinal fluid may represent a promising biomarker of cognitive dysfunction.

#### 3.5.3 Polyphenols

Dietary polyphenols have been associated with reduced risk of a large number of cardiovascular disorders, based mainly on studies in vitro [116]. These compounds are generally metabolized by the gut microbiota into a large number of substances that may exert a protective function against neurodegeneration, and there is interest in these molecules as potential biomarkers of cognitive aging. For example, 3-hydroxybezoic acid and 3-(3'-hydroxyphenyl)propionic acid, two metabolites derived from microbiota degradation of anthocyanidins contained in grape seeds, accumulate in brain tissues of rats and exhibit the capacity of interfering with the assembly of  $\beta$ -amyloid into neurotoxic aggregates [117]. Polyphenol metabolites can also modulate cognitive resilience in mice, but this capacity is dependent on the gut microbiota composition and functionality, the so-called metabotype [118]. Therefore, the administration of polyphenols as food supplements or nutraceuticals may have different effects in different individuals, depending on the gut microbiota.

An example of this inter-individual variability of polyphenol metabolism, depending on gut microbiota composition, is the ellagitannins, a group of polyphenol compounds found mainly in pomegranates and walnuts [119, 120]. In human beings, two different gut microbiota metabotypes (A and B) have been identified, and each of these is able to produce different metabolites from ellagitannins, called urolithin-A and urolithin-B, respectively [119, 120]. Urolithin-A has been associated with reduced cardiometabolic risk factors and with beneficial health effects, unlike urolithin-B [119]. Thus, the gut microbiota urolithin-related metabotype may contribute to the observed variability of health benefits of pomegranate extracts found in humans [120].

A recent study performed in APP/PS1 mouse models of Alzheimer's disease has demonstrated that the administration of urolithin A ameliorated cognitive performance and positively influenced several pathophysiological mechanisms of dementia, including amyloid- $\beta$  deposition, neuronal apoptosis and neuroinflammation [121]. Urolithin A also seems to be able to enhance mitophagy, which is a mechanism protecting against cognitive impairment [122]. In a small human randomized trial, the administration of 8 ounces of pomegranate juice for 4 weeks to older patients reporting cognitive complaints resulted in a significant improvement in verbal memory tasks, which was paralleled by a significant increase in plasma concentrations of urolithin A-glucuronide [123]. These data underline the point that urolithin A and urolithin A-metabotype of gut microbiota may represent interesting biomarkers of cognitive aging viaidentification of a subset of patients that are susceptible to nutritional intervention.

#### 3.5.4 Vitamin K

Vitamin K metabolism and the administration of vitamin K antagonists as oral anticoagulants have been recently studied as possible risk factors for cognitive decline in older individuals [124, 125]. Some studies support a positive association between vitamin K levels and cognitive performance [79, 124], while the administration of vitamin K antagonists to rats resulted in altered cognition [126]. However, the effect of vitamin K antagonists on cognitive function may depend on the pathophysiological mechanism of dementia, as it may be protective in cerebrovascular forms of dementia [125] and probably detrimental in neurodegenerative forms [79, 124]. Patients with cognitive impairment who need the administration of vitamin K antagonists are more likely to have multimorbidity and more severe forms of cognitive dysfunction at baseline, so that the putative association between vitamin K antagonists and progression to dementia may be spurious [127]. In this scenario, the intestinal microbiome could play an important role, since vitamin K is physiologically produced by intestinal bacteria [128]. However, no study has investigated the correlation between production of vitamin K by the gut microbiota and cognitive outcomes to date. Vitamin K could however represent another promising biomarker of cognitive aging meriting future investigations.

## 4 Conclusions

Gut microbiota composition and microbiota-derived metabolites or substances represent a promising area of research for the identification of novel biomarkers of cognitive aging. However, the current literature state-of-the-art does not support the implementation of microbiome-related biomarkers of cognitive aging for use in clinical practice. The study of the relationship between gut microbiota composition and functionality in mild cognitive impairment and different types of dementia should be translated from animal models to patients. In addition, these studies should be focused on alpha diversity, Firmicutes/Bacteroidetes ratio, abundance of specific key taxa and microbial metabolism of amino acids, SCFAs, bile acids, vitamin K and nutrients such as polyphenols and choline.

The use of new microbiome-related biomarkers of cognitive aging could bring to several advantages. If these biomarkers show alterations in the early phases of dementia, they could assist physicians in the diagnostic process, or even lead to early diagnosis. They could also help identifying those patients with a quicker evolution towards dementia, deserving more aggressive treatments and strict follow-up. From the patient's perspective, it could bring improved management of cognitive diseases and give important information on neurological diseases, facilitating a comprehensive classification of health status. Finally, it could also have economic relevance, since the microbiome analyses are generally less expensive than brain imaging examinations. For all these reasons, more research is urgently needed in this field.

Acknowledgements All authors have no conflict of interest to declare. No funding is reported for this manuscript.

## References

- Schmidt TSB, Raes J, Bork P (2018) The human gut microbiome: from association to modulation. Cell 172(6):1198–1215
- Gentile CL, Weir TL (2018) The gut microbiota at the intersection of diet and human health. Science 362:776–780
- Ticinesi A, Nouvenne A, Tana C, Prati B, Cerundolo N, Miraglia C et al (2018) The impact of intestinal microbiota on bio-medical research: definitions, techniques and physiology of a "new frontier". Acta Biomed 89(9S):52–59

- Milosevic I, Vujovic A, Barac A, Djelic M, Korac M, Radovanovic Spurnic A et al (2019) Gut-liver axis, gut microbiota, and its modulation in the management of liver disease: a review of the literature. Int J Mol Sci 20(2):395. pii: E395. https://doi.org/10.3390/ijms20020395
- Cosola C, Rocchetti MT, Sabatino A, Fiaccadori E, Di Iorio BR, Gesualdo L (2019) Microbiota issue in CKD: how promising are gut-targeted approaches? J Nephrol 32(1):27–37
- Ticinesi A, Milani C, Guerra A, Allegri F, Lauretani F, Nouvenne A et al (2018) Understanding the gut-kidney axis in nephrolithiasis: an analysis of the gut microbiota composition and functionality of stone formers. Gut 67(12):2097–2106
- 7. Quach D, Britton RA (2017) Gut microbiota and bone health. Adv Exp Med Biol 1033:47-58
- Ticinesi A, Lauretani F, Milani C, Nouvenne A, Tana C, Del Rio D et al (2017) Aging gut microbiota at the cross-road between nutrition, physical frailty, and sarcopenia: is there a gut-muscle axis? Nutrients 9(12):E1303. pii: E1303. https://doi.org/10.3390/nu9121303
- Junges VM, Closs VE, Nogueira GM, Valle Gottlieb MG (2018) Crosstalk between gut microbiota and the central nervous system: a focus for Alzheimer's disease. Curr Alzheimer Res 15:1–12.43
- Ticinesi A, Tana C, Nouvenne A, Prati B, Lauretani F, Meschi T (2018) Gut microbiota, cognitive frailty and dementia in older individuals: a systematic review. Clin Interv Aging 13:1497–1511
- Kowalski K, Mulak A (2019) Brain-gut-microbiota axis in Alzheimer's disease. J Neurogastroenterol Motil 25(1):48–60
- 12. Cattaneo A, Cattane N, Galluzzi S, Provasi S, Lopizzo N, Festari C et al (2017) Association of brain amyloidosis with pro-inflammatory gut bacterial taxa and peripheral inflammation markers in cognitively impaired elderly. Neurobiol Aging 49:60–68
- Vogt NM, Kerby RL, Dill-McFarland KA, Harding SJ, Merluzzi AP, Johnson SC et al (2017) Gut microbiome alterations in Alzheimer's disease. Sci Rep 7(1):13537. https://doi. org/10.1038/s41598-017-13601-y
- 14. Manderino L, Carroll I, Azcarate-Peril MA, Rochette A, Heinberg L, Peat C et al (2017) Preliminary evidence for an association between the composition of gut microbiome and cognitive function in neurologically-healthy older adults. J Int Neuropsychol Soc 23(8):700–705
- 15. Zhuang ZQ, Shen LL, Li WW, Fu X, Zeng F, Gui L et al (2018) Gut microbiota is altered in patients with Alzheimer's disease. J Alzheimers Dis 63(4):1337–1346
- 16. Saji N, Niida S, Murotani K, Hisada T, Tsuduki T, Sugimoto T et al (2019) Analysis of the relationship between the gut microbiome and dementia: a cross-sectional study conducted in Japan. Sci Rep 9:1008. https://doi.org/10.1038/s41598-018-38218-7
- Desmedt O, Broers VJV, Zamariola G, Pachikian B, Delzenne N, Luminet O (2019) Effects of prebiotics on affect and cognition in human intervention studies. Nutr Rev 77(2):81–95
- Leblhuber F, Steiner K, Schuetz B, Fuchs D, Gostner JM (2018) Probiotic supplementation in patients with Alzheimer's dementia – An explorative intervention study. Curr Alzheimer Res 15:1106–1113
- Tran TTT, Cousin FJ, Lynch DB, Menon R, Brulc J, Brown JRM et al (2019) Prebiotic supplementation in frail older people affects specific gut microbiota taxa but not global diversity. Microbiome 7:39. https://doi.org/10.1186/s40168-019-0654-1
- Agahi A, Hamidi GA, Daneshvar R, Hamdieh M, Soheili M, Alinaghipour A et al (2018) Does severity of Alzheimer's disease contribute to its responsiveness to modifying gut microbiota? A double blind clinical trial. Front Neurol 9:662. https://doi.org/10.3389/fneur.2018.00662
- Inoue T, Kobayashi Y, Mori N, Sakagawa M, Xiao JZ, Moritani T et al (2018) Effect of combined bifidobacteria supplementation and resistance training on cognitive function, body composition and bowel habits of healthy elderly subjects. Benef Microbes 9(6):843–853
- Ticinesi A, Lauretani F, Tana C, Nouvenne A, Ridolo E, Meschi T (2019) Exercise and immune system as modulators of intestinal microbiome: implications for the gut-muscle axis hypothesis. Exerc Immunol Rev 25:84–95
- Ticinesi A, Milani C, Lauretani F, Nouvenne A, Mancabelli L, Lugli GA et al (2017) Gut microbiota composition is associated with polypharmacy in elderly hospitalized patients. Sci Rep 7(1):11102. https://doi.org/10.1038/s41598-017-10734-y

- Jackson MA, Verdi S, Maxan ME, Shin CM, Zierer J, Bowyer RCE et al (2018) Gut microbiota associations with common diseases and prescription medications in a population-based cohort. Nat Commun 9(1):2655. https://doi.org/10.1038/s41467-018-05184-7
- Davenport ER, Mizrahi-Mian O, Michelini K, Barreiro LB, Ober C, Gilad Y (2014) Seasonal variation in human gut microbiome composition. PLoS One 9(3):e90731. https://doi. org/10.1371/journal.pone.0090731
- 26. O'Toole PW, Jeffery IB (2015) Gut microbiome and aging. Science 350(6265):1214–1215
- Vandeputte D, Falony G, Vieiera-Silva S, Tito RY, Joossens M, Raes J (2016) Stool consistency is strongly associated with gut microbiota richness and composition, enterotypes and bacterial growth rates. Gut 65(1):57–62
- Vandeputte D, Kathagen G, D'hoe K, Vieira-Silva S, Valles-Colomer M, Sabino J et al (2017) Quantitative microbiome profiling links gut community variation to microbial load. Nature 551(7681):507–511
- Zhernakova A, Kurilshikov A, Bonder MJ, Tigchelaar EF, Schirmer M, Vatanen T et al (2016) Population-based metagenomics analysis reveals markers for gut microbiome composition and diversity. Science 352(6285):565–569
- 30. Falony G, Vandeputte D, Caenepeel C, Vieira-Silva S, Daryoush T, Vermeire S et al (2019) The human microbiome in health and disease: hype or hope. Acta Clin Belg 74(2):53–64
- Ticinesi A, Tana C, Nouvenne A (2019) The intestinal microbiome and its relevance for functionality in older persons. Curr Opin Clin Nutr Metab Care 22(1):4–12
- 32. Mancuso C, Santangelo R (2018) Alzheimer's disease and gut microbiota modifications: the long way between preclinical studies and clinical evidence. Pharm Res 129:329–336
- Lozupone M, La Montagna M, D'Urso F, Daniele A, Greco A, Seripa D et al (2019) The role of biomarkers in psychiatry. Adv Exp Med Biol 1118:135–162
- Ruan Q, D'Onofrio G, Sancarlo D, Greco A, Lozupone M, Seripa D et al (2017) Emerging biomarkers and screening for cognitive frailty. Aging Clin Exp Res 29:1075–1086
- Ventura M, Turroni F, Canchaya C, Vaughan EE, O'Toole PW, Van Sinderen D (2009) Microbial diversity in the human intestine and novel insights from metagenomics. Front Biosci 14:3214–3221
- Milani C, Hevia A, Foroni E, Duranti S, Turroni F, Lugli GA et al (2013) Assessing the fecal microbiota: an optimized ion torrent 16S rRNA gene-based analysis protocol. PLoS One 8(7):e68739. https://doi.org/10.1371/journal.pone.0068739
- Kim BR, Shin J, Guevarra RB, Lee JH, Kim DW, Seol KH et al (2017) Deciphering diversity indices for a better understanding of microbial communities. J Microbiol Biotechnol 27(12):2089–2093
- Schloss PD, Handelsman J (2006) Introducing SONS, a tool for operational taxonomic unit-based comparisons of microbial community memberships and structures. Appl Environ Microbiol 75:7537–7541
- 39. Ticinesi A, Milani C, Lauretani F, Nouvenne A, Tana C, Ventura M et al (2019) Gut microbiome in the elderly hospitalized patient: a marker of disease and prognosis? In: Faintuch J, Faintuch S (eds) Microbiome and metabolome in diagnosis, therapy, and other strategic applications. Associated Press, London, pp 287–296. isbn:978-0-12-815249-2
- 40. Milani C, Ticinesi A, Gerritsen J, Nouvenne A, Lugli GA, Mancabelli L et al (2016) Gut microbiota composition and Clostridium difficile infection in hospitalized elderly individuals: a metagenomic study. Sci Rep 6:25945. https://doi.org/10.1038/srep2594
- Ianiro G, Tilg H, Gasbarrini A (2016) Antibiotics as deep modulators of gut microbiota: between good and evil. Gut 65(11):1906–1911
- 42. Perez-Cobas AE, Gosalbes MJ, Friedrichs A, Knecht H, Artacho A, Eismann K et al (2013) Gut microbiota disturbance during antibiotic therapy: a multi-omic approach. Gut 62(11):1591–1601
- 43. Zhan G, Yang N, Li S, Huang N, Fang X, Zhang J et al (2018) Abnormal gut microbiota composition contributes to cognitive dysfunction in SAMP8 mice. Aging 10(6):1257–1267
- 44. Hoffman JD, Parikh I, Green SJ, Chlipala G, Mohney RP, Keaton M et al (2017) Age drives distortion of brain metabolic, vascular and cognitive functions, and the gut microbiome. Front Aging Neurosci 9:298. https://doi.org/10.3389/fnagi.2017.00298

- 45. Fröhlich EE, Farzi A, Mayerhofer R, Reichmann F, Jačan A, Wagner B et al (2016) Cognitive impairment by antibiotic-induced gut dysbiosis: analysis of gut microbiota-brain communication. Brain Behav Immun 56:140–155
- 46. Minter MR, Zhang C, Leone V, Ringus DL, Zhang X, Oyler-Castillo P et al (2016) Antibioticinduced perturbations in gut microbial diversity influences neuro-inflammation and amyloidosis in a murine model of Alzheimer's disease. Sci Rep 6:30028. https://doi.org/10.1038/ srep30028
- 47. Minter MR, Hinterleitner R, Meisel M, Zhang C, Leone V, Zhang X et al (2017) Antibioticinduced perturbations in microbial diversity during post-natal development alters amyloid pathology in an aged APP<sub>SWE</sub>/PS<sub>ΔE9</sub> murine model of Alzheimer's disease. Sci Rep 7:10411. https://doi.org/10.1038/s41598-017-11047-w
- Harach T, Marungruang N, Duthilleul N, Cheatham V, McCoy KD, Frisoni G et al (2017) Reduction of Abeta amyloid pathology in APPPS1 transgenic mice in the absence of gut microbiota. Sci Rep 7:41802. https://doi.org/10.1038/srep41802
- 49. Bajaj JS, Fagan A, Sikaroodi M, White MB, Sterling RK, Gilles H et al (2017) Liver transplant modulates gut microbial dysbiosis and cognitive function in cirrhosis. Liver Transpl 23:907–914
- Bajaj JS, Vargas HE, Reddy KR, Lai JC, O'Leary JG, Tandon P et al (2019) Association between intestinal microbiota collected at hospital admission and outcomes of patients with cirrhosis. Clin Gastroenterol Hepatol 17:756–765
- 51. Fernandez-Real JM, Serino M, Blasco G, Puig J, Daunis-i-Estadella J, Ricart W et al (2015) Gut microbiota interacts with brain microstructure and function. J Clin Endocrinol Metab 100(12):4505–4513
- 52. Zhang F, Yang J, Ji Y, Sun M, Shen J, Sun J et al (2019) Gut microbiota dysbiosis is not independently associated with neurocognitive impairment in people living with HIV. Front Microbiol 9:3352. https://doi.org/10.3389/fmicb.2018.03352
- 53. Verdi S, Jackson MA, Beaumont M, Bowyer RCE, Bell JT, Spector TD et al (2018) An investigation into physical frailty as a link between the gut microbiome and cognitive health. Front Aging Neurosci 10:398. https://doi.org/10.3389/fnagi.2018.00398
- 54. Araos R, Andreatos N, Ugalde J, Mitchell S, Mylonakis E, D'Agata EMC (2018) Fecal microbiome among nursing home residents with advanced dementia and *Clostridium difficile*. Dig Dis Sci 63(6):1525–1531
- 55. Costea PI, Hildebrand F, Arumugam M, Bäckhed F, Blaser MJ, Bushman FD et al (2018) Enterotypes in the landscape of gut microbial community composition. Nat Microbiol 3(1):8–16
- Mariat D, Firmesse O, Levenez F, Guimarâes V, Sokol H, Doré J et al (2009) The Firmicutes/ Bacteroidetes ratio of the human microbiota changes with age. BMC Microbiol 9:123. https:// doi.org/10.1186/1471-2180-9-123
- 57. Sze MA, Schloss PD (2016) Looking for a signal in the noise: revisiting obesity and the microbiome. mBio 7(4):e01018–e01016. https://doi.org/10.1128/mBio.01018-16
- Indiani CMDSP, Rizzardi KF, Castelo PM, Ferraz LFC, Darrieux M, Parisotto TM (2018) Childhood obesity and Firmicutes/Bacteroidetes ratio in the gut microbiota: a systematic review. Child Obes 14(8):501–509
- 59. Koliada A, Syzenko G, Moseiko V, Budovska L, Puchkov K, Perederiy V et al (2017) Association between body mass index and Firmicutes/Bacteroidetes ratio in an adult Ukranian population. BMC Microbiol 17(1):120. https://doi.org/10.1186/s12866-017-1027-1
- 60. Sanguinetti E, Collado MC, Marrachelli VG, Monleon D, Selma-Royo M, Pardo-Tendero MM et al (2018) Microbiome-metabolome signatures in mice genetically prone to develop dementia, fed a normal or fatty diet. Sci Rep 8:4907. https://doi.org/10.1038/ s41598-018-23261-1
- Zhang P, Yu Y, Qin Y, Zhou Y, Tang R, Wang Q et al (2019) Alterations to the microbiotacolon-brain axis in high-fat-diet-induced obese mice compared to diet-resistant mice. J Nutr Biochem 65:54–65

- Bäuerl C, Collado MC, Diaz Cuevas A, Viña J, Pérez Martìnez G (2018) Shifts in gut microbiota composition in APP/PSS1 transgenic mouse model of Alzheimer's disease during lifespan. Lett Appl Microbiol 66(6):464–471
- 63. Bonfili L, Cecarini V, Berardi S, Scarpona S, Suchodolski JS, Nasuti C et al (2017) Microbiota modulation counteracts Alzheimer's disease progression influencing neuronal proteolysis and gut hormones plasma levels. Sci Rep 7:2426. https://doi.org/10.1038/s41598-017-02587-2
- 64. Shen L, Liu L, Ji HF (2017) Alzheimer's disease histological and behavioral manifestations in transgenic mice correlate with specific gut microbiome state. J Alzheimers Dis 56(1):385–390
- 65. Tran TTT, Corsini S, Kellingray L, Hegarty C, Le Gall G, Narbad A et al (2019) APOE genotype influences the gut microbiome structure and function in humans and mice: relevance for Alzheimer's disease pathophysiology. FASEB J 33:8221–8231
- 66. Blasco G, Moreno-Navarrete JM, Rivero M, Pérez-Brocal V, Garre-Olmo J, Puig J et al (2017) The gut metagenome changes in parallel to waist circumference, brain iron deposition, and cognitive function. J Clin Endocrinol Metab 102:2962–2973
- Bajaj JS, Ridlon JM, Hylemon PB, Thacker LR, Heuman DM, Smith S et al (2012) Linkage of gut microbiome with cognition in hepatic encephalopathy. Am J Physiol Gastrointest Liver Physiol 302:G168–G175
- 68. Bajaj JS, Ahluwalia V, Steinberg JL, Hobgood S, Boling PA, Godschalk M et al (2016) Elderly patients have an altered gut-brain axis regardless of the presence of cirrhosis. Sci Rep 6:38481. https://doi.org/10.1038/srep38481
- 69. Bajaj JS, Fagan A, White MB, Wade JB, Hylemon PB, Heuman DM et al (2019) Specific gut and salivary microbiota patterns are linked with different cognitive testing strategies in minimal hepatic encephalopathy. Am J Gastroenterol 114:1080–1090
- 70. Qian Y, Yang Y, Xu S, Wu C, Song Y, Qin N et al (2018) Alteration of the fecal microbiota in Chinese patients with Parkinson's disease. Brain Behav Immun 70:194–202
- Barichella M, Severgnini M, Cilia R, Cassani E, Bolliri C, Caronni S et al (2019) Unraveling gut microbiota in Parkinson's disease and atypical parkinsonism. Mov Disord 34(3):396–405
- 72. Franceschi F, Ojetti V, Candelli M, Covino M, Cardone S, Potenza A et al (2019) Microbes and Alzheimer's disease: lessons from *H. pylori* and GUT microbiota. Eur Rev Med Pharmacol Sci 23:426–430
- Alkasir R, Li J, Li X, Jin M, Zhu B (2017) Human gut microbiota: the links with dementia development. Protein Cell 8(2):90–102
- 74. Giau VV, Wu SY, Jamerlan A, An SSA, Kim SY, Hulme J (2018) Gut microbiota and their neuroinflammatory implications in Alzheimer's disease. Nutrients 10:1765. https://doi. org/10.3390/nu10111765
- 75. Sanz Y, Romani-Perez M, Benitez-Pàez A, Portune KJ, Brigidi P, Rampelli S et al (2018) Towards microbiome-informed dietary recommendations for promoting metabolic and mental health: opinion papers of the MyNewGut project. Clin Nutr 37:2191–2197
- Westfall S, Lomis N, Kahouli I, Dia SY, Singh SP, Prakash S (2017) Microbiome, probiotics and neurodegenerative diseases: deciphering the gut-brain axis. Cell Mol Life Sci 74:3769–3787
- 77. Friedland RP, Chapman MR (2017) The role of microbial amyloid in neurodegeneration. PLoS Pathog 13(12):e1006654. https://doi.org/10.1371/journal.ppat.1006654
- Li S, Shao Y, Li K, HuangFu C, Wang W, Liu Z et al (2018) Vascular cognitive impairment and the gut microbiota. J Alzheimers Dis 63(4):1209–1222
- 79. Kiely A, Ferland G, Ouliass B, O'Toole PW, Purtill H, O'Connor EM (2018) Vitamin K status and inflammation are associated with cognition in older Irish adults. Nutr Neurosci 1–9
- Quigley EMM (2016) Leaky gut concept or clinical entity? Curr Opin Gastroenterol 32(2):74–79
- 81. Liebisch G, Ecker J, Roth S, Schweizer S, Öttl V, Schött HF et al (2019) Quantification of fecal short-chain fatty acids by liquid chromatography tandem mass spectrometry-investigation of pre-analytic stability. Biomol Ther 9(4):E121. https://doi.org/10.3390/biom9040121

- Belizário JE, Faintuch J, Garay-Malpartida M (2018) Gut microbiome dysbiosis and immunometabolism: new frontiers for treatment of metabolic diseases. Mediat Inflamm 2018:2037838–2037812. https://doi.org/10.1155/2018/2037838
- 83. López P, Sánchez M, Perez-Cruz C, Velázquez-Villegas LA, Syeda T, Aguilar-López M et al (2018) Long-term genistein consumption modifies gut microbiota, improving glucose metabolism, metabolic endotoxinemia, and cognitive function in mice fed a high-fat diet. Mol Nutr Food Res 62:1800313
- 84. Hayashi K, Hasegawa Y, Takemoto Y, Cao C, Takeya H, Komohara Y et al (2019) Continuous intracerebroventricular injection of Porphyromonas gingivalis lipopolysaccharide induces systemic organ dysfunction in a mouse model of Alzheimer's disease. Exp Gerontol 120:1–5
- Zhao Y, Cong L, Jaber V, Lukiw WJ (2017) Microbiome-derived lipopolysaccharide enriched in the perinuclear region of Alzheimer's disease brain. Front Immunol 8:1064. https://doi. org/10.3389/fimmu.2017.01064
- Zhan X, Stamova B, Jin LW, DeCarli C, Phinney B, Sharp FR (2016) Gram-negative bacterial molecules associate with Alzheimer's disease pathology. Neurology 87(22):2324–2332
- Zhao Y, Jaber V, Lukiw WJ (2017) Secretory products of the human GI tract microbiome and their potential impact on Alzheimer's disease (AD): detection of lipopolysaccharide (LPS) in AD hippocampus. Front Cell Infect Microbiol 7:318. https://doi.org/10.3389/ fcimb.2017.00318
- Andreadou E, Pantazaki AA, Daniilidou M, Tsolaki M (2017) Rhamnolipids, microbial virulence factors, in Alzheimer's disease. J Alzheimers Dis 59(1):209–222
- Zgoda-Pols JR, Chowdhury S, Wirth M, Milburn MV, Alexander DC, Alton KB (2011) Metabolomics analysis reveals elevation of 3-indoxyl sulfate in plasma and brain during chemically-induced acute kidney injury in mice: investigation of nicotinic acid receptor agonists. Toxicol Appl Pharmacol 255:48–56
- Weber D, Oefner PJ, Hiergeist A, Koestler J, Gessner A, Weber M et al (2015) Low urinary indoxyl sulfate levels early after transplantation reflect a disrupted microbiome and are associated with poor outcome. Blood 126:1723–1728
- 91. Ramos-Chávez LA, Roldán-Roldán G, Garcia-Juárez B, González-Esquivel D, Pérez de la Cruz G, Pineda B et al (2018) Low serum tryptophan levels as an indicator of global cognitive performance in nondemented women over 50 years old. Oxidative Med Cell Longev 2018:8604718. https://doi.org/10.1155/2018/8604718
- 92. Hafstad-Solvang SE, Nordrehaug JE, Tell GS, Nygård O, McCann A, Ueland PM et al (2019) The kynurenine pathway and cognitive performance in community-dwelling older adults. The Hordaland Health Study. Brain Behav Immun 75:155–162
- Castillo-Rodriguez E, Fernandez-Prado R, Esteras R, Perez-Gomez MV, Gracia-Iguacel C, Fernandez-Fernandez B et al (2018) Impact of altered intestinal microbiota on chronic kidney disease progression. Toxins (Basel) 10(7):E300. https://doi.org/10.3390/toxins10070300
- 94. Lin YT, Wu PH, Lee HH, Mubanga M, Chen CS, Kuo MC et al (2019) Indole-3 acetic acid increased risk of impaired cognitive function in patients receiving hemodialysis. Neurotoxicology 73:85–91
- 95. Karu N, McKercher C, Nichols DS, Davies N, Shellie RA, Hilder EF et al (2016) Tryptophan metabolism, its relation to inflammation and stress markers and association with psychological and cognitive functioning: Tasmanian Chronic Kidney Disease pilot study. BMC Nephrol 17:171. https://doi.org/10.1186/s12882-016-0387-3
- 96. Yeh YC, Huang MF, Liang SS, Hwang SJ, Tsai JC, Liu TL et al (2016) Indoxyl sulfate, not p-cresyl sulfate, is associated with cognitive impairment in early-stage chronic kidney disease. Neurotoxicology 53:148–152
- 97. Lin YT, Wu PH, Tsai YC, Hsu YL, Wang HY, Kuo MC et al (2019) Indoxyl sulfate induces apoptosis through oxidative stress and mitogen-activated protein kinase signalinig pathway inhibition in human astrocytes. J Clin Med 8(2):E191. https://doi.org/10.3390/jcm8020191
- 98. Paley EL, Merkulova-Rainon T, Faynboym A, Shestopalov VI, Aksenoff I (2018) Geographical distribution and diversity of gut microbial NADH-Ubiquinone oxidoreductase sequence associated with Alzheimer's disease. J Alzheimers Dis 61:1531–1540

- Canfora EE, Jocken JW, Blaak EE (2015) Short-chain fatty acids in control of body weight and insulin sensitivity. Nat Rev Endocrinol 11(10):577–591
- 100. Levy M, Kolodziejczyk AA, Thaiss CA, Elinav E (2017) Dysbiosis and the immune system. Nat Rev Immunol 17(4):219–232
- 101. Russo R, Cristiano C, Avagliano C, De Caro C, La Rana G, Mattace Raso G et al (2017) Gutbrain axis: role of lipids in the regulation of inflammation, pain and CNS diseases. Curr Med Chem 24:1–22
- 102. Stilling RM, van de Vouw M, Clarke G, Stanton C, Dinan TG, Cryan JF (2016) The neuropharmacology of butyrate: the bread and butter of the microbiota-gut-brain axis? Neurochem Int 99:110–132
- 103. Romo-Araiza A, Gutiérrez-Salmeán G, Galván EJ, Hernández-Frausto M, Herrera-López G, Romo-Parra H et al (2018) Probiotics and prebiotics as a therapeutic strategy to improve memory in a model of middle-aged rats. Front Aging Neurosci 10:416. https://doi.org/10.3389/ fnagi.2018.00416
- 104. Liu J, Sun J, Wang F, Yu X, Ling Z, Li H et al (2015) Neuroprotective effects of *Clostridium butyricum* against vascular dementia in mice via metabolic butyrate. Biomed Res Int 2015:412946. https://doi.org/10.1155/2015/412946
- 105. Sun J, Wang F, Ling Z, Yu X, Chen W, Li H et al (2016) *Clostridium butyricum* attenuates cerebral ischemia/reperfusion injury in diabetic mice via modulation of gut microbiota. Brain Res 1642:180–188
- 106. Kobayashi Y, Sugahara H, Shimada K, Mitsuyama E, Kuhara T, Yasuoka A et al (2017) Therapeutic potential of *Bifidobacterium breve* strain A1 for preventing cognitive impairment in Alzheimer's diseases. Sci Rep 7:13510. https://doi.org/10.1038/s41598-017-13368-2
- 107. Nguyen TTT, Fujimura Y, Mimura I, Fujii Y, Nguyen NL, Arakawa N et al (2018) Cultivable butyrate-producing bacteria of elderly Japanese diagnosed with Alzheimer's disease. J Microbiol 56(10):760–771
- Staley C, Weingarden AR, Khoruts A, Sadowsky MJ (2017) Interaction of gut microbiota with bile acid metabolism and its influence on diseases states. Appl Microbiol Biotechnol 101(7):47–64
- 109. Keene CD, Rodrigues CM, Eich T, Chhabra MS, Steer CJ, Low WC (2002) Tauroursodeoxycholic acid, a bile acid, is neuroprotective in a transgenic animal model of Huntington's disease. Proc Natl Acad Sci U S A 99(16):10671–10676
- 110. Nunes AF, Amaral JD, Lo AC, Fonseca MB, Viana RJ, Callaerts-Vegh Z et al (2012) TUDCA, a bile acid, attenuates amyloid precursor protein processing and amyloid- $\beta$  deposition in APP/PS1 mice. Mol Neurobiol 45(3):440–454
- 111. Dionisio PA, Amaral JD, Ribeiro MF, Lo AC, D'Hooge R, Rodrigues CM (2015) Amyloid-β pathology is attenuated by tauroursodeoxycholic acid treatment in APP/PS1 mice after disease onset. Neurobiol Aging 36(1):228–240
- 112. Olzarán J, Gil-de-Gómez L, Rodríguez-Martín A, Valentí-Soler M, Frades-Payo B, Marín-Muñoz J et al (2015) A blood-based, 7-metabolite signature for early diagnosis of Alzheimer's disease. J Alzheimers Dis 45(4):1157–1173
- 113. Nho K, Kueider-Paisley A, MahmoudianDehkordi S, Arnold M, Risacher SL, Louie G et al (2019) Altered bile acid profile in mild cognitive impairment and Alzheimer's disease: relationship to neuroimaging and CSF biomarkers. Alzheimers Dement 15:232–244
- 114. Li D, Ke Y, Zhan R, Liu C, Zhao M, Zeng A et al (2018) Trimethylamine-N-oxide promotes brain aging and cognitive impairment in mice. Aging Cell 10:e12768. https://doi.org/10.1111/ acel.12768. [Epub ahead of print]
- 115. Vogt NM, Romano KA, Darst BF, Engelman CD, Johnson SC, Carlsson CM et al (2018) The gut microbiota-derived metabolite trimethylamine N-oxide is elevated in Alzheimer's disease. Alzheimers Res Ther 10:124. https://doi.org/10.1186/s13195-018-0451-2
- 116. Poti F, Santi D, Spaggiari G, Zimetti F, Zanotti I (2019) Polyphenol health effects on cardiovascular and neurodegenerative disorders: a review and meta-analysis. Int J Mol Sci 20:351. https://doi.org/10.3390/ijms20020351

- 117. Wang D, Ho L, Faith J, Ono K, Janle EM, Lachcik PJ et al (2015) Role of intestinal microbiota in the generation of polyphenol-derived phenolic acid mediated attenuation of Alzheimer's disease β-amyloid oligomerization. Mol Nutr Food Res 59:1025–1040
- 118. Frolinger T, Sims S, Smith C, Wang J, Cheng H, Faith J et al (2019) The gut microbiota composition affects dietary polyphenols-mediated cognitive resilience in mice by modulating the bioavailability of phenolic acids. Sci Rep 9:3546. https://doi.org/10.1038/ s41598-019-39994-6
- 119. Selma MV, González-Sarrías A, Salais-Salvadó J, Andrés-Lacueva C, Alasalvar C, Örem A et al (2018) The gut microbiota metabolism of pomegranate or walnut ellagitannins yields two urolithin-metabotypes that correlate with cardiometabolic risk biomarkers: comparison between normoweight, overweight-obesity and metabolic syndrome. Clin Nutr 37(3):897–905
- 120. González-Sarrías A, García-Villalba R, Romo-Vaquero M, Alasalvar C, Örem A, Zafrilla P et al (2017) Clustering according to urolithin metabotype explains the interindividual variability in the improvement of cardiovascular risk biomarkers in overweight-obese individuals consuming pomegranate: a randomized clinical trial. Mol Nutr Food Res 61(5):1600830. https://doi.org/10.1002/mnfr.201600830
- 121. Gong Z, Huang J, Xu B, Ou Z, Zhang L, Lin X et al (2019) Urolithin A attenuates memory impairment and neuroinflammation in APP/PS1 mice. J Neuroinflammation 16(1):62. https:// doi.org/10.1186/s12974-019-1450-3
- 122. Fang EF, Hou Y, Palikaras K, Adriaanse BA, Kerr JS, Yang B et al (2019) Mitophagy inhibits amyloid-β and tau pathology and reverses cognitive deficits in models of Alzheimer's disease. Nat Neurosci 22(3):401–412
- 123. Bookheimer SY, Renner BA, Ekstrom A, Li Z, Henning SM, Brown JA et al (2013) Pomegranate juice augments memory and FMRI activity in middle-aged and older adults with mild memory complaints. Evid Based Complement Alternat Med 2013:946298. https:// doi.org/10.1155/2013/946298
- 124. Alisi L, Cao R, De Angelis C, Cafolla A, Caramia F, Cartocci G et al (2019) The relationship between vitamin K and cognition: a review of current evidence. Front Neurosci 10:239. https://doi.org/10.3389/fneur.2019.00239
- 125. Mongkhon P, Naser AY, Fanning L, Tse G, Lau WCY, Wong ICK et al (2019) Oral anticoagulants and risk of dementia: a systematic review and meta-analysis of observational studies and randomized controlled trials. Neurosci Biobehav Rev 96:1–9
- 126. Tamadon-Nejad S, Ouliass B, Rochford J, Ferland G (2018) Vitamin K deficiency induced by warfarin is associated with cognitive and behavioral perturbations, and alterations in brain sphingolipids in rats. Front Aging Neurosci 10:213. https://doi.org/10.3389/fnagi.2018.00213
- 127. Brangier A, Ferland G, Rolland Y, Gautier J, Féart C, Annweiler C (2018) Vitamin K antagonists and cognitive decline in older adults: a 24-month follow-up. Nutrients 10:666. https:// doi.org/10.3390/nu10060666
- 128. LeBlanc JG, Milani C, de Giori GS, Sesma F, van Sinderen D, Ventura M (2013) Bacteria as vitamin suppliers to their host: a gut microbiota perspective. Curr Opin Biotechnol 24(2):160–168



## Chapter 9 The Impact of Herbal Products in the Prevention, Regeneration and Delay of Skin Aging

Mega Ferdina Warsito and Idha Kusumawati

## 1 Introduction

The skin is the outermost and the largest organ of the human body, representing one sixth of the total body weight [1]. Skin has a role as a physical barrier against harmful microorganisms, toxic substances and ultraviolet radiation, and it regulates water loss and body temperature [2, 3]. Changes in the skin will affect an individual not only physiologically but also psychologically, which are both significant for quality of life. Along with aging, the skin like all human organs changes progressively. This natural change is called chronological aging. However, as the outermost organ and as a body barrier, the skin will also be subject to exposure to stressors and dangers from the environment outside the body. Changes in the skin due to external factors are referred to as premature aging, and because they are often caused by Ultra Violet radiation (UVR), this is also called photoaging [4–6].

Skin reflects the intrinsic and extrinsic aging process in the human body, which further causes changes in its function and structure [7]. The dermatological concern correlated with skin aging has grown as the aging population in the world has increased. Skin aging not only affects appearance but it can also influence an individual's social behavior and reproductive status [2]. Thus, some people pay a considerable amount of expense for cosmetics and pharmaceuticals that could prevent, regenerate and delay skin aging [8].

M. F. Warsito

I. Kusumawati (🖂)

© Springer Nature Switzerland AG 2019

Research Center for Biotechnology, Indonesian Institute of Sciences (LIPI), Cibinong, West Java, Indonesia

Pharmacognosy and Phytochemistry Department, Faculty of Pharmacy, Airlangga University, Surabaya, East Java, Indonesia e-mail: idha-k@ff.unair.ac.id

P. C. Guest (ed.), *Reviews on Biomarker Studies in Aging and Anti-Aging Research*, Advances in Experimental Medicine and Biology 1178, https://doi.org/10.1007/978-3-030-25650-0\_9

Aging phenomena are known but the aging process is not clearly understood. Aging is associated with a decrease in function and character of physiological, physical and appearance of the body [9–11]. In 1956, Harman developed the theory of free radicals (reactive oxygen species; ROS) as a cause of aging. An imbalance in the amount of exposure and elimination of ROS by antioxidants will affect many physiological processes in the body. Environmental insults to the skin such as those caused by UVR and pollutants can cause the formation of ROS [5, 6]. ROS are unstable molecules that can react easily with electron acceptors, such as oxygen, and turn into free radicals. The accumulation of ROS that are not eliminated by the antioxidant system in cells produces what is called oxidative stress. Clinical manifestations of oxidative stress on the skin due to accumulation of ROS include increased skin weakness, xerosis, wrinkles, pigmentation irregularities, benign growths (such as seborrheic keratosis) and/or malignant neoplasms such as basal or squamous cell carcinoma. At the cellular level, there can be a decrease in keratinocyte proliferation, reduced generation of the stratum corneum, lower regeneration of the protective layer caused by reduced lipid synthesis, and thermoregulation damage due to changes in responsive blood vessels and the autonomic nervous system.

The rate of skin aging is affected by intrinsic and extrinsic factors and almost every aspect of biological function [3]. Intrinsic skin aging factors can be affected by ethnicity, anatomical variation and hormonal changes in cutaneous tissues. Some of the extrinsic factors in skin aging include poor nutrition, smoking, nicotine, pollution, and UVR exposure.

Herbal extracts are complex mixtures of natural compounds with various structures and origins. These have been used in skin care and cosmetic products since ancient times. The strategies of herbal product for the treatment and prevention of aging skin are variously based on their potential activities as antioxidants [1], antiphoto-aging agents [12, 13], anti-inflammation activity [14, 15], as skin cell proliferation promotors [16], modulators of collagen and elastin synthesis [17], or as inhibitors of melanin production [18].

The increasing concerns regarding quality of life and appearance have led to an increase in the market of natural compounds with anti-aging and photoprotective properties [19]. Plant extracts and natural compounds are able to protect or ameliorate the deleterious effects on aged skin, and many of them are used as either oral dietary supplements or topical cosmetic formulations. The importance of cosmetical preparation is not only related to improving the overall appearance of the skin during aging but it also offers a better quality of life through prevention and treatment of skin disorders related to the aging process [20].

This review will summarize the skin aging characteristics in each layer, the usage of herbal products in the prevention, regeneration and delay of the skin aging process, and herbal formulations to increase the effectivity of cosmeceutical products.

## 2 Characteristic of Aging Skin

Aged skin has the characteristics of being thinner and pigmented, with increased laxity, coarseness, wrinkling, sallow coloration, telangiectasia (apparent widened venules), dryness, fragility, ease of bruising, and cutaneous malignancies. Intrinsic skin aging is a normal physiological process that occurs due to genetic or metabolic factors [3]. The most noticeable histological changes occur in the basal cell layer as the epidermis becomes thinner, the dermal-epidermal junction (DEJ) area decreases, and basal cell proliferative capacity weakens [21].

The skin consists of different cell types and compartments with different functions [22–25]. The salient features of aged skin occur throughout the epidermis, dermis and subcutaneous tissue, which in turn manifest in the alteration of skin topography. The basal layer contains pigment-producing melanocytes, which determine skin color and possess photo-protecting properties. The DEJ connects the epidermis to the underlying dermis, which contains dermal fibroblasts and appendages such as hair follicles, sebaceous glands and sweat glands. As shown schematically in Fig. 9.1, the aged skin dermis is relatively avascular and acellular compared with younger skin. It also has lower production and disorganization of collagen, fragmented elastic fibers, as well as lower glycosaminoglycans and melanocytes.



Fig. 9.1 Clinical manifestation of aged skin

Signs of skin aging include thin and dry skin, coarse wrinkles, decreased elasticity, laxity, aberrant pigmentation and a rough-textured appearance. Decreased proliferation and renewal capacity of basal keratinocytes and reduced epidermal stem cell number are also several factors that can cause epidermal, DEJ and dermal thinning [3]. The extracellular membrane (ECM) generates fibroblasts, which are responsible for maintaining skin integrity and elasticity in the dermis area. Therefore, alterations and degradation of the ECM during dermal thinning can cause increased numbers of wrinkles and decreased skin elasticity [26].

## 2.1 Epidermis

The epidermis is the outermost part of the skin, composed of four strata known as the basal, spinous, granular and cornified layers [4, 5, 27]. The dominant cell types in the epidermis are keratinocytes and melanocytes. It has a function to prevent water loss, protect the body from toxic chemicals and pathogenic microorganism, abrasions and UVR.

Regeneration of the epidermal cells occurs continuously in young skin. This regeneration begins with multiplication of proliferative cells in the deepest layer, and these are pushed out by differentiation and cell division [4, 5, 21, 28]. Differentiation causes molecular, structural and functional changes of keratinocytes so that these will occupy different layers. The peak of differentiation will produce corneocytes which are dead cells. Corneocytes, which are rich in proteins, are embedded in the matrix, which is composed of ceramides, cholesterol and fatty acids [23, 29].

In the cornification process, calcium has an important role. The lowest calcium level is found in the basal stratum and increases up to the stratum corneum. In young skin, the stratum granulosum has high calcium levels. In old skin, the distribution of calcium in each layer is so irregular that the composition of the protective proteins of corneocytes changes. This is also thought to be a mechanism of protection against aging [4, 5]. The speed and regeneration ability of epidermal cells is influenced by age. Aging leads to decreased numbers of cells that proliferate in each layer and dead cells accumulate so that the regenerative capacity of the tissue is lower. The structure and shape of cells change to become more porous and the functions of structural organizations become less effective [28].

The normal aging process does not affect dermis thickness, but there are noticeable changes in skin structure. For example, the skin becomes more susceptible to irritation and has lower permeability, which results in reduced trans-epidermal water flux [23, 29]. In some cases, the whole epidermal layer will gradually decrease in thickness with age at a rate of approximately 6.4% per decade. In photoaged conditions, this reduction in thickness will occur faster. The form of keratinocytes also becomes shorter and fatter, while the corneocytes become larger as a result of decreased epidermal turnover. In aged skin, there is a decreased water binding capacity due to changes in amino acid composition and lipid levels so that the capacity of natural skin moisturizing factors are reduced. Disorders due to the aging process can also affect the barrier function of the skin, due to the global reduction of lipids in the stratum corneum, which will affect corneocyte binding in the matrix. In addition, the DEJ can be flattened, causing reduced interdigitation between the epidermis and dermis which results in reduced supply of nutrients and oxygen.

Melanin has a role as sun protector in the skin [23]. In photoaging, the production of melanin decreases due to a decrease in the functional amount of melanocytes in the basal layer of the epidermis (stratum basale). Therefore, older people are more susceptible to UVR exposure which can cause sun-induced cancer [23]. Aging of the skin causes uneven pigmentation with a tendency towards hyperpigmentation, even though there is a decrease in the number of melanocytes. In addition, there is also loss of melanocytes in certain areas and changes in interactions between melanocytes and keratinocytes [27].

## 2.2 Dermis

The dermis contains nerves, blood vessels, lymphatics and secretory organs [27]. Fibroblasts are the main cell type found in the dermis and these function to synthesize and degrade the ECM. The ECM structure consists of highly organized, elastic and reticular collagen fibers. The three primary structural components of the dermis are collagen, elastin and glycosaminoglican (GAG). Collagen functions to give skin its strength and maintain tissue integrity, whereas elastin provides elasticity and resilience. The extrafibrillar matrix consists of a complex mixture of proteoglycans, glycoproteins, glycosaminoglycans, water and hyaluronic acid.

Many of the components and characteristics of the dermis decrease with age, such as thickness, vascularization and cellularity, number of mast cells and fibroblasts, as well as the levels of glycosaminoglycans, hyaluronic acid produced by fibroblasts and the number of interfibrillary ground substances [22, 25, 27, 29]. Decreased elastin and collagen turnover caused by a decrease in fibroblasts and collagen synthesis also occurs due to aging. Photoaged skin shows disorganized collagen fibrils, fragmented elastic fiber and accumulation of abnormal elastin-containing material. The clinical manifestations of dermal aging include increased skin stiffness because the molecular integrity of the dermis is lost, decreased extensibility of torsion, reduced elasticity, and increased susceptibility to tearing-type injuries [29].

#### 2.2.1 Collagen

Collagen is the most abundant protein in skin, constituting about 70% of dry skin mass, and it is responsible for conferring strength and gives support to human skin. Normal aging is characterized by atrophy of epidermis and dermis and also flattened rete ridges [23]. Aged skin has thickened fibrils in rope-like bundles and

it has a decreased collagen synthesis level [23, 30, 31]. The ratio of collagen types in human skin is also an indicator of the skin age, because this ratio is altered in aged skin. In young skin, 80% of collagen is type I and 15% is type III. In contrast, old skin contains more type III collagen due to decreased levels of type I collagen. In photoaged skin, studies have shown that there is 59% reduction of type I collagen. Collagen IV also plays an important role as the framework for other molecules and it is also responsible for maintaining the mechanical stability of DEJ, which is assumed to be correlated with wrinkle formation. However, there are no significant differences in collagen type IV levels in sunexposed and uv-exposed skin. Collagen VII functions by anchoring the fibrils that attach the basement membrane to the underlying pappilary dermis. Sunexposed skin also shows significantly lower numbers of anchoring fibrils compared to normal skin. Therefore, it is assumed that wrinkles may form as a result of a weakened bond between the dermis and epidermis, due to degradation of anchoring fibrils. Loss of collagen type IV and VII marks the base region of the wrinkle. Overall, the collagen abundance in the skin decreases at a rate of approximately 1% per year.

Skin aging, either caused by extrinsic or intrinsic factors, is marked by an elevation of activator protein-1(AP-1) activity, matrix metalloproteinase (MMP) gene expression, inhibition of transforming growth factor (TGF)- $\beta$  signalling, reduction of collagen synthesis and increased collagen degradation [32]. UVR exposure has been found to up-regulate the synthesis of MMPs through increased amounts of transcription factors such as c-jun and c-fos. MMPs are increased even without UVR exposure and this leads to the formation of AP-1. AP-1 then activates the MMP genes and stimulates the production of collagenase, gelatinase and stromelysin. Thus, AP-1 activation and (TGF)- $\beta$  signaling inhibition mediates collagen degradation via increased proteolysis. MMPs, especially collagenase and gelatinase, are known to be produced within hours of UVB exposure. Long term elevation of MMPs likely leads to disorganized and clumped collagen, and it is assumed that this causes the lower collagen I levels in photoaged skin [23, 30, 31].

#### 2.2.2 Elastin

Photo-aged skin causes the accumulation of amorphous elastin material, called elastosis [23, 30, 31]. UV exposure induces elastin fibre thickening and coiling in the pappilary dermis and reticular dermis, reduction of microfibril numbers, increases in the interfibrillar area, complexed shaped and arrangement of the fibres and increased numbers of electron-dense inclusions. Aged skin also has small amounts of sugar and lipids and abnormally high levels of polar amino acids. MMP-2 is thought to be the protease responsible for degradation of elastin.

Sun exposure level affects the magnitude of the hyperplastic response or an increased amount of elastic tissue. Photo-aged skin has been found to have lower

elasticity and resiliency [31]. This skin is characterized by changes in the normal pattern of oxytalan fibers in the pappilary dermis by the formation of the fibrous network in young skin that ascends perpendicullary from the uppermost section of the pappilary dermis to just beneath the basement membrane. The loss of elasticity manifests as sagging skin in the elderly person [23, 30, 31].

#### 2.2.3 Glucosaminoglycans

Glucosaminoglycans (GAGs) are polysaccharide chains with repeating disaccharide units attached to a core protein, which are responsible for conferring the outward appearance of the skin by its capability to bind water uo to 1000 times their volume [31, 33]. Hyaluronic acid (HA), dermatan sulphate, and chondroitin sulphate are types of GAGs, which give skin its plump, soft, and hydrated appearance. They also help to maintain the salt and water balance. There are conflicting reports regarding the effect of UVR exposure on GAG concentrations in the skin, especially regarding HA. Some studies have reported that the GAG concentration in the skin decreased following UVR exposure, but others have suggested that there are no UVR-induced changes in GAG levels. It is assumed that this discrepancy may have occurred because GAGs are produced in both the epidermal and dermal areas. In intrinsically aged skin, the HA level in the dermis remains stable but the epidermal HA diminishes almost completely. Reduced levels of HA can be observed in the photoaged skin and scars. HA can be found in young skin at the periphery of collagen and elastin fibers and also at the intercept of these fibers in the both dermal and epidermal area, but not in the stratum corneum or stratum granulosum [34]. Wrinkle formation, sagging skin, reduced turgidity and capacity to support microvasculature in the skin correlate with decreased HA levels.

## 2.3 Vasculature

Blood flow to the skin is decreased in aged skin, with the reduction reported to be around 40% between the ages of 20 to 70 years [31]. This results in lower nutrient exchange, inhibition of thermoregulation, decreased temperature in the skin surface and skin pallor. The anatomical and physiological changes in the microcirculation that have been reported include impaired microvascular reactivity, increased vascular stiffness, decreased vascular density and impaired vascular organization [35]. Acute and chronic UV radiation stimulates skin angiogenesis through upregulation of vascular endothelial growth factors and inhibition of thrombospondin-1, a potent angiogenesis inhibitor. In particular, a decreased vascular network appears in the pappilary dermis, as shown by the disappearance of the vertical capillary loops which is the cause of the effects described above [36].

## 2.4 Subcutaneous Tissue

The subcutaneous layer maintains the structural integrity and function of blood vessels needed by the dermal and epidermal layers [30, 31, 36]. Subcutaneous fat is higher in females compared to males. In females, it represents 85–90% of the total body fat [36]. Subcutaneous fat is highly partitioned in distinct, independent compartments separated by septal barriers [37]. Fat redistribution in the aged skin results in a reduced subcutaneous:visceral fat ratio [38], which has been found to be correlated with physiological alteration of adipocyte metabolism and adipokine synthesis [39–41]. Skin aging also caused diminished subcutaneous fat in the face, as well as in the dorsal area of the hands and shins.

## 3 Herbal Products for Skin Anti-aging Strategies

The use of oral and topical exogenous antioxidants is one means of preventing and repairing oxidatively-damaged skin. Various plant species have been widely used as sources of oral and topical antiaging agents. In addition, a number of plant second-ary metabolites have activities that may lessen the effects of aging on the skin, such as antioxidants, photoprotective agents, anti-inflammatory molecules, modulators of collagen/elastin synthesis and inhibitors of melanin synthesis.

## 3.1 Antioxidants

The secondary metabolites in plants include antioxidant compounds (especially phenolic compounds; catechins, isoflavones, proanthocyanidins, and anthocyanins), phenolic acids (benzoic, gallic, and cinnamic acids) and stilbenes [42–44]. These are derived from plants such as tea, grape, bergamot, fernblock, rooibos, grapefruit, and red orange, and are currently widely used in cosmetic formulations. Based on their structure, polyphenol compounds can act as antioxidants through radical scavenging or as oxidized prooxidants via phenoxyl reactive radicals or intermediate quinone or quinone metides. The antioxidant activities of polyphenols appear to be more effective than vitamin E and vitamin C. Their use in topical formulas has been shown to be effective as antiaging agents through antioxidant activities [42]. Flavonoids and phenolic acids have the activity of capturing free radicals and acting as chelating metal ions such as iron and copper, which can initiate the formation of reaction free radicals. In addition, flavonoids can inhibit the activity of some redox enzymes so they can inhibit cell damage caused by free radicals [45, 46].

Genistein and daidzein are effective antioxidant compounds. Both of these compounds are isoflavonoids, estrogen-like molecules, which are found in soybeans [45]. The glycoside group is not estrogenically active and can be used for topical applications. Also, the epigallocatechin-3-galate (EGCC) contained in tea leaves is a polyphenol with strong antioxidant activity and topical preparations of tea leaf extract containing this compound can inhibit the effects of UVR exposure, including protecting the body from UVR-induced immunosuppressant activity. Various plant species from the coffee family also contain polyphenols such as proanthrocyanidins, quinic acid, caffeic acid, caffeine and chlorogenic acid, which are also powerful antioxidants [31, 45]. Ferulic acid is another antioxidant found in whole grains, spinach, parsley, grapes, rhubarb, and cereal grains (especially wheat) [47]. Topical use of grape seeds has been proven to prevent or modrate UVR-related damages, such as the thickening of the epidermal layer, erythema, pigmentation, inflammatory neutrophil infiltration and collagen degradation, as well as the increased expression of genes associated with skin aging such as those for cyclooxygenase 2 (COX-2), nuclear factor erythroid 2-related factor 2 (NRF-2) and heme oxygenase 1 (HO-1). The main content of grape seed is resveratrol which is a natural stilbene compound. There are also other antioxidant compounds which may have use as skin anti-aging products, such as quercetin, catechin, epi-catechin, gallic acid and oligomeric proanthocyanidins [43].

## 3.2 Anti-photo-aging Agents

It is well established that UVR can cause skin disorders such as sunburn and nonmelanoma skin cancer. A photoprotective agent can be used to minimize the harmfull effects of UVR exposure to the bare skin. Sun protection factor (SPF) describes the effectiveness level of bioactive compounds that act as UV protectants. It is defined as the minimal erythema dose ratio between sunscreen-protected and unprotected skin [48]. Sunscreens slow the progression of photo-aging and prevent aging effects on the skin such as wrinkles, actinis keratosis reduction, solar elastosis and squamous skin carcinoma. Sunscreen preparations contain organic UV filters that block either UV-A and/or UV-B radiation by absorbing the radiation [49, 50].

Organic sunscreens that have been commonly used are aminobenzoate (e.g., p-aminobenzoic acid esters), cinnamate (e.g., octinoxate), salicylate, actocrylene and benzophenone (e.g., oxybenzone), avobenzone and mexoryl SX [48]. Many plant polyphenols have been evaluated as potential sun protection agents, since they are able to absorb UV-B radiation [12–14]. For example, the phenolic compounds of Schinus terebinthifolius crude extract, such as ethyl gallate, gallic acid and several flavonoid compounds, may be useful as natural sunscreen products [12]. In addition to their antioxidant activities, EGCG and resveratrol also have activities that can reduce the size of erythema caused by UVR exposure. High dosages of flavanol compounds from cocoapowder have a similar capability. Finally, carotenoids such as  $\beta$ -carotene, lycopene, canthaxan-thin, and lutein derived from tomatoes, carrots and algae, are antioxidants that exhibit photoprotective activity and are capable of reducing the size of erythema cause by UVR exposure [43].

## 3.3 Anti-inflammatory Agents

As noted above, UVR induces oxidative stress in epidermal cells which causes lipid peroxidation and cell damage. In this scenario, the complement system recognizes the oxidation-specific epitopes and causes inflammation, which leads to macrophage infiltration and activation that removes the damaged cells and oxidized lipids [51, 52]. Macrophage activation leads to MMP release that degrades the ECM. Thus, long term exposure to UVR can lead to overactivation of the complement system, which causes DEJ damage and overburdens macrophages with oxidized lipids, which further causes chronic inflammation and long-term damage to the dermis due to proinfammatory cytokines and ROS release [51, 53]. In cases of inflammation caused by photoinduced oxidative stress, phenolic compounds can help by attenuating the activity of inflammatory mediators such as interleukin 6 (IL-6) and prostaglandin-E2 (PGE2). In addition, the natural compounds such as veratric acid [15], dihydrochalcone phloretin, afzelin [14] and luteolin [54] can decrease expression of the inflammation-related molecules.

## 3.4 Promotion of Skin Cell Proliferation

Epidermal skin cells possess the ability for self-renewal and dead cell replacement with turnover times of 40-56 days, but aging makes the capacity decline [55]. Bioactive agents from herbal products can be used to restore this regenerative capacity to some extent. Several plants that had been proven to have skin cell renewal capacity include Populus nigra buds, oak woodm mate leaf, and benjoin resin [16]. Nuclear magnetic resonance (NMR) spectroscopy and mass spectrometry (MS) analysis of Populus nigra buds extracts revealed the presence of 6 phenolic acids (caffeic, -coumaric, isoferrulic, di-O-methylcaffeic, cinnamic), three flavonoids (pinocembrin, pinobanksin, and its derivate) and salicin [16], which all have potent antioxidant, antiinflammatory and skin regeneration activity. This extract modulates transcription of genes, such as those for Kruppel-like factor 10 (KLF10), E2F-4 transcription factor (E2F4), and epidermal growth factor (EGF) response factor 1(ZFP36L1), which are involved in skin regeneration through effects on proliferation, differentiation, survival, and DNA synthesis. Expression of the catalase (CAT) gene was also found to be up-regulated, which is responsible for antioxidative effects, and the chemokine ligand 5 (CCL5) gene down-regulated, which is linked to inflammatory processes. The ellagitannin compounds of oak wood, caffeoyl derivatives from mate leaf and phenolic acids in benjoin resin have also been identified as compounds capable of modulating skin-aging related genes. All of these compounds have a demonstrated ability to down-regulate the proinflammatory CCL5 gene and up-regulate proliferation marker genes (KLF10, E2F4 and ZFP36L1) [16].

## 3.5 Modulation of Collagen and Elastin Synthesis

The ECM proteins such as collagen and elastin are increasingly proteolyzed with aging by the MMPs. MMPs in normal cells have been found to be regulated at the transcriptional level by specific protein inhibitors. The homeostatic imbalance of synthesis and degradation of the ECM causes loss of skin integrity, and may result in wrinkle formation. However, inhibition of MMP activity facilitates maintenance of skin structure and the phenolic compounds are known to preserve skin structure through this mechanism [56].

Crude grape pomace from white wine contains phenolic compounds such as cathecin, epicatechin, procyanidins B1 and B2, gallic acid, caftaric acid, quercetin glycosides and stilbene trans-resveratrol which all act as inhibitors of MMPs. The strongest activity was found in a fraction containing gallic acid, and this was followed by procyanidins and catechin which had proteolytic inhibitory activity, especially to collagenase [56]. Unfortunately, the use of these compounds is limited due to their high molecular weights and low permeabilities.

Gallic acid and elaeocarousin in an Emblica officinalis Gaertn ethanolic extract has been reported to enhance proliferation and collagen production in fibroblasts [17]. The extract was capable of minimalizing wrinkle formation through collagen synthesis, MMP inhibition and revival of damaged collagen fibers. Tannins such as emblicanin, pedunculagin and punigluconin are assumed to have similar activity as green tea tannin (EGCG), which modulates the expression and production of MMPs via AP-1 and nuclear factor kappa B (NF- $\kappa$ B) activation [57].

Depletion of collagen production in the aged skin can also be treated with collagen synthesis enhancers, such as vitamin C and glycolic acid, which are claimed to have anti-wrinkle capacities, although this has not been proven.

## 3.6 Melanin Production Inhibitors

Melanin is capable of absorbing 50–75% of the UVR exposure to the skin [58, 59]. Therefore, it can play a significant role as a skin protector. However, the hyperpigmentation caused by melanin overexpression can also lead to dermal disorders. Melanin is natural pigment produced by melanosomes in melanocytes and it is transferred to keratinocytes through dendrites. Melanin is a tyrosine derivative that forms during oxidative stress reactions through the activity of the tyrosinase enzyme. Melanin synthesis is also regulated by tyrosine-related protein 1 and 2 (TRP-1, TRP-2) and cellular signaling.

Melanin inhibitors work through inhibition of tyrosinase [60, 61]. Phenolic and flavonoids which have a structure similar to tyrosine, with regards to the presence of aromatic purine rings, and these can act as substrate analog inhibitors in melanogenesis [62, 63]. The compounds Brazilin and isoflavonoiud-4-O-methylsappanol were identified in a Sappanwood methanolic extract as having the ability to inhibit melanogenesis [64] in faskulin-treated HMV-II human melanoma cells. Resveratrol and oxyresveratrol are also tyrosinase inhibitors [65, 66]. Other herbal products that have melanogenesis inhibitor activity include Morus alba leaf extract [67], Arthrophytum scoparium [18], and Trifolium nigrescens Subsp. The petrisavi [68] and Morus alba leaf extract inhibitory effects are due to the presence of 20 phenolic compounds (8 benzofurans, 10 flavonoids, one stilbenoid and one chalcone) [67]. Catechol and tetrahydro-isoquinoline in Arthrophytum scoparium are responsible for the inhibition of tyrosinase and TRP-1 as well as down-regualtion of the tyrosinase microphthalmia-associated transcription factor (*MITF*) and melanocortin 1 receptor (*MC1R*) genes [18]. In addition, trifolium nigrescens phenolic compounds showed inhibitory effects on mushroom tyrosinase, and one of these containing a glucoside residue was found to have the highest activity [68].

## 4 Herbal Skin Care Product with Anti-aging Properties

The increasing proportion of the aging population globally correlates with the increasing demand of anti-aging products. The need for a beautiful, healthy and youthful appearance has become a trend and the cosmeceutical industry has become an ever-growing market in order to meet these demands. In line with this, new technologies and innovations have been brought into play to increase quality and effectivity of these products, and to attract more customers.

Presently, there are many topical and oral skin care products that function by improving or maintaining skin health [69–72]. Topical products are known as "cosmeceuticals", while oral products are known as "nutricosmeceuticals". Both of these are backed by scientific evidence of effectiveness and safety although these do not follow such strict rules as those demanded in the traditional drug discovery industry (Fig. 9.2).



## 4.1 Cosmeceutical Products

Cosmeceutical is a portmanteau of the words cosmetic and pharmaceutical, which means that it is a cosmetic preparation that also has a pharmaceutical function (Fig. 9.2) [69–74]. Cosmeceuticals are intended to improve and reduce skin imperfections, compared to cosmetics that only hide or mask such imperfections. Similar to cosmetic products, cosmeceutical products are applied topically and intended for daily use. Such products are used to resolve facial aging problems, in particular wrinkles, mottled pigmentation, roughness of skin, rhytides, erythema, skin tone loss, dryness, sallowness, furrows, solar elastosis and black spots. The bioactive ingredients can comprise sunscreens and antioxidants to enhance the protection level of the skin against ROS and thus prevent photodamage and improve skin malignancies [74–77].

There are various forms of cosmeceutical products for antiaging treatments including lotions, creams, gels and liquids. These products usually function as moisturizers, antioxidants, anti-wrinkle agents, whitening agents, skin supplements (vitamins) or growth factors (Fig. 9.3). These products are often intended for daily use and thus the efficacy and safety aspects must be guaranteed. Apart from this requirement, there are several things about the quality that must also be fulfilled, such as it should not be irritative, it should be suitable for various or selected skin types [74, 76, 77].

Aged skin is more vulnarable to dryness. This is due to decreased production and increased degradation of GAG which reduces the ability of the skin to bind water. To improve this condition, regular use of a moisturizer could protect and strengthen the skin. It has been shown that this hydration ability of cosmeceutical products makes the skin softer and smoother [1, 76, 78, 79].



Fig. 9.3 Cosmeceutical product activity

In order to meet these consumer demands, the cosmeceutical is aiming for product specifications with higher efficacy, immediate effects and guaranteed levels of safety. Firstly, the material should have complete function in the form of natural product extracts, active compound isolates or synthesized or semi-synthesized compounds [25, 77]. Secondly, there is the need for development of new technologies for transdermal delivery systems so that the active ingredients can easily and quickly traverse the skin. Finally, the development of technologies that can objectively measure the penetration and effectiveness of the material should be developed in parallel as a means of quality and efficiency monitoring [74, 77, 79].

In the development of product cosmeceuticals, the mechanism of penetration into the skin layer is the basis for developing a drug delivery system. In general, the drug enters the target site through damaged skin However, the active ingredients must be able to penetrate into normal skin in the same manner as cosmetic products. Of the three main skin layers described earlier, the stratum corneum acts as a barrier layer against penetration. The choice of the drug delivery system technology applied to the material must be enable the active ingredient to traverse this barrier [11, 77].

Besides the skin penetration properties, the purpose of a drug delivery system is to increase the duration of action and stability, prevent incompatibility with other ingredients in the formulation and prevent the occurrence of undesirable effects either locally or systemic [73, 79, 80]. Various forms of drug delivery systems that are widely used include vesicles (liposomes, niozomes, nanosomes, phytosomes, herbosomes, marinosom, oleosom, aquasom, ultrasom, photozom, ethosome, transferosome, sphingosome, colloidosomes), emulsions (microemulsions, nanoemulsions, liquid crystals) and particulates (microparticles, nanocapsules, microspheres, nanocrystalline and cyclodextrin). The development of a drug delivery system for cosmeceutical products has been shown to increase its usefulness, but because the product is used daily for a long time, attention should be paid regarding the possibility of unwanted side effects. Therefore, product development must also take in to account the safety aspects [79, 80].

## 4.2 Nutricosmeceutical Products

The desire to have a youthful appearance may be met with good nutrition in line with the pharase, "beauty from inside". This has led to new innovations regarded as "nutricosmeceuticals", which are nutritional products that have both cosmetic and antiaging effects (Fig. 9.2) [70, 81–83]. As physical beauty is reflection of skin health status, the nutricosmeceutical industry aims to develop food products that have been proven to maintain healthy bodies and also have good effects on the skin. Recently, a number of skin food products have been developed, containing carotenoids, polyphenols, vitamins, essential fatty acids and pre-and probiotic substances, which have been used previously via topical applications to improve skin conditions (Fig. 9.4) [1, 82].



Fig. 9.4 Ingredients in nutricosmeceutical products

Since free radical formation appears to be linked to aging, antioxidant intake may help to restore homeostasis. Foods that are rich in antioxidants have also been shown to have a role in maintaining health and in delaying the aging process [75, 78, 81, 82]. These were shown to have a protective effect against free radicals found in the skin, through activation of endogenous antioxidants including enzymatic (reduced glutathione, superoxide dismutase and catalase) and non-enzymatic (vitamin E isoform, vitamin C and ubiquinol) ones. Thus both exogenous and endogenous antioxidants may have an important role in counteracting oxidative stress. In line with this, several endogenous antioxidants are known to be stimulated by a number of nutritional factors derived from fruits and vegetables.

Some carotenoid derivatives of vitamin A, such as  $\beta$ -carotene, astaxanthin, lycopene and retinol, are known to be effective anti-oxidants [1, 75, 78, 81, 82]. Tomatoes, watermelons, guavas, rosehips, and pink grapefruits can reduce oxidative damage because these all contain lycopene which acts as a singlet oxygen quencher. In addition, carrots, pumpkins, yams, mangoes and papayas, which are sources of β-carotene, can protect the skin from erythema caused by UV induction. The structure of polyphenols also shows strong anti-oxidant properties. Other molecules that have the basic structure of polyphenols include phenolic acids, flavonoids, stilbenes and lignans. Polyphenol compounds are found in fruits, vegetables, cereals, chocolate, dried beans and vegetables, as well as in drinks such as fruit juice, tea, coffee and red wine. Food sources containing linolenic acid and linoleic acid include fish and shellfish, flaxseed, flax oil, soybean oil, canola oil, chia seeds, pumpkin seeds, sunflower seeds, leafy vegetables, walnuts, sesame seeds, avocado seeds, avocados, salmon and albacore tuna. These foods can act as anti-inflammatory agents and thereby provide significant protection against UV radiation-induced erythema, for example. Probiotics and prebiotics can be consumed in various forms of food and beverage products, and studies have shown that these products can accelerate the recovery of human skin immune homeostasis after UV-induced immunosuppression.

The association between nutrition and skin conditions has led to the use of nutricosmeceutical products to maintain and regain body homeostasis through nutrients that enter the body, so that these can improve skin health and beauty. A complete and balanced nutrition can also positively affect the condition and appearance of the skin. Therefore, nutritional supplements in the form of nutricosmeceutical products can be used to help optimize diets and improve quality of life.

## 5 Conclusions

Chronological aging naturally affects and causes changes in the body, especially in the appearance and characteristics of the skin. Exposure to external factors will worsen this change. Considerable efforts have been made to improve, maintain and delay the effects of aging due to both intrinsic and external factors through invasive and non-invasive methods. In line with this, antiaging products have developed rapidly. The consumers demand products that can show the effects of rapid change, but they also realize that the results will not be as fast as surgery. Nevertheless, they prefer antiaging products that can "improve", "maintain" and "delay" the effects of aging on their appearance. Products that can meet these demands are nutricosmeceuticals and cosmeceuticals, which offer a complementary means of acquiring skin health from inside and outside.

Acknowledgements Idha Kusumawati was supported by an International Research Collaboration and Scientific Publication grant from The Ministries of Research, Technology, and Higher Education, Republic of Indonesia (Grant no. 597/UN3.14/LT/2017), with the project entitled "The Value of Indonesian Herbal Extract as Natural Ingredients and Novel Technologies Anti-Aging". Special thanks to Dr. Paul C. Guest for the nice picture.

## References

- Pérez-Sánchez A, Barrajón-Catalán E, Herranz-López M, Micol V (2018) Nutraceuticals for skin care: a comprehensive review of human clinical studies. Nutrients 10(4). pii: E403. https://doi.org/10.3390/nu10040403
- 2. Blanpain C, Fuchs E (2006) Epidermal stem cells of the skin. Annu Rev Cell Dev Biol 22:339–373
- 3. Zhang S, Duan E (2018) Fighting against skin aging: the way from bench to bedside. Cell Transplant 27(5):729–738
- Rinnerthaler M, Bischof J, Streubel MK, Trost A, Richter K (2015) Oxidative stress in aging human skin. Biomol Ther 5(2):545–589
- 5. Ahsanuddin S, Lam M, Baron ED (2016) Skin aging and oxidative stress. AIMS Mol Sci 3(2):187–195
- 6. Tu Y, Quan T (2016) Oxidative stress and human skin connective tissue aging. Cosmetics 3(3):28. https://doi.org/10.3390/cosmetics3030028
- Helfrich YR, Sachs DL, Voorhees JJ (2008) Overview of skin aging and photoaging. Dermatol Nurs 20(3):177–183; quiz 184

- 9 Impact of Herbal Products on Skin Aging
- Kazanci A, Kurus M, Atasever A (2017) Analyses of changes on skin by aging. Skin Res Technol 23(1):48–60
- 9. Oliveira BF, Nogueira-Machado JA, Chaves MM (2010) The role of oxidative stress in the aging process. ScientificWorldJournal 10:1121–1128
- Avantaggiato A, Bertuzzi G, Pascali M, Candotto V, Carinci F (2015) The theories of aging: reactive oxygen species and what else? J Biol Regul Homeost Agents 29(3 Suppl 1):156–163
- Jadoon S, Karim S, Bin Asad MH, Akram MR, Khan AK, Malik A et al (2015) Anti-aging potential of phytoextract loaded-pharmaceutical creams for human skin cell longetivity. Oxidative Med Cell Longev 2015:709628. https://doi.org/10.1155/2015/709628
- Bulla MK, Hernandes L, Baesso ML, Nogueira AC, Bento AC, Bortoluzzi BB et al (2015) Evaluation of photoprotective potential and percutaneous penetration by photoacoustic spectroscopy of the Schinus terebinthifolius raddi extract. Photochem Photobiol 91(3):558–566
- Almeida IF, Pinto AS, Monteiro C, Monteiro H, Belo L, Fernandes J et al (2015) Protective effect of C. sativa leaf extract against UV mediated-DNA damage in a human keratinocyte cell line. J Photochem Photobiol B 144:28–34
- 14. Shin SW, Jung E, Kim S, Kim JH, Kim EG, Lee J et al (2013) Antagonizing effects and mechanisms of afzelin against UVB-induced cell damage. PLoS One 8(4):e61971. https://doi. org/10.1371/journal.pone.0061971
- Shin SW, Jung E, Kim S, Lee KE, Youm JK, Park D (2013) Antagonist effects of veratric acid against UVB-induced cell damages. Molecules 18(5):5405–5419
- Dudonné S, Poupard P, Coutière P, Woillez M, Richard T, Mérillon JM et al (2011) Phenolic composition and antioxidant properties of poplar bud (Populus nigra) extract: individual antioxidant contribution of phenolics and transcriptional effect on skin aging. J Agric Food Chem 59(9):4527–4536
- Fujii T, Wakaizumi M, Ikami T, Saito M (2008) Amla (Emblica officinalis Gaertn.) extract promotes procollagen production and inhibits matrix metalloproteinase-1 in human skin fibroblasts. J Ethnopharmacol 119(1):53–57
- Chao HC, Najjaa H, Villareal MO, Ksouri R, Han J, Neffati M et al (2013) Arthrophytum scoparium inhibits melanogenesis through the down-regulation of tyrosinase and melanogenic gene expressions in B16 melanoma cells. Exp Dermatol 22(2):131–136
- 19. Allemann IB, Baumann L (2009) Botanicals in skin care products. Int J Dermatol 48(9):923-934
- Kraft JN, Lynde CW (2005) Moisturizers: what they are and a practical approach to product selection. Skin Therapy Lett 10(5):1–8
- Makrantonaki E, Zouboulis CC (2007) Molecular mechanisms of skin aging: state of the art. Ann N Y Acad Sci 1119:40–50
- 22. El-Domyati M, Attia S, Saleh F, Brown D, Birk DE, Gasparro F et al (2002) Intrinsic aging vs. photoaging: a comparative histopathological, immunohistochemical, and ultrastructural study of skin. Exp Dermatol 11(5):398–405
- 23. Tobin DJ (2017) Introduction to skin aging. J Tissue Viability 26(1):37-46
- Baumann L (2018) How to use oral and topical cosmeceuticals to prevent and treat skin aging. Facial Plast Surg Clin North Am 26(4):407–413
- 25. Kusumawati I, Kurniawan KO, Rullyansyah S, Prijo TA, Widyowati R, Ekowati J et al (2018) Anti-aging properties of Curcuma heyneana Valeton & Zipj: a scientific approach to its use in Javanese tradition. J Ethnopharmacol 225:64–70
- Stearns HC, Sneeden VD (1966) Observations on the clinical and pathologic aspects of the pelvic congestion syndrome. Am J Obstet Gynecol 94(5):718–732
- 27. Costin GE, Hearing VJ (2007) Human skin pigmentation: melanocytes modulate skin color in response to stress. FASEB J 21(4):976–994
- Lorencini M, Brohem CA, Dieamant GC, Zanchin NI, Maibach HI (2014) Active ingredients against human epidermal aging. Ageing Res Rev 15:100–115
- Farage MA, Miller KW, Elsner P, Maibach HI (2013) Characteristics of the aging skin. Adv Wound Care (New Rochelle) 2(1):5–10
- 30. Calleja-Agius J, Muscat-Baron Y, Brincat MP (2007) Skin ageing. Menopause Int 13(2):60-64

- 31. Baumann L (2007) Skin ageing and its treatment. J Pathol 211(2):241-251
- 32. Rittié L, Fisher GJ (2002) UV-light-induced signal cascades and skin aging. Ageing Res Rev 1(4):705–720
- Elsner P, Maibach HI (2005) Cosmeceuticals and active cosmetics: drugs versus cosmetics. Taylor & Francis, Boca Raton. ASIN: B01K0U81P0
- 34. Sakai S, Yasuda R, Sayo T, Ishikawa O, Inoue S (2000) Hyaluronan exists in the normal stratum corneum. J Invest Dermatol 114(6):1184–1187
- Bentov I, Reed MJ (2015) The effect of aging on the cutaneous microvasculature. Microvasc Res 100:25–31
- 36. Rittié L, Fisher GJ (2015) Natural and sun-induced aging of human skin. Cold Spring Harb Perspect Med 5(1):a015370. https://doi.org/10.1101/cshperspect.a015370
- Rohrich RJ, Pessa JE (2008) The retaining system of the face: histologic evaluation of the septal boundaries of the subcutaneous fat compartments. Plast Reconstr Surg 121(5):1804–1809
- Hughes MC, Williams GM, Baker P, Green AC (2013) Sunscreen and prevention of skin aging: a randomized trial. Ann Intern Med 158(11):781–790
- Guo WJ, Datta S, Band V, Dimri GP (2007) Mel-18, a polycomb group protein, regulates cell proliferation and senescence via transcriptional repression of Bmi-1 and c-Myc oncoproteins. Mol Biol Cell 18(2):536–546
- Kim EJ, Kim YK, Kim JE, Kim S, Kim MK, Park CH et al (2011) UV modulation of subcutaneous fat metabolism. J Invest Dermatol 131(8):1720–1726
- Sepe A, Tchkonia T, Thomou T, Zamboni M, Kirkland JL (2011) Aging and regional differences in fat cell progenitors - a mini-review. Gerontology 57(1):66–75
- 42. Stojiljković D, Pavlović D, Arsić I (2014) Oxidative stress, skin aging and antioxidant therapy. Acta Fac Medicae Naissensis 31(4):207–217
- Petruk G, Del Giudice R, Rigano MM, Monti DM (2018) Antioxidants from plants protect against skin photoaging. Oxidative Med Cell Longev 2018:1454936–1454911. https://doi. org/10.1155/2018/1454936
- 44. Kusumawati I, Indrayanto G (2013) Natural antioxidants in cosmetics. In: Atta-ur-Rahman (ed) Studies in natural products chemistry, vol 40, 40th ed, pp 485–505. Elsevier; Amsterdam, Netherlands. ISBN-10: 0444596038
- 45. Simo A, Kawal N, Paliyath G, Bakovic M (2014) Botanical antioxidants for skin health in the world of cosmeceuticals. Int J Adv Nutr Health Sci 2(1):67–88
- 46. Kasote DM, Katyare SS, Hegde MV, Bae H (2015) Significance of antioxidant potential of plants and its relevance to therapeutic applications. Int J Biol Sci 11(8):982–991
- Zduńska K, Dana A, Kolodziejczak A, Rotsztejn H (2018) Antioxidant properties of ferulic acid and its possible application. Skin Pharmacol Physiol 31(6):332–336
- Gilaberte Y, González S (2007) Update on photoprotection. Actas Dermosifiliogr 101(8):659–672
- 49. Liang Y, Wu M, Chen Y (2019) Selected medicinal herbs and functional peptides for protection against photoaging of the skin. J Cosmet Dermatological Sci Appl 9(1):1–18
- 50. Kullavanijaya P, Lim HW (2005) Photoprotection. J Am Acad Dermatol 52(6):937-958
- DeBuys HV, Levy SB, Murray JC, Madey DL, Pinnell SR (2000) Modern approaches to photoprotection. Dermatol Clin 18(4):577–590
- Nash JF (2006) Human safety and efficacy of ultraviolet filters and sunscreen products. Dermatol Clin 24(1):35–51
- 53. Fourtanier A, Moyal D, Seité S (2008) Sunscreens containing the broad-spectrum UVA absorber, Mexoryl SX, prevent the cutaneous detrimental effects of UV exposure: a review of clinical study results. Photodermatol Photoimmunol Photomed 24(4):164–174
- 54. Wölfle U, Esser PR, Simon-Haarhaus B, Martin SF, Lademann J, Schempp CM (2011) UVBinduced DNA damage, generation of reactive oxygen species, and inflammation are effectively attenuated by the flavonoid luteolin in vitro and in vivo. Free Radic Biol Med 50(9):1081–1093
- 55. Koster MI (2009) Making an epidermis. Ann NY Acad Sci 1170:7-10

- Wittenauer J, Mäckle S, Sußmann D, Schweiggert-Weisz U, Carle R (2015) Inhibitory effects of polyphenols from grape pomace extract on collagenase and elastase activity. Fitoterapia 101:179–187
- 57. Chowdhury A, Sarkar J, Chakraborti T, Chakraborti S (2015) Role of Spm-Cer-S1P signalling pathway in MMP-2 mediated U46619-induced proliferation of pulmonary artery smooth muscle cells: protective role of epigallocatechin-3-gallate. Cell Biochem Funct 33(7):463–477
- Hah YS, Cho HY, Lim TY, Park DH, Kim HM, Yoon J et al (2012) Induction of melanogenesis by rapamycin in human MNT-1 melanoma cells. Ann Dermatol 24(2):151–157
- 59. Liu WS, Kuan YD, Chiu KH, Wang WK, Chang FH, Liu CH et al (2013) The extract of Rhodobacter sphaeroides inhibits melanogenesis through the MEK/ERK signaling pathway. Mar Drugs 11(6):1899–1908
- 60. Zucca P, Pintus M, Manzo G, Nieddu M, Steri D, Rinaldi AC (2015) Antimicrobial, antioxidant and anti-tyrosinase properties of extracts of the Mediterranean parasitic plant Cytinus hypocistis. BMC Res Notes 8:562. https://doi.org/10.1186/s13104-015-1546-5
- Roh HJ, Noh HJ, Na CS, Kim CS, Kim KH, Hong CY et al (2015) Phenolic compounds from the leaves of Stewartia pseudocamellia maxim. And their whitening activities. Biomol Ther 23(3):283–289
- 62. Sugumaran M (2002) Comparative biochemistry of eumelanogenesis and the protective roles of phenoloxidase and melanin in insects. Pigment Cell Res 15(1):2–9
- 63. Boissy RE, Manga P (2004) On the etiology of contact/occupational vitiligo. Pigment Cell Res 17(3):208–214
- 64. Mitani K, Takano F, Kawabata T, Allam AE, Ota M, Takahashi T et al (2013) Suppression of melanin synthesis by the phenolic constituents of sappanwood (Caesalpinia sappan). Planta Med 79(1):37–44
- Park J, Park JH, Suh HJ, Lee IC, Koh J, Boo YC (2014) Effects of resveratrol, oxyresveratrol, and their acetylated derivatives on cellular melanogenesis. Arch Dermatol Res 306(5):475–487
- 66. Kim BH (2010) Safety evaluation and anti-wrinkle effects of retinoids on skin. Toxicol Res 26(1):61-66
- 67. Jeong JY, Liu Q, Kim SB, Jo YH, Mo EJ, Yang HH et al (2015) Characterization of melanogenesis inhibitory constituents of Morus alba leaves and optimization of extraction conditions using response surface methodology. Molecules 20(5):8730–8741
- Demirkiran O, Sabudak T, Ozturk M, Topcu G (2013) Antioxidant and tyrosinase inhibitory activities of flavonoids from Trifolium nigrescens Subsp. petrisavi. J Agric Food Chem 61(51):12598–12603
- 69. Oricha BS (2010) Cosmeceuticals: a review. Afr J Pharm Pharmacol 4(4):127-129
- Anunciato TP, da Rocha Filho PA (2012) Carotenoids and polyphenols in nutricosmetics, nutraceuticals, and cosmeceuticals. J Cosmet Dermatol 11(1):51–54
- 71. Kadam Vaishali S, Chintale Ashwini G, Deshmukh Kshitija P, Nalwad Digambar N (2013) Cosmeceuticals an emerging concept: a comprehensive review. Int J Res Pharm Chem 3(2):308–316
- Wanjari N, Waghmare J (2015) A review on latest trend of cosmetics-cosmeceuticals. Int J Pharma Res Rev 4(5):45–51
- Millikan LE (2001) Cosmetology, cosmetics, cosmeceuticals: definitions and regulations. Clin Dermatol 19(4):371–374
- Bellad KA, Nanjwade BK, Kamble MS, Srichana T, Idris NF (2017) Development of cosmeceuticals. World J Pharm Pharm Sci 6(4):643–691
- 75. Pandel R, Poljšak B, Godic A, Dahmane R (2013) Skin photoaging and the role of antioxidants in its prevention. ISRN Dermatol 2013:930164. https://doi.org/10.1155/2013/930164
- 76. McCook JP (2016) Topical products for the aging face. Clin Plast Surg 43(3):597-604
- 77. Shin KO, Park HS (2019) Antiaging cosmeceuticals in Korea and open innovation in the era of the 4th industrial revolution: from research to business. Sustain 11(3):898. https://doi.org/10.3390/su11030898
- 78. Hodges AL, Walker DK (2017) Skin care for women. Nurs Womens Health 20(6):609-613
- 79. Kaul S, Gulati N, Verma D, Mukherjee S, Nagaich U (2018) Role of nanotechnology in cosmeceuticals: a review of recent advances. J Pharm (Cairo) 2018:3420204. https://doi. org/10.1155/2018/3420204
- Algin Yapar E (2017) Herbal cosmetics and novel drug delivery systems. Indian J Pharm Educ Res 51(3):S152–S158
- Vranesić-Bender D (2010) The role of nutraceuticals in anti-aging medicine. Acta Clin Croat 49(4):537–544
- Marini A, Krutmann J (2015) Beauty from the inside. In: Neves D (ed) Anti-ageing nutrients: Evidence-Based of Age-Associated Disease, 1st ed, pp 299–314. John Wiley & Sons Ltd; Chicago, USA. ISBN:9781118733271
- Ramasubramania Raja R (2016) Nutraceuticals and cosmeceuticals for human beings an overview. Am J Food Sci Health 2(2):7–17

# Chapter 10 Epigenetic Clock: Just a Convenient Marker or an Active Driver of Aging?



Vasily V. Ashapkin, Lyudmila I. Kutueva, and Boris F. Vanyushin

## 1 Introduction

Why we age has been a question of heated debate for a very long time. More or less common view today is that aging has multiple causes [1]. In a recent seminal paper, nine hallmarks of aging were proposed to contribute to aging and collectively to determine the aging phenotype [2]. Epigenetic systems control gene activity and thus, directly or indirectly, affect all other hallmarks. Once differentiated, all cells in metazoan organisms must "remember" their appropriate pattern of gene expression for the organism to survive and function normally. Studies of developmental biology have established a role for epigenetic systems in establishing and maintaining these differentiated states of cells. For many years the epigenome was thought of as a static entity; once a cell becomes differentiated, and its genome is appropriately methylated and chromatin configured, no further changes in the epigenome were supposed to occur. Unexpectedly, multiple studies in the last years showed that epigenome is a dynamically regulated system involved in aging, or at least affected by it, and responsive to various external and internal factors [3–5].

An ability of epigenetic systems to affect all other hallmarks of aging puts them into a key position to affect the basic mechanisms (driving forces) of aging. On the other hand, active components of the epigenetic systems are encoded in the genome and, thus, are themselves under epigenetic control. Many links, both feedback, and feed-forward exist between different parts of the epigenome. Transcripts of genes encoding proteins of DNA methylation and histone modification systems could be targeted and controlled by miRNAs, whereas expression of genes encoding miR-NAs are controlled by cytosine methylation and histone modification. DNA

V. V. Ashapkin (🖂) · L. I. Kutueva · B. F. Vanyushin

Belozersky Institute of Physico-Chemical Biology, Lomonosov Moscow State University, Moscow, Russia

e-mail: ashapkin@genebee.msu.ru

<sup>©</sup> Springer Nature Switzerland AG 2019

P. C. Guest (ed.), *Reviews on Biomarker Studies in Aging and Anti-Aging Research*, Advances in Experimental Medicine and Biology 1178, https://doi.org/10.1007/978-3-030-25650-0\_10

methylation and histone modification systems are known to be coupled at both gene expression and target recognition levels. Which alterations in DNA methylation and other epigenetic marks play a causal role in aging and which are just manifestations of aging have yet to be elucidated. The links between epigenome and aging are complicated and probably mutual. The epigenome is changed by various environmental and internal factors that affect aging, but it also could directly or indirectly affect aging. Similar to other cell systems, the epigenome is prone to gradual degradation due to genome damage, stressful agents and other aging factors. However, unlike mutations and other hallmarks of aging, age-related epigenetic changes could be fully or partially reversed to a "young" state.

# 2 Changes in DNA Methylation Levels During Aging

The first experimental proof of age-dependent changes in DNA methylation was obtained in our lab about half a century ago. In humpback salmon and rat, levels of cytosine methylation (5mC/C) in DNA of different organs have been found to gradually decrease with age [6, 7]. In mammals, the age-dependent loss of methylation mainly affected the heavily methylated repeat fraction of the genome [8] possibly leading to a global genome destabilization, since this fraction predominantly consists of transposable elements. Repression of such elements is widely believed to be one of the main defensive functions of DNA methylation. Indeed, activation of retrotransposons has been shown to occur at late cellular senescence leading to an interferon type I (IFN1-type) inflammation response, probably contributing to aging-associated chronic inflammation (inflammaging) – a known hallmark of aging [9, 10].

DNA methylation levels in different mice organs have shown a correlation with the chronological age [11, 12]. The rates of methylation loss upon aging in DNA of two mouse species was found to be inversely correlated with their maximal lifespans. A progressive decrease in DNA methylation levels was also observed in mouse, hamster and human fibroblast cultures [13]. In these cases, the rates of DNA hypomethylation were also inversely correlated to the cell lifespans (maximal number of the cell population doublings before senescence). Errors in the maintenance DNA methylation were considered to be the main cause of the progressive 5mC loss in dividing cells since the fidelity of the maintenance DNA methylation is ~95% per cell generation [14, 15]. Such loss of 5mC, essentially stochastic as it seemed to be, could affect aging at both cellular and organismal levels by increasing the transcriptional noise and eventually leading to aberrant transcription of various genes. However, no DNA hypomethylation was observed in dividing immortal cell lines [13]. Since the fidelity of the maintenance DNA methylation could hardly be higher in these immortal cell lines compared with their normal counterparts, the age-related loss of 5mC could not be explained by the maintenance methylation errors alone. Indeed, similar losses of 5mC upon aging occurred in mouse tissues with widely different proliferative activities (liver, brain, and small intestine mucosa) [11].

Moreover, it was found that besides global DNA hypomethylation, selective hypermethylation of some genes occurs upon aging [16–18]. The tumor suppressor genes, known to be hypermethylated in tumor cells, are also hypermethylated in normal cells upon aging [17, 18]. The promoter CpG island (CGI) sequences of *LOX, CDKN2A* (also known as *p16, INK4a* or *p16<sup>INK4a</sup>*), *RUNX3*, and *TIG1* genes were found to be mainly unmethylated in human normal stomach epithelial cells before the 50-year age but progressively methylated between 50 and 80 years [19]. Such progressive methylation could explain the increase in tumor occurrences known in aged people. Since the age-dependent increase in DNA methylation levels was non-linear in this case, it evidently should be caused by some specific process, not just an accumulation of stochastic errors.

Discordance in DNA methylation profiles between monozygotic (MZ) twins can be used to discriminate between stochastic and "programmed" age-associated changes in DNA methylation. The methylation levels of total lymphocyte DNA were found to be practically identical between MZ twins in 65% of the pairs and significantly different in 35% of the pairs [20]. Identical DNA methylation levels were typical in younger pairs, whereas aged pairs usually had the most divergent DNA levels. DNA methylation differences between MZ twins gradually increased with age. Most discordant DNA methylation patterns were characteristic of the older twin pairs that lived separately and had different lifestyles. Thus, phenotypic and epigenetic differences could be due both to the effects of various factors (smoking, physical activity, dietary preferences) and stochastic methylation errors (epigenetic drift).

#### **3** Changes of Genome Methylation Patterns Upon Aging

DNA methylation patterns in solid human tissues were found to be dependent both on tissue and age [21]. A distinct correlation between age and methylation was observed, loci in CGIs gained methylation with age, whereas loci not in CGIs lost it. This pattern was consistent across tissues and in blood-derived DNA. Gene-loci that earlier have been reported to be associated with aging, *ESR1*, *GSTP1*, *IGF2*, *MGMT*, *MYOD1*, *RARB*, and *RASSF1*, displayed significant methylation alterations (mainly increases) with age. Genes involved in epigenetic regulation (*LAMB1*, *DNMT1*, *DNMT3B*, *HDAC1*, *HDAC7*), telomere maintenance (*TERT*, *ERCC1*, *RAD50*), and premature aging (*WRN*), all showed age-related methylation alterations.

Out of approximately 3600 mouse genes studied, 21% displayed hypermethylation, and 13% displayed hypomethylation with age in small intestine cells [22]. Among genes devoid of the promoter CGIs, 7% displayed hypermethylation and 11% hypomethylation. The genes known to be hypermethylated in colon cancer (*Cdh13*, *Dok5*, *Esr1*, *Igf2*, *Myod1*, *Nkx2-5*, *Cdkn2a*, *Pgr*, and *Tmeff2*) showed agerelated hypermethylation in normal small intestine. Globally, the hypermethylated gene group was enriched for genes involved in development and differentiation, whereas the hypomethylated gene group was not enriched for any specific functional category. These age-dependent gene methylation patterns were partially conserved between humans and mice, and the conserved genes were mainly among the hypermethylated ones.

About 350 CpG loci were found to be differentially methylated with age (aCpGs) in human peripheral blood leukocytes (PBLs), with approximately 200 hypermethylated (hyper-aCpGs) and 150 hypomethylated (hypo-aCpGs) [23]. More than 95% of aCpGs were located within 500 bp of transcriptional start sites (TSS). More than 60% of hyper-aCpGs found in PBLs were also present in purified fractions of CD14+ monocytes (short-lived cells of the myeloid lineage) and CD4+ T-cells (long-lived cells of the lymphoid lineage). Unlike hyper-aCpGs, the hypo-aCpGs set was significantly reproduced in T-cells but not in monocytes. The conservative hyper-aCpGs group was enriched for genes with tissue-specific functions, including neural cell-related processes. Most of these genes are inactive or moderately active in PBL. These hyper-aCpGs significantly overlap with genes that are hypermethylated in various human cancers. Similarly to cancer cells, their methylation could decrease differentiation potential and increase the self-renewal of stem cells. Possibly, this is one of the mechanisms that enhance cancer frequency at advanced ages.

Of 476,366 CpGs covered by the Illumina Human Methylation450 BeadChip, about 29% were found to be differentially methylated across ages 14-94 years in PBLs [24]. Of these aCpGs, 60.5% were hypomethylated and 39.5% hypermethylated with age. For most sites, DNA methylation changed approximately linearly with age. The strongest positive correlation of the methylation level with age was seen for a CpG site located in the ELOVL2 gene promoter CGI (methylation levels were from 0.35 in the youngest to 0.89 in the oldest individuals). Among the sites that become hypermethylated with age, 81.4% were located in CGIs, compared to only 2.8% for the hypomethylated sites. A larger fraction of CpGs located in gene bodies and 3'-un-translated regions (UTRs) became hypomethylated with aging. The genes with hypermethylated CGIs were over-represented in molecular functions of DNA binding and transcriptional regulation, developmental processes, and neurological functions. The genes that became hypomethylated with age were not enriched for any specific functional categories. Interestingly, both genes that became hypermethylated and hypomethylated with age were more likely to decrease than increase expression with age. Thus, both hypermethylation and hypomethylation with age can decrease expression, probably via different regulatory mechanisms.

In contrast to total PBLs, 12,275 and 48,876 aCpGs were identified in purified CD4+ and CD8+ T cell samples [25]. The majority of hyper-aCpGs were located in CGIs at silent gene promoter regions that were enriched for repressive histone methylation marks. Hypo-aCpGs were located more at the borders of CGIs and in gene bodies. In a subset of genes expressed in CD8+ T cells, a negative correlation between DNA methylation and transcription levels was observed. Many genes that are essential for the differentiation and function of T cells were among these, explaining possible causes of the age-related decline in the immune response.

In brain samples from variously aged human donors, 589 CpGs showed agedependent methylation in one division, 167 in two, 86 in three, and only 10 in all four divisions studied [26]. Of all aCpGs found, approximately 82% were located within CGIs, 11% were not within CGIs, and 7% were in regions that could not be unequivocally defined as CGIs or non-CGIs. All 10 aCpGs that showed significant correlations with age in four brain divisions were located within CGIs, and their methylation levels were increased with age. Four of these sites were identified previously as aCpGs in other tissues [21, 23]. A positive correlation between methylation level and age was observed for the majority (>95%) of aCpGs. Of the aCpGs located within CGIs, 98% were hypermethylated at advanced ages. Compared with other tissues, the brain contains mainly hyper-aCpGs and most of these are located in the promoter-associated CGIs of genes that encode transcription factors. Apparently, the age-dependent hypermethylation of specific genes could be a part of the epigenetic program affecting genome expression in aging cells, whereas the age-dependent loss of DNA methylation seems to be a manifestation of stochastic epigenetic drift. Indeed, intergenic regions and CpG poor promoters were found to be more likely to undergo changes in methylation during early life in MZ twin pairs, whereas the methylation of CGIs and CpG-rich promoters was most stable [27]. Since environmental and lifestyle differences between MZ twins are mostly absent at these ages, the divergence of their DNA methylation patterns should be a result of stochastic methylation errors.

The sites located within CGIs were found to be predominant among hyperaCpGs in various human organs (brain, blood, kidney, and skeletal muscle), whereas hypo-aCpGs were most characteristic beyond CGIs, and their methylation levels were more variable between tissues [28]. Hence, age-related variations in methylation common between different tissues are mainly observed in CGIs and represented by increases of methylation levels. On the other hand, tissue-specific variations in methylation are more characteristic of CpGs located beyond CGIs and represented by decreased methylation levels. Among the genes hypermethylated with age in various tissues, most are functionally connected with the regulation of transcription and control of morphogenesis. Skeletal muscle showed the lowest overlap in total aCpGs with other tissues and had the strongest association of aCpGs with tissue-specific gene expression. In a study on human skeletal muscle approximately 6000 CpGs were found to be differentially methylated between the younger (18–27 years) and older (68–89 years age) male adults [29]. Of these aCpGs, 92% (5518) were hyper-aCpGs, while the remaining 8% were hypo-aCpGs. Interestingly, in aging human skeletal muscle, a strong preference was found for the aCpGs to be located within gene bodies, while no preference was observed for promoter regions.

In epidermal samples of variously aged humans, only 58,995 CpGs of those covered by the Illumina Human Methylation450 BeadChip showed a statistically significant change in methylation [30]. The large majority of these aCpGs showed only minor (<0.1) quantitative methylation differences. Of different genomic substructures (CGIs, shores, shelves, and open sea regions), only CGIs showed robust hypermethylation in older samples. Of the 2000 most variably methylated aCpGs, about half showed the consistent age-related difference. Interestingly, a substantial

fraction of age-related methylation changes accumulated during narrowly defined time windows. Step-wise hypomethylation of 256 CpGs occurred between 30 and 50 years, whereas step-wise hypermethylation of 768 CpGs occurred in two pronounced peaks around 40 and 55 years. Within-sample methylation variance for all aCpGs was higher in younger samples, indicating that the dynamic range of skin DNA methylation becomes reduced with age. The methylation patterns appeared highly homogeneous among the younger epidermis samples, but correlation coefficients were markedly reduced between older samples. Similar correlation analysis for specific gene networks showed a distinct age-related loss of correlated gene expression. Skin aging appeared to be accompanied by a reduced fidelity of epigenetic regulation. Age-related erosion of DNA methylation patterns manifested itself in two distinct features. The first one is that sharply demarcated regions of almost complete and almost absent methylation of young methylomes appeared to be less clearly defined in old methylomes, which is reflected by the significantly reduced variance and spatial correlation within methylomes. The second one is that interindividual similarity between young methylomes was essentially lost in older methylomes. Thus, DNA methylation patterns become more homogeneous with age within individuals but more heterogeneous between them. These findings define important features of the epigenetic drift associated with aging. The age-related erosion of methylation patterns is accompanied by a reduced fine-tuning of the transcriptional circuitry.

A comparative study of DNA methylation changes in blood cells of mice, rhesus monkeys, and humans showed that similar epigenetic changes occur in all three species [31]. Age-related gains of methylation were most pronounced in CGI sites that were unmethylated in young individuals, whereas age-related hypomethylation occurred at highly methylated non-CGI sites. Ingenuity® Pathway Analysis (IPA®) of hypermethylated genes showed enrichment for developmental processes, cancer and cardiovascular disease, cell development, signaling, growth, and maintenance, in all three species. For 328 human genes that changed expression by more than two-fold upon aging, for which promoter methylation data were available, a significant negative correlation between changes in methylation and gene expression was observed. Genes showing gains of expression with age were mostly hypomethylated, whereas those with reduced expression were often hypermethylated. When the rates of epigenetic changes were compared between three species, a negative correlation with respective lifespans was observed, at approximately 5% per year in mice, 0.5% per year in monkeys and 0.1% per year in humans. The effects of calorie restriction (CR) on methylation were analyzed in 2.7-3.2-year-old mice exposed to 40% CR starting at 0.3-years, and in rhesus monkeys exposed to 30% CR starting in middle age (7-14 years) and analyzed at 22-30 years (CR treatment period -15-21 years). CR-exposed rhesus monkeys showed attenuation of the age-related methylation changes compared to ad libitum (AL)-fed controls by approximately 7 years. Even stronger effects were seen in CR-mice that had a 0.8-year DNA methylation age at a 2.8-year chronological age. The effects of CR in mice were detectable in different tissues, although significant variability in the extent of the age-related changes and CR impact was observed.

No changes of global DNA methylation level were observed with age in mouse hippocampus in either CpG or CpH contexts [32]. Age-related changes in methylation at individual CpGs (aCpGs) were equally distributed between hypermethylation and hypomethylation events, the majority (>90%) of these being different between males and females. Age-related differentially methylated CpHs (aCpHs) were observed in males and females, with a greater number of hypermethylated sites. Most of these were also sex-specific. Unlike in other organs, enrichment of aCpGs outside of CGI units in intergenic regions and introns, was evident in hippocampus, whereas CGIs and promoter regions were under-represented among hypermethylation events. Similar enrichment patterns were found for aCpHs.

A small increase of global methylation with age was found in a recent wholegenome bisulfite sequencing (WGBS) study of DNA methylation in mouse liver [33]. Out of 3176 aCpGs found, 1945 gained and 1231 lost methylation. Mapping of these aCpGs relative to gene bodies showed 241 hypomethylated and 275 hypermethylated genes. Several hyper-aCpGs were found in the body of the *Elovl2* gene, known to be hypermethylated with aging in humans and mice. Age-related hypermethylated genes showed functional enrichment for the regulation of transcription, circadian rhythm, regulation of ERK and Ras signaling, and response to growth factor signaling, whereas hypomethylated genes were enriched for regulation of the MAPK cascade, xenobiotic, amino acid, and lipid metabolism. In CR mice, only 2250 aCpGs were found, with 1512 hypermethylated and 738 hypomethylated. Globally, an amelioration of aCpGs by CR was observed. Of 571 aCpGs significantly ameliorated by CR, 439 were hypo-aCpGs and 132 hyper-aCpGs. CR-ameliorated aCpGs were enriched in genes involved in the regulation of transcription and ketone, acetyl-CoA, and long-chain fatty acid metabolism. Notably, aCpGs, especially hypo-aCpGs, were strongly enriched in open chromatin domains (containing histone marks H3K9ac, H3K4me1 and H3K27ac). Hypo-aCpGs were strongly enriched in distal and intragenic active enhancers, active promoters, and a wide range of transcription factor binding sites. Thus, open chromatin domains appear to be most prone to methylation changes, especially methylation loss. HyperaCpGs were found to be significantly enriched in bivalent chromatin domain CGIs containing both activating H3K4me3 and repressive H3K27me3 marks. Bivalent CGIs were also enriched for CR-ameliorated aCpGs. Hyper-aCpGs were also enriched in repressed H3K27me3- domains, whereas no enrichment was evident for CR-ameliorated aCpGs. Globally, the observed changes in DNA methylation with aging were not associated with concomitant changes in the expression of differently methylated genes. In old CR mice, 56 genes showed an inverse relationship between gene expression and DNA methylation. These genes code for several key enzymes of hepatic lipid metabolism and are direct targets of the transcription factors Srebf1 and ChREBP, two well-known regulators of hepatic lipogenesis. The Srebf1 gene itself was hypermethylated and downregulated upon CR treatment. Hence, longterm CR-induced changes of DNA methylation and gene expression are specifically associated with hepatic lipid homeostasis.

Genome-wide age-related changes in DNA methylation in blood cells of variously aged mice (between 3-month-old "young adults" and 35-month-old

10%-survival adults) were studied by reduced representation bisulfite sequencing (RRBS) [34]. A slight, but significant decrease of global methylation with age was found. Out of approximately 800,000 CpGs studied, 21.2% were found to be correlated with age, including 10.2% that gained methylation and 11% that lost it. Of these aCpGs, 86% gained (41.6%) or lost (44.3%) methylation predominantly in late life, and 14% changed methylation during early adulthood. Sites with increasing methylation were characterized by the low methylation status, and those with decreasing methylation by high methylation, compared with the average methylation levels. A strong tendency to gain methylation with age was observed in CGIs, 5'-UTRs, first exons, and promoters. Introns and 3'-UTRs tended to lose methylation. Regulatory elements, such as miRNA and lncRNA targets, DNasehypersensitive sites, and transcription factor binding sites tended to gain methylation. Increasing methylation in evolutionarily conserved regions and decreasing methylation in intergenic and repetitive regions (retrotransposons) were also observed. The gene pathway analysis showed 102 pathways to be associated with loss of promoter methylation and 1162 pathways associated with its gain. Among the former ones were pathways related to DNA repair, immunity, and inflammation, whereas the latter showed pathways related to development regulation. In addition, pathways were significantly enriched related to aging and prolongevity interventions, such as the response to IGF and TGFB, MAPK, WNT and Notch signaling, stem cell regulation, estradiol response, fatty acid metabolism, and transcriptional regulation. The DNA methylation pathway was also in that group, including the DNMT1 gene. CR starting at the age of 4 months led to a rapid shift in methylation of 14,516 sites, and most of these were in the same direction as during aging. Long-term effects of CR were essentially opposite, slowing down the influence of aging on the methylome.

When the rates of DNA methylation changes with aging were studied in mammalian species with highly variable maximal lifespans, an inverse correlation was observed [35]. This relationship was observed not only between extremely different in longevity species, such as mouse and human, but also for species that have more similar lifespans, such as dog, naked mole rat, and rhesus macaque. Moreover, this relationship held when two different breeds of dogs were compared, namely the miniature long-haired dachshund (average lifespan 12–15 years) and flat-coated retriever (average lifespan 8–10 years). In a transchromosomic mouse strain that harbors a freely segregating human chromosome 21, the rate of change of methylation with age for the same 15 aCpGs in mouse was about 21-fold faster than in humans. Thus, aCpGs on a chromosome from a human showed greatly accelerated methylation dynamics in mouse, indicating that such aCpGs could serve as sensors of the aging rate.

The human lifespan is widely believed to be genetically determined by 20–30% [36]. In the general human population, the searches for gene loci affecting longevity have been essentially unsuccessful. However, analysis of sibling pairs surviving to beyond 100 years has shown that the genetic influence upon survival increases with older and older ages, particularly beyond the one percentile of survival [37]. Moreover, in a genome-wide association study of extreme human longevity comprising 801 centenarians with a median age of 104 years and 914 genetically

matched controls, 281 independent single nucleotide polymorphisms (SNPs) associated with extreme human longevity were discovered [37]. The possible significance of DNA methylation as a factor controlling lifespan in humans was studied in leukocytes from female centenarians, as well as their daughters of approximately 70-years old selected from pairs, where the father was also long-lived (>77 years), females of approximately 70 years old, whose parents were both non-long-lived, and a control group of young (17-34 years) women [38]. The clinical histories showed that in general, the centenarians' daughters have a much better health status than daughters of non-long-lived parents. Evidently, the probability of becoming long-lived is inheritable to a considerable degree. Global DNA methylation levels were significantly decreased in all three aged groups compared with the control group but to different extents. Maximum hypomethylation was observed in daughters of non-long-lived parents, minimal in centenarians' daughters, and intermediate in centenarians themselves. About 700 CpGs located in approximately 600 genes were hypermethylated in all three aged groups – to similar extents in both daughter groups, and to a greater extent in centenarians. This set of hypermethylated loci was enriched for genes functionally related to organ development, cell differentiation, and transcriptional regulation. A total of 330 CpGs located in 326 genes were hypomethylated in aged groups - to similar extents in both daughter groups and to a greater extent in centenarians. This set of hypomethylated loci was enriched for genes involved in defense responses, acute inflammation, and signal transduction. Approximately 150 CpGs located in 124 genes were significantly hypermethylated in daughters of centenarians compared with daughters of non-long-lived parents. This set was enriched for genes functionally related to nucleotide metabolism, nucleic acids synthesis, and cellular signaling. On the other hand, 67 CpGs located in 65 genes were significantly hypomethylated in the centenarians' daughters compared with daughters of non-long-lived parents. This set was enriched for genes related to downstream processes of the signal transmission. The most strongly pronounced differences in methylation levels between the two daughter groups were found for twelve CpGs (ten hypermethylated and two hypomethylated in centenarians' daughters) located in nine genes. Six of the hypermethylated genes (SLC38A4, SLC22A18, MGC3207, ECRG4, ATP13A4, and AGPAT2) are involved in metabolic processes, one hypermethylated (DUSP22) is a tumor-suppressor gene, and another hypermethylated gene (ZNF169) encodes a zinc finger DNA binding protein with unknown function. The function of the only hypomethylated gene (FLJ32569) is also unknown. Obviously, genome methylation in the daughters of centenarians is more stable compared with daughters of non-long-lived parents. Thus, epigenome stability and a more robust epigenetic control for nucleotide metabolism, nucleic acids synthesis, and signal transduction may contribute to increasing lifespan and healthy aging in centenarians.

Interestingly, the longevity-promoting mutation Ames dwarf *Prop1*<sup>df/df</sup> in mice has been found to increase the stability of the liver DNA methylome through chronological aging [39]. Significantly more age-related differentially methylated regions (aDMRs) were found in the wild-type mice compared with *Prop1*<sup>df/df</sup> mice. At shared hyper-aDMRs, the magnitude of the methylation increase with age was about the

same in wild type (WT) and dwarf mice. At hypo-aDMRs restricted to WT mice, young and old dwarf mice showed methylation levels comparable to young WT mice, while at the hypo-aDMRs restricted to Ames mice, young and old WT mice showed methylation levels comparable to old mutant mice. At shared hypo-aDMRs, dwarf mice showed consistently higher methylation than WT in both age groups, while the magnitude of the change with age was about the same. In dwarf mice, hypo-aDMRs were biased towards a higher methylation level. At hypo-aDMRs shared between WT and dwarf mice, the latter ones exhibited a higher initial level of methylation in young animals, thus potentially buffering them against the effects of age-associated hypomethylation. When hypo-aDMRs were compared between ad libitum (AL)-fed, CR and rapamycin-treated young and old mice, the methylation loss with chronological aging was suppressed by CR and, to a lesser extent, by rapamycin. At hyper-aDMRs, the methylation gain was also suppressed by CR. The protective effect of rapamycin was not significant in the whole population of hyper-aDMRs but was detectable in some individual aDMRs.

## 4 DNA Methylation as an Age Predictor

The aging rates are non-equal in different persons. Women are known to have a longer average lifespan compared with men. Aging can be accelerated by unhealthy life habits, such as smoking, or slowed down by the good ones, such as physical training. Molecular markers of biological, rather than chronological age, are needed to evaluate more precisely the degree of age-dependent deterioration in physical welfare. The age-related variance of DNA methylation seems to be a good contender for this role. Unfortunately, the age-related DNA methylation changes are masked by methylation variations caused by other factors or stochastic errors. In saliva samples of MZ twin pairs, 88 CpGs were found to have methylation levels significantly correlated with age, with 69 of these being positively correlated and 19 negatively correlated [40]. Most (83%) of these aCpGs were located within promoter CGIs. Three genes that showed a most clear correlation with age and had the widest range of methylation values, Edaradd, NPTX2 and Tom1L1, showed a clear correlation with age in men, but only two (Edaradd and Tom1L1) were correlated in women. The methylation levels of *Edaradd* and *Tom1L1* were linearly decreased with age, whereas the methylation level of NPTX2 was increased. Based on the methylation levels of just two CpGs (Edaradd and NPTX2), the age of the test subjects could be predicted with a 5-6 year accuracy, whereas the addition of one more site (located in ELN gene) reduced the mean absolute error (MAE) to ~3.5 years. No epigenetic drift was detected in the promoter regions for the CpGs studied.

Comparative studies of age-related methylation patterns in various tissues showed these patterns to be highly tissue-specific. Nevertheless, some loci have methylation levels significantly correlated with age in various tissues. These common methylated loci could be of the highest relevance to the internal mechanisms of aging, and their methylation status could be used as an epigenetic signature to estimate the biological age. The first non-cell-type-dependent epigenetic aging signature was deciphered based on the DNA methylation datasets from several independent studies that used the Illumina HumanMethylation27 BeadChip platform [41]. Of more than 450 aCpG sites found, most were hypermethylated with age and only 25 hypomethylated. This is in accord with the view noted above that hypermethylation is a dominant trend of specific methylation changes upon aging, whereas hypomethylation is less stringently regulated. The most accurate age predictions were obtained when a set of four hyper-aCpGs (TRIM58, KCN01DN, NPTX2 and GRIA2) were used. To further enhance the prediction accuracy, a hypoaCpG (BIRC4BP) was added to the set. When all five loci were used, the average prediction accuracy across all datasets was ±12.7 years, whereas the use of only three of the most reliable ones (NPTX2, GRIA2, KCNO1DN) enhanced the accuracy to  $\pm 11.4$  years. It should be noted that in this case the age prediction applied to various tissues and was gender-independent, whereas in the study described above [40] the prediction was based only on saliva samples. When the blood samples were investigated, the set of aCpGs with high predictive capability could be narrowed down to just three (ITGA2B, ASPA and PDE4C), and the accuracy of age prediction was  $\pm 4.5$  years [42]. Similar accuracy was obtained with blood samples using another set of five aCpGs (ELOVL2, Clorf132, TRIM59, KLF14 and FHL2) [43]. These same CpGs appeared to be informative to predict age in blood, saliva and buccal swab samples simultaneously with high accuracy (~4 years) [44]. Even the use of two of the most informative aCpGs located in ELOVL2 enabled age prediction in blood cells with reasonable accuracy (MAE = 7 years) [45]. For reasons that are still unknown, the methylation of a few CpG sites in ELOVL2 appeared to be the most robust single-gene predictor of age [46]. In a study of methylation in whole blood, DNA samples obtained from variously aged (9-99 years) subjects and cord blood samples, the highest correlation (0.92) with age was observed for CpG sites in the promoter CGI of the ELOVL2 gene, followed by promoter CGIs of FHL2 (0.80) and PENK (0.63). ELOVL2 showed a progressive increase in methylation beginning from the first stages of life and displayed the widest methylation range from 7% to 91%. Respective ranges for FHL2 and PENK were 12-53% and 1-27%. ELOVL2 encodes an enzyme involved in the synthesis of long ω3 and ω6 polyunsaturated fatty acids (PUFAs). Taking into account the important role that PUFAs play in the regulation of metabolism, inflammation and membrane integrity, it would be interesting to elucidate the possible role of *ELOVL2* expression in aging.

In a simple model that used a quadratic regression for *ELOVL2* methylation and linear regressions for methylation of *ASPA*, *PDE4C* and *EDARADD*, highly accurate age predictions for blood (MAE = 3.75 years) and tooth (MAE = 4.86 years) samples were obtained [47]. An epigenetic clock with reasonable accuracy (MAE~5 years) was built for human semen samples using linear regression of methylation levels of just three CpGs [48].

In a large-scale investigation using the Illumina HumanMethylation450 BeadChip platform, methylation levels of 485,577 CpG sites were analyzed in blood DNA from more than 650 volunteers of 19- to 101-years old [49]. An age prediction model ("Hannum clock") was built using a set of 71 aCpGs. The mean

error of age prediction by this model was  $\pm 3.9$  years (96% correlation between the chronological age and the predicted age). Nearly all markers in the model were located within or near genes with known functions in age-related conditions, such as Alzheimer's disease, cancer, tissue degradation, DNA damage and oxidative stress. The model was capable not only of predicting the age but also of revealing the factors that affect the personal rate of aging. For example, gender was found to affect age significantly, with epigenetic "aging" in men being approximately 4% faster than in women. Body mass index (BMI) was found not to affect the aging rate of blood, adipose and muscle cells, whereas age acceleration by approximately 2.2 years per each additional 10 BMI units was observed in the liver [49, 50]. When the model was used to estimate the age of tumors, these appeared to be around 40%more aged than respective normal cells of the same person. The age prediction model worked with DNA samples from other organs (breast, lung, kidney, and skin) with the same accuracy as with blood samples, when a linear offset specific for each organ was used. When the epigenetic predictive models were constructed using the same algorithm but based on the age-related methylation data from other organs (breast, lung and kidney), the main difference was in the sets of most informative CpG sites chosen. Only two CpGs near *ELOVL2* appeared to be common. Not only the methylation levels of aCpGs were changing with age, but also the variation limits of these methylation levels between different persons became larger for most sites. For any specific person, the extent of deviation of these values from the population averages appeared to be a fairly accurate measure of the individual aging rate.

An "ultimate" biological age predictor ("Horvath clock") was built based on all publicly available data concerning age-related variations of DNA methylation in various tissues and cell lines (nearly 8000 samples, 51 tissues and cell types) [51]. A total of 353 methylation sites were chosen by elastic net regression analyses that allowed the most reliable age prediction for various tissues and cells (96% correlation to chronological age; MAE = 3.6 years). Blood cells that have different life spans, CD14+ monocytes (myeloid lineage) living several weeks at the most, and CD4+ T cells (lymphoid lineage) living for months to years, were shown to have identical epigenetic ages in blood samples from healthy male subjects. Hence, epigenetic age reflects some internal methylome features, related to the chronological age of the person, not just the age-dependent peculiarities of the respective blood cells. The mean epigenetic age was highly correlated with the chronological age in most tissues. The variations in epigenetic age between different tissues of the same person were small. The notable exceptions were breast tissue in women (epigenetically older compared with other tissues) and sperm in men (epigenetically younger compared with other tissues). Surprisingly, though all brain regions of the same person have similar epigenetic ages in subjects younger than 80 years, the cerebellum and to a lesser extent the occipital cortex exhibit progressively negative epigenetic age acceleration in the older persons, i.e., these brain regions are younger than others [52]. This relative resistance of cerebellum to aging correlates with overexpression of 1239 genes compared with other brain regions, two RNA helicase superfamilies (SF1 and SF2) being the most enriched among these genes.

As could be expected the epigenetic age of embryonic stem cells (ESCs) is close to zero [51]. Interestingly, induced pluripotent stem cells (iPSCs) do not differ from ESCs by their epigenetic age. Therefore, epigenetic age is reset when iPSCs are produced from differentiated somatic cells. When cells are maintained in culture, ESCs and iPSCs included, their epigenetic age increases with each passage. However, the epigenetic age is clearly not a reflection of the mitotic age, since it tracks chronological age in tissues that are widely different in proliferative potential, including post-mitotic neurons. It is also not related to cell senescence since its correlation with chronological age is observed in immortal cell lines, such as ESCs.

For reasons that are still unclear, the pan-tissue epigenetic clock appeared not to be a good predictor for the age of skin fibroblasts. Moreover, no aging acceleration was found by this clock in fibroblast lines derived from the skin of patients with Hutchinson Gilford Progeria Syndrome (HGPS) – a disease with obvious manifestations of accelerated aging. By analyzing DNA methylation data sets obtained from human fibroblasts, keratinocytes, buccal cells, endothelial cells, blood and saliva, a new version of the epigenetic clock, referred to as the "skin & blood clock," was built [53]. A total of 391 CpGs were selected, of which 45 were shared with the blood-based Hannum clock and 60 CpGs – with the pan-tissue Horvath clock. This new clock outperformed the pan-tissue clock in fibroblasts, microvascular endothelial cells, buccal epithelial cells and keratinocytes. It also showed that fibroblasts from HGPS cases exhibit accelerated epigenetic aging. In blood samples, it appeared to be more accurate than both "classic" Horvath and Hannum clocks. It also correctly estimated gestational age when applied to cord blood samples. Regarding other cell types and tissues, including sorted neurons and glial cells, brain samples and liver samples, the new clock performed similarly to classic Horvath clock. The new clock correctly estimated the age of neonatal fibroblasts to be close to zero, whereas the pan-tissue clock estimated their age to be >10 years. The proliferation of human fibroblasts or keratinocytes in culture led to a continuous increase in epigenetic age measured by both new and classic clocks but the accuracy of the new clock was clearly better.

While the popular epigenetic clocks by Hannum and Horvath produce fairly good estimates of chronological age, their predictive power of aging acceleration associated with many age-related diseases is limited. This feature is probably a natural consequence of their having been built based on chronological age as the reference for selection of the most relevant CpGs. By definition, CpGs with methylations that do not display strict correlations with chronological age were excluded. On the other hand, the most relevant methylation sites to predict biological age could be those that signal the departure of biological age from chronological age and thus account for differences of health and physiological status between individuals of similar chronological ages. Such differences could be quantitatively described by "phenotypic age" derived from various clinical biomarkers that strongly predict relative risks of mortality, physical and cognitive disability, and visual manifestations of aging between same-aged individuals. Along these lines, a mathematical model of the phenotypic age [54]. When validated on a 12 years

follow-up cohort of more than 6200 individuals, phenotypic age was highly correlated with chronological age (r = 0.94). A 1 year increase in phenotypic age compared with chronological age was associated with a 9% increase in the risk of all-cause mortality and mortality from aging-related diseases, a 10% increase in the risk of mortality from cardiovascular disease, a 7% increase in the risk of mortality from cancer, and a 20% increase in the risk of mortality from diabetes. Furthermore, phenotypic age was highly correlated with comorbidity count and physical function measures. Thus, a more powerful epigenetic biomarker of aging could be developed by replacing the prediction of chronological age with the prediction of phenotypic age. A "PhenoAge" epigenetic clock has been constructed by selecting 513 CpGs whose methylation in blood samples appeared to be most predictive of phenotypic age [54]. Of these 513 CpGs, 41 were common with the classic Horvath clock, 6 were common with the Hannum clock and 5 with both clocks. The strongest correlation to age was observed for149 CpGs located in CGIs. Among genes that tended to be over-expressed in epigenetically older individuals, functional enrichment was observed for pro-inflammatory, response to nutrients, and growth hormone signaling pathways, multicellular organism growth, and regulation of DNA methylation. Among genes that were down-regulated in epigenetically older persons, enrichment was observed for processes involving the transcriptional and translational machinery, and damage recognition and repair. Epigenetic PhenoAge was highly correlated with phenotypic age and significantly associated with mortality risks when tested in five independent large human cohorts. A 1 year increase in epigenetic PhenoAge was associated with a 4.5% increase in the risk of all-cause mortality. In persons representing the top 5% (fastest agers) the risk of mortality was around 1.6 times higher than in the average person of similar chronologic age and approximately 2.6 times higher than in the bottom 5% (slowest agers). Irrespective of chronological age, higher epigenetic PhenoAge was associated with an increased number of coexisting morbidities, decreased likelihood of being disease-free, increased physical disabilities and increased coronary heart disease. It was significantly different between those who had never smoked, current smokers, and former smokers. Epigenetic PhenoAge appeared to differ significantly between racial groups, non-Hispanic blacks having the highest PhenoAge, and non-Hispanic whites the lowest. Individuals with higher education and income appeared younger. Increased exercise and fruit and vegetable consumption (rich in carotenoids) were associated with lower PhenoAge. PhenoAge acceleration was positively correlated with the circulating levels of C-reactive protein (CRP), insulin, glucose and triglycerides, and with the waist-to-hip ratio. Thus, the epigenetic PhenoAge clock appears to capture organismal functional state, beyond the that explained by chronological age, and accurately predicts morbidity and mortality risks.

Another epigenetic clock, DNAm "GrimAge," was built using 1030 unique CpGs to predict with maximal accuracy the negative outcomes of smoking and a selection of plasma proteins that have previously been associated with mortality or morbidity (adrenomedullin, CRP, PAI1, and GDF15) [55]. Among the CpGs selected, significant enrichment was found for 28 functional pathways, including MHC class II receptor activity, cytokine-mediated signaling pathway, response to

IFNy, regulation of protein sumoylation, endoderm formation, epigenetic regulation of gene expression and fatty acid transmembrane transport. A novel measure of epigenetic age acceleration, AgeAccelGrim, was defined as the difference between the DNAm GrimAge and chronological age. A positive value of AgeAccelGrim indicates that the DNAm GrimAge is higher than chronological age. In a 13.7 years follow-up study assessing time-to-death due to all-cause mortality, DNAm GrimAge was superior relative to all other epigenetic clocks in lifespan prediction. Previous epigenetic clocks showed a negative correlation between epigenetic aging acceleration and plasma biomarkers of vegetable consumption, but AgeAccelGrim was more accurate in this aspect. The self-reported proportion of carbohydrate consumption was associated with lower AgeAccelGrim, whereas an increased proportion of fat intake was associated with increased AgeAccelGrim. Triglyceride, insulin, glucose and plasma CRP levels were positively correlated with AgeAccelGrim, whereas HDL cholesterol levels were negatively correlated. BMI and waist-to-hip ratio were associated with increased AgeAccelGrim, whereas higher education and income were associated with lower AgeAccelGrim. AgeAccelGrim appeared to be superior in detecting the beneficial effects of physical exercise. Polyunsaturated omega-3 fatty acid intake was negatively correlated with AgeAccelGrim, the effect being more pronounced in males than in females.

A special version of the epigenetic clock was devised to predict the gestational age (GA) at birth based on DNA methylation in umbilical cord blood samples [56]. Of 148 aCpGs chosen by elastic net regression, only six were in common with those in classic Horvath clock. The accuracy of GA prediction by the clock was within 1.5 weeks (MAE = 1.24 weeks), which is about the same as errors of clinical estimates of GA based on ultrasound investigation.

Since "classic" versions of the epigenetic clock showed age-related DNA methylation to change more rapidly during childhood and adolescence, a special version of the clock "fine-tuned" to predict the age of children and adolescents with maximal precision was built [57]. Of 6350 aCpGs from the epigenome-wide association analysis, 116 of the previously known age-related sites were confirmed in children, but 83 novel aCpGs were selected as the best predictors. These sites collectively predict the chronological age in children and adolescents with high accuracy (correlation = 0.93, MAE = 0.62 years). This higher accuracy compared with "adult" clocks is most likely a consequence of rapid changes of aCpG methylation levels during childhood and adolescence noted above. Besides, children and adolescents have less confounding factors with regards to lifestyle.

As noted above, the age of sperm cells cannot be predicted accurately with the classic multi-tissue epigenetic clock [51]. It has been shown that sperm cells have a distinct pattern of age-associated changes in DNA methylation [58]. A longitudinal study of sperm DNA methylation in fertile donors at time points 9–19 years apart showed 139 aCpGs that lose methylation and only 8 aCpGs that gain methylation with age. The average methylation levels of these aCpGs were used to construct a linear regression model of age prediction [59]. Testing the model on several independent sperm samples showed a high accuracy for age prediction, with an average

standard deviation of only 0.877 years. When sperm ages were compared between individuals who had never smoked, smokers and long-term (>10 years) smokers of similar chronological ages, an increase of around 1.5% was observed in all smokers and an increase of 2.5% in long-term smokers compared with those who had never smoked. Thus, the model is capable not only of correctly predicting age based on sperm DNA methylation but also in predicting the aging-acceleration effects of smoking.

The epigenetic age could serve as a convenient marker in the assessment of the rejuvenation treatments efficiency. A comparative analysis of DNA methylation levels of peripheral blood mononuclear cells between semi-supercentenarians (mean age: 105.6 years), their offspring (mean age: 71.8 years), and age-matched controls (meanage: 69.8 years) showed that the offspring of semi-supercentenarians have a lower epigenetic age than age-matched controls (epigenetic age difference = 5.1 years), and that the epigenetic age of centenarians is approximately 8.6 years lower than their chronological ages [60]. The aging rates of different tissues of the same person can be used to identify those with diseases, such as cancer. Even more, the increased epigenetic aging in blood cells appeared to be highly predictive of future cancer [61-64]. An interesting example of the application of the epigenetic age concept was an estimation of the mortality risk [65]. It was found that a 5-year acceleration of epigenetic aging versus chronological aging results in a 16% increase of mortality risk, irrespective of general health, lifestyle and genetic factors. Similar results were obtained in another study which found that a 5-year higher epigenetic age compared with chronological age was associated with a 15–20% increase of all-cause, cancer and cardiovascular disease mortalities [66]. Generally, the epigenetic age appears to be a more robust predictor of biological aging and all-cause mortality compared with chronological age [67]. HIV infection, both chronic and recent, was shown to increase the epigenetic age by about 5 years in blood cells [68, 69] and by 7.4 years in brain tissue [68]. Down syndrome is known to be associated with segmental manifestations of accelerated aging, such as declining immune function and precocious Alzheimer's disease. It was found that Down syndrome subjects exhibit a significant epigenetic age acceleration (by 6.6 years on average) in blood and brain tissue but not in the buccal epithelium [70]. In addition, a strong correlation between epigenetic age acceleration in brain tissue and features of Alzheimer's disease was observed [71]. A significant epigenetic aging acceleration was also observed in blood samples of Parkinson's disease patients [72] and senior women with insomnia [73]. Memory decline, especially episodic memory, is strongly associated with aging. A lower epigenetic age was observed both at baseline and after a 15-year follow-up in subjects with maintained memory functions compared to those with memory decline [74]. Moreover, epigenetic age at follow-up was significantly predictive for the dementia occurrence, whereas chronological age was not. Excessive and chronic stress is widely believed to be associated with accelerated aging and increased occurrence of aging-related pathologies, such as cardiovascular disease, immune dysfunction and neuropsychiatric disorders. In a highly traumatized cohort of African Americans aged 18-77 years, cumulative life-time stress, but not childhood trauma or current stress alone, was found to be associated with accelerated epigenetic aging [75]. Since glucocorticoids are primary molecular mediators of the stress responses, the authors further studied their possible role in epigenetic aging acceleration. Of the 353 classic Horvath clock CpGs, 110 showed significant methylation changes 3 h after oral exposure to dexamethasone, with 98 of these decreased and 12 increased. Werner syndrome is one of the most extensively studied progeroid syndromes with multiple features of accelerated aging in early adulthood. As expected, a study showed that was also associated with significant acceleration of epigenetic aging by both Hannum and Horvath clocks [76].

Diets rich in whole grain and dietary fibers, fish and omega-3 fatty acids, fruits and vegetables, as well as increased physical activity, moderate alcohol consumption, and higher educational attainment have all been linked to good health and longevity. The effects of these lifestyle factors on the rate of epigenetic aging in blood cells were studied using both the Horvath (internal epigenetic aging) and Hannum (external epigenetic aging) clocks [77]. Small but statistically significant acceleration of epigenetic aging by the Hannum clock was correlated with plasma levels of triglycerides, CRP, insulin, glucose and  $\gamma$ -tocopherol, as well as with systolic blood pressure, waist-to-hip ratio and BMI. An inverse correlation was observed between Hannum age acceleration and fish intake, alcohol consumption, plasma levels of carotenoids, such as  $\alpha$ - and  $\beta$ -carotenes, lutein and zeaxanthin, and  $\beta$ -cryptoxanthin, and with education, income and exercise. The Horvath epigenetic aging rate showed weaker correlations with dietary and lifestyle factors.

A meta-analysis of blood DNA methylation data from 16,245 individuals from 18 cohort studies showed an association of education, smoking, obesity, alcohol intake and physical activity with the aging rate using the number of stochastic epimutations (SEMs) and three versions of the epigenetic clock [78]. The total number of SEMs takes into account whole-genome epigenetic deregulation during aging (epigenetic drift), whereas epigenetic clocks serve as markers of biological age. The level of education was significantly associated with the four DNA methylation markers investigated. Individuals with a lower education level had a higher number of SEMs, and higher epigenetic aging acceleration as assessed by the Horvath, Hannum, and PhenoAge clocks. The observed associations were still significant after the adjustment for smoking, BMI, alcohol and physical activity, but the estimated effects were moderately reduced. Current smokers had higher numbers of SEMs and higher Hannum and PhenoAge epigenetic aging acceleration compared with those who had never smoked. Former smokers had intermediate outcomes. The estimated effect size of the association between smoking and epigenetic aging biomarkers was comparable to those observed for education, except for the magnitude of the association with PhenoAge, which was significantly higher. A similar pattern of associations was observed for the effects of obesity. Obese individuals (BMI  $\ge$  30) had higher Horvath, Hannum, and PhenoAge epigenetic aging acceleration. The estimated effects of obesity were comparable to those of education, except in the case of PhenoAge, which was significantly higher. No significant difference was found between alcohol abstainers and occasional drinkers, but habitual drinkers had higher Horvath, Hannum and PhenoAge epigenetic aging acceleration. As observed

for the other risk factors, the higher estimated effects were observed by the PhenoAge clock. A negative effect of low physical activity was detectable with Horvath clock only. No significant differences in associations of each risk factor with epigenetic aging were detected between men and women. A significant interaction with age was found for the association of SEMs with education, smoking and obesity, with stronger effects in older individuals.

The predictive accuracy of epigenetic clocks can be enhanced when less universal models are used. For example, for blood cells, this reached a value of 2.6 years when 17 aCpGs were used [79]. In a follow-up study of the same persons 8 years later, the predicted increases in methylation levels of hypermethylated sites and decreases in methylation levels of hypomethylated sites were observed. Recently, longitudinal dynamics of the epigenetic age estimates by Horvath clock has been studied using two Finnish follow-up groups – Vitality 90+ (age 90 years at baseline, 4-year follow-up) and Young Finns (age 15–24 years at baseline, 25-year follow-up) [80]. The most notable finding was that the epigenetic age acceleration was stable across the lifespan, its rate being set before adulthood. At the age of 15 years, the slope of epigenetic aging was already set at an individual level that remained essentially the same over the 25-year follow-up period. At older ages, the rate of epigenetic aging was also steady.

A genome wide analysis of DNA methylation in liver biopsies from 45 donors aged between 13 and 90 years using the Illumina HumanMethylation450 BeadChip platform, showed that 8823 CpGs were differentially methylated with age, and most of these (5772) were positively correlated with age [81]. These hyper-aCpGs were enriched in bivalent chromatin domains. Multidimensional scaling of 8823 aCpGs showed a linear correlation with chronological age up to 60 years, while a trend to a plateau was observed at older ages. The same trend was confirmed when epigenetic age of liver samples was estimated by the Horvath epigenetic clock. Thus, slower epigenetic aging of the liver occurs after 60 years of age. In search of compact regions containing multiple aCpGs, 75 differentially methylated regions (aDMRs) were identified that contained 687 aCpGs in 89 genes, all positively correlated with aging. Not unexpectedly, the top-ranking aDMR mapped to the CGI of the ELOVL2 gene. The age-dependent DNA methylation signature identified appeared to be liver-specific, although for some aDMRs the age-association was similar between liver and several other tissues (for example, the CGI of ELOVL2). Of 687 aCpGs in the 75 aDMR signature, 80 aCpGs in 25 aDMRs were significantly associated with BMI. Lean subjects (BMI < 25) had lower DNA methylation levels compared with obese subjects of the same chronological age. Since all 75 aDMRs were hypermethylated with age, this result showed accelerated epigenetic aging in livers of obese subjects. Transcriptome-wide analysis of gene expression in the same liver samples showed 56 genes to be differentially expressed with age (aDEs). Of these, 47 were positively correlated with age. Unlike the methylation data, multidimensional scaling of the expression values of the 56 aDEs presented an almost linear association with chronological age between 13 and 90 years. Several functional gene sets were identified to be significantly enriched among positively correlated aDEs, namely those involved in allograft rejection, IFNy response, inflammatory response, epithelial-mesenchymal transition (EMT) and myogenesis, whereas genes with a negative correlation with age were enriched in metabolic processes. Of 56 aDEs, 11 genes that displayed age-associated changes in both methylation and expression were identified. Of these 11 genes, only three (*ZIC1*, *TSPYL5*, and *FZD2*) showed a significant correlation between methylation and expression levels.

To build an epigenetic clock model in the humpback whale, methylation of CpG sites in regulatory regions of eight genes, known to be differentially methylated with age in other mammals, was studied in skin DNA in a small population of whales with known ages [82]. Three sites were chosen that showed the strongest correlation with age, namely *TET2\_CpG* + 31 (methylation decreases with age), *CDKN2A\_CpG* + 297 and *GRIA2\_CpG* + 202 (methylation increases with age) (positions relative to the start codon). The predictive accuracy of the clock appeared to be fair (MAE = 3.75 years). The precision of the PyroMark system used to measure methylation levels was estimated to be about 2.2 years. Thus, a substantial share of the clock error could be attributable to error in measurement of methylation levels. Hence, it would be higher if more accurate methods of the methylation quantitation are used.

An epigenetic clock model for canid species was built by RRBS analysis of genome wide DNA methylation patterns in blood DNA samples of variously aged domestic dogs and gray wolves [83]. Statistically significant conservation of aCpGs between dogs, wolves and humans was observed for around 9000 CpGs. After an initial pre-selection of sites with the absolute correlation between methylation and age above 0.3, an elastic net regression was used to build an age-predictive model. The correlation between predicted and actual ages was 0.8, and the MAE was 0.8 years. The average number of CpGs in the individual regression models was approximately 120.

Several epigenetic clocks have been constructed based on DNA methylation data of mice. Using RRBS DNA methylomes of mouse blood samples, a robust epigenetic clock (mDNAm clock) was defined by a weighted average of DNA methylation of 90 CpG sites [84]. The mean square difference between the epigenetic age and the chronological age was ~16 days. The CpG sites contributing to the mDNAm clock were distributed across the chromosomes and did not match the sites observed in human clocks. Interventions known to increase lifespan and delay age-dependent phenotypic changes in mice (CR, growth hormone receptor knockout, Snell dwarf mutation) showed consistent reductions of epigenetic age compared with chronological age. Several iPSC lines obtained from kidney and lung fibroblasts of 10-week-old mice were calculated to be less than 1 month old by the mDNAm clock, whereas the original fibroblasts were about 300 days old by the same mDNAm clock. Thus, the mDNAm clock measured the biological age rather than the chronological age.

Using 107 liver methylomes of variously aged mice, a subset of 148 aCpGs was selected as an epigenetic clock in mouse liver [85]. Analysis of methylomes from mice subjected to lifespan-extending conditions (*Prop1df/df* dwarf mutation, CR, rapamycin treatment) showed lower epigenetic ages compared with age-matched

controls. Since previous studies have shown that these interventions not only extend lifespan but also improve tissue and physical functioning with age, the slowing of the epigenetic clock in livers appears to reflect the slowing of biological aging.

A multi-tissue epigenetic clock was constructed based on RRBS DNA methylomes of liver, lung, muscle, spleen, heart and brain samples from newborn to 31–41-week-old mice [86]. Only a small share of the age-correlated methylation changes was common to all tissues, suggesting that changes in DNA methylation are primarily driven by tissue-specific processes. The final multi-tissue epigenetic clock was based on methylation levels of 329 aCpGs. The clock performed well across all tissues and ages, with a MAE of around 3 weeks that corresponded to less than 8.5% error relative to the oldest age analyzed. Similar to human clocks, the performance of mouse age-predictor was more accurate in younger animals.

Four versions of the new epigenetic clock were built using combined data on DNA methylation in mouse tissues obtained from several RRBS and WGBS studies [87]. The first two clocks were built based on all aCpGs using two different regression methods. Two other clocks were constructed using the same two regression methods but based on 952 aCpGs located in highly conserved stretches of DNA. The accuracy of the chronological age prediction appeared to be maximal in clocks built by the elastic net regression on all aCpGs (MAE = 2.5 months). Two clocks based on conserved aCpGs were clearly inferior to those based on all CpGs. Nevertheless, their accuracy was impressive (MAE = 3.8 months). All epigenetic clocks appeared to be of use for multiple tissues, estimating with high accuracy the age in all tissues. One of the most notable conclusions of this study was that many clocks of similar performance could be constructed based on different sets of aCpGs. The data from 3 CR experiments showed significant slowing of aging by both versions of the elastic net regression clocks, but the largest anti-aging effect was observed with the ridge regression clock based on all aCpGs. This latter clock also detected a delay of epigenetic aging in three types of long-lived dwarf mutant mice (growth hormone receptor knock-out, Snell dwarf, and Ames dwarf).

When nine genomic regions of published predictors [84, 86] that contain compact groups of multiple aCpGs were analyzed by pyrosequencing in blood samples from variously aged (11–117 weeks) mice, the best correlation with age was observed for 15 aCpGs from five regions ( $R^2 = 0.99$ ; mean absolute deviation (MAD) = 2.76 weeks) [88]. Moreover, using just three aCpGs in the *Hsf4* gene, it was possible to predict mouse age with high accuracy ( $R^2=0.95$ ; MAD = 5.24 weeks). Similar results were obtained for three of the most informative aCpGs in the *Prima1*, *Hsf4* and *Kcns1* genes. Thus, age predictions based on a few CpG sites have similar precision, as previously described using multi-CpG predictors [84–86].

Since the mouse epigenetic multi-tissue clock [86] was tuned for tissues of younger mice, a new version was built based on RRBS DNA methylomes of different tissues collected from mice aged from 1 week to 35 months [89]. This new epigenetic clock included methylation of 435 CpGs and showed high accuracy in mice of both sexes across the entire lifespan ( $R^2 = 0.89$ , MAE = 72.7 days). Most new clock CpGs (58%) were located in open sea, 5% in shelves, and 11% in shore regions of the genome, and 26% were localized to CGIs. Of different genes in the

mouse genome, Kcns1 contained clock CpGs present both in the new clock and blood epigenetic clock [80]. This gene was represented by 6 CpGs in the blood clock and 11 CpGs in the new multi-tissue clock. As noted above, the CpGs of this gene have a high correlation with age [88]. By using DNA methylation data from variable sets of samples, 100 varieties of the new multi-tissue clock were built along the same lines. Of the first version multi-tissue clock of 435 CpGs, five CpGs were found in the blood clock [84] and multiple new varieties of the multi-tissue clock. A CpG site with the highest absolute weight, chr8:110168311, was found in all new clocks as well as in the blood clock. This site is located in the promoter region of the CALB2 gene. Three other sites from the same region were found in the new multitissue clock and had high weights. The CpG site chr2:164168131, which appeared in the new multi-tissue clock and the three-CpGs pyrosequencing-based epigenetic clock [88] was found in 95 clock varieties out of 100. At the same time, the majority of the new multi-tissue clock sites were detected in less than half of the 100 clock varieties. Thus, globally, methylation changes during aging allow for selection different subsets of aCpGs informative as epigenetic clocks.

### 5 Epigenetic Clock Resetting: A Root to Rejuvenation?

Unlike many other hallmarks of aging, age-related epigenetic changes could be fully or partially reversed to a "young" state. A natural example of the full reversal of aging is fertilization, a process when the fusion of haploid oocyte and sperm cells results in a diploid cell (zygote) that has a zero age. A similar age reset occurs upon somatic cell nuclear transfer (SCNT) when the nucleus of a somatic cell is transferred to cytoplasm of an enucleated oocyte, resulting in the development of a new individual. The oocyte cytoplasm seems to possess a capability to erase all of the aging features that have accumulated in the nucleus of the somatic cell. Thus, agerelated features of the cell nuclei, whatever their nature, are principally reversible.

Regarding epigenetic aging markers, the main difficulty in their resetting is probably the "needle in a haystack" problem. The epigenome of any cell is a complex mosaic of markers, in which the age-related ones are intermixed with a plethora of others. An easy way to reset the age-related epigenetic changes is to fully erase all existing epigenetic information and then rebuild it from scratch in a form corresponding to the zero age. Such erasure occurs during the first hours following fertilization [90, 91].

Another example of epigenetic age resetting is the production of induced pluripotent stem cells [92]. Recent observations have shown that iPSCs generated from senescent cells or centenarian donor cells rejuvenate telomeres, gene expression profiles, oxidative stress, and mitochondrial metabolism to the levels characteristic of ESCs, and can re-differentiate into fully rejuvenated cells [93–95]. Also, such iPSCs have an epigenetic age close to zero, as noted earlier [51]. Interestingly, fibroblasts from the extremely long-leaved rodent naked mole rat were found to be highly resistant to reprogramming into iPSCs [96]. Not only was the frequency of iPSCs colonies extremely low, but their ability to induce teratomas when injected into immunodeficient mice was much lower compared with mouse iPSCs. It is likely that this stable epigenome contributes to the extreme longevity and cancer resistance of the naked mole rat.

In all cases described above, resetting of the aging clock was coupled to cell dedifferentiation. This raises the question of is it possible to epigenetically reprogram cells to a more youthful state without disturbing its differentiation status? Two distinct transcriptional waves or phases have been distinguished during cellular reprogramming. The first stochastic phase is driven by c-Myc and Klf4, and the second deterministic phase is driven by Oct4, Sox2 and Klf4 [97, 98]. Importantly, changes in active and repressive histone markers, such as H3K4me3 and H3K27me3, occur during the first phase, changes in miRNA expression are observed during both phases, and alterations in DNA methylation take place mostly during the second phase [98]. Early in the reprogramming process, the four factors induce epigenetic changes by a stochastic mechanism, leading to an intermediate, partially reprogrammed state. Activation of endogenous Sox2 that occurs eventually in a small fraction of intermediate state cells, triggers a gene activation cascade that drives these cells to the pluripotent state. Since epigenetic reprogramming of somatic cells to iPSCs involves an intermediate state, it is possible that a partial rejuvenation could be achieved at this stage without complete reprogramming to iPSCs. Indeed, short-term expression of the Yamanaka factors Oct4, Sox2, Klf4 and c-Myc in fibroblasts derived from a premature aging mouse model has been found to ameliorate multiple age-associated hallmarks of aging, including the accumulation of DNA damage, cellular senescence markers (p16<sup>INK4a</sup>, p21<sup>CIP1</sup>), epigenetic dysregulation (changed levels of H3K9me3 andH4K20me3) and nuclear envelope defects, without loss of their cellular identity [99]. Although Yamanaka factor expression in vivo has been shown to lead to cancer development, their short-term cyclic induction ameliorated hallmarks of aging and extended the lifespan in a mouse model of premature aging. Additionally, this improved the regenerative capacity of pancreas and muscle tissue following injury in physiologically aged mice. To understand whether or not the partial reprogramming with Yamanaka factors is indeed a bona fide rejuvenation approach, the dynamics of epigenetic age was evaluated [100]. To this end, subpopulations of human dermal fibroblasts successfully transformed after Yamanaka factor transfection and expressing the TRA-1-60 antigen (a specific marker of hPSCs) were isolated and analyzed for gene expression and DNA methylation at regular time points over 49 days. A steady decrease of epigenetic age by 3.8 years per day was observed starting on day 3, which reached a zero level at day 20. While at first glance these data contradict those cited above [98], the populations of potential target sites of DNA methylation were different between these two studies. Partially reprogrammed cells at days 7 and 11 are known to express pluripotency markers at high levels but also have a high propensity to revert to the somatic state [101]. Since the epigenetic age is already significantly decreased at these time points, partial reprogramming of aging without affecting cellular identity is possible in principal.

In an experimental procedure known as heterochronic parabiosis, a shared circulatory system between young and old mice is established, thus exposing an old mouse to factors present in young serum. Heterochronic parabiosis was found to restore the activity of the Notch signaling pathway, as well as the proliferative and regenerative capacity of the aged skeletal muscle satellite cells [102]. Furthermore, heterochronic parabiosis increased proliferation of aged hepatocytes and restored their cEBP- $\alpha$  levels to values seen in young animals. Thus, age-related decline of progenitor cell activity could be modulated by systemic factors that change with age. Another study showed that the number of newly born neurons, proliferating cells and neural progenitors in the dentate gyrus of hippocampus decreased in the young heterochronic parabiont mice and increased in the old ones [103]. These findings suggest that the systemic milieu appears to affect the biological age of cells. In the adult brain, neural stem cells (NSCs) reside in a heterogeneous niche where they are in direct contact with blood vessels and the cerebrospinal fluid. The vasculature influences neural stem cell proliferation and differentiation by providing signaling molecules secreted from endothelial cells and by delivering systemic regulatory factors. In the aged niche, the vasculature deteriorates with a consequent reduction in blood flow and the neurogenic potential of NSCs declines, leading to reduced neuroplasticity and impaired cognition. In a mouse heterochronic parabiosis model, remodeling of the aged cerebral vasculature in response to young systemic factors was observed, producing noticeably higher blood flow [104]. Thus, circulating factors have diverse positive effects in aging mice, including enhancing neurogenesis and improving the vasculature in the cortex and other parts of the brain. It has been shown that the age-dependent decrease in adult neurogenesis in mouse hippocampus and concomitant cognitive impairment could be caused by loss of Tet2 activity and reduced hm5C levels [105]. An increase of Tet2 has been found in older parabionts after exposure to young blood compared with age-matched isochronic parabionts exposed to old blood. As noted above, hypermethylation of promoter-associated CGIs in genes encoding transcription factors is a prevailing feature of the aged brain methylome [26]. Thus, reactivation of Tet2 DNA demethylase appears to be relevant to brain rejuvenation.

About 70 circulatory proteins from young parabiont mice were found in muscle tissue of old parabiont mice [106]. Many of these proteins possess rejuvenating promyogenic properties and are expected to synergize when reaching the old muscle tissue simultaneously. LIF-1 has been shown to enhance the repair of injured muscle. Cripto and cerberus1 act as antagonists of TGF $\beta$ 1 that increases with age and this inhibits regeneration of old muscle. GDF5 is a TGF $\beta$  family member that promotes muscle innervation known to decline with age. Follistatin might counteract the effect of myostatin known to inhibit muscle stem cell proliferation. Cerberus1 and DKK-1 antagonize the age-elevated Wnt pathway activity. These "young" proteins also have known beneficial effects on other tissues. Of special note, leptin was identified as one of these "young" proteins and it broadly regulates hormonal networks, including those controlling reproduction and metabolism, and it has numerous anti-aging effects. Leptin interacts with oxytocin in its positive effects on the health of muscle, bone, brain and in reduction of obesity. Moreover, since both

leptin and oxytocin activity decline with aging, the increase of leptin/oxytocin axis might represent a key event in mammalian aging that is rescued by heterochronic parabiosis.

Genome-wide microarray analysis of hippocampi showed a distinct gene expression profile between aged isochronic (aged-aged) and aged heterochronic (aged-young) parabiont mice [107]. Furthermore, genes related to synaptic plasticity signaling pathways, including *Creb*, were among the top gene ontology enrichment categories associated with heterochronic parabiosis. These changes in gene expression were correlated with increased numbers of cells expressing the immediate early genes *Egr1* and *c-Fos* and with increased dendritic spine density on granule cell neurons in the dentate gyrus of heterochronic compared with isochronic parabionts. Collectively, the data described above show that exposure to young blood counteracts aging at the molecular, structural, functional and cognitive levels in the aged brain.

Effects of the temporal inhibition of NF-kB activity on cell aging were tested in chronologically aged transgenic mice with the NF- $\kappa B$  gene conditionally repressed in epidermal skin cells [108]. Comparison of the global expression profiles between young and old skin samples revealed 414 genes that were significantly altered, with most of these upregulated, in old skin. About 50% of these age-dependent genes were putative direct targets of NF- $\kappa$ B, consistent with the proposed role of NF- $\kappa$ B in aging. Upon 2-week NF-xB blockade in old skin, the expression of 225 of the 414 age-related genes returned to levels indistinguishable from those in the young skin samples. Globally, the expression profile of the NF-kB-quenched aged skin was more similar to that of young skin than to the control aged skin. Thus, NF-KB activity is required to maintain a substantial portion of the global gene expression program induced with age in murine skin. In addition to alterations in gene expression, aged skin is characterized by epidermal atrophy, decreased proliferative capacity and increased frequency of cellular senescence. Aged murine tissues also exhibit increased expression of SA-β-gal and p16<sup>INK4A</sup>, two markers of cellular senescence. The  $NF \cdot \kappa B$  blockade increased epidermal thickness in old skin to a degree intermediate between young and old skin, increased cell proliferation and significantly decreased expression of SA-β-gal and p16<sup>INK4A</sup>. In addition, skin constitution and general condition improved. Importantly, a reversal of skin cell senescence and increased proliferative capacity induced by NF- $\kappa B$  blockade occurred with preservation of normal tissue homeostasis and differentiation. Cell proliferation occurred predominantly in the basal layer of the epidermis, the normal proliferative compartment, and the spatial organization of the mature epidermal stratification was intact. Thus, NF-kB activity is required to maintain cellular senescence associated with chronological aging in murine skin. These data suggest that many features of mammalian aging may not be due to the passive accumulation of stochastic cellular damages and errors but rather are actively enforced by special gene expression programs and therefore can be substantially reversed by selective gene expression interventions. The NF-kB action in skin aging appears to be cell-autonomous since the reversion of the aging features was possible in limited patches of the epidermis in otherwise old animals. Since NF-kB is a transcription factor responsive to oxidative stress, DNA damage, growth signals and immune activation, and it acts on a number of target genes, it seems to be in an ideal position to transduce diverse extracellular signals to adaptive changes in gene expression and tissue homeostasis. The contribution of specific NF- $\kappa$ B target genes to aging remains unknown but it seems likely that its biological effects may be mediated by combined effects of a large number of the NF- $\kappa$ B responsive genes.

HSC activity decreases with age, manifesting in reduced self-renewal, hematopoiesis and lymphopoiesis [109]. In mammals, the target of rapamycin (mTOR) pathway integrates multiple signals from nutrients, growth factors, and oxygen to regulate cell growth, proliferation and survival [110]. Conditional deletion of the *Tsc1* gene in HSCs of young adult mice drove them from quiescence into a state of rapid cycling, increased mitochondrial biogenesis and elevated levels of reactive oxygen species [111]. Importantly, this deletion dramatically reduced both hematopoiesis and self-renewal of HSCs and led to constitutive activation of mTOR. In murine HSCs, mTOR activity increases with age, whereas treatment of old mice with the mTOR inhibitor rapamycin significantly increased their life span [112]. Moreover, the treatment caused a significant increase in the proliferative activity of HSCs and a decrease in the expression of  $p16^{lnk4a}$  and  $p19^{Arf}$ , known markers of cell aging. In addition, old mice have impaired B cell generation due to decreased numbers of pre-B cells and rapamycin treatment enhanced the generation of B cells due to a four-fold increase in the number of pre-B cells.

Thus, the rejuvenation of differentiated or committed cells can be achieved without disturbing their differentiation programs. These results suggest the possibility of targeted therapies to reverse individual features of aging and to alleviate age-related pathologies in the elderly.

## 6 Conclusions

Both epigenetic drift and the epigenetic clock contribute to age-related changes in DNA methylation but in fundamentally different ways [113]. Epigenetic drift is a result of essentially stochastic events and, therefore, leads to increased discordance between individual epigenomes across the lifespan. Conversely, the epigenetic clock refers to programmed changes in methylation of specific CpG sites that are consistently related to age between individuals of the same species. Despite this fundamental difference, both kinds of DNA methylation changes could be affected by environmental and internal stimuli or occur without any obvious cause. Environmental exposures are generally believed to be highly variable between individuals, leading to diverged epigenetic changes that influence the rate of the epigenetic clock. However, the inter-individual consistency of epigenetic clocks suggests that they are a manifestation of some special "epigenetic program of aging." Both kinds of DNA methylation changes with age probably affect age-related phenotypes. Epigenetic drift leads in a global decrease in precision of epigenetic regulation with

age, probably resulting in increased transcriptional noise and tissue dysfunction. Epigenetic clocks could be useful in multiple ways as an accurate molecular biomarker of aging. The crucial question is whether epigenetic clocks are just manifestations or driving forces of aging. An appealing hypothesis is that epigenetic drift is just a by-product of the aging process, whereas epigenetic clocks are at the heart of the aging mechanisms, reflecting an evolutionary selected programmed mechanism. Methylation changes underlying epigenetic clocks start well before the time when the phenotypic manifestations of aging are evident. The genes most closely linked to epigenetic clocks are enriched for those involved in development and differentiation, such as polycomb targets and bivalent chromatin domains. The apparent continuity of epigenetic clocks across development and aging may reflect a common program of development that involves an aging step at older ages. Since, both phenotypic manifestations of aging and epigenetic age can be alleviated by prolongevity treatments, such as CR, it is tempting to suggest that slowing of the clock contributes to lifespan extension. Studies in which rejuvenation has been achieved via targeting the epigenome [99, 100] support this view and make epigenetic intervention appear to be one of the most promising ways to long life and healthy aging.

Aging is generally believed to be an inevitable and essentially irreversible process in all living organisms. However, multiple studies described above have shown that it could be completely reversed at the cellular level and reversed to a considerable extent at the organismal level. Cellular reprogramming has been shown to reset the age of somatic cells to zero. On the whole organism level, manipulations of specific signaling pathways, such as the insulin/IGF-1, NF $\kappa$ B, mTOR, AMPK and sirtuin networks, and external interventions, including CR, physical activity and functional foods, were shown to extend the life span of model animals and slow down epigenetic aging. Heterochronic parabiosis experiments have shown that at the organismal level aging not only could be slowed down but also reversed to a significant extent, at least in some organs. It would be interesting to determine whether these rejuvenating effects of parabiosis are caused by epigenetic changes.

Studies of the aging details using predictive epigenetic models could have many practical implications, from health assessment to forensic analysis. As the predictive accuracy of the models improves, it seems likely that the biological age, measured by epigenetic clocks, might become more useful in clinical practice than chronological age. Finally, epigenetic clocks could become indispensable as markers of biological age in evaluating the efficiency of the new rejuvenation procedures.

Conflict of Interest The authors confirm that this article content has no conflict of interest.

#### References

- 1. Holliday R (2006) Aging is no longer an unsolved problem in biology. Ann N Y Acad Sci 1067:1–9
- Lopez-Otin C, Blasco MA, Partridge L, Serrano M, Kroemer G (2013) The hallmarks of aging. Cell 153(6):1194–1217

- Ashapkin VV, Kutueva LI, Vanyushin BF (2015) Aging epigenetics: accumulation of errors or realization of a specific program? Biochemistry (Moscow) 80(11):1406–1417
- Ashapkin VV, Kutueva LI, Vanyushin BF (2017) Aging as an epigenetic phenomenon. Curr Genomics 18(5):385–407
- Sen P, Shah PP, Nativio R, Berger SL (2016) Epigenetic mechanisms of longevity and aging. Cell 166(4):822–839
- Berdyshev GD, Korotaev GK, Boyarskikh GV, Vanyushin BF (1967) Nucleotide composition of DNA and RNA from somatic tissues of humpback salmon and its changes during spawning. Biokhimiia 32(5):988–993
- Vanyushin BF, Nemirovsky LE, Klimenko VV, Vasiliev VK, Belozersky AN (1973) The 5-methylcytosine in DNA of rats: tissue and age specificity and the changes induced by hydrocortisone and other agents. Gerontologia 19(3):138–152
- Romanov GA, Vanyushin BF (1981) Methylation of reiterated sequences in mammalian DNAs: effects of the tissue type, age, malignancy and hormonal induction. Biochim Biophys Acta 653(2):204–218
- De Cecco M, Criscione SW, Peckham EJ, Hillenmeyer S, Hamm EA, Manivannan J et al (2013) Genomes of replicatively senescent cells undergo global epigenetic changes leading to gene silencing and activation of transposable elements. Aging Cell 12(2):247–256
- De Cecco M, Ito T, Petrashen AP, Elias AE, Skvir NJ, Criscione SW (2019) L1 drives IFN in senescent cells and promotes age-associated inflammation. Nature 566(7742):73–78
- Wilson VL, Smith RA, Mag S, Cutler RG (1987) Genomic 5-methyldeoxycytidine decreases with age. J Biol Chem 262(21):9948–9951
- Singhal RP, Mays-Hoopes LL, Eichhorn GL (1987) DNA methylation in aging of mice. Mech Ageing Dev 41(3):199–210
- Wilson VL, Jones PA (1983) DNA methylation decreases in aging but not in immortal cells. Science 220(4601):1055–1057
- 14. Wigler M, Levy D, Perucho M (1981) The somatic replication of DNA methylation. Cell 24(1):33–40
- Stein R, Gruenbaum Y, Pollack Y, Razin A, Cedar H (1982) Clonal inheritance of the pattern of DNA methylation in mouse cells. Proc Natl Acad Sci U S A 79(1):61–65
- Issa J-PJ, Ottaviano YL, Celano P, Hamilton SR, Davidson NE, Baylin SB (1994) Methylation of the oestrogen receptor CpG island links ageing and neoplasia in human colon. Nat Genet 7(4):536–540
- 17. Ahuja N, Li Q, Mohan AL, Baylin SB, Issa J-PJ (1998) Aging and DNA methylation in colorectal mucosa and cancer. Cancer Res 58(23):5489–5494
- Waki T, Tamura G, Sato M, Motoyama T (2003) Age-related methylation of tumor suppressor and tumor-related genes: an analysis of autopsy samples. Oncogene 22(26):4128–4133
- So K, Tamura G, Honda T, Homma N, Waki T, Togawa N et al (2006) Multiple tumor suppressor genes are increasingly methylated with age in non-neoplastic gastric epithelia. Cancer Sci 97(11):1155–1158
- Fraga MF, Ballestar E, Paz MF, Ropero S, Setien F, Ballestar ML et al (2005) Epigenetic differences arise during the lifetime of monozygotic twins. Proc Natl Acad Sci U S A 102(30):10604–10609
- Christensen BC, Houseman EA, Marsit CJ, Zheng S, Wrensch MR, Wiemels JL et al (2009) Aging and environmental exposures alter tissue-specific DNA methylation dependent upon CpG island context. PLoS Genet 5:e1000602. https://doi.org/10.1371/journal.pgen.1000602
- 22. Maegawa S, Hinka G, Kim HS, Shen L, Zhang L, Zhang J et al (2010) Widespread and tissue specific age-related DNA methylation changes in mice. Genome Res 20(3):332–340
- Rakyan VK, Down TA, Maslau S, Andrew T, Yang T-P, Beyan H et al (2010) Human agingassociated DNA hypermethylation occurs preferentially at bivalent chromatin domains. Genome Res 20(4):434–439
- Johansson A, Enroth S, Gyllensten U (2013) Continuous aging of the human DNA methylome throughout the human lifespan. PLoS One 8:e67378. https://doi.org/10.1371/journal. pone.0067378

- 25. Tserel L, KoldeR LM, Tretyakov K, Kasela S, Kisand K et al (2015) Age-related profiling of DNA methylation in CD8+ T cells reveals changes in immune response and transcriptional regulator genes. Sci Rep 5:13107. https://doi.org/10.1038/srep13107
- 26. Hernandez DG, Nalls MA, Gibbs J, Arepalli S, van der Brug M, Chong S et al (2011) Distinct DNA methylation changes highly correlated with chronological age in the human brain. Hum Mol Genet 20(6):1164–1172
- 27. Martino D, Loke YJ, Gordon L, Ollikainen M, Cruickshank MN, Saffery R et al (2013) Longitudinal, genome-scale analysis of DNA methylation in twins from birth to 18 months of age reveals rapid epigenetic change in early life and pair-specific effects of discordance. Genome Biol 14:R42. https://doi.org/10.1186/gb-2013-14-5-r42
- 28. Day K, Waite LL, Thalacker-Mercer A, West A, Bamman MM, Brooks JD et al (2013) Differential DNA methylation with age displays both common and dynamic features across human tissues that are influenced by CpG landscape. Genome Biol 14:R102. https://doi. org/10.1186/gb-2013-14-9-r102
- 29. Zykovich A, Hubbard A, Flynn JM, Tarnopolsky M, Fraga MF, Kerksick C et al (2014) Genome-wide DNA methylation changes with age in disease-free human skeletal muscle. Aging Cell 13(2):360–366
- Bormann F, Rodriguez-Paredes M, Hagemann S, Manchanda H, Kristof B, Gutekunst J et al (2016) Reduced DNA methylation patterning and transcriptional connectivity define human skin aging. Aging Cell 15(3):563–571
- Maegawa S, Lu Y, Tahara T, Lee JT, Madzo J, Liang S et al (2017) Caloric restriction delays agerelated methylation drift. Nat Commun 8:539. https://doi.org/10.1038/s41467-017-00607-3
- 32. Masser DR, Hadad N, Porter HL, Mangold CA, Unnikrishnan A, Ford MM et al (2017) Sexually divergent DNA methylation patterns with hippocampal aging. Aging Cell 16(6):1342–1352
- 33. Hahn O, Gronke S, Stubbs TM, Ficz G, Hendrich O, Krueger F et al (2017) Dietary restriction protects from age-associated DNA methylation and induces epigenetic reprogramming of lipid metabolism. Genome Biol 18:56. https://doi.org/10.1186/s13059-017-1187-1
- 34. Sziraki A, Tyshkovskiy A, Gladyshev VN (2018) Global remodeling of the mouse DNA methylome during aging and in response to calorie restriction. Aging Cell 17:e12738. https:// doi.org/10.1111/acel.12738
- 35. Lowe R, Barton C, Jenkins CA, Ernst C, Forman O, Fernandez-Twinn DS et al (2018) Ageing-associated DNA methylation dynamics are a molecular readout of lifespan variation among mammalian species. Genome Biol 19:22. https://doi.org/10.1186/s13059-018-1397-1
- 36. Hjelmborg JB, Iachine I, Skytthe A, Vaupel JW, McGue M, Koskenvuo M et al (2006) Genetic influence on human lifespan and longevity. Hum Genet 119(3):312–321
- 37. Sebastiani P, Nussbaum L, Andersen SL, Black MJ, Perls TT (2016) Increasing sibling relative risk of survival to older and older ages and the importance of precise definitions of "aging," "life span," and "longevity.". J Gerontol A Biol Sci Med Sci 71(3):340–346
- Gentilini D, Mari D, Castaldi D, Remondini D, Ogliari G, Ostan R et al (2013) Role of epigenetics in human aging and longevity: genome-wide DNA methylation profile in centenarians and centenarians' offspring. Age 35(5):1961–1973
- 39. Cole JJ, Robertson NA, Rather MI, Thomson JP, McBryan T, Sproul D et al (2017) Diverse interventions that extend mouse lifespan suppress shared age-associated epigenetic changes at critical gene regulatory regions. Genome Biol 18:58. https://doi.org/10.1186/ s13059-017-1185-3
- 40. Bocklandt S, Lin W, Seh ME, Sanchez FJ, Sinsheimer JS, Horvath S et al (2011) Epigenetic predictor of age. PLoS One 6:e14821. https://doi.org/10.1371/journal.pone.0014821
- Koch CM, Wagner W (2011) Epigenetic aging signature to determine age in different tissues. Aging 3(10):1018–1027
- 42. Weidner CI, Lin Q, Koch CM, Eisele L, Beier F, Ziegler P et al (2014) Aging of blood can be tracked by DNA methylation changes at just three CpG sites. Genome Biol 15:R24. https:// doi.org/10.1186/gb-2014-15-2-r24

- 43. Zbieć-Piekarska R, Spólnicka M, Kupiec T, Parys-Proszek A, Makowska Ż, Pałeczka A et al (2015) Development of a forensically useful age prediction method based on DNA methylation analysis. Forensic Sci Int Genet 17:173–179
- 44. Jung SE, Lim SM, Hong SR, Lee EH, Shin KJ, Lee HY (2019) DNA methylation of the ELOVL2, FHL2, KLF14, Clorf132/MIR29B2C, and TRIM59 genes for age prediction from blood, saliva, and buccal swab samples. Forensic Sci Int Genet 38:1–8
- 45. Zbieć-Piekarska R, Spólnicka M, Kupiec T, Makowska Ż, Spas A, Parys-Proszek A et al (2015) Examination of DNA methylation status of the ELOVL2 marker may be useful for human age prediction in forensic science. Forensic Sci Int Genet 14:161–167
- 46. Garagnani P, Bacalini MG, Pirazzini C, Gori D, Giuliani C, Mari D et al (2012) Methylation of *ELOVL2* gene as a new epigenetic marker of age. Aging Cell 11(6):1132–1134
- Bekaert B, Kamalandua A, Zapico SC, Van de Voorde W, Decorte R (2015) Improved age determination of blood and teeth samples using a selected set of DNA methylation markers. Epigenetics 10(10):922–930
- Lee HY, Jung S-E, Oh YN, Choi A, Yang WI, Shin K-J (2015) Epigenetic age signatures in the forensically relevant body fluid of semen: a preliminary study. Forensic Sci Int Genet 19:28–34
- Hannum G, Guinney J, Zhao L, Zhang L, Hughes G, Sadda S et al (2013) Genome-wide methylation profiles reveal quantitative views of human aging rates. Mol Cell 49(2):359–367
- 50. Horvath S, Erhart W, Brosch M, Ammerpohl O, von Schönfels W, Ahrens M et al (2014) Obesity accelerates epigenetic aging of human liver. Proc Natl Acad Sci U S A 111(43):15538–15543
- 51. Horvath S (2013) DNA methylation age of human tissues and cell types. Genome Biol 14:R115. https://doi.org/10.1186/gb-2013-14-10-r115
- Horvath S, Mah V, Lu AT, Woo JS, Choi O-W, Jasinska AJ et al (2015) The cerebellum ages slowly according to the epigenetic clock. Aging (Albany NY) 7(5):294–305
- Horvath S, Oshima J, Martin GM, Lu AT, Quach A, Cohen H et al (2018) Epigenetic clock for skin and blood cells applied to Hutchinson Gilford Progeria Syndrome and ex vivo studies. Aging (Albany NY) 10(7):1758–1775
- Levine ME, Lu AT, Quach A, Chen BH, Assimes TL, Bandinelli S et al (2018) An epigenetic biomarker of aging for lifespan and healthspan. Aging (Albany NY) 10(4):573–591
- 55. Lu AT, Quach A, Wilson JG, Reiner AP, Aviv A, Raj K et al (2019) DNA methylation GrimAge strongly predicts lifespan and healthspan. Aging (Albany NY) 11(2):303–327
- 56. Knight AK, Craig JM, Theda C, Bækvad-Hansen M, Bybjerg-Grauholm J, Hansen CS et al (2016) An epigenetic clock for gestational age at birth based on blood methylation data. Genome Biol 17:206. https://doi.org/10.1186/s13059-016-1068-z
- 57. Li C, Gao W, Gao Y, Yu C, Lv J, Lv R et al (2018) Age prediction of children and adolescents aged 6-17 years: an epigenome-wide analysis of DNA methylation. Aging (Albany NY) 10(5):1015–1026
- Jenkins TG, Aston KI, Pflueger C, Cairns BR, Carrell DT (2014) Age-associated sperm DNA methylation alterations: possible implications in offspring disease susceptibility. PLoS Genet 10:e1004458. https://doi.org/10.1371/journal.pgen.1004458
- Jenkins TG, Aston KI, Cairns BR, Smith A, Carrell DT (2017) Paternal germ line aging: DNA methylation age prediction from human sperm. BMC Genomics 19:763. https://doi. org/10.1186/s12864-018-5153-4
- 60. Horvath S, Pirazzini C, Bacalini MG, Gentilini D, Di Blasio AM, Delledonne M et al (2015) Decreased epigenetic age of PBMCs from Italian semi-supercentenarians and their offspring. Aging (Albany NY) 7(12):1159–1170
- Levine ME, Hosgood HD, Chen B, Absher D, Assimes T, Horvath S (2015) DNA methylation age of blood predicts future onset of lung cancer in the women's health initiative. Aging (Albany NY) 7(9):690–699
- 62. Zheng Y, Joyce BT, Colicino E, Liu L, Zhang W, Dai Q et al (2016) Blood epigenetic age may predict cancer incidence and mortality. EBioMedicine 5:68–73

- 63. Ambatipudi S, Horvath S, Perrier F, Cuenin C, Hernandez-Vargas H, Le Calvez-Kelm F et al (2017) DNA methylome analysis identifies accelerated epigenetic ageing associated with postmenopausal breast cancer susceptibility. Eur J Cancer 75:299–307
- 64. Durso DF, Bacalini MG, Sala C, Pirazzini C, Marasco E, Bonafé M et al (2017) Acceleration of leukocytes' epigenetic age as an early tumor- and sex-specific marker of breast and colorectal cancer. Oncotarget 8(14):23237–23245
- 65. Marioni RE, Shah S, McRae AF, Chen BH, Colicino E, Harris SE et al (2015) DNA methylation age of blood predicts all-cause mortality in later life. Genome Biol 16:25. https://doi. org/10.1186/s13059-015-0584-6
- 66. Perna L, Zhang Y, Mons U, Holleczek B, Saum K-U, Brenner H (2016) Epigenetic age acceleration predicts cancer, cardiovascular, and all-cause mortality in a German case cohort. Clin Epigenetics 8:64. https://doi.org/10.1186/s13148-016-0228-z
- 67. Chen BH, Marioni RE, Colicino E, Peters MJ, Ward-Caviness CK, Tsai P-C et al (2016) DNA methylation-based measures of biological age: meta-analysis predicting time to death. Aging (Albany NY) 8(9):1844–1859
- Horvath S, Levine AJ (2015) HIV-1 infection accelerates age according to the epigenetic clock. J Infect Dis 212(10):1563–1573
- 69. Gross AM, Jaeger PA, Kreisberg JF, Licon K, Jepsen KL, Khosroheidari M et al (2016) Methylome-wide analysis of chronic HIV infection reveals five-year increase in biological age and epigenetic targeting of HLA. Mol Cell 62(2):157–168
- Horvath S, Garagnani P, Bacalini MG, Pirazzini C, Salvioli S, Gentilini D (2015) Accelerated epigenetic aging in Down syndrome. Aging Cell 14(3):491–495
- 71. Levine ME, Lu AT, Bennett DA, Horvath S (2015) Epigenetic age of the pre-frontal cortex is associated with neuritic plaques, amyloid load, and Alzheimer's disease related cognitive functioning. Aging (Albany NY) 7(12):1198–1211
- Horvath S, Ritz BR (2015) Increased epigenetic age and granulocyte counts in the blood of Parkinson's disease patients. Aging (Albany NY) 7(12):1130–1142
- 73. Carroll JE, Irwin MR, Levine M, Seeman TE, Absher D, Assimes T et al (2016) Epigenetic aging and immune senescence in women with insomnia symptoms: findings from the women's health initiative study. Biol Psychiatry 81(2):136–144
- 74. Degerman S, Josefsson M, Adolfsson AN, Wennstedt S, Landfors M, Haider Z et al (2017) Maintained memory in aging is associated with young epigenetic age. Neurobiol Aging 55:167–171
- 75. Zannas AS, Arloth J, Carrillo-Roa T, Iurato S, Röh S, Ressler KJ et al (2015) Lifetime stress accelerates epigenetic aging in an urban, African American cohort: relevance of glucocorticoid signaling. Genome Biol 16:266. https://doi.org/10.1186/s13059-015-0828-5
- Maierhofer A, Flunkert J, Oshima J, Martin GM, Haaf T, Horvath S (2017) Accelerated epigenetic aging in Werner syndrome. Aging (Albany NY) 9(4):1143–1152
- Quach A, Levine ME, Tanaka T, Lu AT, Chen BH, Ferrucci L et al (2017) Epigenetic clock analysis of diet, exercise, education, and lifestyle factors. Aging (Albany NY) 9(2):419–446
- Fiorito G, McCrory C, Robinson O, Carmeli C, Rosales CO, Zhang Y et al (2019) Socioeconomic position, lifestyle habits and biomarkers of epigenetic aging: a multi-cohort analysis. Aging (Albany NY) 11(7):2045–2070
- 79. Florath I, Butterbach K, Müller H, Bewerunge-Hudler M, Brenner H (2014) Cross-sectional and longitudinal changes in DNA methylation with age: an epigenome-wide analysis revealing over 60 novel age-associated CpG sites. Hum Mol Genet 23(5):1186–1201
- 80. Kananen L, Marttila S, Nevalainen T, Kummola L, Junttila I, Mononen N et al (2016) The trajectory of the blood DNA methylome ageing rate is largely set before adulthood: evidence from two longitudinal studies. Age (Dordr) 38:65. https://doi.org/10.1007/s11357-016-9927-9
- Bacalini MG, Franceschi C, Gentilini D, Ravaioli F, Zhou X, Remondini D et al (2019) Molecular aging of human liver: an epigenetic/transcriptomic signature. J Gerontol A Biol Sci Med Sci 74(1):1–8

- Polanowski AM, Robbins J, Chandler D, Jarman SN (2014) Epigenetic estimation of age in humpback whales. Mol Ecol Resour 14(5):976–987
- Thompson MJ, von Holdt B, Horvath S, Pellegrini M (2017) An epigenetic aging clock for dogs and wolves. Aging (Albany NY) 9(3):1055–1068
- Petkovich DA, Podolskiy DI, Lobanov AV, Lee S-G, Miller RA, Gladyshev VN (2017) Using DNA methylation profiling to evaluate biological age and longevity interventions. Cell Metab 25(4):954–960
- Wang T, Tsui B, Kreisberg JF, Robertson NA, Gross AM, Yu MK et al (2017) Epigenetic aging signatures in mice livers are slowed by dwarfism, calorie restriction and rapamycin treatment. Genome Biol 18:57. https://doi.org/10.1186/s13059-017-1186-2
- Stubbs TM, Bonder MJ, Stark A-K, Krueger F, Ageing Clock Team BI, von Meyenn F et al (2017) Multi-tissue DNA methylation age predictor in mouse. Genome Biol 18:68. https:// doi.org/10.1186/s13059-017-1203-5
- Thompson MJ, Chwiałkowska K, Rubbi L, Lusis AJ, Davis RC, Srivastava A et al (2018) A multi-tissue full lifespan epigenetic clock for mice. Aging (Albany NY) 10(10):2832–2854
- Han Y, Eipel M, Franzen J, Sakk V, Dethmers-Ausema B, Yndriago L et al (2018) Epigenetic age-predictor for mice based on three CpG sites. elife 7:e37462. https://doi.org/10.7554/ eLife.37462
- Meer MV, Podolskiy DI, Tyshkovskiy A, Gladyshev VN (2018) A whole lifespan mouse multi-tissue DNA methylation clock. elife 7:e40675. https://doi.org/10.7554/eLife.40675
- Albert M, Peters AH (2009) Genetic and epigenetic control of early mouse development. Curr Opin Genet Dev 19(2):113–121
- Zhou L, Dean J (2015) Reprogramming the genome to totipotency in mouse embryos. Trends Cell Biol 25(2):82–91
- Yamanaka S, Blau HM (2010) Nuclear reprogramming to a pluripotent state by three approaches. Nature 465(7299):704–712
- Marion RM, Blasco MA (2010) Telomere rejuvenation during nuclear reprogramming. Curr Opin Genet Dev 20(2):190–196
- 94. Lapasset L, Milhavet O, Prieur A, Besnard E, Babled A, Ait-Hamou N et al (2011) Rejuvenating senescent and centenarian human cells by reprogramming through the pluripotent state. Genes Dev 25(21):2248–2253
- Soria-Valles C, Lopez-Otin C (2016) iPSCs: on the road to reprogramming aging. Trends Mol Med 22(8):713–724
- 96. Tan L, Ke Z, Tombline G, Macoretta N, Hayes K, Tian X et al (2017) Naked mole rat cells have a stable epigenome that resists iPSC reprogramming. Stem Cell Reports 9(5):1721–1734
- Buganim Y, Faddah DA, Cheng AW, Itskovich E, Markoulaki S, Ganz K et al (2012) Singlecell expression analyses duringcellular reprogramming reveal an early stochastic and a late hierarchic phase. Cell 150(6):1209–1222
- Polo JM, Anderssen E, Walsh RM, Schwarz BA, Nefzger CM, Lim SM et al (2012) A molecular roadmap of reprogramming somatic cells into iPS cells. Cell 151(7):1617–1632
- 99. Ocampo A, Reddy P, Martinez-Redondo P, Platero-Luengo A, Hatanaka F, Hishida T et al (2016) In vivo amelioration of age-associated hallmarks by partial reprogramming. Cell 167(7):1719–1733
- 100. Olova N, Simpson DJ, Marioni RE, Chandra T (2019) Partial reprogramming induces a steady decline in epigenetic age before loss of somatic identity. Aging Cell 18:e12877. https://doi.org/10.1111/acel.12877
- 101. Tanabe K, Nakamura M, Narita M, Takahashi K, Yamanaka S (2013) Maturation, not initiation, is the major roadblock during reprogramming toward pluripotency from human fibroblasts. Proc Natl Acad Sci U S A 110(30):12172–12179
- 102. Conboy IM, Conboy MJ, Wagers AJ, Girma ER, Weissman IL, Rando TA (2005) Rejuvenation of aged progenitor cells by exposure to a young systemic environment. Nature 433(7027):760–764

- 103. Villeda SA, Luo J, Mosher KI, Zou B, Britschgi M, Bieri G et al (2011) The ageing systemic milieu negatively regulates neurogenesis and cognitive function. Nature 477(7362):90–94
- 104. Katsimpardi L, Litterman NK, Schein PA, Miller CM, Loffredo FS, Wojtkiewicz GR et al (2014) Vascular and neurogenic rejuvenation of the aging mouse brain by young systemic factors. Science 344(6184):630–634
- 105. Gontier G, Iyer M, Shea JM, Bieri G, Wheatley EG, Ramalho-Santos M et al (2018) Tet2 rescues age-related regenerative decline and enhances cognitive function in the adult mouse brain. Cell Rep 22(8):1974–1981
- 106. Liu Y, Conboy MJ, Mehdipour M, Liu Y, Tran TP, Blotnick A et al (2017) Application of bio-orthogonal proteome labeling to cell transplantation and heterochronic parabiosis. Nat Commun 8:643. https://doi.org/10.1038/s41467-017-00698-y
- 107. Villeda SA, Plambeck KE, Middeldorp J, Castellano JM, Mosherm KI, Luo J et al (2014) Young blood reverses age-related impairments in cognitive function and synaptic plasticity in mice. Nat Med 20(6):659–663
- 108. Adler AS, Sinha S, Kawahara TLA, Zhang JY, Segal E, Chang HY (2007) Motif module map reveals enforcement of aging by continual NF-κB activity. Genes Dev 21(24):3244–3257
- 109. Rossi DJ, Bryder D, Zahn JM, Ahlenius H, Sonu R, Wagers AJ et al (2005) Cell intrinsic alterations underlie hematopoietic stem cell aging. Proc Natl Acad Sci U S A 102(26):9194–9199
- 110. Wullschleger S, Loewith R, Hall MN (2006) TOR signaling in growth and metabolism. Cell 124(3):471–484
- 111. Chen C, Liu Y, Liu R, Ikenoue T, Guan K-L, Liu Y et al (2008) TSC mTOR maintains quiescence and function of hematopoietic stem cells by repressing mitochondrial biogenesis and reactive oxygen species. J Exp Med 205(10):2397–2408
- 112. Chen C, Liu Y, Liu Y, Zheng P (2009) mTOR regulation and therapeutic rejuvenation of aging hematopoietic stem cells. Sci Signal 2:ra75. https://doi.org/10.1126/scisignal.2000559
- 113. Jones MJ, Goodman SJ, Kobor MS (2015) DNA methylation and healthy human aging. Aging Cell 14(6):924–932

# Chapter 11 From White to Brown – Adipose Tissue Is Critical to the Extended Lifespan and Healthspan of Growth Hormone Mutant Mice



Justin Darcy and Andrzej Bartke

# 1 Introduction

The growth hormone (GH) and insulin-like growth factor 1 (IGF-1) axis (collectively known as the somatotropic axis) was demonstrated to be a major determinant of mammalian longevity more than 20 years ago [1, 2]. Since then, numerous laboratories have attempted to elucidate mechanisms underlying the role of this axis in longevity, leading to an ever-growing list of possible mechanisms to disentangle [3, 4]. During the same timeframe, the growing obesity epidemic in the developed world has resulted in a dramatic increase in adipose tissue (AT) research. Since GH plays integral roles in the physiology of AT, there has been a surge in research attempting to understand how AT impacts longevity in GH-mutant mice. This review is aimed to give an overview of AT, GH, and how the interplay between the two influences longevity.

## 2 Adipose Tissue

Traditionally, AT was believed to be metabolically inactive, solely acting as a tissue to store excess calories. However, paradigm-shifting studies demonstrated that AT secretes adiponectin [5], leptin [5, 6] and resistin [7], which paved the way for

J. Darcy (🖂)

A. Bartke

© Springer Nature Switzerland AG 2019

Section on Integrative Physiology and Metabolism, Joslin Diabetes Center, Harvard Medical School, Boston, MA, USA e-mail: justin.darcy@joslin.harvard.edu

Department of Internal Medicine, Southern Illinois University School of Medicine, Springfield, IL, USA

P. C. Guest (ed.), *Reviews on Biomarker Studies in Aging and Anti-Aging Research*, Advances in Experimental Medicine and Biology 1178, https://doi.org/10.1007/978-3-030-25650-0\_11

	WAT	Beige AT	BAT
Color	White	White	Brown
Lipid storage	Unilocular	Unilocular	Multilocular
Mitochondria	Few	Some	Many
Innervation	Normal	Increased	High
Thermogenic capacity	Low	Medium	High

 Table 11.1
 Different types of adipose tissue. The major similarities and differences between white adipose tissue (WAT), brown adipose tissue (BAT), and beige adipose tissue

future work on AT as an endocrine organ. Since then, our understanding of AT has been greatly expanded. We now know that there are at least three distinct types of adipose tissue: white adipose tissue (WAT); brown adipose tissue (BAT); and beige AT. As predicted, each type of AT depot has specific functions. Further differentiating these types of AT is the significant cellular heterogeneity within an AT depot itself [8]. Despite this heterogeneity, AT is largely made-up of postmitotic adipocytes and their replicative precursors, termed preadipocytes. The differentiation of preadipocytes into mature adipocytes is transcriptionally controlled through the coordination of CCAAT/enhancer-binding proteins (C/EBPs) and peroxisome proliferator-activated receptor gamma (PPAR $\gamma$ ) [9]. This section is dedicated to defining the similarities and differences between WAT, BAT and beige AT, which are summarized in Table 11.1.

### 2.1 WAT

The defining characteristic of WAT in both humans and mice is the storage of excess energy. Morphologically, WAT is characterized by a large, unilocular lipid droplet, with few mitochondria. WAT has both similarities and differences in mice and humans. In mice, WAT is present in superficial subcutaneous depots, mainly in the scapular and inguinal regions. WAT is also present in the intra-abdominal region of mice in the form of perigonadal (epididymal and paraovarian in males and females, respectfully), mesenteric, and retroperitoneal AT. In humans, WAT is more widely distributed and is present subcutaneously in the gluteal, femoral, clavicular, and abdominal regions. WAT in humans is also present intra-abdominally in intraperitoneal, retroperitoneal, mesenteric, and omental AT depots. The major difference in WAT distribution between mice and humans is the large perigonadal depot in mice, and the large omental depot in humans. Most mouse studies in the context of metabolism and aging use inguinal WAT (iWAT) to represent subcutaneous AT, and perigonadal AT to represent the so-called visceral AT. Although this practice is widely used and accepted, it is worth mentioning that some investigators prefer a more stringent use of the term "visceral" to include only AT that directly drains into the portal vein, rather than any intra-abdominal AT depot. By this definition, only mesenteric AT in mice would be considered "visceral" [10].

In obese subjects, there are critical changes in WAT physiology. AT itself is surrounded by a thick extracellular matrix (ECM). During weight gain, the ECM in WAT must expand to accommodate hypertrophic adipocytes. However, this results in poor vascularization [11] and subsequent hypoxia. Hypoxia in the adipocyte is just one of several instances that cause an increased secretion of proinflammatory cytokines to be released from WAT during obesity [12]. Another unfavorable impact of obesity on WAT is ectopic lipid distribution. For example, intra-myocellular lipids [13–17] and intra-hepatic lipids [18–21] are associated with insulin resistance, while epicardial fat is associated with an increased risk of coronary artery disease [22–24].

#### 2.2 Bat

The differences between BAT and WAT begin during development, as BAT comes from a mesoderm lineage that is myogenic factor 5 (MYF5) positive [25]. Unlike WAT, BAT is characterized by multilocular lipid droplets, many mitochondria, and is rich in both innervation and microvasculature. These anatomical traits are important for the main function of BAT, thermogenesis. Sympathetic nerves provide a source of norepinephrine (NE) to stimulate thermogenesis, blood vessels provide nutrients to the tissue as well as aid in heat dissipation, and multilocular lipid droplets have an increased surface area to facilitate an increased rate of lipolysis. The thermogenic circuit relies on several transcriptional inputs including peroxisome proliferator-activated receptor gamma coactivator 1-alpha (PGC-1a) [26] and PR domain containing 16 (PRDM16) [27]. Moreover, cues from other transcriptional machinery such as thyroid hormone receptor [28] and retinoic acid receptor [29] play important roles in the thermogenic circuit. Readers interested in learning more about the transcriptional control of the thermogenic program are directed to the following reviews [30-32]. Central to the function of BAT is uncoupling protein 1 (UCP1), which dissociates the electron transport chain to release chemical energy in the form of heat.

Thermogenesis itself begins with the release of NE from the sympathetic nervous system, which acts on  $\beta$ 3-adrenergic receptors, which are associated with G-protein coupled receptors (GPCRs) of the Gs subtype [33, 34]. Subsequently, a rise in cytosolic cAMP results in the activation of protein kinase A (PKA) [35], which has several functions including activating mitogen-activated protein kinase (MAPK) p38 [36] and increasing cytosolic free fatty acid (FFA) levels in the cell by phosphorylating perilipin [37]. This, in turn, causes the release of comparative gene identification-58 (CGI-58) to activate adipose triglyceride lipase (ATGL), the major triglyceride lipase in BAT [38–40]. The breakdown of triglycerides into FFAs in BAT is critical for two reasons. First, the resulting FFAs can be shuttled into the mitochondria where they can undergo  $\beta$ -oxidation to produce ATP and reduced electron carriers to maintain thermogenesis [30]. Second, the FFAs act as activators of UCP1 [41, 42]. Conversely, purine nucleotides act as inhibitors of UCP1 [41].
It was long believed that BAT in humans was non-existent past adolescence. However, a decade ago, seminal studies that "rediscovered" BAT in adult humans proved otherwise [43–45]. Because a few hundred milligrams of BAT can oxidize up to 60% of consumed glucose and lipids in a cold-acclimated mouse [46, 47], investigators became interested in using BAT therapeutically to combat the growing obesity epidemic. Although humans do possess BAT, it differs in several ways from BAT in mice. Mice have several distinct BAT depots including the large interscapular depot, as well as the axillary, cervical, paraaortic, cardiac, and perirenal depots. In humans, brown adipocytes appear interspersed in WAT, mainly in the supraclavicular region, but are also present in the para-aortic, cervical, axillary, perirenal, and paravertebral regions. It is worth noting that recently a supraclavicular BAT depot was discovered in mice [48]. There is an interscapular BAT depot in humans, although it disappears during adolescence. Currently, the gold standard for assessing the location and activity of BAT in humans is positron emission tomography coupled with computed tomography (PET/CT) during the infusion of radiolabeled 18-fluoro-deoxyglucose (18FDG) in a patient either wearing a "cold vest" or receiving a β3-agonist, such as mirabegron [49]. However, this method does not always accurately reflect BAT activity, and has led to vastly different estimates of the total volume of BAT present in humans, ranging over two orders of magnitude from only a few, to a few hundred milliliters [50]. The measurement of BAT in humans, along with a detailed description of its pitfalls has been reviewed elsewhere [49]. Regardless, we know that BAT is present in adult humans and has a significant impact on metabolism. A clear example of this is a study in which type 2 diabetics spent several hours a day over a 10-day period at 15 °C, which resulted in a significant increase in their glucose infusion rate (GIR) during a euglycemic clamp [51].

One of the biggest advances in our understanding of thermogenic AT in the past few years is the presence of UCP1-independent thermogenic mechanisms. For example, both brown and beige AT thermogenesis can occur through a creatinebased substrate cycle [52–54]. Moreover, beige AT thermogenesis can be controlled through ATP-dependent calcium cycling [55]. These findings can begin to explain why UCP1 null mice only become obese under thermoneutral temperatures [56, 57], while BAT-deficient mice are obese and insulin resistant at standard room temperature [58, 59]. To-date, however, these UCP1-independent forms of thermogenesis have not been examined in GH mutant mice.

### 2.3 Beige AT

Beige AT is distinct from both WAT and BAT. Beige adipocytes reside within WAT depots, contain mitochondria expressing UCP1, and are therefore thermogenic. Under basal conditions, thermogenic output of beige adipocytes is relatively low, however, stimulants such as cold exposure, exercise, or treatment with PPAR $\gamma$  agonists significantly increase the expansion and energy expenditure in these cells in a

process termed beiging. Although beiging was described more than 30 years ago [60, 61], only recently have the specific lineages and molecular regulators that give rise to beige AT been worked out [62, 63]. It is worth noting that this is an area of ongoing investigation, with no clear consensus. For example, some studies have shown that beige AT derives from a lineage distinct from BAT that is positive for myosin heavy chain 11 (MYH11) and platelet derived growth factor receptor alpha (PDGFR $\alpha$ ) in mice [64–67], while other beige adipocytes have been found to be positive for paired box 3 (PAX3) and MYF5 [68]. Adding further complexity are conflicting studies in which some investigators demonstrate that beige adipocytes are formed de novo in response to external cues [66, 69], while others argue they arise from the transdifferentiation of white adipocytes [70, 71]. Regardless of the developmental origin of beige adipose tissue, during cold-exposure, thermogenic beige adipocytes replace non-thermogenic white adipocytes, which is reversible when cold-acclimated mice are placed at thermoneutrality [72].

### **3** Properties of Adipose Tissue

### 3.1 Adipose Tissue Heterogeneity

Beyond the previously discussed inter-depot differences, AT is highly heterogenous within each depot. By volume, the majority of AT is composed of mature adipocytes. These adipocytes have a turnover rate of approximately 10% annually in humans, with a much faster turnover rate of 5% daily in mice [73, 74]. Because of this, preadipocytes, or committed adipocyte progenitors, are critical to the AT niche. Work has already been done to identify markers of white, brown, and beige adipocytes, however, studies identifying novel cell surface markers of preadipocytes that give rise to these adipocytes is mostly lacking, although some markers have been identified [75]. For example, sorting preadipocytes with high CD29 expression enriches for a population of preadipocytes are also a major source of tumor necrosis factor  $\alpha$  (TNF- $\alpha$ ), suggesting their role in AT extends beyond acting as a precursor cell [76].

Immune cells are another cell type with a large role in AT. It has been appreciated for years that macrophages are present in AT, and that their presence increases with obesity. However, considerable improvements have been made to our understanding of different subpopulations of macrophages in AT [77–79]. For example, there is good evidence that both M1 (referred to as classically activated) and M2 (referred to as alternatively activated) macrophages exist within AT [80, 81]. Along with macrophages, natural killer cells are recruited to AT during obesity, causing insulin resistance [82]. Changes in resident immune cells may also play a beneficial role in the physiology of AT. For example, cold-exposure causes an influx of M2 macrophages and eosinophils that aid in thermogenesis [83–85].

AT stores energy as triglycerides during caloric excess, and must liberate FFAs during periods of caloric demand through lipolysis. Since lipolysis is stimulated by the sympathetic release of NE, it makes sense that nerves are a core component of AT [86]. To transport nutrients to AT, or dissipate heat and FFAs (in BAT and WAT, respectfully), there must also be the presence of microvasculature, including endothelial cells and smooth muscle cells [8]. Certainly, AT heterogeneity has garnered attention in recent years. With advances in technologies such as single cell RNA-sequencing, the makeup and functional importance of the AT niche is sure to be further developed in the near future.

## 3.2 Adipose Tissue as a Secretory Organ

Studying WAT as an endocrine organ began 20 years ago with the discovery of leptin, resistin, and adiponectin being secreted from WAT [5–7]. WAT has also been known to secrete proinflammatory cytokines such as TNF- $\alpha$  and IL-6, which facilitate the development of insulin resistance during obesity. What has been much less studied, however, is the role of BAT as a secretory organ. Recently, this has changed as secreted factors from BAT (referred to as batokines) have been demonstrated to have both autocrine/paracrine and endocrine effects. For example, the lipokine 12,13-diHOME has been demonstrated to have an autocrine effect on BAT that results in increased lipid uptake [87]. Other factors that have paracrine/autocrine action in BAT are vascular endothelial growth factor A (VEGFa) and nitric oxide (NO), which increase angiogenesis [88, 89]. Moreover, fibroblast growth factor 2 (FGF2) and nerve growth factor (NGF) increase innervation and the recruitment of preadipocytes [90-92]. Endocrine factors that are secreted from BAT include insulin-like growth factor-binding protein 2 (IGFBP2) [93], WNT10b [93], and FGF21 [94]. BAT also secretes microRNAs. For example, both mice and humans show an inverse relationship between BAT activity and circulating levels of miR-92a [95]. Readers interested in learning more about the secretory function of BAT are encouraged to read a relevant review [96].

### 3.3 Alterations in AT during Aging

The main changes in WAT during aging is the gradual decline in tissue mass, the redistribution from subcutaneous to intra-abdominal depots, and the ectopic distribution of lipids in organs such as the liver and muscle [97–100]. Metabolically, aged WAT has a decline in its sensitivity to insulin and fatty acids [97, 101–103]. Moreover, aged WAT has an increased secretion of harmful proinflammatory cytokines such as TNF- $\alpha$  and IL-6 [104, 105]. There does appear to be an increase in macrophage infiltration in subcutaneous WAT, although this does not seem to apply to intra-abdominal WAT [106]. A final means through which WAT changes during

aging is through the preadipocyte pool. Preadipocytes from aged tissue have lower levels of the transcription factors PPAR $\gamma$  C/EBP $\alpha$ , and their target genes [107, 108], which may explain the reduction in their capacity to differentiate into mature adipocytes. Moreover, senescent preadipocytes accumulate in aged WAT, which could contribute to metabolic impairment and increased systemic inflammation [97]. Readers interested in learning more about WAT remodeling during aging are directed to a relevant review [97].

Some of the changes mentioned above apply to BAT. For example, BAT mass decreases with age. Although senescence in BAT has been understudied, it is plausible to assume that brown preadipocytes also senesce, and lose the ability to differentiate into mature brown adipocytes with age. Many of the age-related changes in BAT relate to impaired thermogenic capacity. One mechanism for this is through the increased visceral AT expression of forkhead box protein A3 (FOXA3), which impairs BAT mass and function [109]. Interestingly, deletion of FOXA3 increases BAT late into life and extends longevity [109]. Another deleterious change in aged BAT is the presence of sympathetic neuron-associated macrophages, which chelate NE, resulting in decreased thermogenic output [110]. Finally, the well-documented age-dependent mitochondrial dysfunction impairs thermogenesis. The decline in the thermogenic function of BAT likely plays a critical role in the metabolic impairment and obesity observed during middle-age.

### 4 Growth Hormone

GH is a 22 kDa peptide hormone that is secreted from somatotrophs in the anterior pituitary. Its secretion is induced by the release of growth hormone releasing hormone (GHRH), and inhibited by the release of somatostatin (SST), both of which are released from the hypothalamus. GH has negative feedback on GH release from the pituitary, as well as on GHRH from the hypothalamus. Another level of feedback is through IGF-1 which acts on both the pituitary and hypothalamus [111]. Ghrelin, a "hunger" hormone is another factor that stimulates the release of GH [112]. AT can regulate GH production through FFAs and leptin which inhibit and stimulate GH production, respectively [113, 114].

In circulation, GH acts by binding to growth hormone receptor (GHR) on target tissues. Mainly, GH acts on the liver to stimulate the production of IGF-1, but GH can also act on other tissues such as muscle and AT [115]. Therefore, GH can elicit direct effects, or indirect effects through the action of IGF-1. Once bound to a homodimerized GHR, there is a conformational change in the receptor structure which brings together the associated janus kinase 2 (JAK2) domains together, allowing for transactivation [116] and subsequent phosphorylation of signal transducer and activator of transcription 5 (STAT5). Activated STAT5 can then enter the nucleus and act as a transcription factor [117]. GH has been demonstrated to signal through other non-canonical pathways including mammalian target of rapamycin (mTOR) and extracellular signaling-regulated kinase (ERK) [118].

### 5 Examples of Altered Growth Hormone Action

## 5.1 Humans

The two main ways that GH is altered in humans is through its overproduction in acromegaly and GH resistance, or through its under production in GH deficiency. Patients with acromegaly suffer from increased GH secretion, and subsequent increases in IGF-1 production. The increased secretion of GH is oftentimes the result of a pituitary adenoma. Acromegaly patients are more prone to cancer [119– 121], diabetes [122], and are often short-lived compared to people with normal GH secretion [123, 124]. GH deficiency has multiple etiologies that influence the age of the onset of disease. In children, congenital GH deficiency is usually the result of mutations in genes encoding GH, GHRH, or other pituitary factors involved in the secretion of GH [125]. In adults, acquired GH deficiency is typically the result of hypopituitarism or irradiation of a pituitary adenoma [126]. Beyond deficiency, patients can be resistant to GH through mutations in the gene encoding GHR [127]. This, disease, termed Laron syndrome, causes patients to have low IGF-1, with elevated levels of GH [128]. Patients with both GH deficiency and resistance demonstrate decreased height, increased obesity, decreased bone mineral density, and altered lipid metabolism [3]. Interestingly, patients with Laron syndrome appear to be protected from cancer [129, 130] and diabetes [131], although Laron syndrome patients from cohorts in Israel and Turkey appear to still develop diabetes [132, 133], making the "protected" status from diabetes less clear.

## 5.2 Mice

To further understand the impact of GH signaling, several transgenic lines overexpressing GH have been created, the most commonly used being the bovine GH (bGH) transgenic line [134, 135]. These mice have a transgene that ectopically expresses GH under a strong promoter such as phosphoenolpyruvate carboxykinase (PEPCK). bGH mice are noticeably larger than their control littermates, and exhibit increased muscle mass. bGH mice demonstrate increased insulin resistance, and severe hyperplasia and hypertrophy of their hepatocytes [136]. Many of these mice die of hepatic cancer. Aging in these mice appears to be accelerated, and lifespan is reduced to around 1 year-of-age [135, 137].

Ames dwarf mice were first described in 1961, and suffer from a spontaneous mutation in the gene encoding prophet of pituitary factor 1 (Prop1) [138]. Snell dwarf mice were first described in the 1929, and suffer from a spontaneous mutation in the gene encoding pituitary factor 1 (Pit1) [139]. Since Prop1 is a transcription factor for the Pit1 gene, the phenotypes of Ames and Snell dwarfs are essentially identical, with both strains of dwarf mice lacking the production of GH, thyroid-stimulating hormone (TSH) and prolactin [140]. The downstream consequences of

these mutations are a decreased production of IGF-1 and the thyroid hormones, T3 and T4. Both Ames and Snell dwarf mice are extremely long-lived, with an ~50% increase in longevity in Ames dwarf mice (1) and ~42% increase in Snell dwarf mice [141]. Along with increased longevity, these animals have a decreased incidence of cancer [142], increased insulin sensitivity [143] and increased adiposity, particularly in the subcutaneous depot [144].

Growth hormone receptor knockout (GHRKO) mice were developed to replicate Laron syndrome in mice by disrupting the gene encoding GHR/GH binding protein [145]. As with Laron syndrome patients, GHRKO mice are small, have increased adiposity, and have increased circulating GH with low circulating IGF-1 [145]. GHRKO mice also live ~38% longer than control mice [146]. Studies of cognitive function [147, 148] and tissue histopathology [149] revealed that GHRKO mice have a delay in aging, similar to that of Ames and Snell dwarf mice. To study the effects of GH action in specific tissues, many tissue-specific lines of GHRKO mice have been created [150–153]. The first line attempting to knockout the GHR gene in AT was done using Cre driven by the adipocyte protein 2 (Ap2) promoter (termed FaGHRKO) [154]. Unlike the global GHRKO mice, these animals had an increase in IGF-1 and had no improvement in insulin sensitivity [154]. It has since been reported that the Ap2 promoter is "leaky" and is expressed in macrophages, endothelial cells, and in the brain [155, 156]. Therefore, an AT-specific GHRKO mouse (termed AdGHRKO) was generated using the more specific adiponectin promoterdriven Cre [157]. These animals show no difference in GH or IGF-1, have increased fat mass, decreased liver triglyceride content, and are insulin sensitive [157].

Other mouse lines such as the GHRH knockout (GHRHKO) mice live ~45% longer than their control littermates, and have decreased IGF-1 production and increased adiposity [158]. "Little" mice have a mutation of the GHRH receptor gene, live ~25% longer than their normal littermates, and develop increased adiposity [141]. Several current reviews discuss mechanisms of altered longevity in all these mouse lines in more detail [4, 159, 160].

# 6 Adipose Tissue in Mice and Humans with Altered Growth Hormone Action

What is clear is that a defining phenotype of GH-related mutations is alterations in adiposity. For example, patients with acromegaly and bGH mice have decreased adiposity [161–164], while patients and mice that are GH-deficient or GH-resistant have increased adiposity [144, 165–167]. Particularly noteworthy is that the iWAT depot in GHRKO mice is equal in mass to that of their control littermates, despite GHRKO mice weighing approximately 66% less [166, 168]. It has also been demonstrated that dwarf and GHRKO mice maintain a higher extra- to intra-peritoneal distribution of lipids [169], which is the opposite of acromegaly patients which have a higher ectopic distribution of lipids [170]. Beyond alterations in mass and

distribution, there appears to be an alteration in the endocrine function of AT in GH-mutant animals. For example, leptin and adiponectin are decreased in bGH mice, and increased in dwarf and GHRKO mice [171, 172]. In fact, altered endocrine function may partially explain an unexpected phenotype, where surgical removal of the epidydimal WAT (eWAT) in dwarf and GHRKO mice results in decreased insulin sensitivity [173, 174]. This finding may be due to the fact that eWAT in long-lived GH-mutant mice secretes less pro-inflammatory cytokines such as TNF- $\alpha$  and IL-6, while secreting more adiponectin [173, 174]. The secretory function of BAT in these animals has not yet been assessed. Lowering senescent cell burden can extend longevity [175]. Therefore, it is important that 18-month-old dwarf and GHRKO mice have a reduced senescent cell burden, and that their pre-adipocytes demonstrate an increased differentiation capacity, suggesting that their AT has a "younger" phenotype [169]. Senescence in BAT of long-lived GH-mutant mice has not yet been investigated.

Although many studies have been conducted on WAT in GH-mutant mice, far less has been done relating to the BAT of these animals. To date, we know that BAT in GHRKO and Ames dwarf mice has an increased expression of Ucp1, and that BAT in bGH mice has a decreased expression of Ucp1 [176, 177]. This is accompanied by an increase or decrease in BAT mass in Ames dwarf/GHRKO and bGH mice, respectively [176, 177]. The highly active BAT observed in Ames dwarf and GHRKO mice may at least partially explain the increased rate of energy expenditure and reduced respiratory quotient observed in these animals [178]. Particularly curious in Ames dwarf mice is the increased BAT activity despite having a depleted thyroid hormone axis. This suggests that other circulating factors may be responsible for the increased BAT activity, although this hypothesis remains to be tested. It does appear that the increased BAT activity of these animals may play a role in several biomarkers of healthy aging. For example, placing Ames dwarf mice at thermoneutrality (30 °C) eliminates differences in oxygen consumption rate and respiratory quotient, and reduces their enhanced insulin sensitivity [179]. Further testing will need to be done to determine if thermoneutral housing also impacts longevity in these animals.

### 7 Final Thoughts

GH has a critical role in metabolism due to its profound effects highly metabolic tissues such as the liver, muscle and AT. Increased adiposity is associated with comorbidities ranging from diabetes to Alzheimer's disease. Therefore, it is of major consequence that the AT in long-lived GH-mutant mice functions in a more metabolically beneficial way. These differences include AT distribution (extrainstead of intraperitoneal), endocrine function (shift from pro- to anti-inflammatory cytokines), and replication and senescence status (differentiates well and has a low senescent cell burden), as well as an increase in thermogenesis and BAT activity. In terms of whole-body physiology, these alterations cannot be understated. For example, both WAT and BAT in these animals most likely significantly contribute to improved glycemic control, which is believed to be a major factor in their improved healthspan and lifespan. There are still areas that should be examined in the context of AT in GH animals. For example, we now know that AT secretes many proteins, metabolites, lipids, miRNAs, and is a rich source of exosomes. Examining how these are altered in GH-mutant mice is a currently unexplored area. Moreover, extrapolating these findings to human subjects would be of considerable interest.

Acknowledgements We sincerely apologize to those whose work we did not reference due to space limitations or inadvertent omissions. This work was supported by the US National Institutes of Health (NIH) grants DK007260 (to JD), and AG019899 and AG051869 (to AB).

### References

- 1. Brown-Borg HM, Borg KE, Meliska CJ, Bartke A (1996) Dwarf mice and the ageing process. Nature 384(6604):33. https://doi.org/10.1038/384033a0
- Steger RW, Bartke A, Cecim M (1993) Premature ageing in transgenic mice expressing different growth hormone genes. J Reprod Fertil Suppl 46:61–75
- 3. Aguiar-Oliveira MH, Bartke A (2018) Growth hormone deficiency: health and longevity. Endocr Rev. https://doi.org/10.1210/er.2018-00216. [Epub ahead of print]
- Bartke A, Quainoo N (2018) Impact of growth hormone-related mutations on mammalian aging. Front Genet 9:586. https://doi.org/10.3389/fgene.2018.00586
- 5. Hu E, Liang P, Spiegelman BM (1996) AdipoQ is a novel adipose-specific gene dysregulated in obesity. J Biol Chem 271(18):10697–10703
- Montague CT, Farooqi IS, Whitehead JP, Soos MA, Rau H, Wareham NJ et al (1997) Congenital leptin deficiency is associated with severe early-onset obesity in humans. Nature 387(6636):903–908
- Steppan CM, Bailey ST, Bhat S, Brown EJ, Banerjee RR, Wright CM et al (2001) The hormone resistin links obesity to diabetes. Nature 409(6818):307–312
- Lynes MD, Tseng YH (2018) Deciphering adipose tissue heterogeneity. Ann N Y Acad Sci 1411(1):5–20
- 9. Farmer SR (2006) Transcriptional control of adipocyte formation. Cell Metab 4(4):263-273
- 10. Cinti S (2005) The adipose organ. Prostaglandins Leukot Essent Fatty Acids 73(1):9–15
- 11. Lemoine AY, Ledoux S, Queguiner I, Calderari S, Mechler C, Msika S et al (2012) Link between adipose tissue angiogenesis and fat accumulation in severely obese subjects. J Clin Endocrinol Metab 97(5):E775–E780
- 12. Rosen ED, Spiegelman BM (2014) What we talk about when we talk about fat. Cell  $156(1{-}2){:}20{-}44$
- Krssak M, Falk Petersen K, Dresner A, DiPietro L, Vogel SM, Rothman DL et al (1999) Intramyocellular lipid concentrations are correlated with insulin sensitivity in humans: a 1H NMR spectroscopy study. Diabetologia 42(1):113–116
- Perseghin G, Scifo P, De Cobelli F, Pagliato E, Battezzati A, Arcelloni C et al (1999) Intramyocellular triglyceride content is a determinant of in vivo insulin resistance in humans: a 1H-13C nuclear magnetic resonance spectroscopy assessment in offspring of type 2 diabetic parents. Diabetes 48(8):1600–1606
- Pan DA, Lillioja S, Kriketos AD, Milner MR, Baur LA, Bogardus C et al (1997) Skeletal muscle triglyceride levels are inversely related to insulin action. Diabetes 46(6):983–988
- Jacob S, Machann J, Rett K, Brechtel K, Volk A, Renn W et al (1999) Association of increased intramyocellular lipid content with insulin resistance in lean nondiabetic offspring of type 2 diabetic subjects. Diabetes 48(5):1113–1119

- Phillips DI, Caddy S, Ilic V, Fielding BA, Frayn KN, Borthwick AC et al (1996) Intramuscular triglyceride and muscle insulin sensitivity: evidence for a relationship in nondiabetic subjects. Metabolism 45(8):947–950
- Korenblat KM, Fabbrini E, Mohammed BS, Klein S (2008) Liver, muscle, and adipose tissue insulin action is directly related to intrahepatic triglyceride content in obese subjects. Gastroenterology 134(5):1369–1375
- Bugianesi E, Gastaldelli A, Vanni E, Gambino R, Cassader M, Baldi S et al (2005) Insulin resistance in non-diabetic patients with non-alcoholic fatty liver disease: sites and mechanisms. Diabetologia 48(4):634–642
- Sanyal AJ, Campbell-Sargent C, Mirshahi F, Rizzo WB, Contos MJ, Sterling RK et al (2001) Nonalcoholic steatohepatitis: association of insulin resistance and mitochondrial abnormalities. Gastroenterology 120(5):1183–1192
- 21. Seppala-Lindroos A, Vehkavaara S, Hakkinen AM, Goto T, Westerbacka J, Sovijarvi A et al (2002) Fat accumulation in the liver is associated with defects in insulin suppression of glucose production and serum free fatty acids independent of obesity in normal men. J Clin Endocrinol Metab 87(7):3023–3028
- 22. Ahn SG, Lim HS, Joe DY, Kang SJ, Choi BJ, Choi SY et al (2008) Relationship of epicardial adipose tissue by echocardiography to coronary artery disease. Heart 94(3):e7. https://doi. org/10.1136/hrt.2007.118471
- 23. Greif M, Becker A, von Ziegler F, Lebherz C, Lehrke M, Broedl UC et al (2009) Pericardial adipose tissue determined by dual source CT is a risk factor for coronary atherosclerosis. Arterioscler Thromb Vasc Biol 29(5):781–786
- 24. Rosito GA, Massaro JM, Hoffmann U, Ruberg FL, Mahabadi AA, Vasan RS et al (2008) Pericardial fat, visceral abdominal fat, cardiovascular disease risk factors, and vascular calcification in a community-based sample: the Framingham Heart Study. Circulation 117(5):605–613
- Seale P, Bjork B, Yang W, Kajimura S, Chin S, Kuang S et al (2008) PRDM16 controls a brown fat/skeletal muscle switch. Nature 454(7207):961–967
- 26. Puigserver P, Wu Z, Park CW, Graves R, Wright M, Spiegelman BM (1998) A cold-inducible coactivator of nuclear receptors linked to adaptive thermogenesis. Cell 92(6):829–839
- Seale P, Kajimura S, Yang W, Chin S, Rohas LM, Uldry M et al (2007) Transcriptional control of brown fat determination by PRDM16. Cell Metab 6(1):38–54
- 28. Cassard-Doulcier AM, Larose M, Matamala JC, Champigny O, Bouillaud F, Ricquier D (1994) In vitro interactions between nuclear proteins and uncoupling protein gene promoter reveal several putative transactivating factors including Ets1, retinoid X receptor, thyroid hormone receptor, and a CACCC box-binding protein. J Biol Chem 269(39):24335–24342
- 29. Alvarez R, de Andres J, Yubero P, Vinas O, Mampel T, Iglesias R et al (1995) A novel regulatory pathway of brown fat thermogenesis. Retinoic acid is a transcriptional activator of the mitochondrial uncoupling protein gene. J Biol Chem 270(10):5666–5673
- Cannon B, Nedergaard J (2004) Brown adipose tissue: function and physiological significance. Physiol Rev 84(1):277–359
- Lynes MD, Tseng YH (2015) The thermogenic circuit: regulators of thermogenic competency and differentiation. Genes Dis 2(2):164–172
- Villarroya F, Peyrou M, Giralt M (2017) Transcriptional regulation of the uncoupling protein-1 gene. Biochimie 134:86–92
- Granneman JG (1988) Norepinephrine infusions increase adenylate cyclase responsiveness in brown adipose tissue. J Pharmacol Exp Ther 245(3):1075–1080
- 34. Marette A, Bukowiecki LJ (1991) Noradrenaline stimulates glucose transport in rat brown adipocytes by activating thermogenesis. Evidence that fatty acid activation of mitochondrial respiration enhances glucose transport. Biochem J 277(Pt 1):119–124
- 35. Thonberg H, Lindgren EM, Nedergaard J, Cannon B (2001) As the proliferation promoter noradrenaline induces expression of ICER (induced cAMP early repressor) in proliferative brown adipocytes, ICER may not be a universal tumour suppressor. Biochem J 354(Pt 1):169–177

- 36. Cao W, Medvedev AV, Daniel KW, Collins S (2001) beta-Adrenergic activation of p38 MAP kinase in adipocytes: cAMP induction of the uncoupling protein 1 (UCP1) gene requires p38 MAP kinase. J Biol Chem 276(29):27077–27082
- Chaudhry A, Granneman JG (1999) Differential regulation of functional responses by betaadrenergic receptor subtypes in brown adipocytes. Am J Phys 277(1):R147–R153
- Granneman JG, Moore HP, Granneman RL, Greenberg AS, Obin MS, Zhu Z (2007) Analysis of lipolytic protein trafficking and interactions in adipocytes. J Biol Chem 282(8):5726–5735
- Granneman JG, Moore HP, Krishnamoorthy R, Rathod M (2009) Perilipin controls lipolysis by regulating the interactions of AB-hydrolase containing 5 (Abhd5) and adipose triglyceride lipase (Atgl). J Biol Chem 284(50):34538–34544
- 40. Zimmermann R, Strauss JG, Haemmerle G, Schoiswohl G, Birner-Gruenberger R, Riederer M et al (2004) Fat mobilization in adipose tissue is promoted by adipose triglyceride lipase. Science 306(5700):1383–1386
- 41. Shabalina IG, Jacobsson A, Cannon B, Nedergaard J (2004) Native UCP1 displays simple competitive kinetics between the regulators purine nucleotides and fatty acids. J Biol Chem 279(37):38236–38248
- 42. Fedorenko A, Lishko PV, Kirichok Y (2012) Mechanism of fatty-acid-dependent UCP1 uncoupling in brown fat mitochondria. Cell 151(2):400–413
- 43. Cypess AM, Lehman S, Williams G, Tal I, Rodman D, Goldfine AB et al (2009) Identification and importance of brown adipose tissue in adult humans. N Engl J Med 360(15):1509–1517
- 44. Nedergaard J, Bengtsson T, Cannon B (2007) Unexpected evidence for active brown adipose tissue in adult humans. Am J Physiol Endocrinol Metab 293(2):E444–E452
- 45. van Marken Lichtenbelt WD, Vanhommerig JW, Smulders NM, Drossaerts JM, Kemerink GJ, Bouvy ND et al (2009) Cold-activated brown adipose tissue in healthy men. N Engl J Med 360(15):1500–1508
- Cannon B, Nedergaard J (2011) Nonshivering thermogenesis and its adequate measurement in metabolic studies. J Exp Biol 214(Pt 2):242–253
- 47. Golozoubova V, Gullberg H, Matthias A, Cannon B, Vennstrom B, Nedergaard J (2004) Depressed thermogenesis but competent brown adipose tissue recruitment in mice devoid of all hormone-binding thyroid hormone receptors. Mol Endocrinol 18(2):384–401
- Mo Q, Salley J, Roshan T, Baer LA, May FJ, Jaehnig EJ et al (2017) Identification and characterization of a supraclavicular brown adipose tissue in mice. JCI Insight 2(11). pii: 93166. https://doi.org/10.1172/jci.insight.93166
- Carpentier AC, Blondin DP, Virtanen KA, Richard D, Haman F, Turcotte EE (2018) Brown adipose tissue energy metabolism in humans. Front Endocrinol (Lausanne) 9:447. https://doi. org/10.3389/fendo.2018.00447
- Lee P, Greenfield JR, Ho KK, Fulham MJ (2010) A critical appraisal of the prevalence and metabolic significance of brown adipose tissue in adult humans. Am J Physiol Endocrinol Metab 299(4):E601–E606
- 51. Hanssen MJ, Hoeks J, Brans B, van der Lans AA, Schaart G, van den Driessche JJ et al (2015) Short-term cold acclimation improves insulin sensitivity in patients with type 2 diabetes mellitus. Nat Med 21(8):863–865
- 52. Kazak L, Chouchani ET, Jedrychowski MP, Erickson BK, Shinoda K, Cohen P et al (2015) A creatine-driven substrate cycle enhances energy expenditure and thermogenesis in beige fat. Cell 163(3):643–655
- 53. Kazak L, Chouchani ET, Lu GZ, Jedrychowski MP, Bare CJ, Mina AI et al (2017) Genetic depletion of adipocyte creatine metabolism inhibits diet-induced thermogenesis and drives obesity. Cell Metab 26(4):660–671.e3
- 54. Bertholet AM, Kazak L, Chouchani ET, Bogaczynska MG, Paranjpe I, Wainwright GL et al (2017) Mitochondrial patch clamp of beige adipocytes reveals UCP1-positive and UCP1negative cells both exhibiting futile creatine cycling. Cell Metab 25(4):811–822.e4
- 55. Ikeda K, Kang Q, Yoneshiro T, Camporez JP, Maki H, Homma M et al (2017) UCP1independent signaling involving SERCA2b-mediated calcium cycling regulates beige fat thermogenesis and systemic glucose homeostasis. Nat Med 23(12):1454–1465

- 56. Enerback S, Jacobsson A, Simpson EM, Guerra C, Yamashita H, Harper ME et al (1997) Mice lacking mitochondrial uncoupling protein are cold-sensitive but not obese. Nature 387(6628):90–94
- Feldmann HM, Golozoubova V, Cannon B, Nedergaard J (2009) UCP1 ablation induces obesity and abolishes diet-induced thermogenesis in mice exempt from thermal stress by living at thermoneutrality. Cell Metab 9(2):203–209
- Ohno H, Shinoda K, Ohyama K, Sharp LZ, Kajimura S (2013) EHMT1 controls brown adipose cell fate and thermogenesis through the PRDM16 complex. Nature 504(7478):163–167
- Cohen P, Levy JD, Zhang Y, Frontini A, Kolodin DP, Svensson KJ et al (2014) Ablation of PRDM16 and beige adipose causes metabolic dysfunction and a subcutaneous to visceral fat switch. Cell 156(1–2):304–316
- Young P, Arch JR, Ashwell M (1984) Brown adipose tissue in the parametrial fat pad of the mouse. FEBS Lett 167(1):10–14
- Loncar D, Afzelius BA, Cannon B (1988) Epididymal white adipose tissue after cold stress in rats. I. Nonmitochondrial changes. J Ultrastruct Mol Struct Res 101(2–3):109–122
- 62. Kajimura S, Spiegelman BM, Seale P (2015) Brown and beige fat: physiological roles beyond heat generation. Cell Metab 22(4):546–559
- Harms M, Seale P (2013) Brown and beige fat: development, function and therapeutic potential. Nat Med 19(10):1252–1263
- Long JZ, Svensson KJ, Tsai L, Zeng X, Roh HC, Kong X et al (2014) A smooth muscle-like origin for beige adipocytes. Cell Metab 19(5):810–820
- 65. Vishvanath L, MacPherson KA, Hepler C, Wang QA, Shao M, Spurgin SB et al (2016) Pdgfrbeta+ mural preadipocytes contribute to adipocyte hyperplasia induced by high-fat-diet feeding and prolonged cold exposure in adult mice. Cell Metab 23(2):350–359
- Berry DC, Jiang Y, Graff JM (2016) Mouse strains to study cold-inducible beige progenitors and beige adipocyte formation and function. Nat Commun 7:10184. https://doi.org/10.1038/ ncomms10184
- Lee YH, Petkova AP, Mottillo EP, Granneman JG (2012) In vivo identification of bipotential adipocyte progenitors recruited by beta3-adrenoceptor activation and high-fat feeding. Cell Metab 15(4):480–491
- Sanchez-Gurmaches J, Guertin DA (2014) Adipocytes arise from multiple lineages that are heterogeneously and dynamically distributed. Nat Commun 5:4099. https://doi.org/10.1038/ ncomms5099
- 69. Wang QA, Tao C, Gupta RK, Scherer PE (2013) Tracking adipogenesis during white adipose tissue development, expansion and regeneration. Nat Med 19(10):1338–1344
- Barbatelli G, Murano I, Madsen L, Hao Q, Jimenez M, Kristiansen K et al (2010) The emergence of cold-induced brown adipocytes in mouse white fat depots is determined predominantly by white to brown adipocyte transdifferentiation. Am J Physiol Endocrinol Metab 298(6):E1244–E1253
- Cinti S (2009) Transdifferentiation properties of adipocytes in the adipose organ. Am J Physiol Endocrinol Metab 297(5):E977–E986
- 72. Loncar D (1991) Convertible adipose tissue in mice. Cell Tissue Res 266(1):149-161
- 73. Spalding KL, Arner E, Westermark PO, Bernard S, Buchholz BA, Bergmann O et al (2008) Dynamics of fat cell turnover in humans. Nature 453(7196):783–787
- Rigamonti A, Brennand K, Lau F, Cowan CA (2011) Rapid cellular turnover in adipose tissue. PLoS One 6(3):e17637
- 75. Xue R, Lynes MD, Dreyfuss JM, Shamsi F, Schulz TJ, Zhang H et al (2015) Clonal analyses and gene profiling identify genetic biomarkers of the thermogenic potential of human brown and white preadipocytes. Nat Med 21(7):760–768
- Cartier A, Lemieux I, Almeras N, Tremblay A, Bergeron J, Despres JP (2008) Visceral obesity and plasma glucose-insulin homeostasis: contributions of interleukin-6 and tumor necrosis factor-alpha in men. J Clin Endocrinol Metab 93(5):1931–1938

- 77. Weisberg SP, McCann D, Desai M, Rosenbaum M, Leibel RL, Ferrante AW Jr (2003) Obesity is associated with macrophage accumulation in adipose tissue. J Clin Invest 112(12):1796–1808
- Xu H, Barnes GT, Yang Q, Tan G, Yang D, Chou CJ et al (2003) Chronic inflammation in fat plays a crucial role in the development of obesity-related insulin resistance. J Clin Invest 112(12):1821–1830
- 79. Ortega Martinez de Victoria E, Xu X, Koska J, Francisco AM, Scalise M, Ferrante AW Jr et al (2009) Macrophage content in subcutaneous adipose tissue: associations with adiposity, age, inflammatory markers, and whole-body insulin action in healthy Pima Indians. Diabetes 58(2):385–393
- Lumeng CN, Bodzin JL, Saltiel AR (2007) Obesity induces a phenotypic switch in adipose tissue macrophage polarization. J Clin Invest 117(1):175–184
- Lumeng CN, DelProposto JB, Westcott DJ, Saltiel AR (2008) Phenotypic switching of adipose tissue macrophages with obesity is generated by spatiotemporal differences in macrophage subtypes. Diabetes 57(12):3239–3246
- Lee BC, Kim MS, Pae M, Yamamoto Y, Eberle D, Shimada T et al (2016) Adipose natural killer cells regulate adipose tissue macrophages to promote insulin resistance in obesity. Cell Metab 23(4):685–698
- Lee MW, Odegaard JI, Mukundan L, Qiu Y, Molofsky AB, Nussbaum JC et al (2015) Activated type 2 innate lymphoid cells regulate beige fat biogenesis. Cell 160(1–2):74–87
- 84. Rao RR, Long JZ, White JP, Svensson KJ, Lou J, Lokurkar I et al (2014) Meteorin-like is a hormone that regulates immune-adipose interactions to increase beige fat thermogenesis. Cell 157(6):1279–1291
- 85. Brestoff JR, Kim BS, Saenz SA, Stine RR, Monticelli LA, Sonnenberg GF et al (2015) Group 2 innate lymphoid cells promote beiging of white adipose tissue and limit obesity. Nature 519(7542):242–246
- Locke AE, Kahali B, Berndt SI, Justice AE, Pers TH, Day FR et al (2015) Genetic studies of body mass index yield new insights for obesity biology. Nature 518(7538):197–206
- Lynes MD, Leiria LO, Lundh M, Bartelt A, Shamsi F, Huang TL et al (2017) The coldinduced lipokine 12,13-diHOME promotes fatty acid transport into brown adipose tissue. Nat Med 23(5):631–637
- Asano A, Kimura K, Saito M (1999) Cold-induced mRNA expression of angiogenic factors in rat brown adipose tissue. J Vet Med Sci 61(4):403–409
- Nisoli E, Tonello C, Briscini L, Carruba MO (1997) Inducible nitric oxide synthase in rat brown adipocytes: implications for blood flow to brown adipose tissue. Endocrinology 138(2):676–682
- Nisoli E, Tonello C, Benarese M, Liberini P, Carruba MO (1996) Expression of nerve growth factor in brown adipose tissue: implications for thermogenesis and obesity. Endocrinology 137(2):495–503
- Nechad M, Ruka E, Thibault J (1994) Production of nerve growth factor by brown fat in culture: relation with the in vivo developmental stage of the tissue. Comp Biochem Physiol Comp Physiol 107(2):381–388
- Yamashita H, Sato Y, Kizaki T, Oh S, Nagasawa J, Ohno H (1994) Basic fibroblast growth factor (bFGF) contributes to the enlargement of brown adipose tissue during cold acclimation. Pflugers Arch 428(3–4):352–356
- Rahman S, Lu Y, Czernik PJ, Rosen CJ, Enerback S, Lecka-Czernik B (2013) Inducible brown adipose tissue, or beige fat, is anabolic for the skeleton. Endocrinology 154(8):2687–2701
- 94. Hondares E, Iglesias R, Giralt A, Gonzalez FJ, Giralt M, Mampel T et al (2011) Thermogenic activation induces FGF21 expression and release in brown adipose tissue. J Biol Chem 286(15):12983–12990
- 95. Chen Y, Buyel JJ, Hanssen MJ, Siegel F, Pan R, Naumann J et al (2016) Exosomal microRNA miR-92a concentration in serum reflects human brown fat activity. Nat Commun 7:11420. https://doi.org/10.1038/ncomms11420

- Villarroya F, Cereijo R, Villarroya J, Giralt M (2017) Brown adipose tissue as a secretory organ. Nat Rev Endocrinol 13(1):26–35
- 97. Tchkonia T, Morbeck DE, Von Zglinicki T, Van Deursen J, Lustgarten J, Scrable H et al (2010) Fat tissue, aging, and cellular senescence. Aging Cell 9(5):667–684
- Visser M, Pahor M, Tylavsky F, Kritchevsky SB, Cauley JA, Newman AB et al (2003) Oneand two-year change in body composition as measured by DXA in a population-based cohort of older men and women. J Appl Physiol (1985) 94(6):2368–2374
- 99. Raguso CA, Kyle U, Kossovsky MP, Roynette C, Paoloni-Giacobino A, Hans D et al (2006) A 3-year longitudinal study on body composition changes in the elderly: role of physical exercise. Clin Nutr 25(4):573–580
- Bertrand HA, Lynd FT, Masoro EJ, Yu BP (1980) Changes in adipose mass and cellularity through the adult life of rats fed ad libitum or a life-prolonging restricted diet. J Gerontol 35(6):827–835
- 101. Yu BP, Masoro EJ, Murata I, Bertrand HA, Lynd FT (1982) Life span study of SPF Fischer 344 male rats fed ad libitum or restricted diets: longevity, growth, lean body mass and disease. J Gerontol 37(2):130–141
- 102. Kirkland JL, Dax EM (1984) Adipocyte hormone responsiveness and aging in the rat: problems in the interpretation of aging research. J Am Geriatr Soc 32(3):219–228
- 103. Yki-Jarvinen H, Kiviluoto T, Nikkila EA (1986) Insulin binding and action in adipocytes in vitro in relation to insulin action in vivo in young and middle-aged subjects. Acta Endocrinol 113(1):88–92
- 104. Morin CL, Pagliassotti MJ, Windmiller D, Eckel RH (1997) Adipose tissue-derived tumor necrosis factor-alpha activity is elevated in older rats. J Gerontol A Biol Sci Med Sci 52(4):B190–B195
- 105. Starr ME, Evers BM, Saito H (2009) Age-associated increase in cytokine production during systemic inflammation: adipose tissue as a major source of IL-6. J Gerontol A Biol Sci Med Sci 64(7):723–730
- 106. Harris TB, Ferrucci L, Tracy RP, Corti MC, Wacholder S, Ettinger WH Jr et al (1999) Associations of elevated interleukin-6 and C-reactive protein levels with mortality in the elderly. Am J Med 106(5):506–512
- 107. Karagiannides I, Tchkonia T, Dobson DE, Steppan CM, Cummins P, Chan G et al (2001) Altered expression of C/EBP family members results in decreased adipogenesis with aging. Am J Physiol Regul Integr Comp Physiol 280(6):R1772–R1780
- Schipper BM, Marra KG, Zhang W, Donnenberg AD, Rubin JP (2008) Regional anatomic and age effects on cell function of human adipose-derived stem cells. Ann Plast Surg 60(5):538–544
- 109. Ma X, Xu L, Gavrilova O, Mueller E (2014) Role of forkhead box protein A3 in ageassociated metabolic decline. Proc Natl Acad Sci U S A 111(39):14289–14294
- 110. Pirzgalska RM, Seixas E, Seidman JS, Link VM, Sanchez NM, Mahu I et al (2017) Sympathetic neuron-associated macrophages contribute to obesity by importing and metabolizing norepinephrine. Nat Med 23(11):1309–1318
- Muller EE (1990) Clinical implications of growth hormone feedback mechanisms. Horm Res 33(Suppl 4):90–96
- 112. Muller AF, Lamberts SW, Janssen JA, Hofland LJ, Koetsveld PV, Bidlingmaier M et al (2002) Ghrelin drives GH secretion during fasting in man. Eur J Endocrinol 146(2):203–207
- 113. Pombo M, Pombo CM, Astorga R, Cordido F, Popovic V, Garcia-Mayor RV et al (1999) Regulation of growth hormone secretion by signals produced by the adipose tissue. J Endocrinol Investig 22(5 Suppl):22–26
- 114. Carro E, Senaris R, Considine RV, Casanueva FF, Dieguez C (1997) Regulation of in vivo growth hormone secretion by leptin. Endocrinology 138(5):2203–2206
- 115. Green H, Morikawa M, Nixon T (1985) A dual effector theory of growth-hormone action. Differentiation 29(3):195–198

- 116. Brown RJ, Adams JJ, Pelekanos RA, Wan Y, McKinstry WJ, Palethorpe K et al (2005) Model for growth hormone receptor activation based on subunit rotation within a receptor dimer. Nat Struct Mol Biol 12(9):814–821
- 117. Wells JA (1996) Binding in the growth hormone receptor complex. Proc Natl Acad Sci U S A 93(1):1–6
- 118. Brooks AJ, Waters MJ (2010) The growth hormone receptor: mechanism of activation and clinical implications. Nat Rev Endocrinol 6(9):515–525
- 119. Jenkins PJ (2006) Cancers associated with acromegaly. Neuroendocrinology 83(3-4):218-223
- 120. Rogozinski A, Furioso A, Glikman P, Junco M, Laudi R, Reyes A et al (2012) Thyroid nodules in acromegaly. Arq Bras Endocrinol Metabol 56(5):300–304
- 121. Rokkas T, Pistiolas D, Sechopoulos P, Margantinis G, Koukoulis G (2008) Risk of colorectal neoplasm in patients with acromegaly: a meta-analysis. World J Gastroenterol 14(22):3484–3489
- 122. Abreu A, Tovar AP, Castellanos R, Valenzuela A, Giraldo CM, Pinedo AC et al (2016) Challenges in the diagnosis and management of acromegaly: a focus on comorbidities. Pituitary 19(4):448–457
- 123. Holdaway IM, Bolland MJ, Gamble GD (2008) A meta-analysis of the effect of lowering serum levels of GH and IGF-I on mortality in acromegaly. Eur J Endocrinol 159(2):89–95
- 124. Melmed S (2009) Acromegaly pathogenesis and treatment. J Clin Invest 119(11):3189–3202
- 125. Mullis PE (2007) Genetics of growth hormone deficiency. Endocrinol Metab Clin N Am 36(1):17–36
- 126. Alatzoglou KS, Webb EA, Le Tissier P, Dattani MT (2014) Isolated growth hormone deficiency (GHD) in childhood and adolescence: recent advances. Endocr Rev 35(3):376–432
- 127. Laron Z, Blum W, Chatelain P, Ranke M, Rosenfeld R, Savage M et al (1993) Classification of growth hormone insensitivity syndrome. J Pediatr 122(2):241
- Laron Z, Pertzelan A, Karp M (1962) Pituitary dwarfism with high serum levels of growth hormone. Isr J Med Sci 4(4):883–894
- 129. Shevah O, Laron Z (2007) Patients with congenital deficiency of IGF-I seem protected from the development of malignancies: a preliminary report. Growth Hormon IGF Res 17(1):54–57
- 130. Steuerman R, Shevah O, Laron Z (2011) Congenital IGF1 deficiency tends to confer protection against post-natal development of malignancies. Eur J Endocrinol 164(4):485–489
- 131. Guevara-Aguirre J, Balasubramanian P, Guevara-Aguirre M, Wei M, Madia F, Cheng CW et al (2011) Growth hormone receptor deficiency is associated with a major reduction in proaging signaling, cancer, and diabetes in humans. Sci Transl Med 3(70):70ra13. https://doi. org/10.1126/scitranslmed.3001845
- 132. Laron Z, Kauli R (2016) Fifty seven years of follow-up of the Israeli cohort of Laron Syndrome patients-From discovery to treatment. Growth Hormon IGF Res 28:53–56
- 133. Agladioglu SY, Cetinkaya S, Savas Erdeve S, Onder A, Kendirci HN, Bas VN et al (2013) Diabetes mellitus with Laron syndrome: case report. J Pediatr Endocrinol Metab 26(9–10):955–958
- 134. Kopchick JJ, Bellush LL, Coschigano KT (1999) Transgenic models of growth hormone action. Annu Rev Nutr 19:437–461
- 135. Kopchick JJ, List EO, Kelder B, Gosney ES, Berryman DE (2014) Evaluation of growth hormone (GH) action in mice: discovery of GH receptor antagonists and clinical indications. Mol Cell Endocrinol 386(1–2):34–45
- 136. Miquet JG, Freund T, Martinez CS, Gonzalez L, Diaz ME, Micucci GP et al (2003) Hepatocellular alterations and dysregulation of oncogenic pathways in the liver of transgenic mice overexpressing growth hormone. Cell Cycle 12(7):1042–1057
- 137. Bartke A (2003) Can growth hormone (GH) accelerate aging? Evidence from GH-transgenic mice. Neuroendocrinology 78(4):210–216
- 138. Sornson MW, Wu W, Dasen JS, Flynn SE, Norman DJ, O'Connell SM et al (1996) Pituitary lineage determination by the Prophet of Pit-1 homeodomain factor defective in Ames dwarfism. Nature 384(6607):327–333

- 139. Snell GD (1929) Dwarf, a new mendelian recessive character of the house mouse. Proc Natl Acad Sci U S A 15(9):733–734
- 140. Bartke A (1965) Influence of luteotrophin on fertility of dwarf mice. J Reprod Fertil 10:93–103
- 141. Flurkey K, Papaconstantinou J, Miller RA, Harrison DE (2001) Lifespan extension and delayed immune and collagen aging in mutant mice with defects in growth hormone production. Proc Natl Acad Sci U S A 98(12):6736–6741
- 142. Ikeno Y, Bronson RT, Hubbard GB, Lee S, Bartke A (2003) Delayed occurrence of fatal neoplastic diseases in Ames dwarf mice: correlation to extended longevity. J Gerontol A Biol Sci Med Sci 58(4):291–296
- 143. Wiesenborn DS, Ayala JE, King E, Masternak MM (2014) Insulin sensitivity in long-living Ames dwarf mice. Age (Dordr) 36(5):9709
- 144. Heiman ML, Tinsley FC, Mattison JA, Hauck S, Bartke A (2003) Body composition of prolactin-, growth hormone, and thyrotropin-deficient Ames dwarf mice. Endocrine 20(1-2):149–154
- 145. Zhou Y, Xu BC, Maheshwari HG, He L, Reed M, Lozykowski M et al (1997) A mammalian model for Laron syndrome produced by targeted disruption of the mouse growth hormone receptor/binding protein gene (the Laron mouse). Proc Natl Acad Sci U S A 94(24):13215–13220
- 146. Coschigano KT, Clemmons D, Bellush LL, Kopchick JJ (2000) Assessment of growth parameters and life span of GHR/BP gene-disrupted mice. Endocrinology 141(7):2608–2613
- 147. Kinney BA, Coschigano KT, Kopchick JJ, Steger RW, Bartke A (2001) Evidence that ageinduced decline in memory retention is delayed in growth hormone resistant GH-R-KO (Laron) mice. Physiol Behav 72(5):653–660
- 148. Kinney-Forshee BA, Kinney NE, Steger RW, Bartke A (2004) Could a deficiency in growth hormone signaling be beneficial to the aging brain? Physiol Behav 80(5):589–594
- 149. Ikeno Y, Hubbard GB, Lee S, Cortez LA, Lew CM, Webb CR et al (2009) Reduced incidence and delayed occurrence of fatal neoplastic diseases in growth hormone receptor/binding protein knockout mice. J Gerontol A Biol Sci Med Sci 64(5):522–529
- 150. Fan Y, Menon RK, Cohen P, Hwang D, Clemens T, DiGirolamo DJ et al (2009) Liver-specific deletion of the growth hormone receptor reveals essential role of growth hormone signaling in hepatic lipid metabolism. J Biol Chem 284(30):19937–19944
- 151. List EO, Berryman DE, Funk K, Jara A, Kelder B, Wang F et al (2014) Liver-specific GH receptor gene-disrupted (LiGHRKO) mice have decreased endocrine IGF-I, increased local IGF-I, and altered body size, body composition, and adipokine profiles. Endocrinology 155(5):1793–1805
- 152. Vijayakumar A, Wu Y, Buffin NJ, Li X, Sun H, Gordon RE et al (2012) Skeletal muscle growth hormone receptor signaling regulates basal, but not fasting-induced, lipid oxidation. PLoS One 7(9):e44777. https://doi.org/10.1371/journal.pone.0044777
- 153. Wu Y, Liu C, Sun H, Vijayakumar A, Giglou PR, Qiao R et al (2011) Growth hormone receptor regulates beta cell hyperplasia and glucose-stimulated insulin secretion in obese mice. J Clin Invest 121(6):2422–2426
- 154. List EO, Berryman DE, Funk K, Gosney ES, Jara A, Kelder B et al (2013) The role of GH in adipose tissue: lessons from adipose-specific GH receptor gene-disrupted mice. Mol Endocrinol 27(3):524–535
- 155. Lee KY, Russell SJ, Ussar S, Boucher J, Vernochet C, Mori MA et al (2013) Lessons on conditional gene targeting in mouse adipose tissue. Diabetes 62(3):864–874
- 156. Krueger KC, Costa MJ, Du H, Feldman BJ (2014) Characterization of Cre recombinase activity for in vivo targeting of adipocyte precursor cells. Stem Cell Rep 3(6):1147–1158
- 157. List EO, Berryman DE, Buchman M, Parker C, Funk K, Bell S et al (2019) Adipocytespecific gh receptor-null (AdGHRKO) mice have enhanced insulin sensitivity with reduced liver triglycerides. Endocrinology 160(1):68–80
- 158. Sun LY, Fang Y, Patki A, Koopman JJ, Allison DB, Hill CM et al (2017) Longevity is impacted by growth hormone action during early postnatal period. elife 6:pii: e24059. https:// doi.org/10.7554/eLife.24059

- 159. Bartke A (2019) Growth hormone and aging: updated review. World J Mens Health 37(1):19–30
- Masternak MM, Darcy J, Victoria B, Bartke A (2018) Dwarf mice and aging. Prog Mol Biol Transl Sci 155:69–83
- 161. Blackburn A, Schmitt A, Schmidt P, Wanke R, Hermanns W, Brem G et al (1997) Actions and interactions of growth hormone and insulin-like growth factor-II: body and organ growth of transgenic mice. Transgenic Res 6(3):213–222
- 162. Olsson B, Bohlooly YM, Fitzgerald SM, Frick F, Ljungberg A, Ahren B et al (2005) Bovine growth hormone transgenic mice are resistant to diet-induced obesity but develop hyperphagia, dyslipidemia, and diabetes on a high-fat diet. Endocrinology 146(2):920–930
- 163. Palmer AJ, Chung MY, List EO, Walker J, Okada S, Kopchick JJ et al (2009) Age-related changes in body composition of bovine growth hormone transgenic mice. Endocrinology 150(3):1353–1360
- 164. Katznelson L (2009) Alterations in body composition in acromegaly. Pituitary 12(2):136–142
- 165. Berryman DE, List EO, Coschigano KT, Behar K, Kim JK, Kopchick JJ (2004) Comparing adiposity profiles in three mouse models with altered GH signaling. Growth Hormon IGF Res 14(4):309–318
- 166. Berryman DE, List EO, Palmer AJ, Chung MY, Wright-Piekarski J, Lubbers E et al (2010) Two-year body composition analyses of long-lived GHR null mice. J Gerontol A Biol Sci Med Sci 65(1):31–40
- 167. Bonkowski MS, Pamenter RW, Rocha JS, Masternak MM, Panici JA, Bartke A (2006) Longlived growth hormone receptor knockout mice show a delay in age-related changes of body composition and bone characteristics. J Gerontol A Biol Sci Med Sci 61(6):562–567
- Berryman DE, List EO, Kohn DT, Coschigano KT, Seeley RJ, Kopchick JJ (2006) Effect of growth hormone on susceptibility to diet-induced obesity. Endocrinology 147(6):2801–2808
- 169. Stout MB, Tchkonia T, Pirtskhalava T, Palmer AK, List EO, Berryman DE et al (2014) Growth hormone action predicts age-related white adipose tissue dysfunction and senescent cell burden in mice. Aging (Albany NY) 6(7):575–586
- 170. Reyes-Vidal CM, Mojahed H, Shen W, Jin Z, Arias-Mendoza F, Fernandez JC et al (2015) Adipose tissue redistribution and ectopic lipid deposition in active acromegaly and effects of surgical treatment. J Clin Endocrinol Metab 100(8):2946–2955
- 171. Wang Z, Al-Regaiey KA, Masternak MM, Bartke A (2006) Adipocytokines and lipid levels in Ames dwarf and calorie-restricted mice. J Gerontol A Biol Sci Med Sci 61(4):323–331
- 172. Troike KM, Henry BE, Jensen EA, Young JA, List EO, Kopchick JJ et al (2017) Impact of growth hormone on regulation of adipose tissue. Compr Physiol 7(3):819–840
- 173. Menon V, Zhi X, Hossain T, Bartke A, Spong A, Gesing A et al (2014) The contribution of visceral fat to improved insulin signaling in Ames dwarf mice. Aging Cell 13(3):497–506
- 174. Masternak MM, Bartke A, Wang F, Spong A, Gesing A, Fang Y et al (2012) Metabolic effects of intra-abdominal fat in GHRKO mice. Aging Cell 11(1):73–81
- 175. Xu M, Pirtskhalava T, Farr JN, Weigand BM, Palmer AK, Weivoda MM et al (2018) Senolytics improve physical function and increase lifespan in old age. Nat Med 24(8):1246–1256
- 176. Li Y, Knapp JR, Kopchick JJ (2003) Enlargement of interscapular brown adipose tissue in growth hormone antagonist transgenic and in growth hormone receptor gene-disrupted dwarf mice. Exp Biol Med (Maywood) 228(2):207–215
- 177. Darcy J, McFadden S, Fang Y, Huber JA, Zhang C, Sun LY et al (2016) Brown adipose tissue function is enhanced in long-lived, male Ames dwarf mice. Endocrinology 157(12):4744–4753
- 178. Westbrook R, Bonkowski MS, Strader AD, Bartke A (2009) Alterations in oxygen consumption, respiratory quotient, and heat production in long-lived GHRKO and Ames dwarf mice, and short-lived bGH transgenic mice. J Gerontol A Biol Sci Med Sci 64(4):443–451
- 179. Darcy J, McFadden S, Fang Y, Berryman DE, List EO, Milcik N et al (2018) Increased environmental temperature normalizes energy metabolism outputs between normal and Ames dwarf mice. Aging (Albany NY) 10(10):2709–2722

# Chapter 12 The Cytoskeleton as a Modulator of Aging and Neurodegeneration



Konstantinos Kounakis and Nektarios Tavernarakis

# 1 Introduction

The cytoskeleton is a cellular entity, encompassing a multitude of filamentous proteins, forming structures that impart mechanical strength, allow intracellular transport and spatial organization, connect the cell to its environment, and generate forces that permit movement [1]. The ubiquitous nature of the cytoskeleton and the breadth of its functionality make it one of the most fascinating aspects of cellular biology, as well as one that is always worth considering when researching or discussing phenomena that affect the cells. In this review, we discuss the relevance of the cytoskeleton to the processes of aging and neurodegeneration, and provide examples that demonstrate its importance.

# 1.1 Components of the Cytoskeleton

Three types of cytoskeleton polymers have been defined: actin microfilaments, microtubules and intermediate filaments [1, 2]. We briefly describe their structure and function below.

K. Kounakis · N. Tavernarakis (⊠)

Institute of Molecular Biology and Biotechnology, Foundation for Research and Technology-Hellas, Heraklion, Greece

Department of Basic Sciences, Medical School, University of Crete, Heraklion, Greece e-mail: tavernarakis@imbb.forth.gr

<sup>©</sup> Springer Nature Switzerland AG 2019

P. C. Guest (ed.), *Reviews on Biomarker Studies in Aging and Anti-Aging Research*, Advances in Experimental Medicine and Biology 1178, https://doi.org/10.1007/978-3-030-25650-0\_12

#### 1.1.1 Actin Microfilaments

Actin filaments (also commonly referred to as F-actin) are about 7 nm in diameter and consist of monomers of globular (G-actin) that interact head-to-tail with each other. G-actin can bind ATP, and this promotes its polymerization to F-actin. ATP is subsequently hydrolyzed to ADP. Actin filaments are polarized with a positive (+) and negative (-) end. Polymerization can occur at both ends but is significantly faster at the + end. The filaments can organize into higher order structures with the help of crosslinkers (Fig. 12.1). Highly aligned actin bundles are responsible for the formation of narrow cell protrusions, such as filopodia, while highly branched bundles take part in larger cellular movements, such as those that occur in phagocytosis. The polarity of the filaments also allows them to support a family of ATP driven motor proteins, the myosins, that contribute to actin network organization and force generation [1, 2].

In neurons, actin forms patches in the initial segment of the axon and at points along its length [3, 4]. It also forms, in association with the actin capping protein



**Fig. 12.1** Organization of actin microfilaments. (**a**) G-actin polymerization. Actin monomers are loaded with ATP with the help of a protein with ATP exchange factor (AEF) activity. This induces their polymerization. (**b**) F-actin. (**c**) Larger scale F-actin organization facilitated by crosslinker proteins

adducin, a series of periodic rings spaced by spectrin that are wrapped around the axonal shaft [5]. Actin is also a major contributor the motility and guidance of the neuronal growth cone. The growth cone has three domains: the central (C), the peripheral (P) and the transition (T) domain [6]. Actin is rich in the P and T domain and its polymerization and recycling allows for the formation of exploratory filopodia. In addition, myosin 2 generates forces that assist in propelling the growth cone forward and steer it towards its targets. Inhibition of these functions does not prevent axonal growth but it significantly reduces its speed and abolishes its ability to respond to guidance cues [7, 8]. It can also act as the driving force for axonal branching, as actin filament patches can initiate the formation of protrusions that subsequently are invaded by microtubules to create new collateral branches that allow the same axon to interact with multiple targets [9]. Furthermore, actin has been connected to synaptic signaling, as it has been implicated in the regulation of synaptic vesicle pools, vesicle docking to the active zone, and even endocytic retrieval of vesicle membranes [10]. Finally, actin contributes to dendritic spine organization [11-13] and, in collaboration with microtubules and the receptorassociated protein gephyrin, contributes to postsynaptic receptor clustering [14].

In oligodendrocytes, the glial cells that are responsible for myelinating the axons of the CNS to facilitate fast action potential conduction, the actin cytoskeleton plays a critical role by allowing these cells to alter their morphology during development [15]. These cells possess protrusions with actin rich filopodia and lamellipodia. Actin in these structures is organized in a fashion mostly similar to growth cones [16]. It acts as a necessary driving force that allows these protrusions to extend towards their target axons and wrap around them [17–20]. Subsequently, actin depolymerization allows these protrusions to convert into sheets by reducing surface tension, enabling proper myelin spreading [19].

#### 1.1.2 Microtubules

Microtubules (MTs) are cylindrical bundles of parallel protofilaments comprised of  $\alpha$ - and  $\beta$ - tubulin. These bundles can have 10–16 individual filaments, with 13 being the most common. They have a typical diameter of about 25 nm. Both tubulins can bind GTP, which promotes polymerization, but eventually hydrolyze it to GDP, weakening their affinity. This leads to what is described as "dynamic instability", as microtubules can switch between stable growth and rapid depolymerization. Microtubules are polarized, with a + and a – end. This polarity becomes particularly apparent during a phenomenon known as treadmilling, during which tubulin is simultaneously removed from the – end of the filament and polymerized to the + end (Fig. 12.2). This polarity also allows microtubules to support ATP driven motor proteins, the kinesins and dyneins, which are responsible for the guided transport of cellular cargo. Microtubules interact with a group of proteins known as MAPs (microtubule associated proteins) that influence their stability and interactions with other cellular components. A subset of MAPs, the +TIPs (plus-end-tracking proteins) interact specifically with growing microtubule ends. There are also – end



Fig. 12.2 Organization of microtubules. (a) GTP is loaded to  $\alpha$ - and  $\beta$ - tubulin with the help of a protein with GTP Exchange Factor (GEF) activity. (b) A tubulin filament. (c) Microtubule structure and dynamics

capping proteins that can prevent depolymerization. Microtubule nucleation often needs to start at a Microtubule organization center (MTOC) where  $\gamma$ -tubulin interacts with  $\alpha$ - and  $\beta$ - tubulin, providing a base for the start of filament extension [1, 2, 21]. The MTOC of mammalian cells is known as the centrosome. It consists of two perpendicular tubulin structures known as centrioles that are surrounded by a centrosomal matrix of proteins involved in microtubule nucleation, anchoring and release. The duplicated centrosome is responsible for the formation of the mitotic spindle, the microtubule structure that segregates chromatids during cell division [22].

In mature neurons, microtubules are arranged with the - end towards the cell body and the + end extending outwards, along the axon. They are discontinuous, with multiple start and stop sites [21]. In this context, there is evidence that microtubules cease to rely on the centrosomal MTOC for their organization [23, 24]. Axonal microtubules extend into growth cones, where they localize primarily in the C domain. However, they can extend even further and they are known to interact with actin, particularly actin bundles that form filopodia in the P zone [6]. These microtubules have dynamic ends and are crucial to growth cone steering [25]. Outside of the growth cone, in cases of interstitial axonal branching formation, some axonal microtubules are reorganized and interact with the newly forming protrusion [26]. Microtubules also extend between the cell body and dendrites. In this case they adopt a mixed orientation, with + and - ends facing towards both directions [27].

#### 1.1.3 Intermediate Filaments

Intermediate filaments (IFs) constitute a diverse family of cytoskeletal proteins that are expressed differentially across cell types. All of these proteins share structural similarity and organize in similar ways to provide mechanical strength and stability to most cell types, especially against tensile forces. IF subunits consist of a globular N-terminal head, an  $\alpha$ -helical core and a variable C-terminal domain. Intermediate filament monomers tend to coalesce in pairs, forming parallel coiled coil dimers. Two antiparallel dimers can also associate to form a tetramer (Fig. 12.3). The higher scale organization depends on the tissue and the actual filament components but typically leads to a filamentous polymer of ~10 nm in diameter. Examples of IFs include keratin, vimentin, the lamins of the nuclear skeleton,  $\alpha$ -internexin, peripherin, synemin, nestin or the light, medium and heavy neurofilaments (NF-L, NF-M and NF-H, respectively). Intermediate filaments are not polarized and therefore do not support molecular motors [1, 2, 28, 29].

Neuronal intermediate filaments (which will be referred as neurofilaments or NFs from now on) represent the main cytoskeletal element of mature neurons. They consist of NF-L, NF-M, NF-H and occasionally  $\alpha$ -internexin and peripherin, with



Fig. 12.3 Intermediate filament organization. (a) Monomer. (b) Dimer of parallel monomers. (c) Tetramer of antiparallel dimers

the actual composition varying by organism or even stage of development. The NF-M and NF-H C-terminal domains are notable for the presence of a large number of lysine-serine-proline (KSP) repeats that represent targets for regulatory phosphorylation. Neurofilaments reside in axons and act as regulators of axonal caliber, which has implications in myelin thickness and the rate of axonal conduction. They are also associated with axonal growth and regeneration [2, 29, 30].

# 2 The Importance of the Cytoskeleton to Aging and Neurodegeneration

### 2.1 Cytoskeleton and Organismal Aging

There is ample experimental evidence connecting the cytoskeleton with the processes of cellular and organismal aging. In yeast, actin has emerged as a regulator of lifespan by regulating the inheritance of mitochondria. During budding, actin cables create a retrograde flow from the bud towards the mother cell, driven by polymerization and myosin activity. This flow pushes mitochondria away from the bud, forcing them to "swim upstream" and ensuring that only healthy mitochondria can reach the new cell, granting it a longer lifespan and healthspan [31, 32].

In C. elegans, the actin cytoskeleton has been observed to deteriorate with aging. HSF-1, the master regulator of the heat shock response that provides thermotolerance and also contributes to organismal longevity, has been shown to act against this deterioration. This effect is mostly mediated through the upregulation of the expression of the calcium binding protein PAT-10. Most notably, loss of pat-10 is sufficient to decrease organismal lifespan and thermotolerance, while overexpression enhances thermotolerance and promotes longevity [33, 34].

In mammals and particularly in humans, oocyte fertility is reduced in aging. This is in part due to deterioration of meiotic spindle integrity. Spindle microtubules lose their ability to accurately interact with meiotic chromosomes and separate them, thus causing aneuploidies. The deterioration of the spindle can be attributed to the reduced activity of enzymes that are responsible for centrosome and microtubule maintenance [22, 35]. Centrosome defects have also been proposed as a possible explanation for the age-related decline of stem cell division [36].

The myelin sheath is crucial to adult neuron performance. Unfortunately, even healthy aging is accompanied by the emergence of defects in myelin composition and structure [37–39]. Thus maintenance mechanisms need to be activated to protect the axons. In the case of CNS oligodendrocytes, there is evidence indicating that the cytoskeleton is a fundamental constituent of these processes. Septins, a family of cytoskeleton associated scaffold proteins, have been shown to form filaments along with anillin in mice that support the myelin sheath and loss of these proteins leads to defects in myelin structure [40]. Additionally, it has been shown that de novo myelination pathways in the CNS remain active in adulthood through

new oligodendrocytes [41–43], the maturation of which is guided by cytoskeleton dynamics. This de novo myelination has been mostly associated with plasticity related adaptations but could also participate in maintenance.

In humans, aging is associated with increased aortic stiffness. This is often considered to preclude myocardial infarction, renal disease or even cognitive decline. A significant part of this aortic stiffness is attributed to vascular smooth muscle cells and particularly to their non-muscle actin cytoskeleton that is responsible for their connection to the extracellular matrix. Decoy peptides that inhibit actin polymerization or the interaction of the actin associated proteins talin and vinculin have been shown to be a potential method for counteracting aortic stiffness [44].

Another issue that emerges with human aging is the deterioration of heart health. Heart failure in particular is one of the most prominent causes of death and disability in the elderly [45]. Actin is critical to heart health, as actin fibers constitute a major component of sarcomers, the mechanical units that drive cardiomyocyte contraction [46]. Experiments in mice and rats have shown a conserved activation of actin remodeling by vinculin during aging. Further experiments in Drosophila have suggested that this is an anti-aging mechanism that improves heart function and overall organismal lifespan [47]. Actin is also relevant to heart health due to its association with the proliferative capacity of cardiac fibroblasts. Aging fibroblasts exhibit reduced levels of the LOX-1 receptor, lose their proliferative capacity and exhibit a disorganized actin network. Restoration of LOX-1 levels re-establishes fibroblast proliferative potential and reinstates actin organization [48].

### 2.2 Cytoskeleton and Neurodegeneration

Considering the prominent presence and important functionality of cytoskeletal proteins in neurons, it comes as no surprise that they have also been heavily implicated over the years in processes underlying their dysfunction. Below, we discuss experimental data that connects the cytoskeleton to neurodegenerative diseases, as well as injury induced neurodegeneration.

#### 2.2.1 Tau Associated Pathologies

Alzheimer's disease (AD) is characterized by extracellular deposits of Aβ peptides and intracellular filamentous aggregates of Tau, a major microtubule associated protein [49–52]. Beyond AD, Tau aggregation has emerged as a common form of phenomenon in more than 20 different types of neurological disease, including Pick's disease, progressive supranuclear palsy, chronic traumatic encelopathy, argyrophilic grain disease, frontotemporal dementia with parkinsonism-17, corticobasal degeneration and Parkinson's disease (PD) [49, 53]. In the human brain, Tau has six isoforms with either 3 or 4 microtubule binding repeats at its C-terminal domain (3R and 4R Tau respectively) [49, 54, 55]. The protein is typically a dipole but post-translational modification, especially phosphorylation, can affect its charges and disrupt its ability to bind microtubules [56–58]. In addition to its microtubule binding abilities, it has been shown to interact with the plasma membrane [59]. It is also capable of interacting with actin, induce its polymerization and promote microtubule and actin co-alignment [60].

Tau assembles into filaments through its repeats forming a cross-beta structure. Thus, the microtubule binding regions are trapped in the core of the aggregate, rendering physiological interaction with microtubules impossible [61–64]. Tau aggregates are commonly referred to as Neurofibrillary Tangles (NFTs), but their actual morphology can vary across different diseases, leading to their sub-characterization into paired helical filaments (PHFs), straight filaments (SFs) and twisted ribbon-like lilaments (TRFs) [49, 51]. Tau is abnormally hyperphosphorylated in all of its aggregates. This has led to the belief that phosphorylation is toxic and induces Tau aggregation. However, this might not be the case as human tauopathies have not been linked to defects in kinases or phosphatases, and kinase inhibition has not been shown to be an effective treatment option [49, 51]. Furthermore, there is evidence of Tau phosphorylation acting in a benign fashion in the process of hibernation [65, 66], without fibril formation and with reversibility.

There are several possible explanations on the causes of Tau associated neuropathology; Tau aggregation could lead to an effective LoF phenotype by preventing the protein form exercising its normal roles [67]. For instance loss of Tau in mouse models of AD (over-expressing mutant APP, the precursor of the Aß peptide) aggravated neurodegeneration and exhibited axonal swellings full of cellular debris and mislocalized organelles, vesicles and even presynaptic terminal components [68]. In addition, Tau KO mice exhibit intracellular iron accumulation, substantia nigra neurodegeneration, brain atrophy and parkinsonism. Supplementation with an iron chelator rescued this phenotype. These observations were attributed to reduced transport of APP onto the neuronal membrane (APP in conjunction with ferroportin acts as the sole iron export system in neurons) due to the altered microtubule dynamics that arise from lack of Tau [69]. Another indication supporting this idea is the observation that microtubule stabilizing drug treatment has had some effectiveness in ameliorating tauopathy [70–72]. An alternate explanation could be that Tau (normal, mutant and/or phosphorylated) represents a toxic threat to cells in a gain of function (GoF) fashion. The protein has, for instance, been implicated in the disruption of mitochondria through the induction of mitochondrial fusion, inhibition of mitophagy and a reduction of ATP production [73, 74]. There are indications suggesting that a GoF threat might arise from non-filamentous forms of Tau [51], as experiments have demonstrated that truncated/cleaved Tau can be toxic [75, 76]. In addition, neurodegeneration can occur before or without Tau filament formation [77, 78] and tangle formation can persist in rescued animal models [79]. In the latter case, NFT formation might act as an attempt from the cell to quarantine dangerous Tau forms. Arguably, it is possible that both explanations are true on a disease by disease basis, or even simultaneously, with aggregation acting as the "lesser evil" that initially protects neuronal cells from toxicity but eventually ends up being deleterious through dysregulation of the cytoskeleton or other effects.

The aforementioned Tau-actin interaction [60] might have a functional implication in neurodegenerative disease, as experiments in Drosophila melanogaster and have shown that mutant forms of Tau associated with human tauopathies are capable of inducing the formation of actin rich structures resembling Hirano bodies (actin aggregates that occur in human patients). Actin was necessary for Tau toxicity in these instances. Tau phosphorylation, as well as transgenic A $\beta$ 42 expression, exacerbated actin aggregation and neuronal death [80]. Recently it was reported that tau can accumulate and form tangles in the medial temporal lobe and particularly in the entorhinal cortex as a pure consequence of normal "healthy" aging indicating a possible mechanism for the aging-associated loss of episodic memory [81].

#### 2.2.2 Other Microtubule Associated Pathologies

The implication of microtubules in neurodegenerative disease extends beyond the role of Tau. Part of the neurotoxicity in Huntington's disease (HD) can be attributed to defects in microtubule based axonal transport, and MT stabilizing acetylation is potentially beneficial [82]. Very similar observations have been made in a model of Charcot-Marie-Tooth disease (CMT) [83]. Experiments in a PD model have shown that intracellular transport could be disrupted due to the reduction of microtubule dynamics, and that this might preclude mitochondrial damage and caspase 3 activation [84]. Disrupted mitochondrial dynamics, along with reduced levels of MAP expression, can also be observed in amyotrophic lateral sclerosis (ALS) patients and models, and pharmacological MT stabilization can delay the progression of the disease in mice [85–87].

#### 2.2.3 Actin Associated Pathologies

ALS is a neurodegenerative disorder associated with the loss of motor neurons in the cerebral cortex, the brainstem, and the ventral horn of the spinal cord [88]. The disease is mainly linked with alterations in genes such as superoxide dismutase 1 (SOD1), fused in sarcoma (FUS) and TAR DNA binding protein (TARDBP / TDP-43) [89]. Spinal muscular atrophy (SMA) is a disorder with phenotypical similarity to ALS that exhibits motor neuron loss exclusively in the ventral horn of the spinal cord [88]. SMA is attributed to loss of function (LoF) of the survival of motor neuron 1 gene (SMN1) [90]. Both ALS and SMA have been linked with altered cytoskeletal dynamics or mutations in known regulators of the cytoskeleton [91–97]. Notably, the actin regulators profilins have been implicated in both diseases [93, 96-98]. Profilins are a family of proteins that can bind monomeric G actin and facilitate the exchange of ADP for ATP. Depending on the cellular conditions, profilins have been suggested to act as either a promoter of actin polymerization and F-actin stabilizer, or as a sequester of G-actin and F-actin destabilizer [88, 99]. Profilin binding activity can be inhibited through phosphorylation by the RhoA kinase (ROCK), an important regulator of actin dynamics [100, 101]. It has been 236

shown that SMN1 binding to profilin 2 reduces its inhibitory effects and promotes actin polymerization [93]. It has also been suggested that this binding protects profilin from ROCK phosphorylation and that the source of cytoskeletal defects in SMA is the loss of this protection [88]. In ALS, profilin 1 has been suggested to contribute to disease pathology through the formation of TDP-43 associated aggregates [97, 102, 103], through loss of its ability to interact with stress granules [104], or through dysregulation of actin dynamics [105, 106]. It is worth mentioning that profilin has also been shown to interact with the polyglutamate protein Huntingtin and inhibit its aggregation. The prevention of profilin inhibition by ROCK has also been demonstrated as a potential therapeutic approach for HD [101]. Beyond its aforementioned potential association with Tau, another connection of actin with AD pathology was revealed recently. The actin cytoskeleton was shown to be compromised in transgenic mouse models early in disease progression in conjunction with dendritic spine effects and a decline of AMPA signaling [107].

Microglia can act as a line of defense against AD by migrating towards extracellular A $\beta$ 42 aggregates, binding them and phagocytosing them. However during aging, Nogo/Ngr signaling reduces the ability of microglia to migrate and adhere to A $\beta$ 42 through Rho-GTPases that regulate actin dynamics and end up preventing protrusion extension and cell polarization [108]. On the other hand, the cytoskeleton might also have an inhibitory role in this interaction, as it has been reported that cytosolic phospholipase A2 (cPLA2), a factor that mediates the A $\beta$ -induced response in glial cells, acts to reduce the cytoskeletal-membrane connectivity that represents a physical barrier against A $\beta$  endocytosis [109].

Alterations in actin dynamics may also play a role in PD, as a-syn has been shown to inhibit cofilin, an actin destabilizer, in experimental models and patients. This leads to actin overstabilization, with potential negative implications for synaptic signaling [110]. Cdc42 is a Rho-GTPase that is involved in the regulation of actin dynamics. Some variants of variants of CMT have been associated with a mutation in Frabin, the GTP exchange factor of Cdc42 [111, 112].

#### 2.2.4 Neurofilament Associated Pathologies

Neurofilaments have also been associated with neurodegenerative disease. Abnormal neurofilament aggregation has been observed in various disorders, such as AD, PD, CMT and ALS. It seems to be connected with deviations from the exact correct NF component stoichiometry, as it can occur in response to both down-regulation or up-regulation of individual NF genes [113–115].

In AD, neurofilaments are another major component of Tau NFTs [116]. In these tangles, they adopt a paired helical filament conformation [117], and exhibit extensive levels of phosphorylation [118].

Neurofilaments are also a primary component of the Lewy bodies, the characteristic protein inclusions of PD [119, 120]. They are extensively phosphorylated in this instance as well [121]. Patient tissues exhibit down-regulation of NF-L and NF-H expression [122]. Mutations in the gene that codes NF-L have emerged as a cause for CMT. These mutations lead to defects of axonal transport, neurofilament disorganization, and usually aggregation [123–131]. Some of the NF-L mutations that cause CMT lead to neurofilament aggregation due to the abolition of protective phosphorylation [129–132].

ALS is characterized by intraneuronalaxonal NF aggregation [133–135]. This is also the case in mice expressing mutant human SOD1, the gene mostly associated with familial cases of ALS [136]. This aggregation might be dispensable for the eventual progression of the disease [137] but its reduction might still be somewhat beneficial. Perhaps unexpectedly, overexpression of NF-H [138–140], or NF-L [138] or downregulation of NF-L [141] were all successful in imparting a partial protective effect that is attributed to a redirection of NF accumulation from the axon to the cell body/perikaryon. The exact mechanism of this protection is, however, uncertain.

#### 2.2.5 Neurodegeneration Due to Injury

Injured axons of CNS neurons degenerate, in a process known as Wallerian degeneration. Fragmentation of microtubules is possibly the earliest step in this process [142]. Axons that are retracting due to injury exhibit a disorganized microtubule network [143]. In cases where axonal regeneration is possible (such as the peripheral nervous system), it is driven by microtubules and requires tubulin deacetylation, a modification that decreases their stability [144, 145]. The levels of expressed and axonally transported neurofilaments are also reduced, and are only restored in axons that can regenerate [146–154]. Microtubule destabilization accompanied by energy depletion precludes neurofilament defects, mitochondrial swelling and axonal degeneration. Artificial energy repletion is effective at stopping this process [155]. Dendrites also degenerate after injury. Experiments in D. Melanogaster showed that this requires microtubule severance by the ATPase fidgetin [156].

# 3 Conclusions

Despite decades of research, our knowledge on the cytoskeleton remains incomplete. There are still numerous questions that need to be addressed regarding cytoskeletal contributions to pathology. In this regard, the cytoskeleton represents a clear challenge for future research, and for the development of potential therapeutic strategies relevant to aging and neurodegeneration.

Acknowledgements We apologize to those colleagues, whose work could not be referenced owing to space limitations. Work in the authors' laboratory is funded by grants from the European Research Council (ERC – GA695190 – MANNA, ERC – GA737599 – NeuronAgeScreen), the European Commission Framework Programmes, and the Greek Ministry of Education. Konstantinos Kounakis is supported by the Greek Foundation of Research and Innovation (ELIDEK).

# References

- Fletcher DA, Mullins RD (2010) Cell mechanics and the cytoskeleton. Nature 463(7280): 485–492
- Eira J, Silva CS, Sousa MM, Liz MA (2016) The cytoskeleton as a novel therapeutic target for old neurodegenerative disorders. Prog Neurobiol 141:61–82
- Spillane M, Ketschek A, Jones SL, Korobova F, Marsick B, Lanier L et al (2011) The actin nucleating Arp2/3 complex contributes to the formation of axonal filopodia and branches through the regulation of actin patch precursors to filopodia. Dev Neurobiol 71(9):747–758
- Watanabe K, Al-Bassam S, Miyazaki Y, Wandless TJ, Webster P, Arnold DB (2012) Networks of polarized actin filaments in the axon initial segment provide a mechanism for sorting axonal and dendritic proteins. Cell Rep 2(6):1546–1553
- 5. Xu K, Zhong G, Zhuang X (2013) Actin, spectrin, and associated proteins form a periodic cytoskeletal structure in axons. Science 339(6118):452–456
- Schaefer AW, Kabir N, Forscher P (2002) Filopodia and actin arcs guide the assembly and transport of two populations of microtubules with unique dynamic parameters in neuronal growth cones. J Cell Biol 158(1):139–152
- 7. Gomez TM, Letourneau PC (2014) Actin dynamics in growth cone motility and navigation. J Neurobiol 129(2):221–234
- Marsh L, Letourneau PC (1984) Growth of neurites without filopodial or lamellipodial activity in the presence of cytochalasin B. J Cell Biol 99(6):2041–2047
- 9. Gallo G (2011) The cytoskeletal and signaling mechanisms of axon collateral branching. Dev Neurobiol 71(3):201–220
- Cingolani LA, Goda Y (2008) Actin in action: the interplay between the actin cytoskeleton and synaptic efficacy. Nat Rev Neurosci 9(5):344–356
- Cohen RS, Chung SK, Pfaff DW (1985) Immunocytochemical localization of actin in dendritic spines of the cerebral cortex using colloidal gold as a probe. Cell Mol Neurobiol 5(3):271–284
- Matus A, Ackermann M, Pehling G, Byers HR, Fujiwara K (1982) High actin concentrations in brain dendritic spines and postsynaptic densities. Proc Natl Acad Sci U S A 79(23):7590–7594
- Korobova F, Svitkina T (2010) Molecular architecture of synaptic actin cytoskeleton in hippocampal neurons reveals a mechanism of dendritic spine morphogenesis. Mol Biol Cell 21(1):165–176
- Kirsch J, Betz H (1995) The postsynaptic localization of the glycine receptor-associated protein gephyrin is regulated by the cytoskeleton. J Neurosci 15(6):4148–4156
- 15. Seixas AI, Azevedo MM, de Faria JP, Fernandes D, Mendes Pinto I, Relvas JB (2019) Evolvability of the actin cytoskeleton in oligodendrocytes during central nervous system development and aging. Cell Mol Life Sci 76(1):1–11
- Fox MA, Afshari FS, Alexander JK, Colello RJ, Fuss B (2006) Growth conelike sensorimotor structures are characteristic features of postmigratory, premyelinating oligodendrocytes. Glia 53(5):563–566
- Song J, Goetz BD, Baas PW, Duncan ID (2001) Cytoskeletal reorganization during the formation of oligodendrocyte processes and branches. Mol Cell Neurosci 17(4):624–636
- Azevedo MM, Domingues HS, Cordelieres FP, Sampaio P, Seixas AI, Relvas JB (2018) Jmy regulates oligodendrocyte differentiation via modulation of actin cytoskeleton dynamics. Glia 66(9):1826–1844
- Nawaz S, Sanchez P, Schmitt S, Snaidero N, Mitkovski M, Velte C et al (2015) Actin filament turnover drives leading edge growth during myelin sheath formation in the central nervous system. Dev Cell 34(2):139–151
- Zuchero JB, Fu MM, Sloan SA, Ibrahim A, Olson A, Zaremba A et al (2015) CNS myelin wrapping is driven by actin disassembly. Dev Cell 34(2):152–167

- 12 Cytoskeleton in Aging and Neurodegeneration
- Penazzi L, Bakota L, Brandt R (2016) Microtubule dynamics in neuronal development, plasticity, and neurodegeneration. Int Rev Cell Mol Biol 321:89–169
- 22. Schatten H, Sun QY (2018) Functions and dysfunctions of the mammalian centrosome in health, disorders, disease, and aging. Histochem Cell Biol 150(4):303–325
- Nguyen MM, Stone MC, Rolls MM (2011) Microtubules are organized independently of the centrosome in Drosophila neurons. Neural Dev 6:38. https://doi.org/10.1186/1749-8104-6-38
- 24. Stiess M, Maghelli N, Kapitein LC, Gomis-Ruth S, Wilsch-Brauninger M, Hoogenraad CC et al (2010) Axon extension occurs independently of centrosomal microtubule nucleation. Science 327(5966):704–707
- Challacombe JF, Snow DM, Letourneau PC (1997) Dynamic microtubule ends are required for growth cone turning to avoid an inhibitory guidance cue. J Neurosci 17(9):3085–3095
- Dent EW, Callaway JL, Szebenyi G, Baas PW, Kalil K (1999) Reorganization and movement of microtubules in axonal growth cones and developing interstitial branches. J Neurosci 19(20):8894–8908
- Conde C, Caceres A (2009) Microtubule assembly, organization and dynamics in axons and dendrites. Nat Rev Neurosci 10(5):319–332
- Herrmann H, Aebi U (2016) Intermediate filaments: structure and assembly. Cold Spring Harb Perspect Biol 8(11):pii: a018242. https://doi.org/10.1101/cshperspect.a018242
- Perrot R, Berges R, Bocquet A, Eyer J (2008) Review of the multiple aspects of neurofilament functions, and their possible contribution to neurodegeneration. Mol Neurobiol 38(1):27–65
- Zhu Q, Couillard-Despres S, Julien JP (2013) Delayed maturation of regenerating myelinated axons in mice lacking neurofilaments. Exp Neurol 148(1):299–316
- Higuchi R, Vevea JD, Swayne TC, Chojnowski R, Hill V, Boldogh IR et al (2013) Actin dynamics affect mitochondrial quality control and aging in budding yeast. Curr Biol 23(23):2417–2422
- 32. Higuchi-Sanabria R, Vevea JD, Charalel JK, Sapar ML, Pon LA (2016) The transcriptional repressor Sum1p counteracts Sir2p in regulation of the actin cytoskeleton, mitochondrial quality control and replicative lifespan in Saccharomyces cerevisiae. Microb Cell 3(2):79–88
- Baird NA, Douglas PM, Simic MS, Grant AR, Moresco JJ, Wolff SC et al (2014) HSF-1-mediated cytoskeletal integrity determines thermotolerance and life span. Science 346(6207):360–363
- 34. Higuchi-Sanabria R, Paul Rd JW, Durieux J, Benitez C, Frankino PA, Tronnes SU et al (2018) Spatial regulation of the actin cytoskeleton by HSF-1 during aging. Mol Biol Cell 29(21):2522–2527
- 35. Wang ZB, Schatten H, Sun QY (2011) Why is chromosome segregation error in oocytes increased with maternal aging? Physiology (Bethesda) 26(5):314–325
- Cheng J, Turkel N, Hemati N, Fuller MT, Hunt AJ, Yamashita YM (2008) Centrosome misorientation reduces stem cell division during ageing. Nature 456(7222):599–604
- 37. Liu H, Yang Y, Xia Y, Zhu W, Leak RK, Wei Z et al (2017) Aging of cerebral white matter. Ageing Res Rev 34:64–76
- Stahon KE, Bastian C, Griffith S, Kidd GJ, Brunet S, Baltan S (2016) Age-related changes in axonal and mitochondrial ultrastructure and function in white matter. J Neurosci 36(39):9990–10001
- 39. Peters A, Kemper T (2012) A review of the structural alterations in the cerebral hemispheres of the aging rhesus monkey. Neurobiol Aging 33(10):2357–2372
- Patzig J, Erwig MS, Tenzer S, Kusch K, Dibaj P, Mobius W et al (2016) Septin/anillin filaments scaffold central nervous system myelin to accelerate nerve conduction. Elife 5:pii: e17119. https://doi.org/10.7554/eLife.17119
- 41. Young KM, Psachoulia K, Tripathi RB, Dunn SJ, Cossell L, Attwell D et al (2013) Oligodendrocyte dynamics in the healthy adult CNS: evidence for myelin remodeling. Neuron 77(5):873–885
- 42. Hill RA, Li AM, Grutzendler J (2018) Lifelong cortical myelin plasticity and age-related degeneration in the live mammalian brain. Nat Neurosci 21(5):683–695

- Hughes EG, Orthmann-Murphy JL, Langseth AJ, Bergles DE (2018) Myelin remodeling through experience-dependent oligodendrogenesis in the adult somatosensory cortex. Nat Neurosci 21(5):696–706
- 44. Nicholson CJ, Singh K, Saphirstein RJ, Gao YZ, Li Q, Chiu JG et al (2018) Reversal of aging-induced increases in aortic stiffness by targeting cytoskeletal protein-protein interfaces. J Am Heart Assoc 7(15):pii: e008926. https://doi.org/10.1161/JAHA.118.008926
- 45. Biernacka A, Frangogiannis NG (2011) Aging and cardiac fibrosis. Aging Dis 2(2):158-173
- 46. Thompson BR, Metzger JM (2014) Cell biology of sarcomeric protein engineering: disease modeling and therapeutic potential. Anat Rec (Hoboken) 297(9):1663–1669
- 47. Kaushik G, Spenlehauer A, Sessions AO, Trujillo AS, Fuhrmann A, Fu Z et al (2015) Vinculin network-mediated cytoskeletal remodeling regulates contractile function in the aging heart. Sci Transl Med 7(292):292ra99. https://doi.org/10.1126/scitranslmed.aaa5843
- Wang X, Khaidakov M, Ding Z, Dai Y, Mercanti F, Mehta JL (2013) LOX-1 in the maintenance of cytoskeleton and proliferation in senescent cardiac fibroblasts. J Mol Cell Cardiol 60:184–190
- 49. Chang HY, Sang TK, Chiang AS (2018) Untangling the tauopathy for Alzheimer's disease and parkinsonism. J Biomed Sci 25(1):54. https://doi.org/10.1186/s12929-018-0457-x
- 50. Iqbal K, Liu F, Gong CX, Grundke-Iqbal I (2010) Tau in Alzheimer disease and related tauopathies. Curr Alzheimer Res 7(8):656–664
- Goedert M, Eisenberg DS, Crowther RA (2017) Propagation of tau aggregates and neurodegeneration. Annu Rev Neurosci 40:189–210
- Lee VM, Goedert M, Trojanowski JQ (2001) Neurodegenerative tauopathies. Annu Rev Neurosci 24:1121–1159
- Williams DR (2006) Tauopathies: classification and clinical update on neurodegenerative diseases associated with microtubule-associated protein tau. Intern Med J 36(10):652–660
- 54. Goedert M, Spillantini MG, Potier MC, Ulrich J, Crowther RA (1989) Cloning and sequencing of the cDNA encoding an isoform of microtubule-associated protein tau containing four tandem repeats: differential expression of tau protein mRNAs in human brain. EMBO J 8(2):393–399
- 55. Goedert M, Spillantini MG, Jakes R, Rutherford D, Crowther RA (1989) Multiple isoforms of human microtubule-associated protein tau: sequences and localization in neurofibrillary tangles of Alzheimer's disease. Neuron 3(4):519–526
- Mietelska-Porowska A, Wasik U, Goras M, Filipek A, Niewiadomska G (2014) Tau protein modifications and interactions: their role in function and dysfunction. Int J Mol Sci 15(3):4671–4713
- Jho YS, Zhulina EB, Kim MW, Pincus PA (2010) Monte carlo simulations of tau proteins: effect of phosphorylation. Biophys J 99(8):2387–2397
- Fischer D, Mukrasch MD, Biernat J, Bibow S, Blackledge M, Griesinger C et al (2009) Conformational changes specific for pseudophosphorylation at serine 262 selectively impair binding of tau to microtubules. Biochemistry 48(42):10047–10055
- Brandt R, Leger J, Lee G (1995) Interaction of tau with the neural plasma membrane mediated by tau's amino-terminal projection domain. J Cell Biol 131(5):1327–1340
- Elie A, Prezel E, Guerin C, Denarier E, Ramirez-Rios S, Serre L et al (2015) Tau co-organizes dynamic microtubule and actin networks. Sci Rep 5:9964. https://doi.org/10.1038/srep09964
- 61. Goedert M, Wischik CM, Crowther RA, Walker JE, Klug A (1988) Cloning and sequencing of the cDNA encoding a core protein of the paired helical filament of Alzheimer disease: identification as the microtubule-associated protein tau. Proc Natl Acad Sci U S A 85(11):4051–4055
- 62. Wischik CM, Novak M, Thogersen HC, Edwards PC, Runswick MJ, Jakes R et al (1988) Isolation of a fragment of tau derived from the core of the paired helical filament of Alzheimer disease. Proc Natl Acad Sci U S A 85(12):4506–4510
- 63. Wischik CM, Novak M, Edwards PC, Klug A, Tichelaar W, Crowther RA (1988) Structural characterization of the core of the paired helical filament of Alzheimer disease. Proc Natl Acad Sci U S A 85(13):4884–4888

- 64. Berriman J, Serpell LC, Oberg KA, Fink AL, Goedert M, Crowther RA (2003) Tau filaments from human brain and from in vitro assembly of recombinant protein show cross-beta structure. Proc Natl Acad Sci U S A 100(15):9034–9038
- 65. Arendt T, Bullmann T (2013) Neuronal plasticity in hibernation and the proposed role of the microtubule-associated protein tau as a "master switch" regulating synaptic gain in neuronal networks. Am J Physiol Regul Integr Comp Physiol 305(5):R478–R489
- 66. Su B, Wang X, Drew KL, Perry G, Smith MA, Zhu X (2008) Physiological regulation of tau phosphorylation during hibernation. J Neurochem 105(6):2098–2108
- 67. Brandt R (2001) Cytoskeletal mechanisms of neuronal degeneration. Cell Tissue Res 305(2):255–265
- Dawson HN, Cantillana V, Jansen M, Wang H, Vitek MP, Wilcock DM et al (2010) Loss of tau elicits axonal degeneration in a mouse model of Alzheimer's disease. Neuroscience 169(1):516–531
- Lei P, Ayton S, Finkelstein DI, Spoerri L, Ciccotosto GD, Wright DK et al (2012) Tau deficiency induces parkinsonism with dementia by impairing APP-mediated iron export. Nat Med 18(2):291–295
- 70. Zhang B, Carroll J, Trojanowski JQ, Yao Y, Iba M, Potuzak JS et al (2012) The microtubulestabilizing agent, epothilone D, reduces axonal dysfunction, neurotoxicity, cognitive deficits, and Alzheimer-like pathology in an interventional study with aged tau transgenic mice. J Neurosci 32(11):3601–3611
- Barten DM, Fanara P, Andorfer C, Hoque N, Wong PY, Husted KH et al (2012) Hyperdynamic microtubules, cognitive deficits, and pathology are improved in tau transgenic mice with low doses of the microtubule-stabilizing agent BMS-241027. J Neurosci 32(21):7137–7145
- Graham WV, Bonito-Oliva A, Sakmar TP (2017) Update on Alzheimer's disease therapy and prevention strategies. Annu Rev Med 68:413–430
- 73. Li XC, Hu Y, Wang ZH, Luo Y, Zhang Y, Liu XP et al (2016) Human wild-type full-length tau accumulation disrupts mitochondrial dynamics and the functions via increasing mitofusins. Sci Rep 6:24756. https://doi.org/10.1038/srep24756
- 74. Hu Y, Li XC, Wang ZH, Luo Y, Zhang X, Liu XP et al (2016) Tau accumulation impairs mitophagy via increasing mitochondrial membrane potential and reducing mitochondrial Parkin. Oncotarget 7(14):17356–17368
- 75. Rao MV, McBrayer MK, Campbell J, Kumar A, Hashim A, Sershen H et al (2014) Specific calpain inhibition by calpastatin prevents tauopathy and neurodegeneration and restores normal lifespan in tau P301L mice. J Neurosci 34(28):9222–9234
- 76. Afreen S, Riherd Methner DN, Ferreira A (2017) Tau45-230 association with the cytoskeleton and membrane-bound organelles: functional implications in neurodegeneration. Neuroscience 362:104–117
- 77. Ozcelik S, Sprenger F, Skachokova Z, Fraser G, Abramowski D, Clavaguera F et al (2016) Co-expression of truncated and full-length tau induces severe neurotoxicity. Mol Psychiatry 21(12):1790–1798
- Wittmann CW, Wszolek MF, Shulman JM, Salvaterra PM, Lewis J, Hutton M et al (2001) Tauopathy in Drosophila: neurodegeneration without neurofibrillary tangles. Science 293(5530):711–714
- Santacruz K, Lewis J, Spires T, Paulson J, Kotilinek L, Ingelsson M et al (2005) Tau suppression in a neurodegenerative mouse model improves memory function. Science 309(5733):476–481
- Fulga TA, Elson-Schwab I, Khurana V, Steinhilb ML, Spires TL, Hyman BT et al (2007) Abnormal bundling and accumulation of F-actin mediates tau-induced neuronal degeneration in vivo. Nat Cell Biol 9(2):139–148
- Maass A, Lockhart SN, Harrison TM, Bell RK, Mellinger T, Swinnerton K et al (2018) Entorhinal tau pathology, episodic memory decline, and neurodegeneration in aging. J Neurosci 38(3):530–543

- 82. Dompierre JP, Godin JD, Charrin BC, Cordelieres FP, King SJ, Humbert S et al (2007) Histone deacetylase 6 inhibition compensates for the transport deficit in Huntington's disease by increasing tubulin acetylation. J Neurosci 27(13):3571–3583
- 83. d'Ydewalle C, Krishnan J, Chiheb DM, Van Damme P, Irobi J, Kozikowski AP et al (2011) HDAC6 inhibitors reverse axonal loss in a mouse model of mutant HSPB1-induced Charcot-Marie-Tooth disease. Nat Med 17(8):968–974
- 84. Cartelli D, Ronchi C, Maggioni MG, Rodighiero S, Giavini E, Cappelletti G (2010) Microtubule dysfunction precedes transport impairment and mitochondria damage in MPP+ -induced neurodegeneration. J Neurochem 115(1):247–258
- Kikuchi H, Doh-ura K, Kawashima T, Kira J, Iwaki T (1999) Immunohistochemical analysis of spinal cord lesions in amyotrophic lateral sclerosis using microtubule-associated protein 2 (MAP2) antibodies. Acta Neuropathol 97(1):13–21
- Farah CA, Nguyen MD, Julien JP, Leclerc N (2003) Altered levels and distribution of microtubule-associated proteins before disease onset in a mouse model of amyotrophic lateral sclerosis. J Neurochem 84(1):77–86
- Fanara P, Banerjee J, Hueck RV, Harper MR, Awada M, Turner H et al (2007) Stabilization of hyperdynamic microtubules is neuroprotective in amyotrophic lateral sclerosis. J Biol Chem 282(32):23465–23472
- Hensel N, Claus P (2018) The actin cytoskeleton in SMA and ALS: how does it contribute to motoneuron degeneration? Neuroscientist 24(1):54–72
- Renton AE, Chio A, Traynor BJ (2014) State of play in amyotrophic lateral sclerosis genetics. Nat Neurosci 17(1):17–23
- Lefebvre S, Burglen L, Reboullet S, Clermont O, Burlet P, Viollet L et al (1995) Identification and characterization of a spinal muscular atrophy-determining gene. Cell 80(1):155–165
- Hadano S, Hand CK, Osuga H, Yanagisawa Y, Otomo A, Devon RS et al (2002) A gene encoding a putative GTPase regulator is mutated in familial amyotrophic lateral sclerosis 2. Nat Genet 29(2):166–173
- 92. Yang Y, Hentati A, Deng HX, Dabbagh O, Sasaki T, Hirano M et al (2001) The gene encoding alsin, a protein with three guanine-nucleotide exchange factor domains, is mutated in a form of recessive amyotrophic lateral sclerosis. Nat Genet 29(2):160–165
- 93. Sharma A, Lambrechts A, Hao le T, Le TT, Sewry CA, Ampe C et al (2005) A role for complexes of survival of motor neurons (SMN) protein with gemins and profilin in neurite-like cytoplasmic extensions of cultured nerve cells. Exp Cell Res 309(1):185–197
- 94. van Bergeijk J, Rydel-Konecke K, Grothe C, Claus P (2007) The spinal muscular atrophy gene product regulates neurite outgrowth: importance of the C terminus. FASEB J 21(7):1492–1502
- 95. Oprea GE, Krober S, McWhorter ML, Rossoll W, Muller S, Krawczak M et al (2008) Plastin 3 is a protective modifier of autosomal recessive spinal muscular atrophy. Science 320(5875):524–527
- 96. Nolle A, Zeug A, van Bergeijk J, Tonges L, Gerhard R, Brinkmann H et al (2011) The spinal muscular atrophy disease protein SMN is linked to the Rho-kinase pathway via profilin. Hum Mol Genet 20(24):4865–4878
- 97. Wu CH, Fallini C, Ticozzi N, Keagle PJ, Sapp PC, Piotrowska K et al (2012) Mutations in the profilin 1 gene cause familial amyotrophic lateral sclerosis. Nature 488(7412):499–503
- 98. Giesemann T, Rathke-Hartlieb S, Rothkegel M, Bartsch JW, Buchmeier S, Jockusch BM et al (1999) A role for polyproline motifs in the spinal muscular atrophy protein SMN. Profilins bind to and colocalize with smn in nuclear gems. J Biol Chem 274(53):37908–37914
- 99. Jockusch BM, Murk K, Rothkegel M (2007) The profile of profilins. Rev Physiol Biochem Pharmacol 159:131–149
- 100. Da Silva JS, Medina M, Zuliani C, Di Nardo A, Witke W, Dotti CG (2003) RhoA/ROCK regulation of neuritogenesis via profilin IIa-mediated control of actin stability. J Cell Biol 162(7):1267–1279

- 101. Shao J, Welch WJ, Diprospero NA, Diamond MI (2008) Phosphorylation of profilin by ROCK1 regulates polyglutamine aggregation. Mol Cell Biol 28(17):5196–5208
- 102. Smith BN, Vance C, Scotter EL, Troakes C, Wong CH, Topp S et al (2015) Novel mutations support a role for Profilin 1 in the pathogenesis of ALS. Neurobiol Aging 36(3):1602. e17–1602.e27. https://doi.org/10.1016/j.neurobiolaging.2014.10.032
- Tanaka Y, Hasegawa M (2016) Profilin 1 mutants form aggregates that induce accumulation of prion-like TDP-43. Prion 10(4):283–289
- 104. Figley MD, Bieri G, Kolaitis RM, Taylor JP, Gitler AD (2014) Profilin 1 associates with stress granules and ALS-linked mutations alter stress granule dynamics. J Neurosci 34(24):8083–8097
- 105. Fil D, DeLoach A, Yadav S, Alkam D, MacNicol M, Singh A et al (2017) Mutant profilin1 transgenic mice recapitulate cardinal features of motor neuron disease. Hum Mol Genet 26(4):686–701
- 106. Yang C, Danielson EW, Qiao T, Metterville J, Brown RH Jr, Landers JE et al (2016) Mutant PFN1 causes ALS phenotypes and progressive motor neuron degeneration in mice by a gain of toxicity. Proc Natl Acad Sci U S A 113(41):E6209–E6218
- 107. Baglietto-Vargas D, Prieto GA, Limon A, Forner S, Rodriguez-Ortiz CJ, Ikemura K et al (2018) Impaired AMPA signaling and cytoskeletal alterations induce early synaptic dysfunction in a mouse model of Alzheimer's disease. Aging Cell 6:e12791. https://doi.org/10.1111/ acel.12791
- 108. Fang Y, Wang J, Yao L, Li C, Wang J, Liu Y et al (2018) The adhesion and migration of microglia to beta-amyloid (Abeta) is decreased with aging and inhibited by Nogo/NgR pathway. J Neuroinflammation 15(1):210. https://doi.org/10.1186/s12974-018-1250-1
- 109. Teng T, Dong L, Ridgley DM, Ghura S, Tobin MK, Sun GY et al (2019) Cytosolic phospholipase A2 facilitates oligomeric amyloid-beta peptide association with microglia via regulation of membrane-cytoskeleton connectivity. Mol Neurobiol 56(5):3222–3234. https://doi. org/10.1007/s12035-018-1304-5
- 110. Bellani S, Mescola A, Ronzitti G, Tsushima H, Tilve S, Canale C et al (2014) GRP78 clustering at the cell surface of neurons transduces the action of exogenous alpha-synuclein. Cell Death Differ 21(12):1971–1983
- 111. Delague V, Jacquier A, Hamadouche T, Poitelon Y, Baudot C, Boccaccio I et al (2007) Mutations in FGD4 encoding the Rho GDP/GTP exchange factor FRABIN cause autosomal recessive Charcot-Marie-Tooth type 4H. Am J Hum Genet 81(1):1–16
- 112. Stendel C, Roos A, Deconinck T, Pereira J, Castagner F, Niemann A et al (2007) Peripheral nerve demyelination caused by a mutant Rho GTPase guanine nucleotide exchange factor, frabin/FGD4. Am J Hum Genet 81(1):158–164
- 113. Xu Z, Cork LC, Griffin JW, Cleveland DW (1993) Increased expression of neurofilament subunit NF-L produces morphological alterations that resemble the pathology of human motor neuron disease. Cell 73(1):23–33
- 114. Wong PC, Marszalek J, Crawford TO, Xu Z, Hsieh ST, Griffin JW et al (1995) Increasing neurofilament subunit NF-M expression reduces axonal NF-H, inhibits radial growth, and results in neurofilamentous accumulation in motor neurons. J Cell Biol 130(6):1413–1422
- 115. Cote F, Collard JF, Julien JP (1993) Progressive neuronopathy in transgenic mice expressing the human neurofilament heavy gene: a mouse model of amyotrophic lateral sclerosis. Cell 73(1):35–46
- 116. Ksiezak-Reding H, Dickson DW, Davies P, Yen SH (1987) Recognition of tau epitopes by anti-neurofilament antibodies that bind to Alzheimer neurofibrillary tangles. Proc Natl Acad Sci U S A 84(10):3410–3414
- 117. Selkoe DJ, Ihara Y, Salazar FJ (1982) Alzheimer's disease: insolubility of partially purified paired helical filaments in sodium dodecyl sulfate and urea. Science 215(4537):1243–1245
- 118. Wang J, Tung YC, Wang Y, Li XT, Iqbal K, Grundke-Iqbal I (2001) Hyperphosphorylation and accumulation of neurofilament proteins in Alzheimer disease brain and in okadaic acidtreated SY5Y cells. FEBS Lett 507(1):81–87

- Goldman JE, Yen SH, Chiu FC, Peress NS (1983) Lewy bodies of Parkinson's disease contain neurofilament antigens. Science 221(4615):1082–1084
- 120. Galloway PG, Mulvihill P, Perry G (1992) Filaments of Lewy bodies contain insoluble cytoskeletal elements. Am J Pathol 140(4):809–822
- 121. Forno LS, Sternberger LA, Sternberger NH, Strefling AM, Swanson K, Eng LF (1986) Reaction of Lewy bodies with antibodies to phosphorylated and non-phosphorylated neurofilaments. Neurosci Lett 64(3):253–258
- 122. Hill WD, Arai M, Cohen JA, Trojanowski JQ (1993) Neurofilament mRNA is reduced in Parkinson's disease substantia nigra pars compacta neurons. J Comp Neurol 329(3):328–336
- 123. Mersiyanova IV, Perepelov AV, Polyakov AV, Sitnikov VF, Dadali EL, Oparin RB et al (2000) A new variant of Charcot-Marie-Tooth disease type 2 is probably the result of a mutation in the neurofilament-light gene. Am J Hum Genet 67(1):37–46
- 124. Brownlees J, Ackerley S, Grierson AJ, Jacobsen NJ, Shea K, Anderton BH et al (2002) Charcot-Marie-Tooth disease neurofilament mutations disrupt neurofilament assembly and axonal transport. Hum Mol Genet 11(23):2837–2844
- 125. Perez-Olle R, Leung CL, Liem RK (2002) Effects of Charcot-Marie-Tooth-linked mutations of the neurofilament light subunit on intermediate filament formation. J Cell Sci 115(Pt 24):4937–4946
- 126. Jordanova A, De Jonghe P, Boerkoel CF, Takashima H, De Vriendt E, Ceuterick C et al (2003) Mutations in the neurofilament light chain gene (NEFL) cause early onset severe Charcot-Marie-Tooth disease. Brain 126(Pt 3):590–597
- 127. Perez-Olle R, Lopez-Toledano MA, Goryunov D, Cabrera-Poch N, Stefanis L, Brown K et al (2005) Mutations in the neurofilament light gene linked to Charcot-Marie-Tooth disease cause defects in transport. J Neurochem 93(4):861–874
- 128. Fabrizi GM, Cavallaro T, Angiari C, Cabrini I, Taioli F, Malerba G et al (2007) Charcot-Marie-Tooth disease type 2E, a disorder of the cytoskeleton. Brain 130(Pt 2):394–403
- 129. Yoshihara T, Yamamoto M, Hattori N, Misu K, Mori K, Koike H et al (2002) Identification of novel sequence variants in the neurofilament-light gene in a Japanese population: analysis of Charcot-Marie-Tooth disease patients and normal individuals. J Peripher Nerv Syst 7(4):221–224
- 130. Fabrizi GM, Cavallaro T, Angiari C, Bertolasi L, Cabrini I, Ferrarini M et al (2004) Giant axon and neurofilament accumulation in Charcot-Marie-Tooth disease type 2E. Neurology 62(8):1429–1431
- 131. Georgiou DM, Zidar J, Korosec M, Middleton LT, Kyriakides T, Christodoulou K (2002) A novel NF-L mutation Pro22Ser is associated with CMT2 in a large Slovenian family. Neurogenetics 4(2):93–96
- 132. Sasaki T, Gotow T, Shiozaki M, Sakaue F, Saito T, Julien JP et al (2006) Aggregate formation and phosphorylation of neurofilament-L Pro22 Charcot-Marie-Tooth disease mutants. Hum Mol Genet 15(6):943–952
- Delisle MB, Carpenter S (1984) Neurofibrillary axonal swellings and amyotrophic lateral sclerosis. J Neurol Sci 63(2):241–250
- Hirano A, Donnenfeld H, Sasaki S, Nakano I (1984) Fine structural observations of neurofilamentous changes in amyotrophic lateral sclerosis. J Neuropathol Exp Neurol 43(5):461–470
- 135. Munoz DG, Greene C, Perl DP, Selkoe DJ (1988) Accumulation of phosphorylated neurofilaments in anterior horn motoneurons of amyotrophic lateral sclerosis patients. J Neuropathol Exp Neurol 47(1):9–18
- 136. Tu PH, Raju P, Robinson KA, Gurney ME, Trojanowski JQ, Lee VM (1996) Transgenic mice carrying a human mutant superoxide dismutase transgene develop neuronal cytoskeletal pathology resembling human amyotrophic lateral sclerosis lesions. Proc Natl Acad Sci U S A 93(7):3155–3160
- 137. Eyer J, Cleveland DW, Wong PC, Peterson AC (1998) Pathogenesis of two axonopathies does not require axonal neurofilaments. Nature 391(6667):584–587

- 12 Cytoskeleton in Aging and Neurodegeneration
- 138. Kong J, Xu Z (2000) Overexpression of neurofilament subunit NF-L and NF-H extends survival of a mouse model for amyotrophic lateral sclerosis. Neurosci Lett 281(1):72–74
- 139. Couillard-Despres S, Zhu Q, Wong PC, Price DL, Cleveland DW, Julien JP (1998) Protective effect of neurofilament heavy gene overexpression in motor neuron disease induced by mutant superoxide dismutase. Proc Natl Acad Sci U S A 95(16):9626–9630
- Beaulieu JM, Julien JP (2003) Peripherin-mediated death of motor neurons rescued by overexpression of neurofilament NF-H proteins. J Neurochem 85(1):248–256
- 141. Williamson TL, Bruijn LI, Zhu Q, Anderson KL, Anderson SD, Julien JP et al (1988) Absence of neurofilaments reduces the selective vulnerability of motor neurons and slows disease caused by a familial amyotrophic lateral sclerosis-linked superoxide dismutase 1 mutant. Proc Natl Acad Sci U S A 95(16):9631–9636
- 142. Zhai Q, Wang J, Kim A, Liu Q, Watts R, Hoopfer E et al (2003) Involvement of the ubiquitinproteasome system in the early stages of wallerian degeneration. Neuron 39(2):217–225
- 143. Erturk A, Hellal F, Enes J, Bradke F (2007) Disorganized microtubules underlie the formation of retraction bulbs and the failure of axonal regeneration. J Neurosci 27(34):9169–9180
- 144. Hall GF, Lee VM, Kosik KS (1991) Microtubule destabilization and neurofilament phosphorylation precede dendritic sprouting after close axotomy of lamprey central neurons. Proc Natl Acad Sci U S A 88(11):5016–5020
- Cho Y, Cavalli V (2012) HDAC5 is a novel injury-regulated tubulin deacetylase controlling axon regeneration. EMBO J 31(14):3063–3078
- 146. Goldstein ME, Weiss SR, Lazzarini RA, Shneidman PS, Lees JF, Schlaepfer WW (1988) mRNA levels of all three neurofilament proteins decline following nerve transection. Brain Res 427(3):287–291
- 147. Oblinger MM, Lasek RJ (1988) Axotomy-induced alterations in the synthesis and transport of neurofilaments and microtubules in dorsal root ganglion cells. J Neurosci 8(5):1747–1758
- Mikucki SA, Oblinger MM (1991) Corticospinal neurons exhibit a novel pattern of cytoskeletal gene expression after injury. J Neurosci Res 30(1):213–225
- 149. Tetzlaff W, Alexander SW, Miller FD, Bisby MA (1991) Response of facial and rubrospinal neurons to axotomy: changes in mRNA expression for cytoskeletal proteins and GAP-43. J Neurosci 11(8):2528–2544
- Hoffman PN, Pollock SC, Striph GG (1993) Altered gene expression after optic nerve transection: reduced neurofilament expression as a general response to axonal injury. Exp Neurol 119(1):32–36
- 151. Hoffman PN, Lasek RJ (1980) Axonal transport of the cytoskeleton in regenerating motor neurons: constancy and change. Brain Res 202(2):317–333
- 152. Hoffman PN, Thompson GW, Griffin JW, Price DL (1985) Changes in neurofilament transport coincide temporally with alterations in the caliber of axons in regenerating motor fibers. J Cell Biol 101(4):1332–1340
- 153. McKerracher L, Essagian C, Aguayo AJ (1993) Temporal changes in beta-tubulin and neurofilament mRNA levels after transection of adult rat retinal ganglion cell axons in the optic nerve. J Neurosci 13(6):2617–2626
- 154. Jiang YQ, Pickett J, Oblinger MM (1994) Comparison of changes in beta-tubulin and NF gene expression in rat DRG neurons under regeneration-permissive and regeneration-prohibitive conditions. Brain Res 637(1–2):233–241
- 155. Park JY, Jang SY, Shin YK, Koh H, Suh DJ, Shinji T et al (2013) Mitochondrial swelling and microtubule depolymerization are associated with energy depletion in axon degeneration. Neuroscience 238:258–269
- 156. Tao J, Feng C, Rolls MM (2006) The microtubule-severing protein fidgetin acts after dendrite injury to promote their degeneration. J Cell Sci 129(17):3274–3281

# Chapter 13 Metabolic Biomarkers in Aging and Anti-Aging Research



Paul C. Guest

# 1 Introduction

Human life expectancy has increased by approximately two-fold over the last 200 years, which has resulted in a significant increase in the proportion of elderly individuals in the population [1]. This increase in lifespan has been predicted to continue rising to reach an anticipated life expectancy of more than 85 years by the year 2030 for people in the developed world [2]. However, it comes as no surprise that advanced age is associated with a decline in physiological status, leading to an increase in age-related diseases [3]. Thus, the increase in life span is significantly associated with increased prevalence of diseases such as diabetes, metabolic disorders, cardiovascular disorders, cancer and neurodegenerative disorders [4]. Reducing the negative impacts of advanced age and increasing the human healthspan has therefore been an important goal of aging and anti-aging research throughout the world [5–7]. With this in mind, we are now beginning to understand the physiological mechanisms underlying aging and the next steps will be to identify and validate biomarkers and to target the underlying cellular and molecular determinants.

Aging can be defined as the time-dependent malfunctioning of molecular and physiological mechanisms in organisms, causing defects such as shortening of telomeres and increased production of damaging reactive oxygen species (ROS), which lead to increased senescence of cells and deterioration of the tissues as well as of the entire organism. Cellular senescence is one process by which cells cease to divide and this is thought to contribute to both tissue and organismal aging, and to be a

P. C. Guest (🖂)

Laboratory of Neuroproteomics, Department of Biochemistry and Tissue Biology, Institute of Biology, University of Campinas (UNICAMP), Campinas, Brazil

<sup>©</sup> Springer Nature Switzerland AG 2019

P. C. Guest (ed.), *Reviews on Biomarker Studies in Aging and Anti-Aging Research*, Advances in Experimental Medicine and Biology 1178, https://doi.org/10.1007/978-3-030-25650-0\_13
protective factor against cancer and tumor cell proliferation [8, 9]. Senescent cells have been found to accumulate in various tissues in mice in an age dependent manner [10]. The presence of these cells in tissues can cause dysfunction of the surrounding cells due to release of pro-inflammatory factors. Thus, aging is characterized by systemic degeneration over time and interventions that counteract this degeneration are expected to augment both the healthspan and the lifespan.

Several differing types of aging interventions have been tested in experimental models as a means of extending healthspan and for prevention or slowing of agerelated diseases. For example, caloric restriction has been shown to decrease agerelated diseases in nonhuman primates [11]. In addition, clinical trials of calorie restriction over a 2 year period found evidence of reduced oxidative damage, suggesting that this approach could also reduce the risk of age-related diseases in humans [12]. As we are now unravelling the mechanisms of how aging occurs, a number of compounds are currently undergoing testing for effectiveness in slowing the aging process in preclinical studies. This includes sirtuin activators [13], mammalian target of rapamycin (mTOR) inhibitors [14] and mitochondrial inhibitors [15].

The mitochondria are now known to be important in regulation of the aging process [16]. The primary function of these organelles is to generate energy for the organism in the form of adenosine triphosphate (ATP) but they are also involved in other physiological processes which have links to aging, such as apoptosis, autophagy and production of reactive oxygen species (ROS). Reduced mitochondrial function and generation of declining amounts of ATP have been observed in various organs and tissues, including skeletal muscle [17], heart [18] and brain [19]. This age-related mitochondrial impairment can be seen at multiple levels including mitochondrial number and morphology, electron transport chain (ETC) activity and ROS formation [20].

This chapter reviews how the complex process of aging may be regulated at the physiological, cellular and molecular levels and how this information is being unravelled by research using model organisms. At present, the most promising results have come from studies of the molecular pathways involved with caloric restriction, insulin/insulin-like growth factor signalling and mitochondrial ROS production, in nematode, fly and rodent models.

# 2 Biomarkers of Aging in Caenorhabditis elegans

C. elegans is a popular model organism for aging and anti-aging studies due to its short lifespan, fully annotated genome and age-dependent physiological changes [21]. Early research found that one gene associated with aging was *DAF-2*, which encodes a homologue of the mammalian insulin/insulin-like growth factor (IGF) family [22, 23]. Decreased DAF-2 signaling leads to translocation of the fork-head transcription factor DAF-16 into the nucleus, and this leads to activation of numerous genes associated with stress response, lipid metabolism, immunity and longevity



Insulin/IGF-like signalling

**Fig. 13.1** Effects on aging via through the insulin/IGF-lik) in C. elegans. Under conditions of high levels of DAF-2 signalling, the transcription factor DAF-16 is phosphorylated and cannot enter the nucleus to activate target genes. Under conditions of low DAF-2 signalling, un-phosphorylated DAF-16 can enter the nucleus and turn on the target genes

(Fig. 13.1). The end result is a worm that can live twice as long as its natural counterparts [22]. There are also several models involving caloric restriction which lead to increased lifespan in C. elegans. EAT-2 mutant animals have dysfunctional pharynges, which results in decreased food intake along with a lifespan increase that is approximately 50% greater than wild type animals [24]. The effects of caloric restriction or impaired insulin/IGF-1-like signalling partially overlap in their downstream signalling processes, which include activation of pathways such as mitochondrial autophagy and inhibition of the mammalian target of rapamycin (mTOR) [23].

Other studies have linked caloric restriction to an improved oxidative stress response, which is mediated by the oxidoreductase enzyme thioredoxin 1 [25]. Another research group identified four genes that extend lifespan specifically in DAF-16 mutants but not in the EAT-2 mutants. These were the genes encoding S-adenosyl methionine synthetase (*SAMS*), Rab-like GTPase (*RAB-10*), dietary restriction response of unknown function (*DRR-1*) and a putative RNA-binding protein (*DRR-2*) [26].

As indicated above, the insulin/insulin-like growth factor-1 (ins/IGF-1) signalling pathway is involved in regulation of longevity and resistance to oxidative stress in C. elegans [27–29]. This is achieved via regulation of the downstream DAF-16 transcription factor [30, 31], which targets genes associated with these pathways during aging of long-lived C. elegans mutants (Fig. 13.1) [32, 33]. Under high insulin/IGF-like signalling conditions, such as following a meal high in fats and sugars, DAF-16 is phosphorylated and cannot enter the nucleus to activate target genes. Under conditions of low insulin/IGF-like signalling, un-phosphorylated DAF-16 can enter the nucleus and turn on the target genes. Intersetingly, a number of these studies have found increased ATP concentrations with reduced insulin/IGF-1 signalling and lower respiratory rates [34–38]. In addition, intracellular ROS are removed more efficiently under these conditions due to the higher activities of antioxidant enzymes such as superoxide dismutase (SOD) and catalase [34, 39].

The C. elegans life cycle consists of larval, dauer larval and adult stages. In the case of the dauer state, no feeding occurs [40, 41]. During the normal larval stages (L2-L4), cell growth and proliferation are driven by the tricarboxylic acid (TCA) cycle and young adult worms have a high tolerance to anoxia and protection against ROS [42, 43]. Significant decreases in oxygen consumption and metabolic rate have been seen in normal worms after these worms reach adulthood, consistent with decay in muscle function [44–46]. A recent study showed that expression of the cyclooxygenase (COX) assembly protein [mammalian gene homolog sco-1 (SCO2)] gene was increased with aging in wild type worms [47], suggesting that mitochondrial components are damaged by ROS during the aging process, inducing a shift from the TCA cycle to aerobic glycolysis. Conversely, the finding of an age-related increase in the levels of phosphoenolpyruvate carboxykinase (PEPCK) induced by calorie restriction in wild type C. elegans, indicates a shift in the balance to gluconeogenesis [47]. PEPCK plays an important role in energy production throughout various life stages of the worm and other invertebrates. The C. elegans PEPCK enzyme is involved in regulation of metabolism associated with cataplerosis, which is the removal of intermediate metabolites from gluconeogenesis and other pathways in anaerobic environments [48, 49]. Yanase et al. found that the upregulation of gluconeogenesis during aging in C. elegans was associated with reduced mitochondrial respiration and increased expression of PEPCK and the NAD-dependent protein deacetylase sir-2.1 [47, 50].

In C. elegans, exposure to the hyperoxia accelerates senescence so that the levels of intracellular ROS increase [51-53]. Mutations of the C. elegans *MEV-1* gene, encoding the large subunit of the cytochrome b succinate dehydrogenase, result in increased production of the mitochondrial superoxide anion (O2-), resulting in a shortened lifespan [54]. Likewise, the levels of cellular metabolites such as lactate and pyruvate are correlated with the switch from mitochondrial respiration to glycolysis during aging [47, 55]. These findings indicate that the cytochrome b succinate dehydrogenase plays an important role in energy metabolism as well as in superoxide anion production that is involved in sensitivity to atmospheric oxygen. Thus, further studies of the physiological and molecular changes in the *MEV-1* mutants might help to elucidate the pathological mechanisms of aging.

# 3 The Role of the Mitochondria in Aging

Mitochondria are the main site of energy production in most eukaryotic cells. This is where the processes of glycolysis and beta-oxidation of lipids occur in the process of generating ATP. This occurs via reduction-oxidation (redox) reactions along the oxidative phosphorylation (OXPHOS) complexes within the inner mitochondrial membrane (Fig. 13.2a) [56]. This is also one of the principal sites of aging as the progressive accumulation of cell damage has been proposed to be due to overproduction of reactive oxygen species (ROS) (Fig. 13.2b) [57, 58]. Catalase is a peroxisomal enzyme which works together with mitochondrial glutathione peroxidise in antioxidant reactions preventing against ROS formation [59]. Studies in mice have shown that targeting of catalase to the mitochondria can result in reduced ROS damage and increased lifespan [60], along with an improvement in exercise performance [61]. With these results in mind, mitochondria-targeted catalase gene therapy has been proposed as a potential co-treatment approach in cases of Duchenne muscle dystrophy, in which the muscle tissues have high levels of ROS production [62]. This approach may also prove useful to slow the effects of sarcopenia, characterized by loss of muscle mass, which can occur during aging. Superoxide SOD2 is



**Fig. 13.2** (a) Generation of ATP via reduction–oxidation (redox) reactions along the oxidative phosphorylation (OXPHOS) complexes within the inner mitochondrial membrane. (b) Reactive oxygen species (ROS) production via the electron transport chain. Anions are produced at complexes I and III producing superoxide ( $O_2$ –). Hydrogen peroxide ( $H_2O_2$ ) is converted into the hydroxyl radical (OH–). Superoxide and the hydroxyl radical are ROS that cause oxidative stress to macromolecules and organelles. Arrows with dashed lines indicate the direction of electron transfer. CoQ coenzyme Q, Cyt c cytochrome c, NAD+ oxidized nicotinamide adenine dinucleotide, NADH reduced nicotinamide adenine dinucleotide, FAD oxidized flavin adenine dinucleotide, FADH<sub>2</sub> reduced flavin adenine dinucleotide

a mitochondrial enzyme that converts superoxide species to  $H_2O_2$  and  $O_2$  [63]. In a heterozygous SOD2 (SOD2<sup>+/-</sup>) knockout model, aged mice had significantly increased oxidative stress in their smooth muscle cells, resulting in a pathological stiffness, similar to that which occurs in atherosclerotic plaque formation during aging [64]. In a human study, suboptimal brain aging was found in subjects with a specific SOD2 variant [65]. These findings indicate an essential role of SOD2 in prevention against oxidative damage.

Coenzyme Q is a component of the mitochondrial electron transport that functions as an electron transporter between oxidative phosphorylation complexes, leading to ATP synthesis and it also serves as an antioxidant factor [66]. Given this, coenzyme Q supplementation has been tested with some success as a potential treatment of a number of disorders, such as cardiovascular disease, metabolic syndrome, neurodegenerative diseases and inflammation [67]. It has also been shown to prevent oxidative stress in a senescence-accelerated mouse model and in aged mice and rats [68, 69]. Along the same lines, the mitochondria-targeted form of coenzyme Q (MitoQ) was found to reduce cognitive decline, oxidative stress and loss of synapses in a mouse model of Alzheimer's disease [70] and to extend the lifespan of a C. elegans Alzheimer's disease model [71].

Caloric restriction appears to work by lowering mitochondrial O<sub>2</sub> consumption, leading to reduced generation of damaging ROS. This has been linked to changes in the mammalian target of rapamycin (mTOR) and sirtuin pathways [72]. Inhibition of mTOR has been found to extend lifespan in multiple species [73-76]. Sirt1 (the mammalian orthologue of Sir2) has been associated with neuroprotection [77], reduction of fat storage [78, 79] and insulin secretion from pancreatic- $\beta$  cells [80]. Sirt1 increases expression of genes involved in fatty acid oxidation in response to low glucose, thereby providing a switch from glucose to a fatty acid oxidation metabolism under low caloric conditions [81, 82]. Some dietary activators of Sirt1 have been identified such as resveratrol and melatonin [83]. Another sirtuin family member (SIRT3) is thought to be involved in increasing the mitochondrial glutathione antioxidant defense system under caloric restriction conditions [84, 85]. One study of a mouse model lacking the p66 adapter protein found that the resulting increase in lifespan might be linked to improved metabolic homeostasis via regulation of Sirt3 activity [86]. Conversely, a Sirt3 knockout mouse (Sirt3<sup>-/-</sup>) model showed increased oxidative stress and mitochondrial protein dysfunction [87].

Caloric restriction has also been shown to reduce the incidence of metabolic disease, cancer and brain atrophy, as well as all-cause mortality in non-human primate species [88, 89]. For example, a two- year caloric restriction study of healthy individuals found that this diet led to enhanced resting energy efficiency and lower systemic oxidative damage, compared to the effects seen for a control group on a normal caloric diet [12]. Another study of healthy elderly subjects found that a caloric restriction diet improved memory performance [90].

In summary, caloric restriction in a number of species has been shown to counteract age-related decline and increase lifespan by inducing a shift from carbohydrate to fatty acid metabolism, enhancing mitochondrial energy production and activating antioxidant defence mechanisms.

# 4 Redox Stress and Aging

The reduced form of nicotinamide adenine dinucleotide phosphate (NADPH) is an essential component in the synthesis of fatty acids, cholesterol and deoxynucleotides as well as being a protective factor against redox stress. The levels of NADPH decrease with age [91, 92] due to oxidative stress resulting from the effects of accumulated mitochondrial electron transport chain dysfunction and inflammation [93, 94]. The mitochondrial theory of aging suggests that aging is associated with the accumulation of damage from increased mitochondrial ROS [58]. The redox theories of aging have proposed the idea that aging results from changes in the redox balance of molecules such as the oxidized and reduced forms of NADP (NADP+/NADPH) and glutathione (GSSG/GSH), as well as by changes in cell signalling [95, 96].

A number of studies have found that the rate of mitochondrial superoxide generation and phospholipid fatty acid saturation levels is negatively-correlated with lifespan in various species [97, 98]. The mitochondrial inner membrane is enriched in certain phospholipids essential for electron transport chain and ADP/ATP transport functions and these phospholipids are vulnerable to damage by ROS [99]. The loss of NAD+ and NADPH appears to be a factor in the aging of Drosophila melanogaster and C. Elegans [100, 101] and addition of NAD+ [102] or nicotinamide riboside [103] to the culture media has been found to extend lifespan. The decreased levels of NAD+ in aged mouse muscle leads to stabilization of the hypoxia inducible factor-1 $\alpha$  (HIF-1 $\alpha$ ) and decreased c-Myc-induced expression of mitochondrial genes involved in electron transport chain function [104]. This leads to increased mitochondrial NADH and decreased NAD+ levels, as well as a reduction of the proton-motive force across the inner mitochondrial membrane. In turn, this would lead to decreased levels of glutathione reductase and, therefore, increased GSSG/ GSH ratios. The increased oxidation within these pathways also results in oxidation of other redox-related molecules [105], thereby leading to oxidation of lipids, proteins and nucleic acids, culminating in the tissue dysfunction associated with the aging process.

Redox stress also appears to be associated with most age-related disorders such as diabetes and cardiovascular conditions. The NADP+/NADPH ratio is the strongest known redox determinant in age-induced oxidation with redox potentials ranging from -400 to -20 mV [106] and this ratio shifts to a more oxidized state with aging [107]. Quantification of oxidative stress in model systems can be performed by measurement of the GSSG/GSH ratio and the oxidation state can be determined in intact cells using the genetically encoded fluorescent probes, such as roGFP or roGFP2 [108]. However, these studies have demonstrated that both oxidizing and reducing changes can affect aging and longevity [109]. These findings suggest that redox measurements in the cytoplasm and mitochondria are important. In C. elegans, Reduced function of the NADPH-generating enzymes in C. elegans has been found to both increase and decrease longevity, most likely due to activation or inhibition of different compensation pathways. Reduction of the cytoplasmic NADP+/NADPH ratio has mostly resulted in increased longevity but this can also result in

reductive stress leading to mitochondrial oxidation and increased ROS generation. Thus, considerable further work is required in this area to fully elucidate the role of the redox potential in aging and longevity.

# 5 I'm Not Dead Yet (INDY)

INDY is a non-electrogenic solute transporter that transports di- and tri-carboxylates across the plasma membrane, as described in studies of D. melanogaster [110]. Reduced expression of INDY has been found to enhance longevity in a way that is similar to the effects of calorie restriction [111, 112]. Knockout of the mammalian homologue of INDY (the sodium-coupled citrate transporter; NaCT) led to protection from obesity and insulin resistance and this effect was found to be mediated by altered mitochondrial metabolism and reduced hepatic lipid generation [113]. Citrate is a vital metabolite that links multiple metabolic pathways such as glycolysis, gluconeogenesis and lipid synthesis [114–118]. Citrate is also an intermediate involved in the TCA pathway, which leads to generation of energy in the form of ATP. Transcription of INDY is regulated by the nutritional status. Calorie restriction was found to reduce expression of INDY in D. melanogaster [119, 120] and of the INDY homologue in mice [113]. On the other hand, administration of large amounts of olive oil increased INDY expression in rats [121]. Other studies have shown that regulation of INDY may occur via epigenetic mechanisms [122–124].

Mutations in the INDY gene are associated with lower levels of body fat, reduced circulating insulin-related proteins and decreased ROS and lifespan extension [111, 119, 125]. Genetic deletion or pharmacological inhibition of INDY in model organisms has been found to reduce the effects of a number of metabolic conditions such as non-alcoholic fatty liver disease (NAFLD), obesity and insulin resistance [113, 126–128]. For this reason, INDY is considered a potential drug target for metabolic diseases [129]. For example, it has been shown that knockdown of INDY expression using a liver-selective siRNA approach resulted in improved insulin sensitivity and reduced triglyceride accumulation [130]. Future studies are required to determine the effectiveness of INDY-directed compounds in the treatment of metabolic diseases and other diseases affected by metabolic disturbances including diabetes, obesity and cardiovascular disorders, as well as neurodegenerative and psychiatric disorders [131]. Ultimately, such compounds should be investigated to determine whether or not they promote healthier aging and increased longevity.

# 6 Obesity and Aging

Obesity has now become a global epidemic with a prevalence that has tripled over the last 30 years [132]. Obesity can increase the risk of numerous disorders such as diabetes, cardiovascular diseases and cancer, thereby increasing the mortality rate

[133–135]. To further our understanding of the effects of obesity, and to identify novel therapeutic approaches, a number of epidemiological approaches have been undertaken including population-based studies, case-control and clinical trials, which have to identification of risk factors, metabolic impacts and potential new treatments [136, 137]. However, such human-based studies are limited by factors such as underreporting and difficulty of discerning the impact of specific components of diets [138, 139]. As an alternative, animal-based studies can be more carefully controlled and these also allow the analysis of different tissues for determination of metabolic and molecular effects [140]. In addition, the pathways that regulate energy balance linked with weight control are highly conserved across the animal kingdom.

There are different types of high fat diets, such as those including 30–85% of the calories coming from fats [141], and these produce physiological effects such as obesity and insulin resistance [139, 142]. Weight gain is commonly used as a simple biomarker for monitoring the outcome of the diet or for determining the effects of an intervention, although body fat composition can give more precise information [139]. For example, one study found that administration of a 40% fat diet for 10 weeks caused rats to gain 10% in body weight but 35–40% in body fat [143]. In addition, the types of dietary fat can be important as several studies have now shown that lard-based diets are more obesogenic compared to oil-based diets [142, 144, 145].

Other studies have used a cafeteria-based diet approach as this closely mimics the "Western diet" [146]. Cafeteria-based foods include biscuits, cheese, processed meats, cakes, chocolate and peanut butter, as examples [147], which tend to induce hyperphagia [148, 149]. Therefore, compared with the high fat diet, the cafeteria diet can induce higher weight and abdominal fat gains, thereby inducing more damage to tissues such as the heart and liver, as well as leading to inflammation, hyperinsulinemia, hyperglycemia and glucose intolerance [150, 151]. Furthermore, a diet that combines both sugar and fats may be more efficient in eliciting metabolic changes and obesity in comparison to high fat diets [152]. The obesogenic effect of the high sugar/high fat diet may be due to the increased levels of saturated fatty acids that are less available as an energy source but instead are acetylated into triacylglycerol and stored in adipose tissue at increased levels [145, 153]. This is compounded by the insulinogenic effect of the rapidly absorbed simple sugars, resulting in a rapid decrease in blood sugar levels [154]. This triggers a neurochemical craving response similar to that seen in cases of addiction [155]. A model of the high sugar high fat diet which incorporates fructose and sweetened condensed milk as a source of sugar and beef tallow as a source of fat resulted in a greater body weight and abdominal fat gain, compared to controls along with induction of metabolic syndrome, and changes in the function of organs, such as heart, liver, and kidneys [156].

Taken together, these findings indicate that increasing our understanding of how obesity can alter cellular physiology and metabolic function could open new therapeutic avenues to extend the period of healthy aging.

# 7 Conclusions and Future Perspectives

Results from epidemiological studies have shown that most of the healthcare costs in developing countries are accounted for by age-related disorders and these costs are expected to increase along with the increasing proportion of the elderly population in developed countries. This is mainly due to the fact that the increase in average life expectancy has not been paralleled by a corresponding increase in healthspan [157]. Thus, considerable research has been underway to understand the process of healthy aging at the physiological and molecular levels. Results from preclinical models and data from human studies suggest that insulin/IGF signalling and efficiency of mitochondrial energy production are key regulators of this process. In addition, studies of these pathways have provided both rationales and potential drug targets for therapeutic interventions. A number of investigations along these lines have already been completed in animal models with the aim of finding a way of slowing the aging process and extending the human healthspan. Interventions such as caloric restriction and exercise, and administration of nutritional compounds and drugs such as antioxidants, omega-3 fatty acids, metformin and aspirin, target the mitochondria to delay or counteract the effects of aging [158]. Many of these approaches have shown early promise and have led to the identification of key biomarkers that can be used for monitoring the effects of aging as well as the efficacy of emerging anti-aging interventions. It is likely that such aging interventions will also delay the development of chronic diseases and thereby extend both the healthspan and the lifespan.

# References

- Christensen K, Doblhammer G, Rau R, Vaupel JW (2009) Ageing populations: the challenges ahead. Lancet 374(9696):1196–1208
- Kontis V, Bennett JE, Mathers CD, Li G, Foreman K, Ezzati M (2017) Future life expectancy in 35 industrialised countries: projections with a Bayesian model ensemble. Lancet 389(10076):1323–1335
- Boss GR, Seegmiller JE (1981) Age-related physiological changes and their clinical significance. West J Med 135(6):434–440
- Rizvi S, Raza ST, Mahdi F (2014) Telomere length variations in aging and age-related diseases. Curr Aging Sci 7(3):161–167
- 5. Melov S (2016) Geroscience approaches to increase healthspan and slow aging. F1000Res 5. pii: F1000 Faculty Rev-785. https://doi.org/10.12688/f1000research.7583.1
- Hansen M, Kennedy BK (2016) Does longer lifespan mean longer healthspan? Trends Cell Biol 26(8):565–568
- 7. Crimmins EM (2015) Lifespan and healthspan: past, present, and promise. Gerontologist 55(6):901–911
- Hayflick L, Moorhead PS (1961) The serial cultivation of human diploid cell strains. Exp Cell Res 25:585–621
- 9. Campisi J (2013) Aging, cellular senescence, and cancer. Annu Rev Physiol 75:685-705

- Wang C, Jurk D, Maddick M, Nelson G, Martin-Ruiz C, von Zglinicki T (2009) DNA damage response and cellular senescence in tissues of aging mice. Aging Cell 8(3):311–323
- Leong I (2018) Sustained caloric restriction in health. Nat Rev Endocrinol 14:322. https://doi. org/10.1038/s41574-018-0008-2
- Redman LM, Smith SR, Burton JH, Martin CK, Il'yasova D, Ravussin E (2018) Metabolic slowing and reduced oxidative damage with sustained caloric restriction support the rate of living and oxidative damage theories of aging. Cell Metab 27(4):805–815.e4
- Mitchell SJ, Martin-Montalvo A, Mercken EM, Palacios HH, Ward TM, Abulwerdi G et al (2014) The SIRT1 activator SRT1720 extends lifespan and improves health of mice fed a standard diet. Cell Rep 6(5):836–843
- Harrison DE, Strong R, Sharp ZD, Nelson JF, Astle CM, Flurkey K et al (2009) Rapamycin fed late in life extends lifespan in genetically heterogeneous mice. Nature 460(7253):392–395
- Zinovkin RA, Zamyatnin AA (2018) Mitochondria-targeted drugs. Curr Mol Pharmacol. https://doi.org/10.2174/1874467212666181127151059. [Epub ahead of print]
- Payne BA, Chinnery PF (2015) Mitochondrial dysfunction in aging: much progress but many unresolved questions. Biochim Biophys Acta 1847(11):1347–1353
- Faitg J, Reynaud O, Leduc-Gaudet JP, Gouspillou G (2017) Skeletal muscle aging and mitochondrial dysfunction: an update. Med Sci (Paris) 33(11):955–962
- Hoppel CL, Lesnefsky EJ, Chen Q, Tandler B (2017) Mitochondrial dysfunction in cardiovascular aging. Adv Exp Med Biol 982:451–464
- Morita M, Ikeshima-Kataoka H, Kreft M, Vardjan N, Zorec R, Noda M (2019) Metabolic plasticity of astrocytes and aging of the brain. Int J Mol Sci 20(4). pii: E941. https://doi. org/10.3390/ijms20040941
- Kim SH, Kim H (2018). Inhibitory effect of astaxanthin on oxidative stress-induced mitochondrial dysfunction-a mini-review. Nutrients 10(9). pii: E1137. https://doi.org/10.3390/ nu10091137
- Son HG, Altintas O, Kim EJE, Kwon S, Lee SV (2019) Age-dependent changes and biomarkers of aging in Caenorhabditis elegans. Aging Cell 18(2):e12853. https://doi.org/10.1111/acel.12853
- 22. Kenyon C, Chang J, Gensch E, Rudner A, Tabtiang R (1993) A C. elegans mutant that lives twice as long as wild type. Nature 366(6454):461–464
- 23. Kenyon CJ (2010) The genetics of ageing. Nature 464(7288):504-512
- Lakowski B, Hekimi S (1998) The genetics of caloric restriction in Caenorhabditis elegans. Proc Natl Acad Sci U S A 95(22):13091–13096
- 25. Fierro-González JC, González-Barrios M, Miranda-Vizuete A, Swoboda P (2011) The thioredoxin TRX-1 regulates adult lifespan extension induced by dietary restriction in Caenorhabditis elegans. Biochem Biophys Res Commun 406(3):478–482
- Hansen M, Hsu AL, Dillin A, Kenyon C (2005) New genes tied to endocrine, metabolic, and dietary regulation of lifespan from a Caenorhabditis elegans genomic RNAi screen. PLoS Genet 1(1):119–128
- 27. Honda Y, Honda S (2002) Life span extensions associated with upregulation of gene expression of antioxidant enzymes in Caenorhabditis elegans; studies of mutation in the AGE-1, PI3 kinase homologue and short-term exposure to hyperoxia. J Am Aging Assoc 24(1):21–25
- Yanase S, Yasuda K, Ishii N (2002) Adaptive responses to oxidative damage in three mutants of Caenorhabditis elegans (age-1, mev-1 and daf-16) that affect life span. Mech Ageing Dev 123(12):1579–1587
- 29. Yanase S, Ishii N (2008) Hyperoxia exposure induced hormesis decreases mitochondrial superoxide radical levels via Ins/IGF-1 signaling pathway in a long-lived age-1 mutant of Caenorhabditis elegans. J Radiat Res 49(3):211–218
- Ogg S, Paradis S, Gottlieb S, Patterson GI, Lee L, Tissenbaum HA et al (1997) The fork head transcription factor DAF-16 transduces insulin-like metabolic and longevity signals in C. elegans. Nature 389(6654):994–999

- 31. Lin K, Dorman JB, Rodan A, Kenyon C (1997) daf-16: an HNF-3/forkhead family member that can function to double the life-span of Caenorhabditis elegans. Science 278(5341):1319–1322
- Kenyon C (2006) Enrichment of regulatory motifs upstream of predicted DAF-16 targets. Nat Genet 38(4):397–398
- 33. Minniti AN, Cataldo R, Trigo C, Vasquez L, Mujica P, Leighton F et al (2009) Methionine sulfoxide reductase A expression is regulated by the DAF-16/FOXO pathway in Caenorhabditis elegans. Aging Cell 8(6):690–705
- 34. Vanfleteren JR (1993) Oxidative stress and ageing in Caenorhabditis elegans. Biochem J 292(Pt 2):605–608
- Murakami S, Johnson TE (1996) A genetic pathway conferring life extension and resistance to UV stress in Caenorhabditis elegans. Genetics 143(3):1207–1218
- 36. Van Voorhies WA, Ward S (1999) Genetic and environmental conditions that increase longevity in Caenorhabditis elegans decrease metabolic rate. Proc Natl Acad Sci U S A 96(20):11399–11403
- Barsyte D, Lovejoy DA, Lithgow GJ (2001) Longevity and heavy metal resistance in daf-2 and age-1 long-lived mutants of Caenorhabditis elegans. FASEB J 15(3):627–634
- Shoyama T, Shimizu Y, Suda H (2009) Decline in oxygen consumption correlates with lifespan in long-lived and short-lived mutants of Caenorhabditis elegans. Exp Gerontol 44(12):784–791
- Larsen PL (1993) Aging and resistance to oxidative damage in Caenorhabditis elegans. Proc Natl Acad Sci U S A 90(19):8905–8909
- 40. Wood WB (1988) Introduction to C. elegans biology. In: Wood WB, The Community of C. elegans Researchers (eds) The nematode Caenorhabditis elegans. Cold Spring Harbor Laboratory Press, New York, pp 1–16. ISBN: 0-87969-433-5
- Riddle DL (1988) The dauer larva. In: Wood WB, The Community of C. elegans Researchers (eds) The nematode Caenorhabditis elegans. Cold Spring Harbor Laboratory Press, New York, pp 393–412. ISBN: 0-87969-433-5
- 42. Wadsworth WG, Riddle DL (1989) Developmental regulation of energy metabolism in Caenorhabditis elegans. Dev Biol 132(1):167–173
- Van Voorhies WA, Ward S (2000) Broad oxygen tolerance in the nematode Caenorhabditis elegans. J Biol Chem 203(Pt 16):2467–2478
- 44. Vanfleteren JR, De Vreese A (1996) Rate of aerobic metabolism and superoxide production rate potential in the nematode Caenorhabditis elegans. J Exp Zool 274(2):93–100
- 45. Klass MR, Johnson TE (1985) Caenorhabditis elegans. In: Lints FA (ed) Non-mammalian models for research on aging. Karger, Basel, pp 164–187. ISBN: 3805540191
- 46. Chow DK, Glenn CF, Johnston JL, Goldberg IG, Wolkow CA (2006) Sarcopenia in the Caenorhabditis elegans pharynx correlates with muscle contraction rate over lifespan. Exp Gerontol 41(3):252–260
- 47. Yanase S, Suda H, Yasuda K, Ishii N (2017) Impaired p53/CEP-1 is associated with lifespan extension through an age-related imbalance in the energy metabolism of C. elegans. Genes Cells 22(12):1004–1010
- Owen OE, Kalhan SC, Hanson RW (2002) The key role of anaplerosis and cataplerosis for citric acid cycle function. J Biol Chem 277(34):30409–30412
- Yang J, Kalhan SC, Hanson RW (2009) What is the metabolic role of phosphoenolpyruvate carboxykinase? J Biol Chem 284(40):27025–27029
- Rodgers JT, Lerin C, Naas W, Gygi SP, Spiegelman BM, Puigserver P (2005) Nutrient control of glucose homeostasis through a complex of PGC-1α and SIRT1. Nature 434(7029):113–118
- 51. Honda S, Matsuo M (1992) Lifespan shortening of the nematode Caenorhabditis elegans under higher concentrations of oxygen. Mech Ageing Dev 63(3):135–246
- Darr D, Fridovich I (1995) Adaptation to oxidative stress in young, but not in mature or old, Caenorhabditis elegans. Free Radic Biol Med 18(2):195–201

- Freeman BA, Crapo JD (1981) Hyperoxia increases oxygen radical production in rat lungs and lung mitochondria. J Biol Chem 256(21):10986–10992
- 54. Ishii N, Fujii M, Hartman PS, Tsuda M, Yasuda K, Senoo-Matsuda N et al (1998) A mutation in succinate dehydrogenase cytochrome b causes oxidative stress and ageing in nematodes. Nature 394(6694):694–697
- 55. Senoo-Matsuda N, Yasuda K, Tsuda M, Ohkubo T, Yoshimura S, Nakazawa H et al (2001) A defect in the cytochrome b large subunit in complex II causes both superoxide anion overproduction and abnormal energy metabolism in Caenorhabditis elegans. J Biol Chem 276(45):41553–41558
- 56. DiMauro S, Schon EA (2003) Mitochondrial respiratory-chain diseases. N Engl J Med 348(26):2656–2568
- 57. Harman D (1956) Aging: a theory based on free radical and radiation chemistry. J Gerontol 11(3):298–300
- 58. Harman D (1972) The biologic clock: the mitochondria? J Am Geriatr Soc 20(4):145-147
- 59. Dai D-F, Chiao Y-A, Martin GM, Marcinek DJ, Basisty N, Quarles EK et al (2017) Mitochondrial-targeted catalase: extended longevity and the roles in various disease models. Prog Mol Biol Transl Sci 146:203–241
- 60. Schriner SE, Linford NJ, Martin GM, Treuting P, Ogburn CE, Emond M et al (2005) Extension of murine life span by overexpression of catalase targeted to mitochondria. Science 308(5730):1909–1911
- 61. Li D, Lai Y, Yue Y, Rabinovitch PS, Hakim C, Duan D (2009) Ectopic catalase expression in mitochondria by adeno-associated virus enhances exercise performance in mice. In: Lucia A (ed). PLoS One 4:e6673. https://doi.org/10.1371/journal.pone.0006673
- Selsby JT (2011) Increased catalase expression improves muscle function in mdx mice. Exp Physiol 96(2):194–202
- Azadmanesh J, Borgstahl GEO (2018) A review of the catalytic mechanism of human manganese superoxide dismutase. Antioxidants (Basel, Switzerland) 7:25. https://doi.org/10.3390/ antiox7020025
- 64. Zhou R-H, Vendrov AE, Tchivilev I, Niu X-L, Molnar KC, Rojas M et al (2012) Mitochondrial oxidative stress in aortic stiffening with age: the role of smooth muscle cell function. Arterioscler Thromb Vasc Biol 32(3):745–755
- 65. Salminen LE, Schofield PR, Pierce KD, Bruce SE, Griffin MG, Tate DF et al (2017) Vulnerability of white matter tracts and cognition to the SOD2 polymorphism: a preliminary study of antioxidant defense genes in brain aging. Behav Brain Res 329:111–119
- Ernster L, Dallner G (1995) Biochemical, physiological and medical aspects of ubiquinone function. Biochim Biophys Acta 1271(1):195–204
- Hernández-Camacho JD, Bernier M, López-Lluch G, Navas P (2018) Coenzyme Q10 supplementation in aging and disease. Front Physiol 9:44. https://doi.org/10.3389/fphys.2018.00044
- 68. Shetty RA, Forster MJ, Sumien N (2013) Coenzyme Q(10) supplementation reverses age-related impairments in spatial learning and lowers protein oxidation. Age (Dordr) 35(5):1821–1834
- 69. Ulla A, Mohamed MK, Sikder B, Rahman AT, Sumi FA, Hossain M et al (2017) Coenzyme Q10 prevents oxidative stress and fibrosis in isoprenaline induced cardiac remodeling in aged rats. BMC Pharmacol Toxicol 18:29. https://doi.org/10.1186/s40360-017-0136-7
- McManus MJ, Murphy MP, Franklin JL (2011) The mitochondria-targeted antioxidant MitoQ prevents loss of spatial memory retention and early neuropathology in a transgenic mouse model of Alzheimer's disease. J Neurosci 31(44):15703–15715
- 71. Ng LF, Gruber J, Cheah IK, Goo CK, Cheong WF, Shui G et al (2014) The mitochondriatargeted antioxidant MitoQ extends lifespan and improves healthspan of a transgenic Caenorhabditis elegans model of Alzheimer disease. Free Radic Biol Med 71:390–401
- 72. Betz C, Hall MN (2013) Where is mTOR and what is it doing there? J Cell Biol 203(4):563-574
- Fontana L, Partridge L, Longo VD (2010) Extending healthy life span--from yeast to humans. Science 328(5976):321–326

- 74. Kapahi P, Chen D, Rogers AN, Katewa SD, Li PW-L, Thomas EL et al (2010) With TOR, less is more: a key role for the conserved nutrient-sensing TOR pathway in aging. Cell Metab 11(6):453–465
- Lamming DW, Ye L, Sabatini DM, Baur JA (2013) Rapalogs and mTOR inhibitors as antiaging therapeutics. J Clin Invest 123(3):980–989
- 76. Xia Y, Sun M, Xie Y, Shu R (2017) mTOR inhibition rejuvenates the aging gingival fibroblasts through alleviating oxidative stress. Oxid Med Cell Longev 2017:6292630. https://doi. org/10.1155/2017/6292630
- Araki T, Sasaki Y, Milbrandt J (2004) Increased nuclear NAD biosynthesis and SIRT1 activation prevent axonal degeneration. Science 305(5686):1010–1013
- Cohen HY, Miller C, Bitterman KJ, Wall NR, Hekking B, Kessler B et al (2004) Calorie restriction promotes mammalian cell survival by inducing the SIRT1 deacetylase. Science 305(5682):390–392
- Nemoto S, Fergusson MM, Finkel T (2004) Nutrient availability regulates SIRT1 through a forkhead-dependent pathway. Science 306(5704):2105–2108
- Bordone L, Motta MC, Picard F, Robinson A, Jhala US, Apfeld J et al (2006) Sirt1 regulates insulin secretion by repressing UCP2 in pancreatic beta cells. In: Dillin A (ed). PLoS Biol 4:e31. https://doi.org/10.1371/journal.pbio.0040031
- Gerhart-Hines Z, Rodgers JT, Bare O, Lerin C, Kim S-H, Mostoslavsky R et al (2007) Metabolic control of muscle mitochondrial function and fatty acid oxidation through SIRT1/ PGC-1alpha. EMBO J 26(7):1913–1923
- Rodgers JT, Lerin C, Gerhart-Hines Z, Puigserver P (2008) Metabolic adaptations through the PGC-1 alpha and SIRT1 pathways. FEBS Lett 582(1):46–53
- Ramis MR, Esteban S, Miralles A, Tan D-X, Reiter RJ (2015) Caloric restriction, resveratrol and melatonin: role of SIRT1 and implications for aging and related-diseases. Mech Ageing Dev 146–148:28–41
- 84. Someya S, Yu W, Hallows WC, Xu J, Vann JM, Leeuwenburgh C et al (2010) Sirt3 mediates reduction of oxidative damage and prevention of age-related hearing loss under caloric restriction. Cell 143(5):802–812
- Hebert AS, Dittenhafer-Reed KE, Yu W, Bailey DJ, Selen ES, Boersma MD et al (2013) Calorie restriction and SIRT3 trigger global reprogramming of the mitochondrial protein acetylome. Mol Cell 49(1):186–199
- 86. Pérez H, Finocchietto PV, Alippe Y, Rebagliati I, Elguero ME, Villalba N et al (2018) p66Shc inactivation modifies RNS production, regulates Sirt3 activity, and improves mito-chondrial homeostasis, delaying the aging process in mouse brain. Oxid Med Cell Longev 2018:8561892. https://doi.org/10.1155/2018/8561892
- 87. Li Y, Ma Y, Song L, Yu L, Zhang L, Zhang Y et al (2018) SIRT3 deficiency exacerbates p53/ Parkin-mediated mitophagy inhibition and promotes mitochondrial dysfunction: implication for aged hearts. Int J Mol Med 41(6):3517–3526
- Colman RJ, Anderson RM, Johnson SC, Kastman EK, Kosmatka KJ, Beasley TM et al (2006) Caloric restriction delays disease onset and mortality in rhesus monkeys. Science 325(5937):201–204
- Colman RJ, Beasley TM, Kemnitz JW, Johnson SC, Weindruch R, Anderson RM (2014) Caloric restriction reduces age-related and all-cause mortality in rhesus monkeys. Nat Commun 5:3557. https://doi.org/10.1038/ncomms4557
- Witte AV, Fobker M, Gellner R, Knecht S, Flöel A (2009) Caloric restriction improves memory in elderly humans. Proc Natl Acad Sci U S A 106(4):1255–1260
- 91. Aw TY (1991) Postnatal changes in pyridine nucleotides in rat hepatocytes: composition and O2 dependence. Pediatr Res 30(1):112–117
- 92. Ghosh D, Levault KR, Brewer GJ (2014) Relative importance of redox buffers GSH and NAD(P)H in age-related neurodegeneration and Alzheimer disease-like mouse neurons. Aging Cell 13(4):631–640

- Lenaz G, D'Aurelio M, Merlo Pich M, Genova ML, Ventura B, Bovina C et al (2000) Mitochondrial bioenergetics in aging. Biochim Biophys Acta 1459(2–3):397–404
- 94. Baciou L, Masoud R, Souabni H, Serfaty X, Karimi G, Bizouarn T et al (2018) Phagocyte NADPH oxidase, oxidative stress and lipids: anti- or pro ageing? Mech Ageing Dev 172:30–34
- 95. Sohal RS, Orr WC (2012) The redox stress hypothesis of aging. Free Radic Biol Med 52(3):539–555
- 96. Go YM, Jones DP (1979) Redox theory of aging: implications for health and disease. Clin Sci (Lond) 131(14):1669–1688
- Barja G (2002) Rate of generation of oxidative stress-related damage and animal longevity. Free Radic Biol Med 33(9):1167–1172
- Schindeldecker M, Stark M, Behl C, Moosmann B (2011) Differential cysteine depletion in respiratory chain complexes enables the distinction of longevity from aerobicity. Mech Ageing Dev 132(4):171–179
- 99. Paradies G, Paradies V, Ruggiero FM, Petrosillo G (2014) Cardiolipin and mitochondrial function in health and disease. Antioxid Redox Signal 20(12):1925–1953
- Pollak N, Dölle C, Ziegler M (2007) The power to reduce: pyridine nucleotides--small molecules with a multitude of functions. Biochem J 402(2):205–218
- 101. Copes N, Edwards C, Chaput D, Saifee M, Barjuca I, Nelson D et al (2015) Metabolome and proteome changes with aging in Caenorhabditis elegans. Exp Gerontol 72:67–84
- 102. Mouchiroud L, Houtkooper RH, Moullan N, Katsyuba E, Ryu D, Cantó C et al (2013) The NAD(+)/Sirtuin pathway modulates longevity through activation of mitochondrial UPR and FOXO signaling. Cell 154(2):430–441
- 103. Gomes AP, Price NL, Ling AJ, Moslehi JJ, Montgomery MK, Rajman L et al (2013) Declining NAD(+) induces a pseudohypoxic state disrupting nuclear-mitochondrial communication during aging. Cell 155(7):1624–1638
- 104. Ren X, Zou L, Zhang X, Branco V, Wang J, Carvalho C et al (2017) Redox signaling mediated by thioredoxin and glutathione systems in the central nervous system. Antioxid Redox Signal 27(13):989–1010
- 105. Veech RL, Bradshaw PC, Clarke K, Curtis W, Pawlosky R, King MT (2017) Ketone bodies mimic the life span extending properties of caloric restriction. IUBMB Life 69(5):305–314
- 106. Veech RL, Eggleston LV, Krebs HA (1969) The redox state of free nicotinamide-adenine dinucleotide phosphate in the cytoplasm of rat liver. Biochem J 115(4):609–619
- 107. Tischler ME, Friedrichs D, Coll K, Williamson JR (1977) Pyridine nucleotide distributions and enzyme mass action ratios in hepatocytes from fed and starved rats. Arch Biochem Biophys 184(1):222–236
- 108. Hanson GT, Aggeler R, Oglesbee D, Cannon M, Capaldi RA, Tsien RY et al (2004) Investigating mitochondrial redox potential with redox-sensitive green fluorescent protein indicators. J Biol Chem 279(13):13044–13053
- 109. Bradshaw PC (2019) Cytoplasmic and mitochondrial NADPH-coupled redox systems in the regulation of aging. Nutrients 11(3). pii: E504. https://doi.org/10.3390/nu11030504
- Inoue K, Zhuang L, Ganapathy V (2002) Human Na+ -coupled citrate transporter: primary structure, genomic organization, and transport function. Biochem Biophys Res Commun 299(3):465–471
- 111. Rogina B, Reenan RA, Nilsen SP, Helfand SL (2000) Extended life-span conferred by cotransporter gene mutations in Drosophila. Science 290(5499):2137–2140
- 112. Anderson RM, Weindruch R (2012) The caloric restriction paradigm: implications for healthy human aging. Am J Hum Biol 24(2):101–106
- 113. Birkenfeld AL, Lee HY, Guebre-Egziabher F, Alves TC, Jurczak MJ, Jornayvaz FR et al (2011) Deletion of the mammalian INDY homolog mimics aspects of dietary restriction and protects against adiposity and insulin resistance in mice. Cell Metab 14(2):184–195
- 114. Gregolin C, Ryder E, Kleinschmidt AK, Warner RC, Lane MD (1966) Molecular characteristics of liver acetyl CoA carboxylase. Proc Natl Acad Sci U S A 56(1):148–155

- 115. Fu JY, Kemp RG (1973) Activation of muscle fructose 1,6-diphosphatase by creatine phosphate and citrate. J Biol Chem 248(3):1124–1125
- 116. Nielsen TT (1983) Plasma citrate in relation to glucose and free fatty acid metabolism in man. Dan Med Bull 30(6):357–378
- 117. Ros S, Schulze A (2013) Balancing glycolytic flux: the role of 6-phosphofructo-2-kinase/ fructose 2,6-bisphosphatases in cancer metabolism. Cancer Metab 1(1):8. https://doi. org/10.1186/2049-3002-1-8
- 118. Huard K, Brown J, Jones JC, Cabral S, Futatsugi K, Gorgoglione M et al (2015) Discovery and characterization of novel inhibitors of the sodium-coupled citrate transporter (NaCT or SLC13A5). Sci Rep 5:17391. https://doi.org/10.1038/srep17391
- 119. Wang PY, Neretti N, Whitaker R, Hosier S, Chang C, Lu D et al (2009) Long-lived Indy and calorie restriction interact to extend life span. Proc Natl Acad Sci U S A 106(23):9262–9267
- 120. Pijpe J, Pul N, van Duijn S, Brakefield PM, Zwaan BJ (2011) Changed gene expression for candidate ageing genes in long-lived Bicyclus anynana butterflies. Exp Gerontol 46(6):426–434
- 121. Martinez-Beamonte R, Navarro MA, Guillen N, Acin S, Arnal C, Guzman MA et al (2011) Postprandial transcriptome associated with virgin olive oil intake in rat liver. Front Biosci (Elite Ed) 3:11–21
- 122. Etcheverry A, Aubry M, de Tayrac M, Vauleon E, Boniface R, Guenot F et al (2010) DNA methylation in glioblastoma: impact on gene expression and clinical outcome. BMC Genomics 11:701. https://doi.org/10.1186/1471-2164-11-701
- 123. Tian Y, Arai E, Gotoh M, Komiyama M, Fujimoto H, Kanai Y (2014) Prognostication of patients with clear cell renal cell carcinomas based on quantification of DNA methylation levels of CpG island methylator phenotype marker genes. BMC Cancer 14:772. https://doi. org/10.1186/1471-2407-14-772
- 124. Díaz M, García C, Sebastiani G, de Zegher F, López-Bermejo A, Ibáñez L (2016) Placental and cord blood methylation of genes involved in energy homeostasis: association with fetal growth and neonatal body composition. Diabetes 66(3):779–784. https://doi.org/10.2337/ db16-0776
- 125. Neretti N, Wang PY, Brodsky AS, Nyguyen HH, White KP, Rogina B et al (2009) Long-lived Indy induces reduced mitochondrial reactive oxygen species production and oxidative damage. Proc Natl Acad Sci U S A 106(7):2277–2282
- 126. Neuschäfer-Rube F, Lieske S, Kuna M, Henkel J, Perry RJ, Erion DM, Pesta D et al (2014) The mammalian INDY homolog is induced by CREB in a rat model of type 2 diabetes. Diabetes 63(3):1048–1057
- 127. von Loeffelholz C, Döcke S, Lock JF, Lieske S, Horn P, Kriebel J et al (2017) Increased lipogenesis in spite of upregulated hepatic 5'AMP-activated protein kinase in human nonalcoholic fatty liver. Hepatol Res 47(9):890–901
- 128. Willmes DM, Helfand SL, Birkenfeld AL (2016) The longevity transporter mIndy (Slc13a5) as a target for treating hepatic steatosis and insulin resistance. Aging (Albany NY) 8(2):208–209
- 129. Willmes DM, Kurzbach A, Henke C, Schumann T, Zahn G, Heifetz A et al (2018) The longevity gene INDY (I'm Not Dead Yet) in metabolic control: potential as pharmacological target. Pharmacol Ther 185:1–11
- 130. Brachs S, Winkel AF, Tang H, Birkenfeld AL, Brunner B, Jahn-Hofmann K et al (2016) Inhibition of citrate cotransporter Slc13a5/mINDY by RNAi improves hepatic insulin sensitivity and prevents diet-induced non-alcoholic fatty liver disease in mice. Mol Metab 5(11):1072–1082
- 131. Guest PC (ed) (2019) Reviews on biomarker studies of metabolic and metabolism-related disorders. In: Advances in experimental medicine and biology, 1st edn. Springer, Cham. ISBN-10: 3030126676
- 132. http://www.who.int/mediacentre/factsheets/fs311/en/
- 133. Flegal KM, Kit BK, Orpana H, Graubard BI (2013) Association of all-cause mortality with overweight and obesity using standard body mass index categories a systematic review and meta-analysis. JAMA 309(1):71–82

- 134. Aune D, Sen A, Prasad M, Norat T, Janszky I, Tonstad S et al (2016) BMI and all cause mortality: systematic review and non-linear dose-response meta-analysis of 230 cohort studies with 3.74 million deaths among 30.3 million participants. BMJ 353:i2156. https://doi. org/10.1136/bmj.i2156
- 135. Di Angelantonio E, Bhupathiraju SN, Wormser D, Gao P, Kaptoge S, de Gonzalez AB et al (2016) Body-mass index and all-cause mortality: individual-participant-data meta-analysis of 239 prospective studies in four continents. Lancet 388(10046):776–786
- 136. Hubert HB, Feinleib M, McNamara PM, Castelli WP (1983) Obesity as an independent risk factor for cardiovascular disease: a 26-year follow-up of participants in the Framingham Heart Study. Circulation 67(5):968–977
- 137. Sacco MR, de Castro NP, Euclydes VLV, Souza JM, Rondó PH (2013) Birth weight, rapid weight gain in infancy and markers of overweight and obesity in childhood. Eur J Clin Nutr 67(11):1147–1153
- Heitmann BL, Lissner L (1995) Dietary underreporting by obese individuals--is it specific or non-specific? BMJ 311(7011):986–989
- 139. Hariri N, Thibault L (2010) High-fat diet-induced obesity in animal models. Nutr Res Rev 23(2):270–299
- Ong TP, Guest PC (2018) Nutritional programming effects on development of metabolic disorders in later life. Methods Mol Biol 1735:3–17
- 141. Mickelsen O, Takahashi S, Craig C (1955) Experimental obesity. J Nutr 57:541-554
- 142. Buettner R, Schölmerich J, Bollheimer LC (2007) High-fat diets: modeling the metabolic disorders of human obesity in rodents. Obesity (Silver Spring) 15(4):798–808
- 143. Woods SC, D'Alessio DA, Tso P, Rushing PA, Clegg DJ, Benoit SC et al (2004) Consumption of a high-fat diet alters the homeostatic regulation of energy balance. Physiol Behav 83(4):573–578
- 144. Fontelles CC, Guido LN, Rosim MP et al (2016) Paternal programming of breast cancer risk in daughters in a rat model: opposing effects of animal- and plant-based high-fat diets. Breast Cancer Res 18:71. https://doi.org/10.1186/s13058-016-0729-x
- 145. Hariri N, Gougeon R, Thibault L (2010) A highly saturated fat-rich diet is more obesogenic than diets with lower saturated fat content. Nutr Res 30(9):632–643
- 146. Febbraio M, Podrez EA, Smith JD, Hajjar DP, Hazen SL, Hoff HF et al (2000) Targeted disruption of the class B scavenger receptor CD36 protects against atherosclerotic lesion development in mice. J Clin Invest 105(8):1049–1056
- 147. Pinheiro-Castro N, Silva LBAR, Novaes GM, Ong TP (2019) Hypercaloric diet-induced obesity and obesity-related metabolic disorders in experimental models. Adv Exp Med Biol 1134:149–161
- 148. Berridge KC, Ho C-Y, Richard JM, DiFeliceantonio AG (2010) The tempted brain eats: pleasure and desire circuits in obesity and eating disorders. Brain Res 1350:43–64
- 149. Castro L, Gao X, Moore AB, Yu L, Di X, Kissling GE et al (2016) A high concentration of genistein induces cell death in human uterine leiomyoma cells by autophagy. Expert Opin Environ Biol 5(Suppl 1). https://doi.org/10.4172/2325-9655.S1-003
- 150. Sampey BP, Vanhoose AM, Winfield HM, Freemerman AJ, Muehlbauer MJ, Fueger PT et al (2011) Cafeteria diet is a robust model of human metabolic syndrome with liver and adipose inflammation: comparison to high-fat diet. Obesity 19(6):1109–1117
- 151. Zeeni N, Dagher-Hamalian C, Dimassi H, Faour WH (2015) Cafeteria diet-fed mice is a pertinent model of obesity-induced organ damage: a potential role of inflammation. Inflamm Res 64(7):501–512
- 152. Maioli TU, Gonçalves JL, Miranda MCG, Martins VD, Horta LS, Moreira TG et al (2016) High sugar and butter (HSB) diet induces obesity and metabolic syndrome with decrease in regulatory T cells in adipose tissue of mice. Inflamm Res 65(2):169–178
- 153. Crescenzo R, Bianco F, Mazzoli A, Giacco A, Cancelliere R, di Fabio GA et al (2015) Fat quality influences the obesogenic effect of high fat diets. Nutrients 7(11):9475–9491

- Aller EEJG, Abete I, Astrup A, Martinez JA, van Baak MA (2011) Starches, sugars and obesity. Nutrients 3(3):341–369
- Lennerz B, Lennerz JK (2018) Food addiction, high-glycemic-index carbohydrates, and obesity. Clin Chem 64(1):64–71
- 156. Panchal SK, Poudyal H, Iyer A, Nazer R, Alam MA, Diwan V et al (2011) Highcarbohydrate, high-fat diet–induced metabolic syndrome and cardiovascular remodeling in rats. J Cardiovasc Pharmacol 57(5):611–624
- 157. Hung WW, Ross JS, Boockvar KS, Siu AL (2011) Recent trends in chronic disease, impairment and disability among older adults in the United States. BMC Geriatr 11:47. https://doi.org/10.1186/1471-2318-11-47
- 158. Madreiter-Sokolowski CT, Sokolowski AA, Waldeck-Weiermair M, Malli R, Graier WF (2018) Targeting mitochondria to counteract age-related cellular dysfunction. Genes (Basel) 9(3):pii: E165. https://doi.org/10.3390/genes9030165

# Index

#### A

Acetoacetate, 80 Acetone, 78, 80 Acetyl-coenzyme A (acetyl-CoA), 78 Actin microfilaments, 228, 229 Activate adipose triglyceride lipase (ATGL), 209 Activator protein-1(AP-1) activity, 160 Adaptive homeostasis, 57 Adenosine triphosphate (ATP), 248 Adipose tissue (AT) alterations, 212 BAT, 209, 210 Beige AT, 210 heterogeneity, 208, 211, 212 mice and humans, 215, 216 secretory organ, 212 WAT, 208, 209 Adrenocorticotropic hormone (ACTH), 61,63 Age-related cardiovascular diseases (CVDs), 114-115 Age-related differentially methylated regions (aDMRs), 183 Age-related diseases Alzheimer's disease, 58 Parkinson's disease, 59 Aging and atherosclerosis impacts of ADF, 118 impacts of CR, 118 impacts of DR, 116-117 Aging biomarker antiaging-strategy discovery, 40 biological and chronological age, 40 interventions (see Anti-aging interventions)

multifactorial effects, 39 population, 39 research, 41 risk factor, 39 Aging gonadotropic axis clinical symptomatologies, 60 female menopausal transition, 60 FSH, 60 HPG. 59 menstrual cycle, 60 testosterone levels, 60 Aldabrachelys gigantea, 12 Alternate-day fasting (ADF) CVDs, 114 experiments, 120-121 "feast day", 114 human study, 118 Alzheimer's disease, 3, 58, 87-89, 190 APP/PS1, 146 Aβ amyloid deposition, 134 bacterial amyloid proteins, 140 Bifidobacterium breve strain A1, 144 biomarkers, 139 butyrate, 144 chronic inflammation/molecular mimicry mechanisms, 139 fecal microbiota, 137 LPS, 142 mild cognitive impairment/dementia, 145 shikimate pathway, 143 TMAO, 145 Alzheimer's Disease Assessment Scalecognitive subscale scores, 89 Amyotrophic lateral sclerosis (ALS), 235

© Springer Nature Switzerland AG 2019 P. C. Guest (ed.), *Reviews on Biomarker Studies in Aging and Anti-Aging Research*, Advances in Experimental Medicine and Biology 1178, https://doi.org/10.1007/978-3-030-25650-0 . . . .

Animal models impacts of DR ADF experiments, 120 aging and atherosclerosis, 116-117 CR experiments, 118-119 neurodegenerative disorders, 115 Anti-aging interventions gut microbiota, 46-48 longevity proteins, 42-46 SASP. 45-46 telomeres, 41-42 Anti-inflammatory agents, 164 Antioxidants, 162 Antioxidative effects, 164 Anti-photo-aging agents, 163 Appendages, 157 Argininosuccinate lyase (ASL), 14 Astrocyte monocarboxylate transporter (MCT), 28 Astrocyte-neuron lactate shuttle (ANLS), 28 Astrocytes, 27, 28 aldolase C and hexokinase 1, 29 ANLS, 28 brain functions, 27 coculture model. 28 connexins, 28 DISC-1, 30 neuroinflammation, 29 and oligodendrocytes, 26, 33 role of. 30 Atrial fibrillation (AF), 114 Attention deficit hyperactivity disorder (ADHD), 81 Autism spectrum disorders (ASD), 82 Autism Treatment Evaluation Checklist scores, 82

## B

Beta-hydroxybutyrate injections, 78, 89 Bifidobacterium, 47 Bile acids, 144, 145 Biodiversity alpha diversity *vs.* cognitive symptoms, 135 Alzheimer's disease, 134 antibiotic therapy, 134 *Bacteroides* and Firmicutes/Bacteroidetes ratio, 130 communities *vs.* host, 133 Deary-Liewald Reaction Time test score, 135 definition, 133 HIV, 134 SAMP8, 134 Simpson Index/Chao1 Index, 133 Biological systems, 57 Blood-brain barrier (BBB), 27 Body mass index (BMI), 186 Bovine GH (bGH), 214 Brain metabolic signatures, 142 Brain tissue, 106 Brown adipose tissue (BAT), 208–210

# С

Caenorhabditis elegans, 9, 10, 41, 248.250 Calorie restriction (CR), 48, 50, 180 cardiovascular morbidity and mortality, 113 definition, 113 delays cardiac aging, 119-120 experiments, 118-119 human study, 118 myocardial ischemia, 121-122 physiological and pathophysiological changes, 113 signaling pathways, 114 Cancer pathogenesis, 5 Carbohydrates, 86 Cardiac aging and age-related left ventricular diastolic dysfunction ADF animal, 120-121 human, 121 CR animal. 118-119 human, 121 Cardiomyocytes, 119 Cardiovascular aging, 114 Cardiovascular disease, 108, 110 Cardiovascular fibrosis, 108 Catalase (CAT) gene, 164, 250, 251 CCAAT/enhancer-binding proteins (C/EBPs), 208 Cellular biology, 227 Cellular energy metabolism, 27 Central nervous system (CNS), 28 Centrosome defects, 232 Cerebrospinal fluid (CSF), 29, 61, 106 Charcot-Marie-Tooth disease (CMT), 235 Chelonoidis abingdonii, 12 Chemokine ligand 5 (CCL5), 164 Chicago Health and Aging Project, 88 Childhood Autism Rating Scale, 82 Cholesterol, 253 Cholinesterase inhibitors, 87 Chondroitin sulphate, 161 Chronological aging, 155 Coenzyme Q, 252

Coenzyme Q10 (CoQ10) antioxidant and anti-inflammatory agent, 103 bezoquinone nucleus and isoprenoid side chain, 104 functions, 104-105 isoprenoid subunits, 103 KISEL-10 study, 109 laboratory measurement, 106-107 lipophilic molecule, 103 MCR. 103 supplementation (see Supplementation) synthesis and deficiency, 105-106 ubiquinones, 103 Cognitive symptoms, 83 Collagen, 159, 160, 165 Comparative gene identification-58 (CGI-58), 209 Computed tomography (CT), 130 Congestive heart failure (CHF), 114 Connective tissue growth factor (CTGF), 108 Corneocytes, 158 Coronary artery disease, 209 Corticotrophin releasing hormone (CRF), 63 Cortisol awakening response (CAR), 64, 66 Cortisol binding globulin (CBG), 63, 65, 66 Cosmeceutical products, 166-168 CpG island (CGI) sequences, 177 Cyclooxygenase (COX), 250 Cyclooxygenase 2 (COX-2), 163 Cytochrome oxidase subunit-I (COI/COX1), 16 Cytoskeleton actin associated pathologies, 235, 236 actin microfilaments, 228, 229 cellular biology, 227 IFs, 231 intracellular transport and spatial organization, 227 microtubule associated pathologies, 235 MTs, 229 neurodegeneration, 237 neurofilaments (NF), 236 organismal aging, 232, 233 Tau associated pathologies, 233, 234 Cytosolic phospholipase A2 (cPLA2), 236

#### D

Dehydroepiandrosterone (DHEA) adrenal fetal zone, 61 adrenal secretion, 61 androgens and estrogens, 62 biological reason, 62 cross-sectional studies, 61 health outcomes, 62

hypotheses, 62 neurosteroidal properties, 61 sulphotransferase, 61 Dementia gut-brain axis animal study, 129 dysbiosis, 129 humans, 130-132 toxic/protective activity, 130 Deoxynucleotides, 253 Dermal-epidermal junction (DEJ), 157 Dermal fibroblasts, 157 Dermatan sulphate, 161 Dermis collagen, 159, 160 elastin, 160 GAGs. 161 Dexamethasone suppression test (DST), 66 Diastolic BP variability (DBPV), 124 Dietary restriction (DR) age-related cardiovascular diseases, 115 aging and atherosclerosis (see Aging and atherosclerosis) animal study, 116-117 autonomic nerve function, 124 human, 121 metabolic cardiomyopathy, 123 methods, 114 myocardial infarction and post-infarct LV remodeling, 122 Dietary restriction response of unknown function (DRR-1), 249 Diphenylhydantoin, 80 Disrupted in schizophrenia-1 (DISC-1), 30 DNA methylation AgeAccelGrim, 189 age-related conditions, 186 biological aging and all-cause mortality, 190 blood cells, 186, 192 childhood and adolescence, 189 chronological age, 187 correlation, 193 diets rich. 191 DNAm GrimAge, 189 Down syndrome, 190 Edaradd and Tom1L1. 184 elastic net regression analyses, 186 epigenetic aging vs. chronological aging, 190.193 excessive and chronic stress, 190 gestational age (GA), 189 glucocorticoids, 191 linear regression model, 189 molecular markers, 184

DNA methylation (*cont.*) multidimensional scaling, 192 multi-tissue clock, 195 *NPTX2*, 184 peripheral blood mononuclear cells, 190 PhenoAge epigenetic clock, 188 PyroMark system, 193 quadratic regression, 185 RRBS and WGBS studies, 194 tissue and physical functioning, 194 tissue-specific processes, 194 transcriptome-wide analysis, 192 unhealthy life habits, 184 Down syndrome, 190

#### E

18-fluoro-deoxyglucose (18FDG), 210 8-Oxo-7,8-dihydroguanosine (8-oxoGsn), 46 Elastin, 160, 165 Elastosis, 160 Electron transport chain (ETC) activity, 248 Embryonic stem cells (ESCs), 187 Endocrine factors, 212 Endothelial cells, 212 Endothelial NO synthase (eNOS), 117 Energy-consuming process, 27 Enzyme-linked immunosorbent assay (ELISA), 46 EPH receptor A2 (EPHA2), 14 Epidermis, 158, 159 Epidydimal WAT (eWAT), 216 Epigallocatechin-3-galate (EGCC), 163 Epigenetic clock cell systems, 176 cellular reprogramming, 196 components, 175 cripto and cerberus1 act, 197 DNA methylation, 175, 176 epigenome, 175 genome-wide microarray analysis, 198 heterochronic parabiosis, 197 histone modification system, 175 iPSCs, 195 mTOR, 199 NF-kB activity, 198 pancreas and muscle tissue, 196 somatic cells, 195, 196 Epigenetic reprogramming, 196 Epilepsy, 80 Extracellular matrix (ECM), 209 Extracellular membrane (ECM), 158 Extracellular signaling-regulated kinase (ERK), 213

Extracellular vesicles (EVs), 46 Extrafibrillar matrix, 159

#### F

Fatty acids, 253 Fibroblast growth factor 2 (FGF2), 212 Fibroblasts, 107, 159 Fibrosis cardiovascular disease, 108 fibroblasts, 108 heart failure, 109 loss of function, 107 mechanism, 107 myocardial infarction, 109 necrotic cells, 108 study, 107-108 wound healing and tissue repair, 107 Fibrous network, 161 Filopodia, 228, 231 Firmicutes/bacteroidetes ratio fecal microbiota, 135 gut microbiota and obesity, 136 human, 136 recognition memory and spatial memory tasks, 136 taxonomic level and calculation, 135 Fluorescence in situ hybridization (FISH), 42 Follicle-stimulating hormone (FSH), 59 Forkhead box protein A3 (FOXA3), 213 Free fatty acid (FFA), 209 Fused in sarcoma (FUS), 235

## G

Gain of function (GoF), 234 Gallic acid. 165 Genome methylation patterns blood-derived DNA, 177 hyper-aCpGs, 178 hypermethylation, 177 solid human tissues, 177 Genome-wide association studies (GWAS), 4 Glial cells, 27 Glial fibrillary acid protein (GFAP), 29 Glucocorticoid receptors (GR), 64 Glucocorticoids, 63 Glucosaminoglycans (GAGs), 161 Glucose infusion rate (GIR), 210 Glycolysis acute and long-term treatment, 29 alterations. 31 astrocytes, 27, 28 glucose, 27

neurodevelopment, 30 and oligodendrocytes, 27 pathway, 33 progenitor cells, 31 Glycosaminoglican (GAG), 159 Gonadotropin-releasing hormone (GnRH), 59 G-protein coupled receptors (GPCRs), 209 Gross domestic product (GDP), 2 Growth factor receptor-bound protein 14 (GRB14). 14 Growth hormone (GH) Alzheimer's disease, 216 humans, 214 longevity, 215 mice, 214, 215 whole-body physiology, 216 Growth hormone receptor (GHR), 213 Growth hormone receptor knockout (GHRKO) mice, 215 Growth hormone releasing hormone (GHRH), 213 Gut microbiota abundance taxa, 136-139 bifidobacterium, 47 bile acids, 144-145 biomarkers of disease, 133 brain metabolic signatures, 142 complex interplay, 143 diversity, 47 environment vs. human pathophysiology, 132 - 133Firmicutes/Bacteroidetes ratio, 135-136 functions, 46 human health and disease, 132 human serum levels, 142 IAA. 143 immunosenescence, 47 indoxyl sulfate, 143 and inflamm-aging, 47 inflammation, 142 kynurenine pathway, 143 LPS, 141 metagenomics analyses, 132 microbial metabolites (see Microbial metabolites) microbiome biodiversity, 133-135 Mini-Mental State Examination test score, 143 neurological toxicity and inflammation, 143 patterns and habits, 132 polyphenols, 145-146 recognition and spatial memory, 142 rhamnolipids, 142 **SCFAs** 

Alzheimer's disease, 144 butyrate and propionate, 144 *Clostridium butyricum*, 144 dysfunction and lower colonic concentrations, 144 physiopathological role, 144 serum and feces, 142 shikimate pathway, 143 study, 130 TMAO, 145 tryptophan, 48, 143 vitamin K, 146

# H

Hair follicles, 157 Heart rate variability (HRV), 124 Heme oxygenase 1 (HO-1), 163 Herbal skin care product cosmeceutical products, 166-168 nutricosmeceutical products, 166, 168-170 Heterochronic parabiosis, 197 High-density-lipoprotein (HDL), 87 High-pressure liquid chromatography (HPLC), 106 High-pressure liquid chromatography with ultraviolet (HPLC-UV), 106 Homeostatic mechanisms, 57 Hormonal status, 58, 60 Human induced pluripotent stem cell (hiPSC), 31 Humans cardiac aging and age-related left ventricular diastolic dysfunction, 121 impacts of ADF, 118 impacts of CR, 118 Huntington's disease (HD), 235 Hutchinson Gilford Progeria Syndrome (HGPS), 187 Hvaluronic acid (HA), 159, 161 Hypermethylated (hyper-aCpGs) age-dependent, 179 age-related, 180 chromosome, 182 DNA methylation, 180, 181 genome-wide association study, 182 human organs, 179 human skeletal muscle, 179 leukocytes, 183 metabolic processes, 183 molecular functions, 178 morphogenesis, 179 mouse hippocampus, 181 open chromatin domains, 181

Hypermethylated (hyper-aCpGs) (cont.) PBLs, 178 regulatory elements, 182 stem cells, 178 T cells, 178 tissue-specific functions, 178 transchromosomic mouse strain, 182 Hyperphagia, 255 Hypomethylation, 177 Hypothalamus, 213 Hypothalamus pituitary gonadal (HPG), 59 Alzheimer's disease, 65 CAR. 64 chronic activation, 64 cognitive processes, 64 dynamics, 64 glucocorticoids, 63, 64 homeostatic mechanism, 63 isoforms, 64 measurement CAR. 66 clinical testing, 66 hair growth, 66 psychosocial protocols, 67 salivary cortisol, 66 substrates, 66 receptors, 64 regulation CBG, 65 clinical investigations, 65 cortisol levels, 65 estradiol, 65 menstrual cycle, 65 Hypoxia, 209 Hypoxia inducible factor-1 $\alpha$  (HIF-1 $\alpha$ ), 253

#### I

Idiopathic pulmonary fibrosis (IPF), 110 Illumina HumanMethylation27 BeadChip platform, 185 Immune cells, 211 Immune system adaptive homeostatic responses, 67 effects, 67 host protection, 67 inflamm-aging, 68 macrophages and dendritic cells, 67 pathogens, 67 risk factor, 68 Immunological response adaptive immune system, 69 estrogen receptors, 68 pro-inflammatory markers, 68 X chromosome, 68

Immunosenescence, 47 I'm not dead yet (INDY), 254 Indole-3 acetic acid (IAA), 143 Induced pluripotent stem cells (iPSCs), 187.195 Inducible NO synthase (iNOS), 117 Inflamm-aging, 68 Ingenuity® Pathway Analysis (IPA®), 180 Inguinal WAT (iWAT), 208 Insulin and insulin-like growth factor (IGF), 5 Insulin/insulin-like growth factor (IGF), 44 Insulin/insulin-like growth factor-1 (ins/IGF-1) signalling pathway, 250 Insulin-like growth factor (IGF) family, 248 Insulin-like growth factor 1 (IGF-1) axis, 207 Insulin receptor substrates (IRS), 8 Insulin resistance, 254 Insulin signalling, 10 Interferon type I (IFN1-type) inflammation response, 176 Interleukin (IL), 67 Interleukin 6 (IL-6), 164 Intermediate filaments (IFs), 231 Intra-hepatic lipids, 209 Intra-myocellular lipids, 209

#### J

Janus kinase 2 (JAK2), 213 Janus kinase signal transducer and activator of transcription (JAK/STAT), 45

## K

Keratinocytes, 158 Ketogenic diet (KD), 48-50 Alzheimer's disease, 87-89 carbohydrate and proteins, 78 epilepsy, 78, 80 fasting/food restriction, 78 free fatty acids, 78 liver and adipose tissue, 78 nutritional interventions, 77 obesity and type II diabetes, 84, 86, 87 Parkinson's disease, 89 pathophysiology of psychiatric conditions, 79 physical and mental health, 77 schizophrenia, 82, 83 side effects, 90, 91 ulcers and haemorrhoids, 77 KISEL-10 study, 109

Index

#### L

Laron syndrome, 214 L-dihydoxyphenylalanine (L-Dopa), 89 Left ventricular hypertrophy (LVH), 114 Lipolysis, 212 Lipopolysaccharide (LPS), 141 Longevity proteins apidonectin, 5 α-Klotho, 44 atherosclerosis, 8 biomarkers, 6, 7 bowhead whales, 12-14 caloric restriction, 9 cytokines, 7 dietary antioxidant, 7 gene polymorphisms, 5 genetic analyses, 5 genetic, environmental and lifestyle components, 4 giant tortoise, 12 glucose metabolism, 8 Greenland shark, 14 healthcare expenditures, 2 hormones, 5, 7 IGF pathway, 44 immortal jellyfish, 15 mice, 8, 9 miRNAs, 5, 10, 11 molecular mechanisms, 3 mortality risk, 5 mTOR, 45 non-communicable diseases, 2 ocean quahog, 14 polyunsaturated lipids, 7 randomized control trial, 7 sirtuins, 44 tumor necrosis factor-alpha, 7 visual memory task, 7 worms, 9, 10 Loss of function (LoF), 235 Low-density-lipoprotein (LDL), 87 Lower glycosaminoglycans, 157 Luteinizing hormone (LH), 59

#### M

Mammalian target of rapamycin (mTOR), 8, 45, 49, 108, 213, 248, 249, 252 Mass spectrometry (MS) analysis, 164 Mast cells, 159 Matrix metalloproteinase (MMP) gene expression, 160 Melanin, 165 Melanocortin 1 receptor (MC1R), 166 Melanocytes, 157, 158 Mercenaria mercenaria, 15 Mesolimbic dopamine pathway, 83 Metabolic biomarkers age-related diseases, 247, 248 Caenorhabditis elegans, 248, 250 caloric restriction, 248 cellular and molecular determinants, 247 cellular senescence, 247 human life expectancy, 247 **INDY. 254** mitochondria, 251, 252 molecular and physiological mechanisms, 247 obesity and aging, 254, 255 organs and tissues, 248 physiological mechanisms, 247 redox stress and aging, 253 senescent cells, 248 Metabolic pattern CR, 48-50 KD, 48-50 NAD+. 48 Microbial metabolites Alzheimer's disease, 140 byproducts/constituents, 139 gut-brain axis, 140 "leaky gut", 140 neurotransmitters, 140 polyphenol and urolithins, 140 SCFAs, 140 TMAO, 140 Microbiota diversity, 47 Microglia, 236 Microphthalmia-associated transcription factor (MITF), 166 Microtubule organization center (MTOC), 230 Microtubules (MTs), 229 Mild cognitive impairment, 130, 131, 145, 147 Mineralocorticoid receptor (MR), 64 Mitochondria, 232, 251, 252 Mitochondrial inhibitors, 248 Mitochondrial respiratory chain (MRC), 103.104 Mitochondria-targeted form of coenzyme Q (MitoQ), 252 Mitogen-activated protein kinase (MAPK), 209 Molecular mechanisms, 13 Monocarboxylic acid transporter, 78 Monozygotic (MZ) twins, 177 Montreal Cognitive Assessment scores, 89 Myelin and lymphocyte protein (MAL), 30 Myelin-associated glycoprotein (MAG), 30 Myogenic factor 5 (MYF5), 209 Myosin heavy chain 11 (MYH11), 211

#### N

Nerve growth factor (NGF), 212 Neural stem cells (NSCs), 197 Neurodegenerative disorders, 79 Neurofibrillary Tangles (NFTs), 234 Neurofilaments (NFs), 231 Neurological disorders, 79, 233 Neuronal cells, 27 Nicotinamide adenine dinucleotide (NAD+), 48.49 Nicotinamide adenine dinucleotide hydride (NADH), 78 Nicotinamide adenine dinucleotide phosphate (NADPH), 253 Nitric oxide (NO), 212 Nitrous oxide (N<sub>2</sub>O), 14 N-methyl d-aspartate (NMDA) receptor, 87 Non-alcoholic fatty liver disease (NAFLD), 254 Norepinephrine (NE), 209 Nuclear factor erythroid 2-related factor 2 (NRF-2), 163 Nuclear factor kappa B (NF-kB), 117, 165 Nuclear magnetic resonance (NMR) spectroscopy, 164 Nutricosmeceutical products, 166, 168-170

## 0

Obesity, 84, 86, 87, 254 Oligodendrocytes, 229 dysfunction, 30 metabolic coupling, 29 MK-801-treated, 31 myelination, 28, 30, 31 NMDA receptors, 29 task, 29 telomeres, 33 Oxidative phosphorylation (OXPHOS) complexes, 251 Oxidative stress, 106–108

## P

Paired box 3 (PAX3), 211 Paired helical filaments (PHFs), 234 Pancreatic stellate cells (PCSs), 108 Parkinson`s disease, 59, 89, 190 Peripheral blood leukocytes (PBLs), 178 Peroxisome proliferator-activated receptor gamma (PPARγ), 9, 208 Phosphoenolpyruvate carboxykinase (PEPCK), 214, 250 Photoaging, 155 Pituitary factor 1 (Pit1), 214 Platelet derived growth factor receptor alpha (PDGFRα), 211 Polymerase chain reaction (PCR), 42 Polyphenols, 169 Polyunsaturated fatty acids (PUFAs), 185 Positive and Negative Syndrome Scales, 84 Positron emission tomography coupled with computed tomography (PET/CT), 210 Positron-emission tomography (PET/CT), 210 Positron-emission tomography (PET), 130 PR domain containing 16 (PRDM16), 209 Preadipocytes, 211, 213 Proinflammatory cytokines, 209, 212 Proliferating cell nuclear antigen (PCNA), 13 Prostaglandin-E2 (PGE2), 164 Protein kinase A (PKA), 209

### R

Rab-like GTPase (RAB-10), 249 Reactive oxygen species (ROS), 8, 42, 104, 156, 247, 248, 251 Reduced representation bisulfite sequencing (RRBS), 182 Reduction–oxidation (redox) reactions, 251 Replication protein A2 (RPA2), 14 RhoA kinase (ROCK), 235, 236

#### S

S-adenosyl methionine synthetase (SAMS), 249 Sarcoplasmic reticulum (SR), 119 Schizophrenia, 82, 83 aging effect elongation, 33 glucose metabolism, 31 telomeres, 32 white matter abnormalities, 33 and astrocytes, 29-30 definition, 25 energy handling, 25 glial cells, 27-29 glucose, 27 glycolysis, 26, 27 and oligodendrocytes, 30-31 Sebaceous glands, 157 Selenocysteine, 10 Senescence-accelerated mouse prone 8 (SAMP8), 134 Senescence-associated secretory phenotype (SASP) detection, 46 DNA damage, 45 8-oxoGsn, 46 EVs, 46

proteins and cytokines, 45 secreted factors, 45 Septins, 232 Serum, 29 Sex differences in aging age-related diseases, 58-59 DHEA. 61-62 endocrine markers, 59 homeostatic biological systems, 69 homeostatic mediators, 58 HPA axis, 63-67 immune system, 67-68 immunological response, 68-69 non-modifiable risk factors, 69 reproductive system, 59-61 steroidal hormones analyses, 62, 63 Sex hormone binding globulin (SHBG), 60 Short-chain fatty acids (SCFAs), 140 Signal transducer and activator of transcription 5 (STAT5), 213 Single nucleotide polymorphisms (SNPs), 4.183 Single telomere length analysis (STELA), 42 Sirtuin activators, 248 Sirtuin pathways, 252 Skin aging anti-aging and photoprotective properties, 156 characteristics basal layer, 157 epidermis, 158, 159 genetic/metabolic factors, 157 skin topography, 157 herbal extracts, 156 intrinsic and extrinsic aging process, 155.156 physical barrier, 155 plant extracts and natural compounds, 156 skin cell proliferation, 164 subcutaneous tissue, 162 vasculature, 161 Smooth muscle cells, 212 Somatic cell nuclear transfer (SCNT), 195 Somatostatin (SST), 213 Spinal muscular atrophy (SMA), 235 Sprague Dawley rats, 124 Stochastic epimutations (SEMs), 191.192 Straight filaments (SFs), 234 Subcutaneous tissue, 162 Substance abuse disorders, 79 Sun protection factor (SPF), 163 Superoxide dismutase (SOD), 250 Superoxide dismutase 1 (SOD1), 235

Supplementation antioxidant and anti-inflammatory functions, 110 dietary deficiencies, 110 IPF, 110 mTOR expression, 108 PCSs, 108 requirements, 110 reverse fibrosis, 108 safety, 109, 110 Survival of motor neuron 1 gene (SMN1), 235 Sweat glands, 157

## Т

Tannins, 165 TAR DNA binding protein (TARDBP/ TDP-43), 235 Tauroursodeoxycholic acid, 145 Taxa animal models/human beings, 137, 138 bacterial taxon, 136 Burkholderiales, 137 Christensenellaceae and Ruminococcaceae, 137 cognitive dysfunction, 137 Helicobacter pylori, 139 Lachnospiraceae and Clostridium XIV b, 139 Lactobacillaceae and Christensenellaceae, 137 over-representation, 137 p-tau and β-amyloid deposition, 139 under-representation, 137 Telomerase and telomeric repeat-containing RNA (TERRA), 41 Telomeres, 43 aging-related biomarkers, 31 attrition, 41 cancer cells, 42 description, 32 erosion. 32, 33 nucleoprotein, 41 shortening, 32, 33, 42 TERRA, 41 Tert-butyl hydroperoxide (TBHP), 15 3'-Un-translated regions (UTRs), 178 Thermotolerance, 232 Thyroid-stimulating hormone (TSH), 214 Transcriptional start sites (TSS), 178 Transdifferentiation, 15 Transforming growth factor  $\beta$  (TGF- $\beta$ ), 45, 108, 160 Tricarboxylic acid (TCA), 27, 28, 250 Trier Social Stress Test (TSST), 67 Trimethylamine N-oxide (TMAO), 140

Tryptophan metabolism, 47, 48 Turritopsis dohrnii, 15 Twisted ribbon-like lilaments (TRFs), 234 Type II diabetes, 84, 86, 87 Tyrosine-related protein 1 (TRP-1), 165 Tyrosine-related protein 2 (TRP-2), 165

#### U

Ultra Violet radiation (UVR), 155 Uncoupling protein 1 (UCP1), 209 Urolithin-A, 146 Urolithin-B, 146

# $\mathbf{V}$

Vascular endothelial growth factor A (VEGFa), 212 Vascular endothelial growth factors (VEGFs), 45 Vasculature, 161

## W

White adipose tissue (WAT), 208, 209 Whole-genome bisulfite sequencing (WGBS) study, 181 Wild type (WT), 184 World Health Organization (WHO), 1